Cancer Metastasis – Biology and Treatment 20

Chris Maxwell Cal Roskelley *Editors*

Genomic Instability and Cancer Metastasis

Mechanisms, Emerging Themes, and Novel Therapeutic Strategies



Cancer Metastasis - Biology and Treatment

Volume 20

Series Editors

Richard J. Ablin Dept. of Pathology, University of Arizona College of Medicine, Tucson, Arizona, USA

Wen G. Jiang Institute of Cancer & Genetics Cardiff University, Cardiff, United Kingdom More information about this series at http://www.springer.com/series/5761

Chris Maxwell • Cal Roskelley Editors

Genomic Instability and Cancer Metastasis

Mechanisms, Emerging Themes, and Novel Therapeutic Strategies



Editors Chris Maxwell Child & Family Research Institute University of British Columbia, Pediatrics Vancouver British Columbia Canada

Cal Roskelley Dept of Cellular & Physiological Sciences University of British Columbia, Dept. of Cellular & Physiological Sciences Vancouver British Columbia Canada

ISSN 1568-2102 ISBN 978-3-319-12135-2 DOI 10.1007/978-3-319-12136-9

ISBN 978-3-319-12136-9 (eBook)

Library of Congress Control Number: 2014956658

Springer Cham Heidelberg New York Dordrecht London © Springer International Publishing Switzerland 2015

This work is subject to copyright. All rights are reserved by the Publisher, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilms or in any other physical way, and transmission or information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed. Exempted from this legal reservation are brief excerpts in connection with reviews or scholarly analysis or material supplied specifically for the purpose of being entered and executed on a computer system, for exclusive use by the purchaser of the work. Duplication of this publication or parts thereof is permitted only under the provisions of the Copyright Law of the Publisher's location, in its current version, and permission for use must always be obtained from Springer. Permissions for use may be obtained through RightsLink at the Copyright Clearance Center. Violations are liable to prosecution under the respective Copyright Law.

The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use.

While the advice and information in this book are believed to be true and accurate at the date of publication, neither the authors nor the editors nor the publisher can accept any legal responsibility for any errors or omissions that may be made. The publisher makes no warranty, express or implied, with respect to the material contained herein.

Printed on acid-free paper

Springer is part of Springer Science+Business Media (www.springer.com)

Preface

In 1975, the mortality rate for all cancers in the United States was 199 deaths per 100,000 people in the population [1]. The most recent statistics available (2006–2010) indicate that the current mortality rate is 176.4 deaths per 100,000 people [2], which represents a modest 11% reduction in the past 35 years. It has been estimated that approximately 90% of cancer deaths arise from the metastatic spread of primary tumors [3]. Thus, any future improvement in the overall cancer mortality rate will depend upon a more thorough understanding of the properties that enable the metastatic process.

Cancers are diverse and complex tissues. Conceptually, it is proposed that cancers possess shared hallmarks [4], but it may be more correct to consider cancer as an evolving, heterogeneous, and dynamic entity [5] that is responsive to a plethora of selective events, both intrinsic to the tumor (e.g., hypoxia) and extrinsic (e.g., combination chemotherapy). Within the framework of an evolving tissue, it is clear that the extent of heterogeneity may provide selective advantages. Moreover, scientists have long known that cancer cells display significant intra-tumoral heterogeneity at the genetic level [6], which falls under the umbrella term of genomic instability. Therefore, it is reasonable to propose the interplay between tumor heterogeneity, which may be enabled by genomic instability, and an enhanced ability to withstand the selection pressures applied throughout the metastatic process.

This book connects cancer metastasis with genomic instability in a comprehensive manner through four sections. Section 1 outlines the fundamental mechanisms that occur at tissue, cellular and molecular levels and regulate the processes of cancer metastasis, genomic stability, and DNA damage response, respectively. Section 2 discusses the model systems that will enable our better understanding of the metastatic process and genomic instability through experimentation performed *in silico*, *in vitro*, and *in vivo*. Section 3 reviews emerging themes and frameworks for the understanding of the contributions of non-tumor cells to the metastatic process (ex., tumor microenvironment, mechanotransduction, and immunomodulation). Finally, Section 4 discusses new therapeutic approaches designed to overcome the unique challenges presented by the heterogeneous and metastatic tumor.

Section 1 takes a reductionist approach to describe the mechanisms responsible for the maintenance of tissue and cellular integrity starting with an examination of the tumor tissue, followed by a discussion of processes that prevent instability at the genome level, and ending with a discussion of molecular pathways that act at the base-pair level to maintain integrity. Drs. Rodenhiser and Chambers (London Regional Cancer Clinic, Canada) introduce the metastatic tumor as an evolving tissue that presents many challenges, including diversity and dynamic heterogeneity, which must be interrogated in the clinic if novel treatments are to be successful. Dr. Connell and colleagues (University of British Columbia, Canada) outline the pathways in the normal cell, or the malignant tumor cell, that are responsible for the prevention of genomic instability and, when compromised, the promotion of heterogeneity. Dr. El-Khamisy and colleagues (University of Sheffield, UK and Helmy Institute, Egypt) review the diverse molecular pathways that fall under the umbrella term of the DNA damage response, and discuss the intimate relationship between loss of function in each of these pathways and cancer predisposition.

More complete knowledge of cancer metastasis, and the complex interplay between regulators that are found in the cancer cell, the tumor tissue and within the organism inflicted with cancer, will be born from studies that encompass in silico, in vitro, and in vivo model systems, each of which are reviewed in Section 2. Dr. Costes and colleagues (Lawrence Berkeley National Laboratories, USA) utilize 3D automated foci detection and computational modeling to understand and track the properties and kinetics for repair of DNA double strand breaks. Chapter 4 outlines how integration of mathematical models with irradiation data, a highly quantitative and reproducible manner to induce DNA damage, enables the synthesis of new knowledge in the fields of cancer initiation, detection and progression. In Chapter 5, Dr. Bennewith and colleagues (BC Cancer Agency and Dalhousie University, Canada) discuss a variety of animal models and the strengths and considerations when using these models to address specific research questions, which span from high-throughput analysis of novel compounds to dissection of the relative contributions of individual gene products during defined stages of cancer metastasis, from local invasion to distal entrenchment and expansion.

In 2000, Hanahan and Weinberg outlined six hallmarks of cancer [7]; only one of the six original hallmarks of cancer (i.e., they stimulate the growth of blood vessels to supply nutrients to the tumor) identified a property that was extrinsic to the tumor cell. It is increasing clear, however, that non-tumor cells, in the cancer tissue and the patient afflicted with cancer, are critical to cancer progression and metastasis, including, but not limited to, the role of tumor-stroma interactions and the tumor microenvironment, mechanical cues provided from the environment to the tumor, and immunomodulation. Section 3 reviews these emerging themes in the field of tumor microenvironment and cancer metastasis. In Chapter 6, Dr. Calvin Roskelley (University of British Columbia, Canada) provides an overview of microenvironment ransduction in the tumor and immunosurveillance are detailed by Dr. Nelson and colleagues (Princeton University, USA) and Dr. Gregor Reid (University of British Columbia, Canada) in Chapters 7 and 8, respectively.

Prevention and treatment of metastatic tumor spread may represent the most significant challenge of medical oncology. In Chapter 8, Dr. Reid reviews the cross-talk between immune cells and cancer cells; the controlled regulation of these dynamic processes through chemical or cell-based therapies may allow for improved immunosurveillance of metastatic cells. While tumor heterogeneity enabled through genome instability likely provides the tumor advantages against conventional chemotherapeutic and irradiation treatments, it is hoped that molecular-targeted therapies can turn the table by targeting pathways that are non-essential in normal tissue but, due to the loss of parallel pathways, are essential to tumor cells. In Chapter 9, Dr. McManus and colleagues (University of Manitoba, Canada) utilize colorectal cancer as the framework within which to introduce the concept of synthetic lethality and review the recent therapeutic advances gained through the targeting of deficient DNA repair pathways. Finally, a significant hurdle to the success of any systemic therapy, including those that may eventually be used to combat metastatic disease, is the efficient and specific delivery of the therapeutic agent to the target cell or tumor. In Chapter 10, Drs. Hauser-Kawaguchi and Luvt (University of Western Ontario, Canada) review the emerging field of nanomedicine and the utility of nanoparticles for improved cancer imaging and drug delivery.

References

- Department of Health and Human Services (2010) Fact Sheet- Cancer. National Institutes of Health. http://report.nih.gov/nihfactsheets/Pdfs/Cancer(NCI).pdf. Oct. 2010
- [2] Howlader NNA, Krapcho M, Garshell J, Miller D, Altekruse SF, Kosary CL, Yu M, Ruhl J, Tatalovich Z, Mariotto A, Lewis DR, Chen HS, Feuer EJ, Cronin KA (eds) SEER cancer statistics review, 1975–2011. National Cancer Institute. Bethesda, MD. http://seer.cncer.gov/csr/1975_2011, based on November 2013 SEER data submission, posted on the SEER web site. April 2014.
- [3] Christofori G (2006) New signals from the invasive front. Nature 441:444-450
- [4] Hanahan D, Weinberg RA (2011) Hallmarks of cancer: the next generation. Cell 144:646–674
- [5] Floor SL, Dumont JE, Maenhaut C, Raspe E (2012) Hallmarks of cancer: of all cancer cells, all the time? Trends Mol Med 18:509–515
- [6] Holland AJ, Cleveland DW (2009) Boveri revisited: chromosomal instability, aneuploidy and tumorigenesis. Nat Rev Mol Cell Biol 10:478–87
- [7] Hanahan D, Weinberg RA (2000) The hallmarks of cancer. Cell 100:57-70

Contents

1	Cancer Metastasis: Tracking and Attacking a Moving Target David I. Rodenhiser and Ann F. Chambers	1
2	The Generation, Detection, and Prevention of Genomic Instability During Cancer Progression and Metastasis Helen Chen, Christopher Maxwell and Marisa Connell	15
3	DNA Damage Response Pathways in Cancer Predisposition and Progression	39
	Mohamed E. Ashour, Lamia El-Shafie and Sherif F. El-Khamisy	
4	Mathematical Modeling for DNA Repair, Carcinogenesis and Cancer Detection	75
	Jonathan Tang, Walter Georgescu, Thomas Deschamps, Steven M. Yannone and Sylvain V. Costes	
5	Animal Models of Metastasis Dawn R Cochrane, Dong Lin, Graham Dellaire, Elizabeth C Halvorsen, Jason N Berman, Yuzhou Wang, David G Huntsman and Kevin L Bennewith	95
6	Microenvironmental Control of Metastatic Progression Calvin D. Roskelley	125
7	Mechanotransduction, Metastasis and Genomic Instability Allison K. Simi, Alexandra S. Piotrowski and Celeste M. Nelson	139
8	Immunomodulation and Genomic Instability Gregor Reid	159

9	Synthetic Genetic Approaches in Colorectal Cancer: Exploiting and Targeting Genome Instability Babu V. Sajesh, Amy L. Cisyk and Kirk J. McManus	179
10	Nanomedicine—Nanoparticles in Cancer Imaging and Therapy Alexandra M. N. Hauser-Kawaguchi and Leonard G. Luyt	205
Ind	ex	245

Chapter 1 Cancer Metastasis: Tracking and Attacking a Moving Target

David I. Rodenhiser and Ann F. Chambers

Abstract The effective treatment of metastatic cancer is complicated by both the diverse set of dysregulated molecular pathways contributing to cancer progression and the challenge of aiming clinical therapies at a seemingly unpredictable moving target. From an evolutionary perspective, metastasis can be considered as a process during which novel cell populations are generated that can exploit the unique tissue environments they encounter at a secondary tissue site and in response to treatment. In this review, we explore metastasis as a consequence of evolution on the scale of tumor cells within the individual patient. The survivability of any individual cancer cell, and as a consequence, the success of any broad-based or targeted therapy to treat that patient, may best be understood in terms of selective advantage and phenotypic changes resulting from genomic drift among cells from the original tumor. These drivers of evolution can generate successful metastatic cells that either survive as dormant cells, or thrive as secondary tumors during the time course of the disease. The metastatic target is thus dynamic, requiring a dynamic approach to treatment. Here we will discuss the growing information about heterogeneity and evolution of metastatic cell populations, and how this information impacts on treatment strategies that will be needed to combat metastatic disease.

A. F. Chambers Department of Pathology, Western University, London, ON, Canada e-mail: ann.chambers@Lhsc.on.ca

Department of Medical Biophysics, Western University, London, ON, Canada

D. I. Rodenhiser (🖂) · A. F. Chambers

Department of Oncology, Western University, London, ON, Canada e-mail: drodenhi@uwo.ca

London Regional Cancer Program, 790 Commissioners Road East, London, ON N6A 4L6, Canada

D. I. Rodenhiser Department of Biochemistry, Western University, London, ON, Canada

Department of Paediatrics, Western University, London, ON, Canada

Children's Health Research Institute, Lawson Research Institute, London, ON Canada

[©] Springer International Publishing Switzerland 2015 C. Maxwell, C. Roskelley (eds.), *Genomic Instability and Cancer Metastasis*, Cancer Metastasis - Biology and Treatment 20, DOI 10.1007/978-3-319-12136-9_1

Keywords Metastasis \cdot Tumor heterogeneity \cdot Darwinian evolution \cdot Clonal selection

Introduction

While the past 30 years of the genomics revolution have dramatically advanced our knowledge of the molecular mechanisms of cancer, successful treatment of metastatic disease has been limited. This is despite the development of landmark synthetic approaches such as the Hallmarks of Cancer [1, 2] that have provided the conceptual framework within which to integrate the complex characteristics of cancer causation and the potential targets for novel treatment strategies. All cancers can be linked together mechanistically as involving interconnected pathways that can be independently (and simultaneously) activated (or disrupted). Paradoxically, the common, acquired molecular variants that are the genesis of these altered hallmarks also define the uniqueness of cancer phenotypes among tumor types as well as among the individuals within whom these cancers develop and thrive.

Notwithstanding these advances in knowledge, and the specific successes made in understanding, diagnosing and treating many primary cancers, two critical roadblocks exist for curing these diseases, in general, and metastatic cancer, in particular. The first involves the cellular and genomic complexity and plasticity of heterogeneous cancer cell populations that exist as dynamic and unstable entities that evolve over time. Second, the timing, rate and sites of these evolutionary processes (within the originating tissue or at a distant secondary site) are dependent on the genomic makeup of that individual cell and the selective pressures that are placed on it. This level of complexity also changes temporally, as cancer cells acquire the ability to metastasize, with new selection pressures yielding novel, more unstable subclones of metastatic cells that can be selected for by that new environment. Unfortunately, the inaccessibility of these dispersed metastatic cells, as either camouflaged micrometastases or dormant cells, makes detection and treatment of micrometastatic disease nearly impossible. This lag period between dispersal and detection of rogue cancer cells permits the development of heterogeneity to go ahead unnoticed, with the genesis of novel clones that differ from the parent tumor and are better suited to survival in the face of new microenvironmental stresses. This paradigm of clonal evolution in the metastatic context offers both challenges and opportunities to effectively detect, treat, control and/or cure metastatic disease.

Introduction to Metastasis—The Clinical Problem

Cancer that is detected early, before it has spread, is more readily treated with success, when compared to cancer that has already metastasized at the time of diagnosis. Metastatic cancer is generally not considered to be curable, although long-term

survival with metastatic disease is increasingly feasible. However, even cancer that is detected "early" can subsequently recur. Once it has metastasized, it is then more difficult to treat with success. Thus, for example, many patients with early breast cancer are treated first with local therapies (surgery, radiation), in some cases followed by often-aggressive adjuvant therapy designed to treat presumed, but clinically undetected, micrometastatic disease, to attempt to prevent metastatic recurrence. The decision to use adjuvant therapy is generally population-based, with groups of patients that are believed to be at higher risk for subsequent recurrence being more likely to be offered (and benefit from) adjuvant therapy. Recent and ongoing efforts at molecular profiling of groups of patients is leading to gene signatures that are both associated with poor vs. good outcome (suggesting the degree of potential benefit from aggressive initial therapy), as well as signatures that are predictive for responses to specific therapies (suggesting which patients should vs. should not be treated with specific therapies) (reviews, [3–6]). In spite of this improved ability to assess the potential for recurrence, as well as likelihood of response to given therapies, these therapy decisions are imprecise, such that many individual patients who likely did not need this treatment may be over-treated, while other patients, with "favorable" tumors, are not treated with adjuvant therapy but may subsequently relapse. Thus, our understanding of the risk of metastatic recurrence in individual patients is inexact and requires information specific to the genotype of the patient's primary tumor as well as the generation of genetic roadmaps laying out the likely paths of clonal development that lead to the genesis of novel metastatic cells in their relevant environmental contexts.

The logistical steps undertaken in cancer metastasis have been well established from extensive data gleaned from experimental studies (reviews, [7, 8]). These steps include the acquisition of a locally invasive phenotype, entry into blood or lymphatic vessels in the primary tumor, escape into these circulatory systems, systemic delivery to new organs, extravasation from these vasculature channels, and re-initiation of growth of the tumor in these new sites. However, many of the steps in metastasis likely have occurred prior to our current ability to diagnose cancer [9], suggesting that the latter steps of the process (primarily growth in ectopic, secondary organs) may be the more appropriate targets for therapeutic intervention [10, 11].

Coupled with the challenges of deciding on initial therapy for localized disease, it is becoming increasingly recognized that some cancers can recur years or decades after apparently successful primary therapy, due to the process of tumor dormancy (reviews, [7, 12–14]). Much ongoing work is aimed at clarifying mechanisms that regulate entry into tumor dormancy, as well as cessation of dormancy and re-initiation of growth of dormant cancer. However, considerable clinical challenges still remain in determining which cancers are likely to enter into clinical dormancy, and what factors (inherent to the tumor cells, the host or specific organ microenvironments, or modifiable lifestyle factors), can lead to release of growth constraints on dormant cancer (reviews, [14–17]). The relationship between cancer dormancy and the evolutionary nature of cancer progression discussed here remains poorly understood. Dormant cells maintain a degree of normalcy in that they possess a benign phenotype programmed by a molecular signature that lacks the final genetic aberrations that tip those cells towards becoming a novel metastatic cancer. Furthermore, those molecular changes that have occurred reflect the constraints of the new environment in which they populate with subsequent changes necessary to permit release from dormancy at that secondary site. However, the fact that cancer can be maintained in a functionally dormant state, often for many years, suggests that understanding the inherent regulatory mechanisms within these dormant cells could offer new therapeutic strategies to maintain dormant cancer in a chronic, indolent state, in spite of the evolutionary forces that are inherent to cancer.

We have learned a great deal over the past decades about molecular aspects of cancer biology. This information has been translated into new medicines targeted to the molecular defects that have been identified and these agents have shown significant clinical benefit, with some having reducing toxicity compared to earlier therapies (reviews, [8, 18]). Much of this success has come primarily for patients with cancer that is localized at the time of diagnosis. Essential to these successes have been technologies that inform early diagnosis. Often, these patients may be treated aggressively, with the hope of totally eradicating the tumor, first by local therapy followed by adjuvant treatment to eliminate presumed but undetected micrometastatic disease [19]. This approach has led to apparent survival benefits for many patients with localized disease. Biologically, it is recognized that some patients may be at risk for very late recurrences, and may benefit from long-term therapy. This has been shown particularly in hormone responsive breast cancer, where patients have been shown to benefit from long-term endocrine therapy, which can have low enough toxicities to warrant long-term use, although the toxicities associated with treating many patients to benefit a sub-population are still of concern (reviews, [13, 20, 21]).

In contrast to cancer that is localized to the primary site, metastatic cancer is often treated quite differently, with less aggressive, sequential use of therapies predicted to have some effect, at least transiently, against a given patient's tumor [22]. This strategy is based on the current recognition that cure of metastatic cancer may not be possible, while prolongation of survival may be achievable. Indeed, some progress has been made in extending life for patients with metastatic disease. For lung, breast, prostate and colorectal cancers, for example, progress has been made in decreasing cancer-specific mortality overall [23], although improved survival for patients with metastatic disease has been only "modest" [24].

In spite of these limited gains, metastatic disease is still regarded as ultimately incurable. Such current paradigms of treatment of metastatic disease are not adequate in successfully tackling this disease. In spite of the significant progress made in understanding molecular and genetic factors that play a role in many different cancer types, we still have major unmet medical needs in treating metastatic cancer, preventing metastatic recurrence in successfully treated local disease and delaying metastatic recurrence. These challenges will require new ways of thinking about the nature of metastatic disease and new therapeutic strategies to counter it.

Clonal Heterogeneity and Evolution as a Hallmark of Cancer and Metastasis

Until recently, the working paradigm of cancer development has been based on the idea that cancer can progress linearly, from benign, early cancers, through increasingly aggressive, locally invasive disease, culminating in frank metastatic disease. This progression was initially noted at the histological level, which still forms the basis of the initial assessment of a tumor and its appropriate therapy. Histological progression is increasingly associated with defined molecular changes that functionally underpin clinical progression, following the paradigm set forth for colorectal cancer by Vogelstein and colleagues in the 1990's [25]. These molecular changes are also used clinically to target appropriate therapies to individual patients, when such targeted therapies are available. The potential drivers of this progression, as outlined in the Hallmarks of Cancer [1, 2], are complex with respect to the cellular pathways upon which they impact, but ultimately these genomic changes result in the generation of diversity and selection for increasingly aggressive and treatment-resistant phenotypes.

One significant conceptual advance towards a better understanding of metastasis is to consider the inherent nature of metastatic disease as a persistent, evolutionary process, possessing the consequences and challenges of successfully treatment of disease that is, by its very nature, temporally and spatially dynamic. Such a model for clonal evolution of a cancer cell population is modeled in the context of Darwinian natural selection, whereby the cancer cells represent somatic cell species that adapt to changing environment. This perspective of an evolutionary process underpinning cancer development was first put forth nearly 40 years ago. In his groundbreaking paper from 1976, Peter Nowell proposed that tumor progression results from acquired genetic variability within an original clone of cancer cells, allowing the sequential selection of more aggressive sublines [26]. He proposed that carcinogen-induced changes in a normal progenitor cell produce daughter cells with growth advantages initiating the process of clonal expansion. In this context, clonal expansion within tumors is not necessarily linear, but rather is branched (Fig. 1.1). Multiple competing subclones of tumor cells are generated that differ cytogenetically, genomically and ultimately differ at the transcriptome level. Genetic instability in these daughter cells leads to the further generation of clonal variants: some of which die, while others establish themselves as subpopulations that are suited to their particular microenvironment. The process repeats itself, as new clones are generated from the surviving progenitors and are selected for through interactions with the host microenvironment. Inherent within this selected, viable cell population are cells with silent genetic variants, and permissive epigenetic changes, which could be selected for at a later time, when the cells are exposed to a newer, more advantageous environment. Recently, Nik-Zainal and colleagues have reconstructed the genomic evolution of 21 breast cancer genomes and have shown that subclonal diversification is prominent, with most mutations being found in just a fraction of tumor cells [27]. Every tumor has a dominant subclonal lineage, representing more



Fig. 1.1 Model of clonal evolution. Cancer cell clones arise and are selected through interactions between the cell's genotype/ phenotype and the environmental pressures impacting on those cells. These selection pressures may be diverse, changing the microenvironment around the cells, altering gene expression, involving agents that damage DNA (generating genomic instability) and may include the chemotherapies that eliminate some cells while letting other clonal populations to expand. Cell colors infer to populations of cells with distinct genomic characteristics. 'X' indicates dead cells. *Grey* boxes (*Niches*) indicate individual environmental conditions within which clones can survive, including as metastatic cells. Some clones (in *yellow*) may be able to fit and evolve in multiple niches. (Figure based on Nowell [26])

than half of the tumor cells, although minimal expansion of these subclones occurs until many hundreds to thousands of mutations have accumulated. This invokes a model of long-lived, but sparse, lineages of cells passively accumulating mutations until poised for tumor dominance. Expansion of the dominant subclone to a detectable tumor mass may therefore represent the final rate-limiting step in breast cancer's development, triggering diagnosis.

Key features of tumor progression, which were noted early on, are heterogeneity within, and between tumors, and the evolution of this heterogeneity over time. Initially, it was considered that a population of cancer cells could include stable phenotypes that would endow some cells with increased ability to metastasize relative to other cells [28]. However, work in the 1980's introduced the idea of 'dynamic heterogeneity', i.e. the idea that variations within a cancer population might themselves be unstable [29–33]. By applying Luria-Delbruck fluctuation analysis techniques to expanding populations of cancer cell lines, these studies instead suggested that a key feature of highly aggressive cancer cell populations was a high degree of phenotypic 'plasticity', when compared to less aggressive populations. This plasticity would endow cancer cells with the adaptability that would favor survival and proliferation in new microenvironments, such as found in ectopic, metastatic sites. Highly 'successful', aggressive cancer cell populations thus may have an increased capacity to generate transient variants that are able to survive and form metastatic colonies. This 'plasticity' is a population-based phenomenon, giving rise perhaps stochastically to individual cells, at any moment in time (and position in the body), that are able to thrive as metastasis-initiating cells. Recently, Meacham and Morrison have discussed the idea of tumor cell plasticity in the context of the cancer stem cell concept, supporting the idea that plasticity of cancer cell populations could support progression to more aggressive states [34].

One might ask why cancer occurs at all, given the cellular surveillance and repair pathways tasked with maintaining genomic and cellular integrity. In their review on clonal evolution in cancer, Greaves and Maley have posited that cells in tissue ecosystem habitats have evolved over millennia to ensure multicellular function of the collective cells comprising the individual while restraining clonal expansion of renegade cells [35]. Thus, a fine balance is struck such that mechanisms of self-renewal, proliferation and growth are regulated within limits that permit their survival, particularly in response to environmental change or damage, without 'excess' leeway in these responses that could tip survival in favor of genetically damaged, rogue cells that can proliferate uncontrollably and put the multicellular organism at risk. As a consequence, extensive and interconnected cellular pathways regulate the cell cycle, proliferation, DNA repair, senescence and differentiation of cells, in an effort to maintain the functionality of the multicellular cell collective, while weeding out any rogue cells that begin to deviate from their appropriate programming. Clinically benign cancer cells that do come into existence have gained and maintained (epi) genetic changes providing a selective growth advantage for those individual cells so that they may eventually expand into multiple clones that co-populate the primary tumor site. It is the exposure to cues from a permissive microenvironment acting on these variant cells within that population that eventually leads to the expansion of a cancer cell clone that is 'successful' in that context of that novel environment.

The challenges of treating metastatic disease are due in good measure to this heterogeneity described above. Recent evidence has suggested that such intratumoral heterogeneity occurs both in the primary tumor and within the metastases themselves, hindering personalized-medicine strategies that depend on results from single tumor-biopsy samples. The literature provides evidence of clonal heterogeneity, genomic diversity and in some cases, the emergence of treatment-resistant subclones within primary tumors and metastases. For example, tumor specimens analyzed from patients with high-grade serous ovarian cancer (HGSC), exhibited widespread intratumoral variation in mutation, copy number and gene expression profiles, with key driver alterations in genes present in only a subset of samples [36]. Furthermore, reconstruction of evolutionary histories indicated that diversity may arise at early stages of tumorigenesis. Similarly, primary triple-negative breast cancers (TNBCs) have been shown to exhibit a wide and continuous spectrum of genomic evolution. While certain somatic mutations (TP53, PIK3CA and PTEN)

seem to be clonally dominant compared to other genes, in some tumors their clonal frequencies are incompatible with founder status [37]. This same group showed that single nucleotide mutational heterogeneity occurred in ER (a)-positive metastatic lobular breast cancer, with multiple mutations identified in metastases that were not identified in the DNA from the primary tumor of the same patient, which arose 9 years earlier [38]. In addition, Yachida and colleagues have shown that clonal heterogeneity contributes to pancreatic cancer progression, in that distant metastases arose from clones that were genetically distinct from the parental clones, yet these metastatic subclones were also present within the primary carcinoma [39]. This suggests that from the mixture of unique but related subclones within a tumor, some clones can preferentially expand and seed future metastatic populations.

In their study on primary and metastatic renal cell carcinomas, Gerlinger and colleagues integrated data from several genomic platforms to show branched evolutionary tumor growth across multiple regions with primary tumors and associated with their metastases [40]. Mutational intratumor heterogeneity was seen for multiple tumor-suppressor genes, with several genes (SETD2, PTEN, and KDM5C) undergoing multiple distinct and spatially separated inactivating mutations within a single tumor. These genomic changes suggested that convergent phenotypic evolution had occurred. Furthermore, a phylogenetic tree could be generated that revealed that metastatic tumors were derived from one of two subclone branches within one sector of the primary tumor, while the other branch diversified into other primary tumor regions. The observation that mutations shared with metastatic sites were detected at higher frequencies than were mutations shared with other primary-tumor regions, has implications for diagnosis and treatment, in that single tumor-biopsy samples likely underestimate the tumor genomics landscape.

Translating Heterogeneity into the Clinic: Tracking and Attacking a Moving Target

Significant gains in knowledge have been made regarding the molecular causes of cancer and in identifying potential targets to attack that are clinically relevant to finding a cure [1, 2]. Yet, as previously described, any overall successes have been muted by the fundamental nature of cancer as a highly mutable set of diseases that are complex intra-tumorally, within a patient (and among patients) and temporally, as tumors evolve and as metastases develop. Furthermore, patients can differentially respond to treatments (based on their cancer genotypes), and likely will experience reoccurrence as new cancer cell subclones repopulate tissue niches abandoned by clones that have succumbed to therapies, or find new previously un-hospitable niches. How then, to outsmart an adversary that constantly camouflages itself and changes when detected?

One strategy is to gather as much information as possible about the tumor prior to treatment. To date, most treatments have failed with respect to long term survival because in many cases the diagnostic criteria have been simplistic (unimodal; i.e. ER status, HER2 status, recorded as 'positive' or 'negative', etc.) thereby failing to accurately characterize a tumor, its inherent heterogeneity and/or its subtypes and somatic driver mutations. Confounding the situation, treatment decisions in the metastatic setting are often based on features that were noted in the primary tumor, although there is a growing recognition of the importance of biopsy and characterizing of metastases, when feasible (e.g., [41, 42]). To this end, attempts to define cancer subgroups and their molecular drivers have recently been published, which in many cases involve generating integrated views of the genome and transcriptome from representative numbers of patients [43-46]. For example, Yachida and colleagues showed in pancreatic cancer that the number of intragenic mutations in KRAS, CDKN2A and TP53 and immunolabeled for CDKN2A, TP53 and SMAD4 protein products in each carcinoma were correlated to clinicopathologic feature [39]. Carcinomas with only one to two driver alterations were enriched for those patients with the longest survival [39]. As well, Curtis and colleagues undertook an integrated analysis of copy number and gene expression in 2000 primary breast tumors and provided a novel molecular stratification of the breast cancer population [47]. These ten integrative subgroups overlapped somewhat with existing PAM50 subgroups and also had distinct clinical outcomes, particular in the context of long term survival. However, their ten subgroups still maintained certain degrees of heterogeneity within their tumor classification.

The keys to these studies have been the use and integration of multiple, high resolution platforms that include but are not limited to exome sequencing, genomic deep sequencing, copy number analysis and gene expression analyses [43]. Furthermore, there is the necessity to undertake multi-regional analysis to identify spatially distinct tumor regions [40]. Correlating spatially and temporally separated tumor specimens within and across patient samples can map tumor diversity within patients, define causative driver mutations, expose potential evolutionary trajectories prior to treatment and better inform personalized therapies in those patients throughout the treatment course [36]. As well, the use and integration of newer technologies to map epigenetic changes at the microRNA and DNA methylation levels will further inform therapeutic choices [48, 49].

Ultimately, two scenarios for success are possible. First, the full weight of these technologies may identify essential driver mutations that may be critical to the survival of the cancer and, once therapeutically targeted, could lead to clinical success with the obliteration of that disease (possessing that unique genomic signature) in that individual. More likely, however, is that such a driver mutation may not be essential to cells in all cancer clones, leaving the possibility of relapse after aggressive treatment due to expansion of a new cancer from the remaining clonal populations in the primary tumor or from until-now dormant metastases (Fig. 1.2). Hence, when necessary, the concept of 'cure' must include developing strategies to turn cancer into a chronic disease. This will necessitate several steps. First, early diagnosis is necessary, since by definition this suggests that clonal heterogeneity is less extensive and the emergence of metastatic cells has been minimized. In addition, early diagnosis should be augmented by manipulating the patient's environment by minimizing environmental exposures (i.e. UV, ionizing radiation, carcinogen exposure)



Fig. 1.2 Clinical relevance of clonal heterogeneity and tumor progression. Here the branching evolution model of cancer cells is overlaid with a clinical context. This model implies that at different times during the course of their disease, patients have different cancers at the genomic level that must be targeted appropriately. Cancer cell clones expand from a progenitor (*Steps 1, 2*), eventually becoming detectable clinically (*Step 3*), and are assessed and treated on the basis of characteristics related to the major clone in the tumor at that point in time. Tumor response to treatment may suggest treatment success, however surviving cells (*Step 4*) may eventually expand until cancer relapse is confirmed. The new, genomically unique tumors require different treatment strategies (*Step 5; different colored lightning bolt*). This process may repeat itself (*Steps 6, 7, 8*) as new cancer clones arise

that may contribute to genomic instability and further drive clonal evolution in preexisting dormant or residual cancer cells, post-treatment. Second, continual surveillance is essential. Promising new methods involving the isolation of circulating tumor cells can provide a real-time liquid biopsy that allows assessment of genetic drift, molecular characterization and identification of actionable genomic targets, within the total population of cancer cells present in the body [50]. In addition, the presence of cell-free tumor-specific DNA in the peripheral blood from patients with metastatic disease can also be used to identify alterations in tumor burden and to monitor therapy response [51].

To conclude, our research efforts to date have led to great strides being made in acquiring mechanistic knowledge related to cancer biology, the synthesis of critical conceptual paradigms such as the cancer hallmarks, and the application of new genomic technologies that can be applied to the growing field of personalized medicine. The role of clonal heterogeneity in cancer offers a conceptual framework within which to consider cancer as an ever-evolving entity requiring integrative approaches that map cancer evolution both spatially and temporally. In some cases, these integrative approaches may identity novel pathways and hallmarks for which off-the-shelf agents do not exist, opening new treatment paradigms for investigation and drug development. Ironically, the present challenges we experience in understanding clonal heterogeneity as a mechanism underpinning metastatic progression may ultimately offer the best opportunities for long term successes in treating cancer patients. Translation of this knowledge into the clinic continues, but as is the case in learning any new language, we are still developing our vocabulary.

Acknowledgements AFC is Canada Research Chair in Oncology, supported by the Canada Research Chairs Program. AFC is Director of the Pamela Greenaway-Kohlmeier Translational Breast Cancer Research Unit, supported in part by the Breast Cancer Society of Canada.

References

- 1. Hanahan D, Weinberg RA (2000) The hallmarks of cancer. Cell 100(1):57-70
- Hanahan D, Weinberg RA (2011) Hallmarks of cancer: the next generation. Cell 144(5):646– 674
- 3. Rodenhiser DI, Andrews JD, Vandenberg TA, Chambers AF (2011) Gene signatures of breast cancer progression and metastasis. Breast Cancer Res 13(1):201
- Duffy MJ, O'Donovan N, Crown J (2011) Use of molecular markers for predicting therapy response in cancer patients. Cancer Treat Rev 37(2):151–159
- Kittaneh M, Montero AJ, Gluck S (2013) Molecular profiling for breast cancer: a comprehensive review. Biomark Cancer 5:61–70
- Sleijfer S, Bogaerts J, Siu LL (2013) Designing transformative clinical trials in the cancer genome era. J Clin Oncol 31(15):1834–1841
- Chambers AF, Groom AC, MacDonald IC (2002) Dissemination and growth of cancer cells in metastatic sites. Nat Rev Cancer 2(8):563–572
- Eccles SA, Welch DR (2007) Metastasis: recent discoveries and novel treatment strategies. Lancet 369 (9574):1742–1757
- Husemann Y, Geigl JB, Schubert F, Musiani P, Meyer M, Burghart E, Forni G, Eils R, Fehm T, Riethmuller G, Klein CA (2008) Systemic spread is an early step in breast cancer. Cancer Cell 13(1):58–68
- Hedley BD, Winquist E, Chambers AF (2004) Therapeutic targets for antimetastatic therapy. Expert Opin Ther Targets 8(6):527–536
- 11. Mina LA, Sledge GW Jr (2011) Rethinking the metastatic cascade as a therapeutic target. Nat Rev Clin Oncol 8(6):325–332
- Aguirre-Ghiso JA (2007) Models, mechanisms and clinical evidence for cancer dormancy. Nat Rev Cancer 7(11):834–846
- Goss PE, Chambers AF (2010) Does tumour dormancy offer a therapeutic target? Nat Rev Cancer 10(12):871–877
- Uhr JW, Pantel K (2011) Controversies in clinical cancer dormancy. Proc Natl Acad Sci U S A 108(30):12396–12400
- Bragado P, Sosa MS, Keely P, Condeelis J, Aguirre-Ghiso JA (2012) Microenvironments dictating tumor cell dormancy. Recent Results Cancer Res 195:25–39
- Sosa MS, Bragado P, Debnath J, Aguirre-Ghiso JA (2013) Regulation of tumor cell dormancy by tissue microenvironments and autophagy. Adv Exp Med Biol 734:73–89
- 17. Barkan D, Green JE, Chambers AF (2010) Extracellular matrix: a gatekeeper in the transition from dormancy to metastatic growth. Eur J Cancer 46(7):1181–1188
- 18. Huang M, Shen A, Ding J, Geng M (2014) Molecularly targeted cancer therapy: some lessons from the past decade. Trends Pharmacol Sci 35(1):41–50

- Goldhirsch A, Winer EP, Coates AS, Gelber RD, Piccart-Gebhart M, Thurlimann B, Senn HJ (2013) Personalizing the treatment of women with early breast cancer: highlights of the St Gallen international expert consensus on the primary therapy of early breast cancer 2013. Ann oncol 24(9):2206–2223
- Strasser-Weippl K, Badovinac-Crnjevic T, Fan L, Goss PE (2013) Extended adjuvant endocrine therapy in hormone-receptor positive breast cancer. Breast 22 Suppl 2:S171–S175
- Jankowitz RC, McGuire KP, Davidson NE (2013) Optimal systemic therapy for premenopausal women with hormone receptor-positive breast cancer. Breast 22 Suppl 2:S165–S170
- 22. Morris PG, McArthur HL, Hudis CA (2009) Therapeutic options for metastatic breast cancer. Expert Opin Pharmacother 10(6):967–981
- Aizer AA, Wilhite TJ, Chen MH, Graham PL, Choueiri TK, Hoffman KE, Martin NE, Trinh QD, Hu JC, Nguyen PL (2014) Lack of reduction in racial disparities in cancer-specific mortality over a 20-year period. Cancer 120(10):1532–1539
- Dawood S, Broglio K, Gonzalez-Angulo AM, Buzdar AU, Hortobagyi GN, Giordano SH (2008) Trends in survival over the past two decades among white and black patients with newly diagnosed stage IV breast cancer. J Clin Oncol 26(30):4891–4898
- Kinzler KW, Vogelstein B (1996) Lessons from hereditary colorectal cancer. Cell 87(2):159– 170
- 26. Nowell PC (1976) The clonal evolution of tumor cell populations. Science 194(4260):23-28
- 27. Nik-Zainal S, Van Loo P, Wedge DC, Alexandrov LB, Greenman CD, Lau KW, Raine K, Jones D, Marshall J, Ramakrishna M, Shlien A, Cooke SL, Hinton J, Menzies A, Stebbings LA, Leroy C, Jia M, Rance R, Mudie LJ, Gamble SJ, Stephens PJ, McLaren S, Tarpey PS, Papaemmanuil E, Davies HR, Varela I, McBride DJ, Bignell GR, Leung K, Butler AP, Teague JW, Martin S, Jonsson G, Mariani O, Boyault S, Miron P, Fatima A, Langerod A, Aparicio SA, Tutt A, Sieuwerts AM, Borg A, Thomas G, Salomon AV, Richardson AL, Borresen-Dale AL, Futreal PA, Stratton MR, Campbell PJ (2012) The life history of 21 breast cancers. Cell 149(5):994–1007
- Fidler IJ (1978) Tumor heterogeneity and the biology of cancer invasion and metastasis. Cancer Res 38(9):2651–2660
- 29. Chambers AF, Hill RP, Ling V (1981) Tumor heterogeneity and stability of the metastatic phenotype of mouse KHT sarcoma cells. Cancer Res 41(4):1368–1372
- Hill RP, Chambers AF, Ling V, Harris JF (1984) Dynamic heterogeneity: rapid generation of metastatic variants in mouse B16 melanoma cells. Science 224(4652):998–1001
- Ling V, Chambers AF, Harris JF, Hill RP (1984) Dynamic heterogeneity and metastasis. J Cell Physiol Suppl 3:99–103
- 32. Chambers AF, Harris JF, Ling V, Hill RP (1984) Rapid phenotype variation in cells derived from lung metastases of KHT fibrosarcoma. Invasion Metastasis 4(4):225–237
- Ling V, Chambers AF, Harris JF, Hill RP (1985) Quantitative genetic analysis of tumor progression. Cancer Metastasis Rev 4(2):173–192
- Meacham CE, Morrison SJ (2013) Tumour heterogeneity and cancer cell plasticity. Nature 501(7467):328–337
- 35. Greaves M, Maley CC (2012) Clonal evolution in cancer. Nature 481(7381):306-313
- 36. Bashashati A, Ha G, Tone A, Ding J, Prentice LM, Roth A, Rosner J, Shumansky K, Kalloger S, Senz J, Yang W, McConechy M, Melnyk N, Anglesio M, Luk MT, Tse K, Zeng T, Moore R, Zhao Y, Marra MA, Gilks B, Yip S, Huntsman DG, McAlpine JN, Shah SP (2013) Distinct evolutionary trajectories of primary high-grade serous ovarian cancers revealed through spatial mutational profiling. J Pathol 231(1):21–34
- 37. Shah SP, Roth A, Goya R, Oloumi A, Ha G, Zhao Y, Turashvili G, Ding J, Tse K, Haffari G, Bashashati A, Prentice LM, Khattra J, Burleigh A, Yap D, Bernard V, McPherson A, Shumansky K, Crisan A, Giuliany R, Heravi-Moussavi A, Rosner J, Lai D, Birol I, Varhol R, Tam A, Dhalla N, Zeng T, Ma K, Chan SK, Griffith M, Moradian A, Cheng SW, Morin GB, Watson P, Gelmon K, Chia S, Chin SF, Curtis C, Rueda OM, Pharoah PD, Damaraju S, Mackey J, Hoon K, Harkins T, Tadigotla V, Sigaroudinia M, Gascard P, Tlsty T, Costello JF, Meyer IM, Eaves CJ, Wasserman WW, Jones S, Huntsman D, Hirst M, Caldas C, Marra MA, Aparicio

S (2012) The clonal and mutational evolution spectrum of primary triple-negative breast cancers. Nature 486(7403):395–399

- 38. Shah SP, Morin RD, Khattra J, Prentice L, Pugh T, Burleigh A, Delaney A, Gelmon K, Guliany R, Senz J, Steidl C, Holt RA, Jones S, Sun M, Leung G, Moore R, Severson T, Taylor GA, Teschendorff AE, Tse K, Turashvili G, Varhol R, Warren RL, Watson P, Zhao Y, Caldas C, Huntsman D, Hirst M, Marra MA, Aparicio S (2009) Mutational evolution in a lobular breast tumour profiled at single nucleotide resolution. Nature 461(7265):809–813
- 39. Yachida S, White CM, Naito Y, Zhong Y, Brosnan JA, Macgregor-Das AM, Morgan RA, Saunders T, Laheru DA, Herman JM, Hruban RH, Klein AP, Jones S, Velculescu V, Wolfgang CL, Iacobuzio-Donahue CA (2012) Clinical significance of the genetic landscape of pancreatic cancer and implications for identification of potential long-term survivors. Clin Cancer Res 18(22):6339–6347
- Gerlinger M, Catto JW, Orntoft TF, Real FX, Zwarthoff EC, Swanton C (2014) Intratumour heterogeneity in urologic cancers: From molecular evidence to clinical implications. Eur Urol 2014 May 2. pii: S0302-2838(14)00395-9. doi: 10.1016/j.eururo.2014.04.014 [Epub ahead of print]
- 41. Stoecklein NH, Klein CA (2010) Genetic disparity between primary tumours, disseminated tumour cells, and manifest metastasis. Int J Cancer 126(3):589–598
- Arslan C, Sari E, Aksoy S, Altundag K (2011) Variation in hormone receptor and HER-2 status between primary and metastatic breast cancer: review of the literature. Expert Opin Ther Targets 15(1):21–30
- Sottoriva A, Spiteri I, Piccirillo SG, Touloumis A, Collins VP, Marioni JC, Curtis C, Watts C, Tavare S (2013) Intratumor heterogeneity in human glioblastoma reflects cancer evolutionary dynamics. Proc Natl Acad Sci U S A 110(10):4009–4014
- 44. Nickerson ML, Im KM, Misner KJ, Tan W, Lou H, Gold B, Wells DW, Bravo HC, Fredrikson KM, Harkins TT, Milos P, Zbar B, Linehan WM, Yeager M, Andresson T, Dean M, Bova GS (2013) Somatic alterations contributing to metastasis of a castration-resistant prostate cancer. Hum Mutat 34(9):1231–1241
- 45. Anaka M, Hudson C, Lo PH, Do H, Caballero OL, Davis ID, Dobrovic A, Cebon J, Behren A (2013) Intratumoral genetic heterogeneity in metastatic melanoma is accompanied by variation in malignant behaviors. BMC Med Genomics 6:40
- 46. Bochtler T, Stolzel F, Heilig CE, Kunz C, Mohr B, Jauch A, Janssen JW, Kramer M, Benner A, Bornhauser M, Ho AD, Ehninger G, Schaich M, Kramer A (2013) Clonal heterogeneity as detected by metaphase karyotyping is an indicator of poor prognosis in acute myeloid leukemia. J Clin Oncol 31(31):3898–3905
- 47. Curtis C, Shah SP, Chin SF, Turashvili G, Rueda OM, Dunning MJ, Speed D, Lynch AG, Samarajiwa S, Yuan Y, Graf S, Ha G, Haffari G, Bashashati A, Russell R, McKinney S, Langerod A, Green A, Provenzano E, Wishart G, Pinder S, Watson P, Markowetz F, Murphy L, Ellis I, Purushotham A, Borresen-Dale AL, Brenton JD, Tavare S, Caldas C, Aparicio S (2012) The genomic and transcriptomic architecture of 2000 breast tumours reveals novel subgroups. Nature 486(7403):346–352
- Dvinge H, Git A, Graf S, Salmon-Divon M, Curtis C, Sottoriva A, Zhao Y, Hirst M, Armisen J, Miska EA, Chin SF, Provenzano E, Turashvili G, Green A, Ellis I, Aparicio S, Caldas C (2013) The shaping and functional consequences of the microRNA landscape in breast cancer. Nature 497(7449):378–382
- Moelans CB, de Groot JS, Pan X, van der Wall E, van Diest PJ (2014) Clonal intratumor heterogeneity of promoter hypermethylation in breast cancer by MS-MLPA. Mod Pathol 27(6):869–874
- Lowes LE, Allan AL (2014) Recent advances in the molecular characterization of circulating tumor cells. Cancers (Basel) 6(1):595–624
- 51. Fackler MJ, Lopez Bujanda Z, Umbricht C, Teo WW, Cho S, Zhang Z, Visvanathan K, Jeter S, Argani P, Wang C, Lyman JP, de Brot M, Ingle JN, Boughey J, McGuire K, King TA, Carey LA, Cope L, Wolff AC, Sukumar S (2014) Novel methylated biomarkers and a robust assay to detect circulating tumor DNA in metastatic breast cancer. Cancer Res 74(8):2160–2170

Chapter 2 The Generation, Detection, and Prevention of Genomic Instability During Cancer Progression and Metastasis

Helen Chen, Christopher Maxwell and Marisa Connell

Abstract Genome stability is tightly regulated through the cell cycle. Aberrations in genome structure and sequence are a hallmark of malignancy and these changes can allow abnormal cells to escape the regulatory mechanisms that would otherwise direct these cells into apoptosis or senescence. When genome instability occurs, it can happen as large or small structural changes in the genome, changes in gene expression, or even changes at the epigenetic level. There are many environmental factors that can induce DNA damage and strain the machinery that is responsible for maintaining genome stability. In some cases, such as UV light or chemical carcinogens, it is possible to avoid these factors and thus reduce the risk of cancer. But, in other instances, hereditary mutations impair the function of genes and their products, which normally protect the stability of the genome. While genomic instability offers selective advantages to the tumor, the tumor-specific loss of these pathways may provide therapeutic opportunities, which could be personalized through knowledge of the specific types of genomic instability that characterize an individual's tumor.

Keywords Genomic instability · Epigenome stability · DNA damage

Abbreviations

- BER Base excision repair
- BFB Break fusion break
- CDK Cyclin dependent kinase
- CGH Comparative genomic hybridization
- CIN Chromosome instability
- CpG C-phosphate-G
- CRC Colorectal cancer

C. Maxwell (\boxtimes) \cdot H. Chen \cdot M. Connell

Department of Pediatrics, Child & Family Research Institute, University of British Columbia, Room 3086, 950 W 28th Ave., Vancouver, BC V5Z 4H4, Canada e-mail: cmaxwell@cfri.ubc.ca

[©] Springer International Publishing Switzerland 2015

C. Maxwell, C. Roskelley (eds.), *Genomic Instability and Cancer Metastasis*, Cancer Metastasis - Biology and Treatment 20, DOI 10.1007/978-3-319-12136-9_2

DDR	DNA damage response
DNA	Deoxynucleic acid
DSB	Double strand break
EMT	Epithelial to mesenchymal transition
HDR	Homology directed repair
LOH	Loss of heterozygosity
MET	Mesenchymal to epithelial transition
MIN or MSI,	Microsatellite instability
MMR	Mismatch repair
mtDNA	Mitochondrial DNA
NER	Nucleotide excision repair
NHEJ	Non-homologous end joining
NIN	Nucleotide instability
PCR	Polymerase chain reaction
SAC	Spindle assembly checkpoint
UV	Ultraviolet

Introduction

Cancer is a genetic disease. Tumor cells contain multiple mutations, ranging from single nucleotide changes to large-scale structural and numerical alterations of chromosomes. Collectively, these mutations are referred to as genome instability, which may be predisposed through inherited, germ-line mutations, as in the case of p53, BRCA1 and BRCA2, or acquired as somatic mutations throughout an individual's lifetime. Among the currently known cancer causing genes, 82 are associated with germ-line mutations, 474 are associated with somatic mutations and 513 are associated with chromosomal alterations [1]. These mutations promote cell growth, inhibit cell death and are propagated through accelerated cell division. Moreover, specific genetic alterations among cancer cells can give rise to sub-populations of cells with growth advantages, as in the case of a metastatic cancer cell or a cancer stem cell. These aggressive cells often arise during later stages of tumorigenesis, and their genome may significantly differ from the initial tumor population. However, regardless of how or when these genetic mutations are acquired, alteration of critical genes can change a once-normal cell into a cancerous cell that divides uncontrollably, and additional genomic changes may allow them to gain further growth advantages and eventually spread throughout the body. Thus, the molecular process behind tumorigenesis can be viewed as the accumulation and evolution of genomic alterations.

During the life cycle of a normal cell, genome maintenance is tightly regulated to prevent neoplastic transformation or tumorigenesis. There are various caretaker processes throughout the cell cycle that strategically minimize genome instability, including high-fidelity DNA replication, accurate chromosome segregation and cell cycle checkpoints. In addition, other mechanisms, such as DNA damage response (DDR) pathways, telomere stability and epigenome maintenance, prevent non-mutational genetic changes. Defects in these gene products and processes will compromise their ability to monitor genomic alterations and execute the appropriate damage responses, which include repair, induction of senescence or apoptosis. In most cases, alteration of a particular gene product is not sufficient for transformation but may fast-track a subset of pre-cancer cells to acquire additional genomic changes that allow them to gain further growth advantages. Accumulation rates and the types of genomic alterations may vary in distinct subsets of cancer cells, contributing to the heterogeneity observed in cancer.

The notion of genomic instability as a hallmark of malignancy has intrigued cancer biologists for over a century, from Theodore Boveri's hypothesis that highlighted chromosomal aberrations as the cause of cancer [2, 3], to the discovery of the "Philadelphia chromosome" that leads to the activation of the *Abl* gene [4] and then the identification of the first familial breast cancer susceptibility gene, *BRCA1* [5, 6]. High-resolution arrays, such as comparative genomic hybridization (CGH) and whole-genome sequencing, have identified recurrent alterations as well as genomic heterogeneity within clinically similar cancers. For example, genomic and epigenomic profiling allows for more precise classification of breast cancer sub-types, as well as the prediction of subtype-specific therapeutic targets [7].

However, despite our technological advances, we are still baffled by questions posited following Boveri's observation of genetic imbalances in sea-urchin eggs, such as: (1) how and when does genome instability occur?; (2) how many mutations does it take to cause cancer?; (3) is genome instability the driver for tumorigenesis, or simply a passenger of disease progression?; (4) what role does genome instability play in cancer evolution and metastasis?; and (5) how do we take advantage of this shared trait of all cancer cells to uncover new paradigms for prevention, diagnosis and responsive therapy?

Genomic Instability in Cancer

Genomic instability is a characteristic of all cancers and encompasses a variety of genetic alterations ranging from single nucleotide differences to large-scale changes at the chromosomal levels [8]. Genomic instability can be divided into three categories based on the degree and type of genetic alteration.

a. Nucleotide Instability (NIN)

Nucleotide instability includes base substitutions, deletions and insertions of one or a few nucleotides. These alterations result when errors occur during DNA replication or when the repair machinery malfunctions, such as nucleotide excision repair (NER) and base excision repair (BER) [8]. These alterations can cause dramatic changes to gene structure and expression. For example, missense mutation in the K-*ras* gene occur in over 80% of primary exocrine pancreatic tumors and their corresponding metastases [9]. NIN may also arise in mitochondrial DNA (mtDNA), and instability of mtDNA occurs in a variety of human cancers, including colorectal (CRC), gastric and lung [10].

b. Microsatellite Instability (MIN or MSI)

Microsatellites are short, two to six base-pair simple, or tandem, sequence repeats of DNA located throughout the genome. MIN occurs when the DNA mismatch repair (MMR) system is impaired, which results in the expansion, contraction, deletion and random insertion of microsatellites [11]. The MMR system identifies and binds to the mismatch, and excises the erroneous nucleotide and repairs the mismatch. MIN has been documented in a variety of cancers, including gastric, ovarian, lung, endometrial, and CRC [12–17]. To date, five MIN markers have been recommended by the National Cancer Institute for disease screening in patients susceptible to Lynch syndrome. MIN occurs in approximately 15% of all CRC, which include both hereditary and sporadic forms of CRC, and are associated with a better prognosis than non-MSI tumors [18].

c. Chromosomal Instability (CIN)

CIN is the most prevalent form of genomic instability, observed in over 90% of all malignancies, and is detected throughout the entire neoplastic transformation process, from premalignant lesions to metastatic lesions [19]. For example, chromosome 10 is often lost in glioblastomas, resulting in the inactivation of the tumor-suppressor gene, *PTEN* [20]. CIN refers to alterations of segments of chromosomes, or whole chromosomes, in terms of their structure or number, including amplifications, deletions, translocations, insertions, inversions, loss of heterozygosity (LOH) and homozygous deletions [8]. Change in chromosome numbers is a condition known as aneuploidy, while chromosome translocation involves the fusion of different chromosome [8]. Finally, the ploidy of the entire genome may deviate from the standard 2N complement of chromosomes and give rise to polyploid cells. CIN in tumor cells alters the expression of thousands of genes, which may help to explain why CIN tumors have a poorer prognosis than either MIN or NIN tumors [21].

Telomere Maintenance in Cancer

Telomeres are unique G-rich repetitive sequences (TTAGGG) located at the ends of the eukaryotic chromosomes [22, 23]. Telomeres protect the ends of the chromosomes and preserve their integrity [24]. As telomeres will gradually shorten with

each round of cell division due to chromosome end-processing, telomere maintenance is necessary for continuous cell division and the addition of telomeric repeats must be catalyzed by telomerase reverse transcriptase (hTERT) [25]. In most somatic cells, insufficient telomerase activity will lead to telomere shortening and the induction of cellular senescence [26]. Thus, telomerase inhibition may be a promising target in cancer therapy [27–29].

Inhibition of telomerase activity in a variety of cancer cell lines resulted in accelerated telomere shortening, cell death and differentiation [27–29]. However, due to mutations in the tumor suppressor p53, cancer cells frequently bypass senescence, and continue to divide, which promotes genome instability due to chromosome fusions [24]. Telomere shortening will signal the DDR pathway [30, 31], which may promote CIN, tumor initiation and progression [32, 33]. As an example, telomerase knockout ($mTR^{-/-}$) mice significantly increase the incidence of spontaneous tumor formation (4-6 fold compared to the wild-type population) due to telomere shortening [34]. These $mTR^{-/-}$ tumors have 3–18 fold more chromosome fusions and a two-fold increase in aneuploidy compared to $mTR^{+/+}$ tumors [34]. Conversely, telomerase activity also promotes tumorigenesis [35]. Thus, telomere-associated tumorigenic processes are stage specific; telomere shortening is critical to the accumulation of genetic mutations needed for cancer initiation while, in later stages, telomerase activity promotes cell proliferation necessary for cancer expansion and metastasis.

Epigenome Instability in Cancer

Epigenetics is defined as all heritable changes that may modify gene expression without affecting the primary DNA sequence, such as DNA methylation and chromatin remodeling. DNA methylation and histone modification are the most well understood epigenetic processes, and preservation of these epigenetic markers during cell division is vital for gene regulation. In cancer, epimutations may result in dysregulation of critical genes either independently, or in conjunction with deleterious genetic mutations. Moreover, these epimutations are inherited through clonal expansion, which can promote cancer initiation and progression [36, 37]. Although it is unclear whether these epigenetic alterations are causative or a consequence of tumorigenesis, it is certain that epigenome instability is a prominent feature in cancer.

a. DNA Methylation in Cancer

DNA methylation is a covalent modification where a methyl group is added to the carbon-5 position of cytosine nucleotides followed by a guanine (CpG) via a group of DNA methyltransferase (DNMT) enzymes [38]. DNA methylation can result in gene silencing, and occurs primarily at CpG islands within heterochromatin [38]. An individual cell's DNA methylation pattern is important for its ability to establish

tissue-specific gene expression or maintain pluripotency. Global DNA hypomethylation and site-specific hypermethylation are two key epimutations that occur in cancer [36]. DNA hypomethylation can lead to aberrant overexpression of oncogenes, such as *R-Ras* in gastric cancer, *S-100* in colon cancer [39] and *IGF2* in Wilms' tumor [40]. Hypomethylation of retrotransposons and specific repeat sequences can also result in genomic instability by promoting chromosome rearrangements [41, 42]. Global hypomethylation in many cancers, such as in the breast, brain and cervix, is positively correlated with increased grades of malignancy [43]. On the other hand, hypermethylation contributes to tumorigenesis by silencing the transcription of tumor suppressor genes, such as *Rb*, *BRCA1* and *p16* [36], which may act as a second hit as described by Knudson's hypothesis [44].

b. Histone Modification in Cancer

At the chromatin level, modifications of the four core histones, H2A, H2B, H3 and H4, regulate gene expression. These histone modifications include acetylation, deacetylation, methylation, phosphorylation and ubiquitination [45]. For example, methylation of histone H3 lysine 9 (H3K9), H3K27, H4K20, and H3 arginine 2 (R2) are indicative of heterochromatin formation and transcriptional repression, while acetylation of H3 and H4 promote euchromatin formation and gene transcription [45].

The global loss of H4K16ac and H4K20me3 has been recognized as a hallmark of almost all human cancers [46], while other changes in histone modification are used as prognostic markers [47–51]. These epigenetic marks are catalyzed by various histone-modifying enzymes, such as lysine methyltransferases, arginine methyltransferases, serine-threonine kinases, histone deacetylases and acetyltransferases [45]. Aberrant expression of these enzymes results in changes in histone modification, which can dysregulate subsequent DNA repair, gene transcription and growth-promotion. For example, histone deacetylase 1 (HDAC1) is reduced in gastric cancer, while reduced HDAC5 and HDAC10 associate with poor prognosis in lung cancer [52]. Changes in histone modification patterns can also lead to an overall change in the chromatin structure, which increases the risk of translocation of random transposons and chromosome breakage during mitosis.

c. Nucleosome Remodeling in Cancer

Nucleosomes are the basic unit for DNA packaging, and the architecture of these nucleosomes determines chromatin structure and the accessibility of regulatory DNA sequences to transcription factors [53]. Nucleosome remodeling includes the repositioning of the nucleosome, as well as changes in the content of histone proteins within the nucleosome [53]. In a cancer epigenome, the sliding of pre-existing nucleosomes and incorporation of new nucleosomes, determines chromatin

accessibility and gene expression [53]. For example, in hereditary nonpolyposis colon cancer, three *de novo* nucleosomes are present within the promoter CpG island of the *MLH1* gene, which is a homolog of the E. coli DNA mismatch repair gene [54]. Moreover, substitution of the canonical histone proteins with non-canonical histone proteins within the nucleosome can also influence nucleosome occupancy and gene expression [53, 54].

Crosstalk Between Genomic and Epigenomic Instability

Despite the differences between DNA mutations and epimutations in cancer, their effects on gene regulation are ultimately the same. Indeed, these two processes work symbiotically and synergistically with genetic mutations in epigenetic regulators resulting in epimutations, and vice versa. For example, hypermethylation of key tumor suppressor genes, such as *Rb*, *PTEN*, *BRCA1* and *MLH1*, is a common phenomenon observed in cancers [54, 55]. On the other hand, genetic mutations of key epigenetic modifiers, such as different DNMTs, histone modification proteins and chromatin remodeling proteins, can also impair epigenome stability [56]. Both genomic instability and epigenomic instability are phenomena observed in almost all cancers at every stage of cancer evolution [56, 57]. The crosstalk between these two phenomena adds complexity to cancer biology but also offers potential novel therapeutic targets [58, 59].

Mechanisms for Genome and Epigenome Stability

For a normal cell, the end goal of cell division is to accurately duplicate its genome and distribute its genetic material evenly between the two daughter cells. To maintain genome integrity during proliferation, four major mechanisms are in place: (1) high-fidelity DNA replication during S-phase; (2) accurate chromosome segregation during mitosis; (3) sporadic DDR throughout the cell cycle; and (4) quality control checkpoints regulating cell cycle progression (Fig. 2.1).

a. Error-Free DNA Replication

The duplication of the genome during S-phase is under tight regulation to ensure copy number and temporal (once per cell cycle) control, known as replication licensing [60, 61]. Replication licensing is highly conserved throughout evolution and is regulated by cyclin dependent kinases (CDKs). Replication is initiated with the assembly of a pre-replication complex assembled at the replication origin. Untimely initiation can cause re-replication and aneuploidy, and low replication-initiation density can lead to unfinished replication of the whole genome [62]. During



Fig. 2.1 Mechanisms to prevent genome instability throughout the cell cycle. The cell employs many mechanisms through the cell cycle to prevent genome instability. The important processes that occur during each phase of the cell cycle are highlighted in *black* while cell cycle checkpoints are highlighted in *red*

the extension phase of DNA replication, replication forks often pause and restart to ensure replication completion. Failure to restart these replication forks will result in double-strand breaks, single-strand DNA breaks and hyper-recombination, which all need to be resolved by the DDR pathway before mitotic entry to maintain genome stability [62]. Replication on the lagging strand occurs through the production of multiple short Okazaki fragments, these fragments need to be ligated to maintain the structure of the lagging strand [63]. Moreover, there is an RNA primer and a short DNA segment at the 5' end of each Okazaki fragment, which all needs to be removed prior to ligation of the Okazaki fragment [63]. Failure to remove these elements affects Okazaki fragment maturation, and can result in genomic alterations [63]. El-Khamisy and colleagues (Chapter 3) provide an in-depth discussion of the molecular control of DDR pathways.

b. Bipolar Spindle Assembly and Accurate Chromosome Segregation During Mitosis

After error-free DNA replication during S-phase, genome stability requires accurate chromosome segregation during mitosis. Chromosome segregation is a multi-step process, which requires faithful centrosome duplication, assembly of a bipolar mitotic spindle, proper attachment between the chromosomes and the mitotic spindle, and completion of cytokinesis. Dysregulation of any of these processes may result in aneuploidy.

The centrosome is the primary microtubule organizing center in the eukaryotic cell and is composed of a pair of centrioles surrounded by a cloud of proteins that promote microtubule nucleation and cilia formation [64, 65]. During mitosis, the centrosome directs mitotic spindle formation at the poles, which contributes to faithful chromosome segregation and cleavage furrow formation [65, 66]. Since the mitotic spindle is a bipolar structure, the centrosome must be duplicated only once during every cell cycle and in a semi-conservative manner. Thus, centrosome duplication occurs simultaneously with DNA replication during S-phase, and is tightly regulated.

Centrosome amplification, the presence of more than two centrosomes, is a common characteristic of almost all cancers, which frequently accompanies aberrant chromosome numbers [67]. The presence of supernumerary centrosomes during mitosis is problematic as it can cause multipolar mitoses, chromosome missegregation, cytokinesis failure and subsequent chromosome imbalances that promote tumorigenesis [67]. Due to dysregulation of various tumor suppressors and oncogenes, supernumerary centrosomes can arise from centrosome overduplication, de novo assembly, or previous mitotic failure [64, 67–70]. Centrosome duplication errors result from multiple daughter centrioles forming around a single mother centriole, or multiple rounds of centrosome duplication and paired centriole duplication [64, 67–70]. Extra centrosomes can also be made from *de novo* assembly, where an acentriolar centrosome is made without a pre-existing centriolar centrosome as the template [71–73]. Various mitotic catastrophes can also result in supernumerary centrosomes. For example, loss of mitotic spindle integrity due to inadequate crosslinking by microtubule-associated proteins can cause a centrosome to split with each individual centriole capable of functioning as a spindle pole, or the formation of an acentriolar spindle pole [74-79]; aborted cytokinesis, mitotic slippage and DNA damage are additional mechanisms for cancer cells to obtain extra centrosomes [80].

The presence of supernumerary centrosomes may not be good news even for cancer cells, as catastrophic aneuploidy can result in nonviable daughter cells [81]. However, cancer cells have developed mechanisms that overcome this fate by clustering multiple centrosomes [70, 81–84]. These mechanisms can dampen high level aneuploidy and extreme CIN to avoid programmed cell death [85, 86]. Given that centrosome clustering may be advantageous for the survival of cancer cells, this process may be an attractive and specific therapeutic target [83, 84, 87].

The mitotic spindle, a microtubule-based bipolar structure, is the cellular machinery responsible for the distribution of genetic material between the progeny cells. The mitotic spindle will capture the chromosomes at the kinetochores, align them along the cell's equator, and then pull them towards each spindle pole. Mitotic spindle integrity is a vital tumor suppressor pathway that requires microtubule crosslinking and motor protein movements to establish spindle length, position and orientation [88–92]. Bipolar chromosome attachment generates tension across the sister kinetochores, and this signals for chromatid separation and mitotic progression. Erroneous attachments engage the spindle assembly checkpoint (SAC), which prevents the onset of anaphase until the errors are corrected. However, if the SAC fails to sense the misattached chromosome, the misattached/unattached chromatid will lag behind, which could mean the loss or gain of whole or part of a chromosome in the daughter cells [93, 94].

Cytokinesis partitions the cytoplasm of the mother cell between the two daughter cells concurrent with nuclear membrane formation. The process of cytokinesis includes the specification of the cleavage plane, ingression of the cleavage furrow, and abscission of the midbody, which are all heavily dependent on microtubule and actin networks. Cytokinesis failure may occur when mitotic spindle elongation and positioning are perturbed, thus disrupting delivery of activation signals to the cortex for cleavage plane formation [95]. Aborted cytokinesis will result in a cell with double the genetic material (tetraploid) and double the number of centrosomes. This tetraploid state is considered an intermediate for the aneuploid state frequently observed in cancer cells [96], and the fate of these tetraploid cells heavily depends on p53 [97–100].

Telomere-associated genome instability occurs when inappropriate DNA repair (i.e. non-homologous end joining (NHEJ) and homology directed repair (HDR)) takes place at dysfunctional telomeres. Inappropriate NHEJ produces dicentric chromosomes [30, 31], which are highly unstable with a tendency to break during mitosis [30, 31]. Repair of these new breaks can propagate new dicentric chromosome formation [30, 31]. This process is called the breakage-fusion-bridge (BFB) cycle, which can continue over multiple cell divisions and generate complex chromosomal rearrangements [101, 102]. This BFB cycle generally has three outcomes: (1) LOH due to breakage sites or asymmetric segregation of chromosomes after breakage; (2) gene amplification due to breakage sites; and (3) unbalanced translocations [30]. During HDR, inversions, deletions and translocations are generated when recombination occurs at a highly homologous stretch of telomeric DNA either on the same chromosome or between two different chromosomes [30]. Activation of the HDR pathway increases the formation of anaphase bridges, which must be resolved to prevent cytokinesis failure and aneuploid progenies [103]. El-Khamisy and colleagues (Chap. 3) provide an in-depth discussion of the molecular control of NHEJ and HDR pathways.

c. Cell Cycle Checkpoints

Cell cycle checkpoints are the cell's quality control mechanisms that coordinate the progression of the cell cycle and delay entry into the next stage in the presence of genome damage. Under circumstances where the detected genome damage is not fixed, these checkpoints will trigger senescence, mitotic catastrophe or apoptosis to eliminate high-risk cells. The G1/S (and post-mitotic) and G2/M checkpoints can recognize DNA lesions and abnormal chromosome structures, whereas the spindle assembly checkpoint is an intra-mitosis surveillance mechanism that monitors the interaction between chromosomes and the mitotic spindle.

The post-mitotic checkpoint monitors the state of the genome after the previous round of the cell cycle, and delays replication in the presence of damaged DNA.

This checkpoint is heavily dependent on p53, which is the most commonly mutated gene in cancer [104]. Cytokinesis failure during the previous round of cell division will result in tetraploid progeny with supernumerary centrosomes, a so-called "double-value cell" [3]. Tetraploid intermediates may play an important role in tumorigenesis [96, 105–109], and precede aneuploidy cells in a premalignant condition called Barrett's oesophagus [106]. Tetraploid cells isolated from p53^{-/-} mouse mammary epithelial cells form tumours in nude mice [100].

p53 is a phosphoprotein found at low levels in normal cells. Upon DNA damage, p53 is stabilized and protein turnover is inhibited through phosphorylation by ATM (Ataxia telangiectasia mutated) and Chk1 (checkpoint kinase-1) [110–113]. The key transcriptional target of p53 is the cdk inhibitor, p21^{Waf1} [111]. p21^{Waf1} binds to the cyclin E/cdk2 complex to inactivate the kinase, which prevents the initiation of DNA synthesis and blocks G1-S progression [114]. Prolonged G1 arrest by p53 can result in cellular apoptosis or senescence [115, 116]. When p53 is compromised, tetraploid cells continue to proliferate leading to increasingly error prone divisions and aneuploid progeny [96].

The G2/M checkpoint senses DSBs and facilitates their repair by preventing mitotic entry. The checkpoint inhibits the mitosis-promoting complex, cyclin B1/ cdc2 kinase, through inhibitory phosphorylation of cdc2 kinase [117–119]. If the G2/M arrest fails, the damaged chromosome will enter mitosis and may initiate a BFB cycle that can generate complex chromosomal rearrangements [101, 102]. Alternatively, the acentromeric fragment of the broken chromosome may be degraded, while the centromeric fragment may be repaired through the addition of a telomere region [120].

The spindle assembly checkpoint is a unique surveillance mechanism that does not sense DNA damage, but rather serves as a preventative measure against genome instability. During mitosis, sister chromatids are kept together by the cohesion complex, which is degraded at the metaphase-anaphase transition in an ubiquitin-dependent manner by the anaphase promoting complex/cyclosome. Correct bipolar attachment of chromosomes generates tension across the sister chromatids at the centromeres [121, 122]; the assembly checkpoint generates a wait signal in the absence of tension due to erroneous attachment, such as syntelic (both sister chromatids are attached to the same pole), monotelic (only one sister chromatid is attached) and/or absence of any attachments from either sister chromatid [123]. Upon assembly checkpoint activation, the mitotic checkpoint complex is loaded onto the kinetochores of the misat-tached chromosome, which inhibits the anaphase promoting complex/cyclosome and allows the cell to fix the erroneous attachment [121, 123].

The assembly checkpoint involves extensive kinase signaling pathways to sense misattachments, delay chromosome segregation and correct attachment errors. For example, Aurora kinase B detects chromosome misattachments [124, 125], but Aurora kinase A can override the checkpoint and enable cells to enter anaphase despite misattached chromosomes [126, 127]. In order for all 23 pairs of chromosomes to establish bipolar attachment during metaphase, a delicate balance is needed between stabilization of the correct kinetochore-microtubule attachment and turnover of misattachments by dynamic spindle microtubules. Polo-like kinase 1 decreases

kinetochore–microtubule dynamics to stabilize initial attachments during prometaphase, and removal of Polo-like kinase 1 from kinetochores during metaphase is necessary to maintain microtubules dynamics [128]. Finally, the checkpoint kinase BUBR1 inhibits the anaphase promoting complex/cyclosome through interactions with the mitotic checkpoint complex and the motor protein CENPE, which is required for microtubule attachment at the kinetochores [129, 130].

Mosaic variegated aneuploidy (MVA; OMIM: 257300) is a rare, hereditary condition due to biallelic mutations in BUBR1 and characterized by mosaic aneuploidy that leads to developmental defects and predisposition to cancer [131–136]. Both frameshift and missense mutations in the gene encoding BUBR1 have been identified in MVA families [135, 136], which result in truncated transcripts and reduced protein abundance [137]. Reduction in BUBR1 levels affect chromosome attachment and assembly checkpoint stringency, which leads to premature chromatid separation. As a result, individuals with MVA have a high incidence of childhood cancers, such as Wilms' tumour, rhabdomyosarcoma and leukemia. The severity of the disease phenotype in MVA patients reinforces the notion that the assembly checkpoint is crucial for the prevention of genomic abnormalities and cancer.

Prevention, Detection, and Prognosis of Genome Instability

Preventing DNA damage and genome instability can reduce the risk of developing cancer. Although there is no one way to prevent genomic damage, measures can be taken to reduce one's exposure to non-inherited sources and limit one's cancer risk.

a. Non-Inherited Sources

Many environmental factors, such as exposure to carcinogens, viruses, and diet, may lead to genome instability and eventually cancer. In the following section, we will discuss some of these environmental factors, how they lead to genome instability, and some methods that reduce exposure to these risk factors.

People are exposed to radiation daily, particularly in the form of solar radiation (ultraviolet (UV) light). UV-A light causes indirect DNA damage by producing free radicals and reactive oxygen species that go on to damage DNA, while UV-B light directly damages DNA by causing the formation of pyrimidine dimers [138]. DNA damage from UV light is a causative factor for melanoma and other skin cancers [139]. Fortunately, it is easy to take steps, like using a sunscreen that protects against both UV-A and UV-B light, which can cut the risk of melanoma in half [140].

Chemical carcinogens are found in the environment and can arise from many different sources. Some chemical carcinogens, like air pollution, cannot be completely avoided and, in these cases, it is important to limit exposure as much as possible. Other chemical carcinogens, like cigarette smoke, can be completely avoided. Cigarette smoke contains more than 20 carcinogenic chemicals [141] and smoking

causes at least 80% of lung cancers and 20–30% of pancreatic cancers [142, 143]. Benzene, a carcinogen in cigarette smoke, is linked to leukemia and other blood-related cancers [144] and is known to cause genome instability by inducing DNA strand breaks and other chromosome damage [145].

Human papillomavirus (HPV) can induce cancer of the cervix, vulva, vagina, penis, oropharynx, and anus. HPV can be passed on to children during birth and has been implicated in cases of sporadic retinoblastoma in children [146]. HPV causes cancer through expression of virus proteins or oncoproteins that promote oncogenesis. In HPV, the E6 and E7 oncoproteins cause aberrant proliferation [147–149], leading to centrosome duplication and eventual genome instability. Ectopic expression of E6 and E7 leads to structural and numerical chromosomal abnormalities, respectively [150–152]; similar abnormalities, such as enlarged nuclei, multinucleate cells, and tripolar mitotic spindles, are observed in clinical samples of cervical lesions. Vaccines that protect against several HPV strains, including HPV-16 and HPV-18 [148, 153], have been developed and their use may significantly reduce the prevalence of HPV-induced cancers.

Diet impacts DDR pathways as certain nutrients act as necessary cofactors. A dietary lack of folate, for example, can lead to the misincorporation of uracil into the genome rather than thymidine [154], which induces chromosome strand breaks and/or impairs excision repair. A reduced dietary intake, or low tissue/plasma levels, of Vitamin B6 associates with a higher risk of developing cancer while Vitamin B12 (folic acid) deficiency is linked to DNA damage, such as chromosome breaks, micronuclei formation, and DNA hypomethylation [155]. Additionally, low dietary intake of calcium, folate, nicotinic acid, vitamin E, retinol, and β -carotene and high intake of panthothenic acid, biotin, and riboflavin have been associated with increased genomic instability [156]. With respect to β -carotene, sufficient intake in one's diet correlated to a lower risk of MSI-H types of colon cancer [157]. Together, these data highlight the importance of a well-balanced diet in the maintenance of genome integrity.

b. Diagnosis of Genome Instability

The type and level of genomic instability may offer opportunities to personalize therapies, which will be discussed in greater detail in Chaps. 9 and 10. Methods to quantify genomic instability, including the structure of the chromosomes, genomic sequences, and/or gene expression, each have intrinsic advantages and disadvantages. In the following sections, and in Table 2.1, we outline some of the common methods used to identify genome instability. Chapter 4 will discuss in more detail some new methods for measuring and modeling DNA damage.

b.i. Diagnosing Large Scale Aberrations

Karyotyping images and arrays all chromosomes to measure aneuploidy, chromosome breaks, translocations, and inversions [158, 159]. Different types of staining
Large aberrations				
	Karyotype	FISH	Array-CGH	Flow cytometry
Technique	Isolate entire chromosomes and determine ploidy with dyes	Label cells with DNA dye and fluorescence determines ploidy	Hybridize non- tumor and tumor samples. Identify balanced or un- balanced signal	Label cells with probes that bind chromosomes
Aberrations detected	Aneuploidy, inversions, chro- mosome breaks, translocations	Changes in ploidy	Changes in copy number	Changes in ploidy
Advantages	High sensitiv- ity for mosaic cultures can detect balanced aberrations		Better resolution	Very accurate for ploidy
Disadvantages	Low resolution	Not automated cell sectioning and overlap	Detection of mutations is limited Low sensitiv- ity in mosaic cultures	Limited to ploidy
Small aberrations				
	ISSR-PCR	AP-PCR	SNP Array	Genome sequencing
Technique	Amplify micro- satellite regions. Compare PCR products between tumor and non-tumor samples	DNA amplified using random primers. Foot- print is compared between tumor and non-tumor	DNA fragments labelled, hybrid- ized to an array with known SNPs	Isolate genomic DNA and sequence
Aberrations detected	MIN, amplifica- tions, deletions, translocations, insertions	Amplifica- tions, deletions, insertions translocations	LOH, copy num- ber changes	All mutations
Advantages		Uses small amounts of sample. No prior sequence knowl- edge required	Detects copy neutral events	Multiple muta- tions at one time
Disadvantages	Exact region dif- ficult to identify	Reproducibility	Low sensitiv- ity in mosaic cultures	Cost

 Table 2.1 Methods of determining genomic instability

	Expression analysis
Technique	mRNA is labelled and hybridized to an array. Fluorescence is compared between tumor and non-tumor samples
Aberrations detected	Mutations that cause changes in gene expression, including epigenetic changes
Advantages	Can easily identify involved genes. Certain profiles have been linked to particular prognoses

Table 2.1 (continued)

can be used to identify different characteristics of the chromatin. For example, Giemsa stain binds to heterochromatin and the banding patterns will indicate insertions, deletions, and other abnormalities. Spectral karyotyping, a type of fluorescence *in situ* hybridization, employs multiple fluorescent probes to uniquely identify chromosome pairs and determine changes in ploidy and rearrangements. Array-CGH also detects changes in copy number, such as insertions, deletions, and amplifications [160, 161] by hybridizing fluorescently-labeled DNA isolated from cancer cells to fluorescently-labeled DNA isolated from non-cancer cell. This method has better resolution than karyotyping or fluorescence *in situ* hybridization, but it cannot detect mutations, such as inversions and translocations that do not change the copy number.

b. ii. Diagnosing Small Scale Aberrations

There are several polymerase chain reaction (PCR)-based assays that identify genomic instability. Inter-simple sequence repeat PCR (ISSR-PCR) uses primers that have regions homologous to dinucleotide repeats to amplify microsatellite regions of DNA [162–166]. By comparing the gains and losses in amplified DNA bands between tumor and normal tissues, regions of genome instability can be identified. The advantage of this method is that it can detect smaller alterations however it may be difficult to find, or sequence, the altered DNA region. Arbitrarily primed PCR (AP-PCR) uses primers composed of random sequences and low stringency conditions to create genome footprints for normal and tumor tissues, which allows detection sequence changes, insertions, deletions, and amplifications [167]. This technique uses small amounts of DNA, and allows one to reamplify, clone, and/or sequence the resulting bands which aids in the identification of abnormal tumor sequences [168], however specificity and reproducibility can be challenges. Single nucleotide polymorphism (SNP) arrays can detect LOH and copy number changes [169, 170]. Although similar to array-CGH, SNP arrays have an advantage in that they can detect copy-neutral events. Due to their high resolution (i.e. over 1.8 million markers that span the entire genome), SNP arrays enabled discoveries such as a reduced level of genome instability in hereditary as compared to sporadic cases of retinoblastoma, which

contrasted with a popular model for hereditary retinoblastoma [171]. However, SNP arrays may not be sufficient to identify mutations in mosaic cultures. Whole genome sequencing will identify point mutations [172–174], but its high cost precludes it broad clinical application at this time.

Genome instability can also be determined by gene expression analyses. Several different profiles can identify cancer cells and some of these profiles are also predictive of prognosis. Habermann et al. [175] identified a 12-gene expression signature that distinguished between breast cancers with high or low genomic instability [175]. This signature can also predict prognosis in other cancer types [176]. A CIN measure, using either the top 25 or top 70 genes (CIN25 and CIN70 signatures respectively), correlated gene expression to the "total functional aneuploidy" in data sets from many different tumor types [177].

c. Genomic Instability and Cancer Prognosis

Genome instability is typically correlated with a poor prognosis [175–177]. Higher CIN is indicative of a poor prognosis in early invasive luminal HER2-negative and node negative breast cancer. Moreover, the 3q8pq20 subtype of oral cancer is prone to chromosome abnormalities and a higher rate of metastasis than non-3q8pq20 patients. However, genome instability is not always correlated with a poor prognosis. In sporadic colorectal cancer, patients with microsatellite instable cancers have better prognosis. Moreover, some cancers appear to have a threshold where genome instability no longer indicates a poor prognosis. In some ER-negative breast cancers, the CIN MCD4 cohort had better survival than patients that were classified in the intermediate and lower MCD cohorts [178]. This improved outcome was also seen in MCD4 patients that were treated with adjuvant chemotherapy. Thus, genomic instability may empower tumor heterogeneity but it may also negatively impact biological fitness.

Genetic Changes and Cancer Evolution

Carcinomas undergo a series of morphological changes to promote local invasion and distant migration. In preparation for local dissemination, tumor cells may undergo an adaption known as the "epithelial-mesenchymal transition" (EMT), which allows these epithelial cells to invade nearby blood and lymphatic vessels and migrate to distant sites [179-181]. Activation of the EMT program in tumor cells requires changes to the genome and/or epigenome, which promotes the turnover of adherens junctions, the expression of extracellular matrix degrading enzymes, and increased cell motility among other changes. Once these metastatic cells have reached their potential secondary colonization sites, they must pass through a reversal process called the "mesenchymal-epithelial transition" (MET). In theory, tumor cells undergoing reversible EMT and MET programs suggests the new tumor colony should be identical to the cells in the primary tumor. However, multiple studies have identified genetic and epigenetic heterogeneity in metastatic cells, suggesting that not all tumor cells are capable of initiating metastasis or able to revert back to their previous epithelial stage during MET [182–184].

The process of neoplastic progression is an evolutionary process, similar to that observed during speciation, where heritable genetic variations can promote the survival of tumor cells through clonal expansion [185–187]. Mutations and epimutations that favor increased proliferation and resistance to apoptosis are highly favorable during clonal expansion of neoplastic cells. For example, amplification of Aurora kinase A can override the assembly checkpoint to prevent mitotic arrest and promote chromosome aberrations [126]. The tumor microenvironment changes as the disease progresses; dense population growth and over-consumption of local resources add selective pressures for dispersal and metastasis. In most solid tumors, the center of the tumor mass is often necrotic, hypoxic and densely populated [188, 189]. Metastatic cells, however, leave the primary tumor site and colonize secondary sites, which require certain fitness advantages. An aneuploid genome, and the heterogeneity it engenders, has been shown to promote metastatic properties [190]. On the other hand, tumor cells at the primary site may be selected for their ability to cope with the stressful local environment through metabolic reprogramming. For example, hypoxic cancer cells uses glucose as fuel and produce lactate as a waste product, which serves as fuel for their oxygenated neighboring cells [191–193]. This intratumor symbiosis enables tumor cell cooperation that promotes tumor growth. Lastly, cancer therapies apply artificial selection pressures over the tumor cell population. Thus, the mosaic genome and epigenome in cancer cells may seem chaotic, but, in its chaos, may in fact coordinate disease progression.

Acknowledgements CAM is supported through a CIHR New Investigator Salary award while HC and MC are supported through a Michael Cuccione Foundation for Childhood Cancer Research scholarship and fellowship, respectively.

References

- 1. Futreal PA et al (2004) A census of human cancer genes. Nat Rev. Cancer 4:177-183
- Hardy PA, Zacharias H (2005) Reappraisal of the Hansemann-Boveri hypothesis on the origin of tumors. Cell Biol Int 29:983–992
- 3. Boveri T (1914) Zur Frage der Entstehung Maligner Tumoren. Gustav Fischer Verlag, Jena
- Nowell P, Hungerford D (1960) Chromosome studies on normal and leukemic human leukocytes. J Natl Cancer Inst 25:85–109
- 5. Miki Y et al (1994) A strong candidate for the breast and ovarian cancer susceptibility gene BRCA1. Science 266:66–71
- Futreal PA et al (1994) BRCA1 mutations in primary breast and ovarian carcinomas. Science 266:120–122
- Korkola J, Gray JW (2010) Breast cancer genomes–form and function. Curr Opin Genet Dev 20:4–14

- Lengauer C, Kinzler KW, Vogelstein B (1998) Genetic instabilities in human cancers. Nature 396:643–649
- 9. Almoguera C et al (1988) Most human carcinomas of the exocrine pancreas contain mutant c-K-ras genes. Cell 53:549–554
- Lee H-C et al (2005) Mitochondrial genome instability and mtDNA depletion in human cancers. Ann N Y Acad Sci 1042:109–122
- 11. Pikor L et al (2013) The detection and implication of genome instability in cancer. Cancer Metastasis Rev 32:341–352
- 12. Risinger JI et al (1993) Genetic instability of microsatellites in endometrial carcinoma. Cancer Res 53:5100–5103
- 13. Leach FS et al (1993) Mutations of a mutS homolog in hereditary nonpolyposis colorectal cancer. Cell 75:1215–1225
- 14. Rhyu MG, Park WS, Meltzer SJ (1994) Microsatellite instability occurs frequently in human gastric carcinoma. Oncogene 9:29–32
- 15. Fishel R et al (1994) The human mutator gene homolog MSH2 and its association with hereditary nonpolyposis colon cancer. Cell 77:1027–1038
- Kim W et al (1999) Microsatellite instability (MSI) in non-small cell lung cancer (NSCLC) is highly associated with transforming growth factor-beta type II receptor (TGF-beta RII) frameshift. Anticancer Res 20:1499–1502
- 17. Halling KC et al (1999) Origin of microsatellite instability in gastric cancer. Am J Pathol 155:205–211
- Boland CR, Goel A (2010) Microsatellite instability in colorectal cancer. Gastroenterology 138:2073–2087
- Gagos S, Irminger-Finger I (2005) Chromosome instability in neoplasia: chaotic roots to continuous growth. Int J Biochem Cell Biol 37:1014–1033
- Wang SI et al (1997) Somatic mutations of PTEN in glioblastoma multiforme. Cancer Res 57:4183–4186
- Jefford CE, Irminger-Finger I (2006) Mechanisms of chromosome instability in cancers. Critical Rev Oncol/Hematol 59:1–14
- 22. Moyzis RK et al (1988) A highly conserved repetitive DNA sequence, (TTAGGG)n, present at the telomeres of human chromosomes. Proc Natl Acad Sci U S A 85:6622–6626
- 23. Meyne J, Ratliff RL, Moyzis RK (1989) Conservation of the human telomere sequence (TTAGGG)n among vertebrates. Proc Natl Acad Sci U S A 86:7049–7053
- Bailey SM, Murnane JP (2006) Telomeres, chromosome instability and cancer. Nucleic Acids Res 34:2408–2417
- 25. Chakhparonian M, Wellinger RJ (2003) Telomere maintenance and DNA replication: how closely are these two connected? Trends Genet 19:439–446
- 26. Harley CB, Futcher AB, Greider CW (1990) Telomeres shorten during ageing of human fibroblasts. Nature 345:458–460
- Kondo S et al (1998) Antisense telomerase treatment: induction of two distinct pathways, apoptosis and differentiation. FASEB J 12:801–811
- Zhang X et al (1999) Telomere shortening and apoptosis in telomerase-inhibited human tumor cells. Genes Dev 13:2388–2399
- Hahn WC et al (1999) Inhibition of telomerase limits the growth of human cancer cells. Nat Med 5:1164–1170
- De Lange T (2005) Telomere-related genome instability in cancer. Cold Spring Harb Symp Quant Biol 70:197–204
- 31. Riha K, Heacock ML, Shippen DE (2006) The role of the nonhomologous end-joining DNA double-strand break repair pathway in telomere biology. Annu Rev Genet 40:237–277
- 32. Maser RS, DePinho RA (2002) Connecting chromosomes, crisis, and cancer. Science 297:565–569
- Feldser DM, Hackett JA, and CW (2003) Greider, telomere dysfunction and the initiation of genome instability. Nat Rev Cancer 3:623–627

- 2 The Generation, Detection, and Prevention of Genomic Instability ...
- Rudolph KL et al (1999) Longevity, stress response, and cancer in aging telomerase-deficient mice. Cell 96:701–712
- Kim NW et al (1994) Specific association of human telomerase activity with immortal cells and cancer. Science 266:2011–2015
- 36. Sharma S, Kelly TK, Jones PA (2010) Epigenetics in cancer. Carcinog 31:27-36
- 37. Sarkies P, Sale JE (2012) Cellular epigenetic stability and cancer. Trends Genet 28:118–127
- Miller OJ et al (1974) 5-Methylcytosine localised in mammalian constitutive heterochromatin. Nature 251:636–637
- Wilson AS, Power BE, Molloy PL (2007) DNA hypomethylation and human diseases. Biochim Biophys Acta 1775:138–162
- 40. Sperandeo M et al (2000) Relaxation of insulin-like growth factor 2 imprinting and discordant methylation at KvDMR1 in two first cousins affected by Beckwith-Wiedemann and Klippel-Trenaunay-Weber syndromes. Am J Hum Genet 66:841–847
- 41. Jones PA, Baylin SB (2007) The epigenomics of cancer. Cell 128:683-692
- Jones PA, Baylin SB (2002) The fundamental role of epigenetic events in cancer. Nat Rev Genet 3:415–428
- 43. Ehrlich M (2002) DNA methylation in cancer: too much, but also too little. Oncogene 21:5400–5413
- 44. Jones PA, Laird PW (1999) Cancer epigenetics comes of age. Nat Genet 21:163-167
- 45. Kouzarides T (2007) Chromatin modifications and their function. Cell 128:693-705
- 46. Fraga MF et al (2005) Loss of acetylation at Lys16 and trimethylation at Lys20 of histone H4 is a common hallmark of human cancer. Nat Genet 37:391–400
- 47. Seligson DB et al (2005) Global histone modification patterns predict risk of prostate cancer recurrence. Nature 435:1262–1266
- Seligson DB et al (2009) Global levels of histone modifications predict prognosis in different cancers. Am J Pathol 174:1619–1628
- Schneider A-C et al (2011) Global histone H4K20 trimethylation predicts cancer-specific survival in patients with muscle-invasive bladder cancer. BJU Int 108:E290–E296
- Ellinger J et al (2010) Global levels of histone modifications predict prostate cancer recurrence. Prostate 70:61–69
- Kurdistani SK (2011) Histone modifications in cancer biology and prognosis. Prog Drug Res 67:91–106
- 52. Osada H et al (2004) Reduced expression of class II histone deacetylase genes is associated with poor prognosis in lung cancer patients. Int J Cancer.112:26–32
- Mavrich TN et al (2008) A barrier nucleosome model for statistical positioning of nucleosomes throughout the yeast genome. Genome Res 18:1073–1083
- 54. Lin JC et al (2007) Role of nucleosomal occupancy in the epigenetic silencing of the MLH1 CpG island. Cancer Cell 12:432–444
- Hatziapostolou M, Iliopoulos D (2011) Epigenetic aberrations during oncogenesis. Cell Mol Life Sci 68:1681–1702
- You JS, Jones PA (2012) Cancer genetics and epigenetics: two sides of the same coin? Cancer Cell 22:9–20
- 57. Barretina J et al (2012) The cancer cell line encyclopedia enables predictive modelling of anticancer drug sensitivity. Nature 483:603–607
- 58. Matei DE, Nephew KP (2010) Epigenetic therapies for chemoresensitization of epithelial ovarian cancer. Gynecol Oncolo 116:195–201
- Juergens RA et al (2011) Combination epigenetic therapy has efficacy in patients with refractory advanced non-small cell lung cancer. Cancer Discov 1:598–607
- 60. Stoeber K et al (2001) DNA replication licensing and human cell proliferation. J Cell Sci 114:2027–2041
- Nishitani H, Lygerou Z (2002) Control of DNA replication licensing in a cell cycle. Genes Cells 7:523–534
- 62. Aguilera A, García-Muse T (2013) Causes of genome instability. Annu Rev Genet 47:1-32
- 63. Shen Z (2011) Genomic instability and cancer: an introduction. J M Cell Biol 3:1-3

- 64. Doxsey S (2001) Re-evaluating centrosome function. Nat Rev Mol Cell Biol 2:688-698
- 65. Nigg EA, Stearns T (2011) The centrosome cycle: Centriole biogenesis, duplication and inherent asymmetries. Nat Cell Biol 13:1154–1160
- 66. Heald R et al (1996) Self-organization of microtubules into bipolar spindles around artificial chromosomes in Xenopus egg extracts. Nature 382:420–425
- 67. Nigg EA, Centrosome A (2002) Cause of consequence of cancer progression? Nat Rev Genet 3:815–815
- Ko MA et al (2005) Plk4 haploinsufficiency causes mitotic infidelity and carcinogenesis. Nat Genet 37:883–888
- Fukasawa K (2005) Centrosome amplification, chromosome instability and cancer development. Cancer Lett 230:6–19
- Pellegrino R et al (2010) Oncogenic and tumor suppressive roles of polo-like kinases in human hepatocellular carcinoma. Hepatology 51:857–868
- Khodjakov A et al (2002) De novo formation of centrosomes in vertebrate cells arrested during S phase. J Cell Biol 158:1171–1181
- 72. La Terra S et al (2005) The de novo centriole assembly pathway in HeLa cells: cell cycle progression and centriole assembly/maturation. J Cell Biol 168:713–722
- 73. Peel N et al (2007) Overexpressing centriole-replication proteins in vivo induces centriole overduplication and de novo formation. Curr Biol 17:834–843
- 74. Merdes A et al (1996) A complex of NuMA and cytoplasmic dynein is essential for mitotic spindle assembly. Cell 87:447–458
- 75. Maxwell CA et al (2003) RHAMM is a centrosomal protein that interacts with dynein and maintains spindle pole stability. Mol Biol Cell 14:2262–2276
- Maxwell CA et al (2005) Receptor for hyaluronan-mediated motility correlates with centrosome abnormalities in multiple myeloma and maintains mitotic integrity. Cancer Res 65:850–860
- Luca MD, Lavia P, Guarguaglini G (2006) A functional interplay between Aurora-A, Plk1 and TPX2 at spindle poles. Cell Cycle 5:296–303
- 78. Haren L et al (2009) NuMA is required for proper spindle assembly and chromosome alignment in prometaphase. BMC Res Notes 2:64
- 79. Silk AD, Holland AJ, Cleveland DW (2009) Requirements for NuMA in maintenance and establishment of mammalian spindle poles. J Cell Biol 184:677–690
- Hut HMJ et al (2003) Centrosomes split in the presence of impaired DNA integrity during mitosis. Mol Biol Cell 14:1993–2004
- Krämer A, Maier B, Bartek J (2011) Centrosome clustering and chromosomal (in)stability: a matter of life and death. Mol Oncol 5:324–335
- Quintyne NJ et al (2005) Spindle multipolarity is prevented by centrosomal clustering. Science 307:127–129
- Marthiens V, Piel M, Basto R (2012) Never tear us apart–the importance of centrosome clustering. J Cell Sci 125:3281–3292
- 84. Ogden A, Rida PC, Aneja R (2012) Let's huddle to prevent a muddle: centrosome declustering as an attractive anticancer strategy. Cell Death Differ 19:1255–1267
- Weaver BA et al (2007) Aneuploidy acts both oncogenically and as a tumor suppressor. Cancer Cell 11:25–36
- Birkbak NJ et al (2011) Paradoxical relationship between chromosomal instability and survival outcome in cancer. Cancer Res 71:3447–3452
- Gergely F, Basto R (2008) Multiple centrosomes: together they stand, divided they fall. Genes Dev 22:2291–2296
- Dechant R, Glotzer M (2003) Centrosome separation and central spindle assembly act in redundant pathways that regulate microtubule density and trigger cleavage furrow formation. Dev Cell 4:333–344
- Bringmann H et al (2007) LET-99, GOA-1/GPA-16, and GPR-1/2 are required for asterpositioned cytokinesis. Curr Biol 17:185–191
- 90. Bird AW, Hyman AA (2008) Building a spindle of the correct length in human cells requires the interaction between TPX2 and Aurora A. J Cell Biol 182:289–300

- 2 The Generation, Detection, and Prevention of Genomic Instability ...
- 91. von Dassow G et al (2009) Action at a distance during cytokinesis. J Cell Biol 187:831-845
- 92. Dunsch AK et al (2012) Dynein light chain 1 and a spindle-associated adaptor promote dynein asymmetry and spindle orientation. J Cell Biol 198: 1039–1054
- Cimini D et al (2001) Merotelic kinetochore orientation is a major mechanism of aneuploidy in mitotic mammalian tissue cells. J Cell Biol 153:517–527
- Cimini D et al (2002) Merotelic kinetochore orientation versus chromosome mono-orientation in the origin of lagging chromosomes in human primary cells. J Cell Sci 115:507–515
- 95. Normand G, King R (2010) Understanding cytokinesis failure. Adv Exp Med Biol 676: 27–55
- 96. Shackney SE et al (1989) Model for the genetic evolution of human solid tumors. Cancer Res 49:3344–3354
- 97. Lanni JS, Jacks T (1998) Characterization of the p53-dependent postmitotic checkpoint following spindle disruption. Mol Cell Biol 18:1055–1064
- Andreassen PR et al (2001) Tetraploid state induces p53-dependent arrest of nontransformed mammalian cells in G1. Mol Biol Cell 12:1315–1328
- Borel F et al (2002) Multiple centrosomes arise from tetraploidy checkpoint failure and mitotic centrosome clusters in p53 and RB pocket protein-compromised cells. Proc Natl Acad Sci U S A 99:9819–9824
- Fujiwara T et al (2005) Cytokinesis failure generating tetraploids promotes tumorigenesis in p53-null cells. Nature 437:1043–1047
- 101. McClintock B (1941) The stability of broken ends of chromosomes in *Zea Mays*. Genetics 26:234–282
- Gisselsson D et al (2000) Chromosomal breakage-fusion-bridge events cause genetic intratumor heterogeneity. Proc Natl Acad Sci U S A 97:5357–5362
- Acilan C, Potter DM, Saunders WS (2007) DNA repair pathways involved in anaphase bridge formation. Genes Chromosomes Cancer 46:522–531
- 104. Vogelstein B (1990) Cancer. A deadly inheritance. Nature 348(6303):681-682
- 105. Levine DS et al (1991) Formation of the tetraploid intermediate is associated with the development of cells with more than four centrioles in the elastase-simian virus 40 tumor antigen transgenic mouse model of pancreatic cancer. Proc Natl Acad Sci U S A 88:6427–6431
- Galipeau PC et al (1996) 17p (p53) allelic losses, 4N (G2/tetraploid) populations, and progression to aneuploidy in Barrett's esophagus. Proc Natl Acad Sci U S A 93:7081–7084
- Pihan GA et al (2003) Centrosome abnormalities and chromosome instability occur together in pre-invasive carcinomas. Cancer Res 63:1398–1404
- Storchova Z, Pellman D (2004) From polyploidy to aneuploidy, genome instability and cancer. Nat Rev Mol Cell Biol 5:45–54
- Olaharski AJ et al (2006) Tetraploidy and chromosomal instability are early events during cervical carcinogenesis. Carcinog 27:337–343
- 110. Maya R et al (2001) ATM-dependent phosphorylation of Mdm2 on serine 395: role in p53 activation by DNA damage. Genes Dev 15:1067–1077
- Wahl GM, Carr AM (2001) The evolution of diverse biological responses to DNA damage: insights from yeast and p53. Nat Cell Biol 3:E277–286
- Shiloh Y (2003) ATM and related protein kinases: safeguarding genome integrity. Nat Rev. Cancer 3:155–168
- 113. Bartek J, Lukas J (2003) Chk1 and Chk2 kinases in checkpoint control and cancer. Cancer cell 3:421–429
- 114. Harper JW et al (1993) The p21 Cdk-interacting protein Cip1 is a potent inhibitor of G1 cyclin-dependent kinases. Cell 75:805–816
- 115. Burns TF, El-Deiry WS (1999) The p53 pathway and apoptosis. J Cell Physiol 181:31-39
- Ferbeyre G et al (2002) Oncogenic ras and p53 Cooperate to induce cellular senescence. Mol Cell Biol 22:3497–3508
- 117. Coleman TR, Dunphy WG (1994) Cdc2 regulatory factors. Curr Opin Cell Biol 6:877-882
- 118. Jin P, Gu Y, Morgan DO Role of inhibitory CDC2 phosphorylation in radiation-induced G2 arrest in human cells. J Cell Biol 134:963–970

- 119. Kommajosyula N, Rhind N (2006) Cdc2 tyrosine phosphorylation is not required for the S-phase DNA damage checkpoint in fission yeast. Cell Cycle 5:2495–24500
- 120. Wilkie AO et al (1990) A truncated human chromosome 16 associated with alpha thalassaemia is stabilized by addition of telomeric repeat (TTAGGG)n. Nature 346:868–871
- Maresca TJ, Salmon ED (2010) Welcome to a new kind of tension: translating kinetochore mechanics into a wait-anaphase signal. J Cell Sci 123:825–835
- 122. Lara-Gonzalez P, Westhorpe FG, Taylor SS (2012) The spindle assembly checkpoint. Curr Biol CB 22:R966–980
- Redondo DM, Meraldi P(2011) The spindle assembly checkpoint: Clock or domino? In: Kubiak JZ (ed) Cell cycle in development. Springer Berlin Heidelberg, Berlin, pp 75–91
- Lens SM, Voest EE, Medema RH (2010) Shared and separate functions of polo-like kinases and aurora kinases in cancer. Nat Rev. Cancer 10:825–841
- 125. Foley EA, Kapoor TM (2013) Microtubule attachment and spindle assembly checkpoint signalling at the kinetochore. Nat Rev Mol Cell Biol 14:25–37
- Anand S, Penrhyn-Lowe S, Venkitaraman AR (2003) AURORA-A amplification overrides the mitotic spindle assembly checkpoint, inducing resistance to Taxol. Cancer Cell 3:51–62
- 127. Jiang Y et al (2003) AuroraA overexpression overrides the mitotic spindle checkpoint triggered by nocodazole, a microtubule destabilizer. Oncogene 22:8293–8301
- Liu D, Davydenko O, Lampson MA (2012) Polo-like kinase-1 regulates kinetochore-microtubule dynamics and spindle checkpoint silencing. J Cell Biol 198:491–499
- 129. Bharadwaj R, Yu H (2004) The spindle checkpoint, aneuploidy, and cancer. Oncogene 23:2016–2027
- Lampson MA, Kapoor TM (2005) The human mitotic checkpoint protein BubR1 regulates chromosome-spindle attachments. Nat Cell Biol 7:93–98
- Jacquemont S et al (2002) High risk of malignancy in mosaic variegated aneuploidy syndrome. Am J Med Genet 109:17–21
- 132. Hanks S et al (2006) Comparative genomic hybridization and BUB1B mutation analyses in childhood cancers associated with mosaic variegated aneuploidy syndrome. Cancer Lett 239:234–238
- 133. Bohers E et al (2008) Gradual reduction of BUBR1 protein levels results in premature sister-chromatid separation then in aneuploidy. Hum Genet 124:473–478
- 134. Karess R, Wassmann K, Rahmani Z (2012) New insights into the role of BubR1 in mitosis and beyond. Int Rev Cell Mol Biol 306:223–273
- 135. Hanks S et al (2004) Constitutional aneuploidy and cancer predisposition caused by biallelic mutations in BUB1B. Nat Genet 36:1159–1161
- Matsuura S et al (2006) Monoallelic BUB1B mutations and defective mitotic-spindle checkpoint in seven families with premature chromatid separation (PCS) syndrome. Am J Med Genet A 140:358–367
- 137. Suijkerbuijk SJE et al (2010) Molecular causes for BUBR1 dysfunction in the human cancer predisposition syndrome mosaic variegated aneuploidy. Cancer Res 70:4891–4900
- Clingen PH et al (1995) Correlation of UVC and UVB cytotoxicity with the induction of specific photoproducts in T-lymphocytes and fibroblasts from normal human donors. Photochem Photobiol 61(2):163–170
- Jhappan C, Noonan FP, Merlino G (2003) Ultraviolet radiation and cutaneous malignant melanoma. Oncogene 22(20):3099–3112
- 140. van der Pols JC et al (2006) prolonged prevention of squamous cell carcinoma of the skin by regular sunscreen use. Cancer Epidemiol Biomark Prev 15:2546–2548
- 141. Hecht SS (1999) Tobacco smoke, carcinogens and lung cancer. J Natl Cancer Inst 91:1194– 210
- 142. Shopland DR (1995) Tobacco use and its contribution to early cancer mortality with a special emphasis on cigarette smoking. Environ Health Perspect 103:131–142
- Lowenfels AB, Maisonneuve P (2004) Epidemiology and prevention of pancreatic cancer. Jpn J Clin Oncol 34:238–244
- 144. Khalade A et al (2010) Exposure to benzene at work and the risk of leukemia: a systematic review and meta-analysis. Environ Health 9:31

- 2 The Generation, Detection, and Prevention of Genomic Instability ...
- Hiraku Y, Kawanishi S (1996) Oxidative DNA damage and apoptosis induced by benzene metabolites. Cancer Res 56:5172–5178
- 146. Orjuela M et al (2000) Presence of human papilloma virus in tumor tissue from children with retinoblastoma : an alternative mechanism for tumor development. Clin Cancer Res 6:4010–4016
- Hawley-Nelson P et al (1989) HPV16 E6 and E7 proteins cooperate to immortalize human foreskin keratinocytes. EMBO J 8:3905–3910
- 148. Münger K et al (1989) The E6 and E7 genes of the human papillomavirus type 16 together are necessary and sufficient for transformation of primary human keratinocytes. J Virol 63:4417–4421
- Halbert CL, Demers GW, Galloway DA (1992) The E6 and E7 genes of human papillomavirus type 6 have weak immortalizing activity in human epithelial cells. J Virol 66:2125– 2134
- Hashida T, Yasumoto S (1991) Induction of chromosome abnormalities in mouse and human epidermal keratinocytes by the human papillomavirus type 16 E7 oncogene. J Gen virol 72:1569–1577
- 151. White AE, Livanos EM, Tlsty TD (1994) Differential disruption of genomic integrity and cell cycle regulation in normal human fibroblasts by the HPV oncoproteins. Genes Dev 8:666–677
- Duensing S, Münger K (2004) Mechanisms of genomic instability in human cancer: insights from studies with human papillomavirus oncoproteins. Int J Cancer 109:157–62
- Muñoz N et al (2003) Epidemiologic classification of human papillomavirus types associated with cervical cancer. New Engl J Med 348:518–27
- 154. Blount BC et al (1997) Folate deficiency causes uracil misincorporations into human DNA and chromosome breakage: implications for cancer and neuronal damage. Proc Natl Acad Sci U S A 94:3290–3295
- 155. Fenech M (2001) The role of folic acid and vitamin B12 in genomic stability of human cells. Mutat Res 475:57–67
- 156. Fenech M et al (2005) Low intake of calcium, folate, nicotinic acid, vitamin E, retinol, beta-carotene and high intake of pantothenic acid, biotin and riboflavin are significantly associated with increased genome instability—results from a dietary intake and micronucleus index survey in South Australia. Carcinog 26:991–999
- 157. Satia JA et al (2005) Diet, lifestyle, and genomic instability in the North Carolina Colon Cancer Study. Cancer Epidemiol Biomark Prev 14:429–436
- 158. Beheshti B et al (2001) Evidence of chromosomal instabilityin prostate cancer determined by spectral karyotyping (SKY) and interphase fish analysis. Neoplasia 3:62–69
- Wan TS, Ma ES (2011) Molecular cytogenetics: an indispensable tool for cancer diagnosis. Chang Gung Med J 35:96–110
- Pinkel D et al (1998) High resolution analysis of DNA copy number variation using comparative genomic hybridization to microarrays. Nat Genet 20(2):207–211
- Ishkanian AS et al (2004) A tiling resolution DNA microarray with complete coverage of the human genome. Nat Genet 36(3):299–303
- Zietkiewicz E, Rafalski A, Labuda D (1994) Genome fingerprinting by simple sequence repeat (SSR)-anchored polymerase chain reaction amplification. Genomics 20:176–183
- 163. Stoler DL et al (1999) The onset and extent of genomic instability in sporadic colorectal tumor progression. Proc Natl Acad Sci U S A 96:15121–15126
- 164. Basik M et al (1997) Genomic instability in sporadic colorectal cancer quantitated by intersimple sequence repeat PCR analysis. Genes Chromosomes Cancer 29:19–29
- 165. Tang JC et al (2001) Detection of genetic alterations in esophageal squamous cell carcinomas and adjacent normal epithelia by comparative DNA fingerprinting using inter-simple sequence repeat PCR. Clin Cancer Res 7:1539–1545
- Stoler DL et al (2002) Genomic instability measurement in the diagnosis of thyroid neoplasms. Head Neck 24:290–295
- 167. Samuelsson JK et al (2010) DNA fingerprinting techniques for the analysis of genetic and epigenetic alterations in colorectal cancer. Mutat Res 693:61–76

- Achille A et al (1996) Chromosome 7q allelic losses in pancreatic carcinoma. Cancer Res 56:3808–3813
- 169. Zhao X et al (2004) An integrated view of copy number and allelic alterations in the cancer genome using single nucleotide polymorphism arrays. Cancer Res 64:3060–3071
- 170. Heinrichs S, Look AT (2007) Identification of structural aberrations in cancer by SNP array analysis. Genome Biol 8:219
- 171. Mol BM et al (2014) High resolution SNP array profiling identifies variability in retinoblastoma genome stability. Genes Chromosomes Cancer 14:1–14
- 172. Wheeler DA et al (2008) The complete genome of an individual by massively parallel DNA sequencing. Nature 452:872–876
- 173. Meyerson M, Gabriel S, Gets G (2010) Advances in understanding cancer genomes through second-generation sequencing. Nat Rev Geneti 11:685–696
- 174. Wang L et al (2012) Whole-exome sequencing of human pancreatic cancers and characterization of genomic instability caused by MLH1 haploinsufficiency and complete deficience. Genome Res 22:208–19
- 175. Habermann JK et al (2009) The gene expression signature of genomic instability in breast cancer is an independent predictor of clinical outcome. Int J Cancer 124:1552–1564
- 176. Mettu RKR et al (2010) A 12-gene genomic instability signature predicts clinical outcomes in multiple cancer types. Int J Biol Markers 25:219–228
- 177. Carter SL et al (2006) A signature of chromosomal instability inferred from gene expression profiles predicts clinical outcome in multiple human cancers. Nat Genet 38:1043–1048
- Roylance R, Endesfelder D, Gorman P (2011) Relationship of extreme chromosomal instability with long-term survival in a retrospective analysis of primary breast cancer. Cancer Epid 20:2183–2194
- 179. Polyak K, Weinberg RA (2009) Transitions between epithelial and mesenchymal states: acquisition of malignant and stem cell traits. Nat Rev. Cancer 9:265–273
- Yilmaz M, Christofori G (2009) EMT, the cytoskeleton, and cancer cell invasion. Cancer Metastasis Rev 28:15–33
- Thiery JP et al (2009) Epithelial-mesenchymal transitions in development and disease. Cell 139:871–890
- Nguyen DX, Massagué J (2007) Genetic determinants of cancer metastasis. Nat Rev Genet 8:341–352
- 183. Campbell PJ et al (2010) The patterns and dynamics of genomic instability in metastatic pancreatic cancer. Nature 467:1109–1113
- 184. Ni X et al (2013) Reproducible copy number variation patterns among single circulating tumor cells of lung cancer patients. Proc Natl Acad Sci U S A 110:21083–21088
- Merlo LMF et al (2006) Cancer as an evolutionary and ecological process. Nat Rev. Cancer 6:924–935
- Hanahan D, Weinberg RA (2011) Hallmarks of cancer: the next generation. Cell 144:646– 674
- Nowell PC (2014) The clonal evolution of tumor cell populations. Acquired genetic lability permits stepwise selection. Soc Invest Dermatol 194:23–28
- Vaupel P, Mayer A (2005) Hypoxia and anemia: effects on tumor biology and treatment resistance. Transfus Clini Biol 12:5–10
- Paszek MJ et al (2005) Tensional homeostasis and the malignant phenotype. Cancer cell 8:241–254
- Hemmer J, Kraft K (2001) Stability of aneuploid clones during oral squamous cell carcinoma metastasis. Anticancer Res 21:1459–1464
- 191. Semenza GL (2008) Tumor metabolism: cancer cells give and take lactate. J Clin Invest 118:3835–3837
- 192. Feron O (2009) Pyruvate into lactate and back: from the Warburg effect to symbiotic energy fuel exchange in cancer cells. Radiother Oncol 92:329–333
- 193. Kennedy KM, Dewhirst MW (2010) Tumor metabolism of lactate: the influence and therapeutic potential for MCT and CD147 regulation. Future Oncol 6:127–148

Chapter 3 DNA Damage Response Pathways in Cancer Predisposition and Progression

Mohamed E. Ashour, Lamia El-Shafie and Sherif F. El-Khamisy

Abstract Cells are continually challenged by DNA assaults from endogenous and exogenous sources. Without appropriate repair, DNA damage can cause genome instability and the development of various diseases such as cancer, Immunodeficiency, neurological abnormalities, and premature aging. To maintain genome integrity, cells have evolved highly conserved defense mechanisms, collectively known as the DNA damage response (DDR), to identify the lesions, signal their presence, and activate the appropriate DNA repair pathway. In this chapter, we will discuss the recent advances in DDR, cancers with heritable defect in DNA repair, and secondary cancers developed following treatment with chemotherapeutic drugs that damage the DNA. We will focus on four major repair pathways with particular attention to the exploitation of recent knowledge to improve cancer therapy.

Keywords Double-strand breaks · Single-strand breaks · Reactive oxygen species · Synthetic lethality · Homologous recombination · Non-homologous end joining · Mismatch repair · DNA end processing · Fanconi Anemia

Abbreviations

5' dRP	5' deoxyribose phosphate
8-oxodG	7,8 dihydro-8-oxo-2'-deoxyguanosine
8-oxoGua	8-oxo-7,8-dihydroguanine
9-1-1	Rad9-Rad1-Hus1
AML	Acute myeloid leukemia
A-NHEJ	Alternative NHEJ
APE-1:	Apurinic endonuclease
aCGH	Array comparative genomic hybridization
B-CLL	B-cell chronic lymphocytic leukemia
BER	Base excision repair

S. F. El-Khamisy (X) · M. E. Ashour · L. El-Shafie

S. F. El-Khamisy Krebs Institute, University of Sheffield, Sheffield, UK

C. Maxwell, C. Roskelley (eds.), Genomic Instability and Cancer Metastasis,

Cancer Metastasis - Biology and Treatment 20, DOI 10.1007/978-3-319-12136-9_3

Center for Genomics, Helmy Institute, Zewail City of Science and Technology, Giza, Egypt e-mail: S.El-Khamisy@sheffield.ac.uk

[©] Springer International Publishing Switzerland 2015

BIR	Break-induced replication
BMF	Bone marrow failure
C-NHEJ	Classical NHEJ
CRC	Colorectal cancer
DDR	DNA damage response
dHJ	Double Holliday junction
DSB	Double strand break
DSBR	DSB repair
dsDNA	Double-stranded DNA
EC	Endometrial cancer
EDM	Exonuclease domain mutations
FA	Fanconi anemia
FapyA	4,6 diamino-5-formamidopyrimidine
FapyG	2,6-diamino-4-hydroxy-5-formamidopyrimidine
FEN-1	Flap endonuclease-1
hESC	Human embryonic stem cell
HSC	Hematopoietic stem cell
HR	Homologous recombination
ICLs	Interstrand crosslinks
IDLs	Insertion-deletion loops
iPSC	Induced pluripotent stem cells
IR	Ionizing radiation
LS	Lynch syndrome
MGMT	Methylguanine methyltransferase
MIN or MSI	Microsatellite instability
MMR	Mismatch repair
MPG	N-methylpurine DNA glycosylase
NEIL1	Endonuclease VIII-like 1
NHEJ	Non-Homologous end joining
OGG1	8-oxoguanine glycosylase
PAR	Poly ADP-ribose
PARG	Poly(ADP-ribose) glycohydrolase
PARP	poly (ADP-ribose) polymerase
PNK	polynucleotide kinase
PUA	α , β -unsaturated aldehyde
ROS	Reactive oxygen species
RPA	Replication protein A
SCE	Sister chromatid exchange
SL	Synthetic lethality
SSA	Single-strand annealing
SSB	Single strand break
SSBR	SSB repair
SSDA	Synthesis-dependent strand annealing
ssDNA	Single-stranded DNA
t-AML	Therapy-related AML

TLS	Translesion synthesis
t-MDS	Therapy-related myelodysplastic syndrome
t-MDS/MPN	Therapy-related myelodysplastic/myeloproliferative neoplasms
t-MN	Therapy-related myeloid neoplasms
UAF1	USP1-associated protein
USP1	Ubiquitin-specific peptidase 1
UV	Ultraviolet
VUS	Variants of uncertain significance

Introduction

Damage to the DNA poses a threat to the survival of the organisms and the faithful transmission of genetic information to offspring. Genome integrity is continually challenged by both endogenous and exogenous factors that can lead to genomic instability. Endogenous DNA damage can arise from attacks by reactive oxygen species (ROS), which are normal cell metabolic byproducts. Additionally, misincorporation of dNTPs during replication and base deamination, depurination, and modification contribute to endogenous DNA breakage. Exogenous DNA damage, on the other hand, can arise from ionizing radiation (IR), ultraviolet (UV) light, and chemical exposure to genotoxic material like industrial chemicals [85, 35]. Additionally, conventional cancer chemotherapeutic agents work by inducing extensive DNA damage and, in turn, cell death. Because these agents do not target cancer cells exclusively, mutations may also be propagated to normal cells leading to the development of secondary cancers [31]. The most common forms of DNA damages are single base lesions and DNA single-strand breaks (SSBs), and the ones with the most deleterious effects are interstrand crosslinks (ICLs) and DNA doublestrand breaks (DSBs) [131, 35]. To avoid such deleterious outcomes, cells have evolved defense mechanisms, collectively known as the DNA damage response (DDR) to identify lesions, signal their presence, and activate the appropriate DNA repair mechanisms that will work on fixing the damage with little to no loss. Distinct DNA lesions are repaired by different DNA repair pathways such as mismatch repair (MMR), SSB repair (SSBR), DSB repair (DSBR), and Fanconi anemia (FA) pathway. While each of these pathways process a particular set of lesions, there is also considerable overlap between the pathways [73, 138].

Cells defective in any of the DNA repair pathways generally demonstrate high sensitivity towards DNA damaging agents and this could be viewed as two sides of the same coin: one negative side, which is the predisposition to diseases like cancer, and the other positive side, which is portrayed in harnessing this knowledge for targeted therapy. In this chapter, we shall discuss the recent advances in selected DNA repair pathways, with particular attention to cancers with heritable defect in DNA repair and secondary cancers developed following treatment with chemotherapeutic drugs.

Mismatch Repair

The MMR pathway is a bidirectional excision-resynthesis system that corrects mismatches generated during DNA replication or homologous recombination (HR). MMR increase the fidelity of DNA replication by 1000-fold [112, 184]. Mismatches fall into two groups: base-base mispairs resulting from incorrect nucleotide insertion by DNA polymerases and insertion-deletion loops (IDLs) resulting from slippage of DNA polymerase at simple sequence repeats (microsatellite) [116, 126].

a.i.-Prokaryotic Mismatch Repair MMR is well conserved from bacteria to mammals. The primary bacterial proteins involved in MMR are designated MutS, MutL, and MutH. MutS recognizes mismatchs as homodimers. MutL (also called molecular matchmaker) facilitates the interaction between DNA-MutS-MutL and MutH [150, 194]. MutH cleaves GATC sequences selectively in the nascent strand, which remains transiently unmethylated because deoxyadenine methylase lags behind the replication fork [121]. The nicked strand is cleaved by one of four single-strand exonucleases (the 5' \rightarrow 3' exonucleases ExoI and ExoX or the 3' \rightarrow 5' exonucleases RecJ and ExoVII) [24]. The single-strand gap is bound by single strand-DNA binding protein (SSB). DNA polymerase III then completes the gap, and DNA ligase seals the nick [88].

a.ii.-Eukaryotic Mismatch Repair MMR in eukaryotes retains many of the key features of the *E. coli* MMR pathway. The MutS equivalents in humans (MSH) exist in two heterodimeric forms: MutS α (MSH2 and MSH6) that identifies base-base mismatches and small loops, and MutS β (MSH2 and MSH3) that identifies larger loops, with some overlap in substrate specificity between the two MutS complexes [127]. Recently, MutS α has been shown to have strong bias for insertion loops repair, while MutS β has an even stronger bias for deletion loops repair [195]. Human cells express more MSH6 than MSH3, leading to a MutS α : MutS β ratio of 10:1 [51, 147]. Despite their redundant activities, both complexes are required for MMR, and defective or abnormal expression of MSH6 or MSH3 leads to spontaneous mutation (mutator phenotype) [51, 52, 79, 148].

Heterodimeric MutS complex (MSH2/6 or MSH2/3) recognizes the substrate, likely by recognizing increased flexibility at the site of the mismatch [95]. MutS complex then recruits MutL. MutL equivalents in humans exist in three heterodimeric forms: MutLa (MLH1–PMS2), MutL β (MLH1–MLH3), and MutL γ (MLH1–PMS1). MutLa is the major MutL homolog that participates in MMR and has endonuclease activity [30]. MutLa binds several MMR proteins and modulates their activity in a mismatch-dependent manner [116, 126]. EXO1, the only eukaryotic exonuclease implicated in MMR to date, has an obligate 5' \rightarrow 3' polarity, which seems inconsistent with the bidirectional MMR, but close analysis revealed that MutLa harbors cryptic endonuclease activity. PMS2 introduces a nick in the daughter strand 5' or 3' of the mismatch, and this nick serves as an entry point for the EXO1 that carries out the excision step [100, 101]. Consistently, MLH1–PMS2 is required for 3' excision but not 5' excision [41, 185, 255]. Replication protein A (RPA) protects the MMR excision intermediate from nuclease degradation, and the

excised DNA strand is resynthesized by Pol δ [116, 126]. The basic human MMR system includes MutS α or MutS β , MutL α , EXOI, PCNA, RFC (loads PCNA onto DNA), RPA, Pol δ , and DNA ligase I [41, 255]. The 5' to 3' mismatch-directed strand excision requires only MutS α , EXOI, and RPA, whereas substrates with a 3' nick also require MutL α , PCNA, and RFC [56].

MMR in eukaryotic cells has to deal with the nucleosome to reach the mismatch. Previous reports have demonstrated that DNA mismatches within tightly associated nucleosomes, in contrast to naked DNA, are poor MMR substrates [125, 202], so there must be a signal that allows for a timely recruitment of MMR onto the nucleosome. Li et al. [124] have made a breakthrough by demonstrating that an epigenetic histone mark, H3K36me3, during G1- and early S-phase recruits the MutSα onto the chromatin before replication, independent of the presence of mismatch. Cells that lack STD2 (H3K36 trimethyltransferase) display microsatellite instability (MSI) and spontaneous mutation frequencies, characteristic of MMR-deficient cells.

a.iii.-Strand Discrimination Eukaryotic cells do not use methylation for strand discrimination, alternatively a nick in DNA can signal for strand-specific eukaryotic MMR *in vitro*. The first biochemical studies carried out with extracts of human or *Drosophila melanogaster* cells showed that covalently closed circular DNA substrates with a single mismatch were refractory to MMR, but a nick in either strand situated up to 1 kb away from the mismatch was necessary and sufficient to activate the MMR process [86, 223]. Discontinuous lagging strand synthesis of Okazaki fragments (aproximately 200 bp long in eukaryotes) introduces a high number of 5' DNA ends that discriminate the nascent lagging strand [25]. On the other hand, the leading strand is replicated in a continuous manner. This raises the question of how the MMR directs the nascent leading strand.

The answer to this question appears to lie in an interaction between MutL α and PCNA. RFC loads PCNA at 3' primer termini (boundaries between double- and single-stranded DNA) with the same side facing the DNA terminus [154]. Mismatch made by DNA polymerase is detected by MutS α or MutS β , which slides along the DNA, interacts with PCNA, and displaces the polymerase. The loading of MutL α generates a protein complex that travels towards the mismatch where MutL α can introduce nicks in the leading strand, which are used as loading sites for EXO1. Since only one strand of DNA has the correct orientation (5' \rightarrow 3' or 3' \rightarrow 5') for hydrolysis, the enzyme will cleave only a single strand [182, 183]. According to this model, on the leading strand, the MutL α /PCNA complex needs to travel from the 3' terminus to the mismatch which could be hundreds of nucleotides away [202], so it was suggested that MMR is less efficient on the leading strand compared to the lagging strand, where strand discontinuities are available [169].

Recently, an additional mechanism was proposed for the nascent leading strand discrimination. During replication, more than one million ribonucleotides are introduced into the mouse genome [84, 193]. A similar situation occurs in *Saccharomyces cerevisiae*, where Pole (the leading strand polymerase) incorporates about four times more ribonucleotides than Polô (the lagging strand polymerase) into the nascent DNA [140, 170, 171]. Recent reports have shown that RNase H2-dependent processing of the ribonucleotides incorporated by Pole acts as a signal

that can direct MMR to the nascent leading strand. Inhibition of RNase H2 has no effect on the lagging strand because of the high number of nicks introduced by Okazaki fragments. This mechanism has limited contribution to MMR fidelity because it requires that the mispair and the ribonucleotide are within 1 kb of each other [72, 139].

b. Mismatch Repair Deficiency, Microsatellite Instability, and Lynch Syndrome Microsatellites are short repetitive DNA sequences 1–6 [60]. Because of their repetitive sequence structure, microsatellites exhibit a particularly high mutation rate. During replication, DNA polymerases often fail to correctly duplicate the microsatellite repeats due to slippage, which results in IDLs [183]. This phenomenon is known as MSI and it is recognized as length changes in the microsatellites. The MSI status is determined using a panel of five microsatellites (BAT25, BAT26, D2S123, D5S346, and D17S250) [155].

In 1993, MSI was detected in about 10–15% of sporadic colorectal carcinomas as well as in >90% of Lynch syndrome (LS) patients, also referred to as hereditary nonpolyposis colorectal cancer (CRC) [183]. The finding that MMR deficiency in Saccharomyces cerevisiae induced MSI led to the suggestion that cancers with MSI might also have defects in MMR [97]. LS is a prevalent autosomal dominant hereditary cancer syndrome caused by heterozygous mutations in one of the MMR genes MSH2, MSH6, MLH1, or PMS2 [191]. The mutations in MMR genes that lead to truncation or deletion can be securely classified as pathogenic, but in a significant fraction of individuals suspected to develop LS, subtle alterations in MMR genes are identified such as missense mutations or mutations in splice sites. These types of mutation are called variants of uncertain significance (VUS) [191]. The pathogenesis of many of these VUS is not clear due to the absence of data on the consequences of these mutations on gene function. Many functional analysis assays have been developed in vitro and in vivo to identify the pathogenicity of VUS [50, 191]. Recently, using yeast as a model system, it has been demonstrated that more than half of the deleterious missense mutation in MSH2 result in lower levels of the protein due to ubiquitin-mediated proteasomal degradation (the primary ubiquitin ligase being san1). Increasing the expression of the unstable variants, deletion of san1, or the of proteasomal inhibitor restores MMR function [9].

MSI is known to occur due to defects in MMR genes, such as germline mutation in MSH2 or MLH1 in most LS cases and epigenetic silencing of MLH1 in most sporadic cases [19, 82, 118, 232]. Nevertheless, many colorectal and several other MSIpositive cancers do not have genetic or epigenetic defects in MMR genes. Recently, Li et al. [124] have shown that depletion of SETD2 impairs MutS α chromatin binding, leads to MSI, and increases the mutation rates. Intriguingly, they have found a renal cell carcinoma and a Burkitt's lymphoma cell line, both without defects in MMR genes but MSI positive, to be mutated in SETD2. This report provides an explanation for the discrepancy between the genotypes and phenotypes of such cancers. In addition, recent studies support the idea that defects in MMR pathway could be independent from defects in MMR genes. POLE and POLD1 are related B family polymerases, and they represent the main catalytic and proofreading subunits of the Pole and Pol δ enzyme complexes [187, 168]. POLE and POLD1 contain a 3'-5' exonuclease (proofreading) domain which recognizes and excises the mispair and in turn increases replication fidelity by approximately 100-fold. Recent reports have shown that POLE and POLD1 exonuclease domain mutations (EDMs) increase the susceptibility to CRC and, in the latter case, to endometrial cancer (EC). In addition, somatic POLE EDMs have been reported in sporadic CRCs and ECs [34, 177].

Microsatellites have been identified within the coding sequences of a number of genes [54, 55]. The DNA polymerase slippage within these coding sequences can induce frameshift mutations. In case of CRC genomes, cancer-associated genes frequently affected by MSI (e.g., TGFBR2, ACVR2A, and BAX) have been investigated [99, 146, 189]. Recently, Kim et al. [108] have provided a comprehensive analysis of the prevalence and functional consequence of MSI in CRC and EC. Using exome and whole genome sequencing, they have shown that recurrent MSI events in coding sequences have (1) elevated frameshift-to-inframe ratios, so they hypothesized that the genes inactivated by recurrent MSI may have tumor suppressor roles, and the high frame shift (nonneutral) could provide selection advantage on coding sequence (2) lower transcript levels than wild-type alleles, which may be due to RNA surveillance pathway that eliminates mRNA containing a premature stop codon, and (3) tumor type specificity.

Mismatch Repair, Monofunctional Alkylating Agent, and Therapy-Related Myeloid Neoplasms

Conventional chemotherapeutic agents used in clinics operate by inducing DNA damage in cancer cells. Unfortunately, normal cells are also targeted by these chemotherapeutic agents, which induce mutations and, in turn, the development of secondary cancers in normal cells. The most prevalent forms are therapy-related myeloid neoplasms (t-MN) account for about 10-20% of myeloid neoplasms and can be subdivided into therapy-related myelodysplastic syndrome (t-MDS), therapy-related acute myeloid leukemia (t-AML), and therapy-related myelodysplastic/myeloproliferative neoplasms (t-MDS/MPN) [218]. Based on the type of chemotherapeutic agents, two main subtypes of t-MN with different characteristic have been identified. The first subtype of t-MN is related to exposure to alkylating agents, and it is characterized by a long latency period of 3–10 years, a preceding myelodysplasia, and loss of all or parts of chromosomes 5 or 7 or both. The second subtype of t-MN is related to exposure to topoisomerase II poisons and is characterized by a short latency period of 1-3 years, often without a preceding myelodysplasia, and balanced chromosomal rearrangements involving MLL at 11q23 and t(15,17)(PML-RARA) [74, 178, 181].

Alkylating agents are divided into monofunctional (e.g., temozolomide, dacarbazine, and methylnitronitrosoguanidine (MNNG)) or bifunctional alkylating agents, such as nitrogen mustards (chlorambucil and cyclophosphamide), and chloroethylating agents (e.g., nimustine (ACNU), lomustine (CCNU), and carmustine (BCNU)) [113]. Base excision repair (BER) can repair the majority of the alkylated DNA adducts induced by monofunctional alkylators except for O6meG, which is largely responsible for the cytotoxicity of this class of chemotherapeutic agents. Methylguanine methyltransferase (MGMT) can directly repair the O6meG by covalent attachment of the methyl group from the O6meG to a cysteine residue on MGMT, leading to irreversible inactivation of MGMT [71, 145, 162]. During replication, DNA polymerases frequently mispair O6meG with thymine, which, in turn, activates the MMR [213]. Interestingly, rather than repairing *O*6meG, MMR induces DNA damage signaling, cell cycle arrest, and apoptosis [83, 98, 158]. This means that the cytotoxicity of monofunctional alkylating agents requires a proficient MMR. Indeed, cells proficient in MMR and deficient in MGMT show high sensitivity to monofunctional alkylating agent, while cells deficient in MMR and MGMT are resistant to cell death and have increased mutation rates [18, 45, 204]. The mutator phenotype that characterizes MMR-deficient cells may accelerate t-MN development. Consistently, t-AML, which arises after exposure to alkylating agents, displays MSI [31].

The mechanism by which MMR mediates the cytotoxicity of monofunctional alkylating agents is not fully understood. Two models have been proposed, the "futile cycle" and "direct signaling" models. The "futile cycle" model suggests that since the MMR machinery can only target the newly synthesized DNA strand containing the mismatched thymine, the O6meG will never be removed and another thymine opposite to O6meG will be inserted in the following replication. The repeated excision and regeneration of O6meG:T mispairs will induce cytotoxic DNA DSBs. In this model, the ATR is indirectly activated after DNA damage [158, 250]. The "direct signaling" model suggests that MMR proteins binding to a O6meG:T mispair acts as scaffold for direct recruitment and activation of ATR DNA damage signaling pathway. This model has been supported by separation of function mutations in mice containing mutations in Msh2 and Msh6 ATPase domains, which are essential for MMR activation but not for MMR-dependent DNA damage-induced apoptosis. These mice showed that MMR activity can be inhibited without affecting MMRinduced DDR [130, 249, 252]. Interestingly, many studies have shown that RPA is not essential for MMR-dependent ATR activation [134, 175]. In contrast to "the direct signaling" model, a recent study showed that O6meG induced ATM and ATR activation, and inhibition of ATM and ATR sensitized the cells to monofunctional alkylating agent [58].

Single-Strand Break Repair

SSBs arise either through insult from the direct action of ROS or topoisomerases or indirectly as a result of an intermediate step in the BER pathway [29] where BER enzymes injure the sugar backbone in the process of removing the damaged base [28, 66]. Oxidations, deaminations, and alkylations occur at a very high rate of about 30,000 damages per cell per day.

a.i-Recognition The first step in SSBR is the detection of the break. The main protein involved in this step is poly (ADP-ribose) polymerase (PARP). PARP is rapidly

recruited to the DNA strand break and consequently activated [28]. PARylation, or the polymerization of ADP-ribose, of target proteins is thought to modulate their recruitment, stabilization, or activity at SSB sites [138]. However, the residence time of PARP at the site of break is very short because as soon as PARP is autoribosylated the charge repulsion allows for its dissociation [138]. PARP is then returned to its original conformation and is free to be recruited on a different break via the degradation of poly ADP-ribose (PAR) by poly(ADP-ribose) glycohydrolase (PARG) [230]. One significant protein in BER, which is thought to be recruited by PARP1, is the scaffold protein XRCC1 [59, 149]. XRCC1 and PARP1 together aid repair by forming complexes with other BER proteins to promote recruitment/retention of repair factors and chromatin modification [152].

a.ii.-Base Excision and End Processing BER works through the excision of the incorrect/damaged base through cleavage of the N-glycosidic bond by DNA glycosylases [152]. Different types of lesions are recognized by different gylcosylases. For example, 8-oxoguanine glycosylase (OGG1) works on 8-oxoGua and 2,6-diamino-4-hydroxy-5-formamidopyrimidine (FapyG) lesions, while Endonuclease VIII-like 1 (NEIL1) has higher selectivity towards FapyG and 4,6 diamino-5-formamidopyrimidine (FapyA) [152]. When removing the damaged base, BER enzymes incise oxidatively damaged DNA to create an SSB. This step is achieved using apurinic endonuclease-1 (APE-1) or bifunctional DNA glycosylases that are capable of both base removal and apurinic site incision. This step leaves behind not only an abasic site but also unconventional termini; these need to be processed and restored to the proper 3'-hydroxyl and 5'-phosphate before the repair via short-patch or long-patch BER proceeds. Next, DNA ends are processed, which is considered the most miscellaneous step of SSBR, where a number of enzymes are involved including polynucleotide kinase (PNK), APE-1, DNA polymerase, and flap endonuclease-1 (FEN-1) [28]. An abasic site formed as a result of a monofunctional glycosylase will be recognized and cleaved by APE-1, leaving behind 3' OH and 5' deoxyribose phosphate (5'dRP) termini [234]. Polß possesses lyase activity. removing the 5'dRP at the nick [151]. However, if the glycosylase is bifunctional, its lyase activity cleaves the sugar-phosphate backbone leaving behind either an α , β -unsaturated aldehyde (PUA) or a phosphate group at the 3' end. APE1 then removes the PUA creating a 3' hydroxyl substrate for Polß, while PNK, stimulated by XRCC1, removes the phosphate group at the 3' end [234]. Both indirect and direct SSBs can be repaired by either of the BER pathways, short- or long-patch. The significant feature that distinguishes the two pathways from each other is the size of the repair patch—single nucleotide for short patch [49] and 2-12 nucleotides for long patch [68].

a.iii.-Gap Filling and Ligation Following restoration of conventional DNA termini, a single nucleotide or more is inserted by polymerases. *In vitro* studies suggest that the major polymerase involved in gap filling of short-patch repair is Polβ, while Polβ and Polδ/Polε are employed in long-patch repair [67, 186]. Whether Polβ requirement *in vivo* is due to its lyase or gap filing activity or both is not clear [215]. DNA ligases are now ready to seal the remaining nick. Long-patch ligation is primarily mediated by Lig1, while short patch is mediated by the Lig3a-XRCC1 complex [28].

Single-Strand Break Repair and Cancer

When a single component of the SSBR is inhibited, repair is compromised. When this happens, mutagenic lesions may accumulate causing risk of cancer development. Cancer is one of the diseases that are closely related to oxidative stress caused by ROS. The result of the oxidation of the highly susceptible guanine base is 7.8 dihydro-8-oxo-2'-deoxyguanosine (8-oxodG). During DNA replication, this modified base favors pairing with adenine instead of cytosine; further DNA replication will then allow adenine to bond with thymine [206]. This leads to mutagenesis via the dreadful GC-TA transversion. Other modified bases as well as other mutations also occur; however, 8-OxoGua and GC-TA transversion are both better studied and occur more frequently. If not fixed, this transversion mutation may deactivate a tumor suppressor gene (e.g. p53) or activate an oncoprotein (e.g. RAS), leading to tumorigenesis [42]. If this SSB is not immediately recognized and properly repaired, it can and will develop into the more deleterious DSB when replication takes place. Sakumi et al. [201] showed that OGG1-knockout mice were predisposed to lung cancer. Another study showed the increased risk of uterine myomas transformation into malignant cells is related to the levels of 8-oxoGua [65].

The aim of studying the different pathways of repair is not only to recognize the factors that increase risk of disease development but also to target these pathways for therapy. The study by Trivedi et al. [227] showed that both elevated N-methvlpurine DNA glycosylase (MPG) expression and siRNA-mediated loss of Polß in human breast cancer cells increases the sensitivity towards temozolomide. An increased accumulation of 5'dRP caused by the overexpression of MPG and loss of the lyase Pol β (mentioned earlier) is suggested to be the cause of this sensitization [227]. Following the induction of alkylating damage, PARP hyperactivation, as a result of Polß deficiency, will lead to BER failure due to the consequent NAD+ and ATP depletion [221]. Because the process of PARylation requires NAD+ as substrate, NAD+ biosynthesis inhibition could be used to sensitize cancer cells towards chemotherapy. Dual inhibition of BER and NAD+ biosynthesis was more effective in sensitizing temozolomide-resistant glioblastoma cells rather than targeting BER alone [75]. Further, PARG, which is responsible for PAR removal, could be a good target for sensitization; however, PARG inhibition is still not well characterized as a target for cancer therapy. Inhibitors of APE1 are divided into two classes: indirect inhibitors, like methoxyamine, which irreversibly bind to the AP site on the DNA and direct inhibitors, like lucanthone, which inhibits APE1 endonuclease activity [1, 44]. In preclinical studies, methoxyamine was found to potentiate the cytotoxic effects of temozolomide and pemetrexed [1, 44].

Emerging studies on the use of BER gene expression as sensitive predictors of cancer prognosis have become popular. Upregulation of APE1 is a feature of several malignancies with higher levels indicative of more aggressive tumors. The expression level of APE1 could also be an indicator of therapeutic response [1]. Higher levels of APE1 may overcome damage caused by IR or chemotherapy [1]; therefore, combination therapy that targets BER candidates for a better therapeutic outcome may be useful. For example, depletion of APE1 along with the TDP1, a protein involved in the repair of stalled topoisomerase I, amplifies the sensitivity to MMS treatment [6]. Germline and tumour-specific polymorphisms in BER genes are related to the risk of cancer development [44], making these polymorphisms useful as predictive biomarkers in cancer. A number of meta-analyses and case-control studies have emerged, uncovering such polymorphisms OGG1 S326C has been associated with an increased risk of CRC [159].

Double-Strand Break Repair

There are two distinct pathways that contribute to the repair of DSBs: HR and nonhomologous end joining (NHEJ).

a. Homologous Recombination Repair of the DSBs where the DNA strand goes on a search for homologous sequences to synthesize the new DNA sequence is termed HR, which is believed to be an error-free process. There are four main models for HR including break-induced replication (BIR), synthesis-dependent strand annealing (SDSA), single-strand annealing (SSA), and double Holliday junction (dHJ). The latter is only seen in meiosis and will not be discussed here. When a nick is recognized, endonucleases resect or chew off the 5' ends of both strands exposing the 3' ends to create 3' overhangs. This is followed by the binding and assembly of recombination proteins to form a presynaptic filament. The 3' overhangs go on a search for similar or homologous sequences to copy and synthesize new DNA sequences. Once the homologous sequence is found, DNA synthesis is initiated by the invasion of the single strand into the double strand homologous DNA, usually a sister chromatid. After successful DNA synthesis is completed, separation, ligation, and removal of extra nucleotides bring this process to an end.

The MRN complex, which is composed of MRE11, RAD50, and NBS1, is employed early on in the process of HR. This complex works at the site of break to create the 3' overhangs and to facilitate DDR signaling [93, 220, 253]. The single-strand DNA formed is then approached by RPA; together they form a recombinogenic intermediate. Both the MRN complex and RPA call on signaling proteins ATM and ATR, members of phosphatidylinositol 3-kinase-related kinases (PIKKs), which, through a series of phosphorylation cascades involving the cell-cycle check-point kinases CHK1 and CHK2, arrest the cell cycle to allow DNA repair to take place [119, 120]. Next, RAD51 displaces RPA on the single-stranded DNA (ssD-NA) [103] and forms the nucleoprotein presynaptic filament in a process mediated by BRCA2 and PALB2 (also known as FANC). RAD51 is essential in the steps of homology searching and double-stranded DNA (dsDNA) invasion [216]; therefore, RAD51 foci are surrogate markers for HR.

a.i.-Break-Induced Replication When replication starts, single-strand lesions, which are very frequent as discussed earlier, may cause stalling of the replication fork. This, in turn, transforms the single-strand nick to the more serious DSB. The one-ended DSBR involved here is often called BIR—a part of the HR pathway. Previously, the BIR pathway was thought to be a semiconservative method where it copies short DNA stretches and would stop once the Holliday junction is resolved and replication would carry on normally. Contrary to the early model, once BIR is initiated, it continues to copy hundreds of kilobases—all the way to the telomere—through what is known as bubble migration [200]. This means that the very accurate S-phase replication process will be replaced by the erroneous BIR. It has been suggested that this error is due to the recruitment of low fidelity Translesion synthesis (TLS) DNA polymerases resulting in BIR-induced mutagenesis [47].

a.ii.-Synthesis-Dependent Strand Annealing Gene conversion by SDSA is a process where the newly synthesized sequence appears only in the repaired DNA, while no exchange in the flanking regions occurs, and thus, SDSA naturally leads to conservative repair [144]. This happens as the resolvases cut in a manner that produces noncrossover products giving rise to a different heteroduplex DNA configuration than that of the dHJ model associated with meiosis [144].

a.iii.-Single-Strand Annealing Another pathway that repairs two-ended DSBs is SSA, which takes place when the DSB occurs between two repeat sequences. The resection of the dsDNA exposes the two repeat sequences, and thus, homology is found within each other. Subsequently, annealing directly commences and the removal of the extra DNA sequences takes place leaving behind a DNA sequence that has lost a significant portion of its nucleotides (one of the repeat sequences and the nucleotides between the repeat sequences) [81], possibly giving rise to loss of heterozygosity.

Homologous Recombination and Cancer

Ongoing research aims to study inhibitory molecules that target different steps in the HR process; some of these inhibitory molecules made it to clinical and preclinical trials. For example, an inhibitor of ATR, NU6027, has been uncovered and found to reduce Rad51 foci and to boost cisplatin's cytotoxicity [180]. Similarly, two more ATR inhibitors have been studied; one is VE821, which was shown to promote sensitization to chemo- and radiotherapy in pancreatic cancer [64], and the second one is ETP46464. A PI3K inhibitor NVPBEZ235 has been identified as a potent ATM and ATR inhibitor [225] and is being tested in clinical trials as a chemo- and radiosensitizer [94]. Downstream of ATR, AZD7762, an inhibitor of CHK1, has been found to decrease the formation of Rad51 foci and decrease gene conversion. BRCA1 has been newly identified as a phosphorylation target of CDK1 and through this phosphorylation event, Rad51 foci are formed and gene conversion occurs. Therefore, inhibition of CDK1 using RO3306 or AG24322 is considered a possible therapeutic target [32]. It is important to note that while the inhibition of kinases is considered a promising therapeutic target for cancer therapy, there are still several problems that need to be addressed. Because kinases do not act only on a single substrate, inhibition of these kinases will result in inhibition of several other pathways. Further, these inhibitors are nonselective, and so using them will definitely inhibit other kinases leading to off-target effects.

Cells that are defective in NBS1 show reduction in sister chromatid exchange (SCE). Moreover, knockdown of NBS1 sensitizes PARP-inhibited cells [153]. This is because PARP is an essential enzyme in SSBR as mentioned earlier, and its inhibition would elevate the level of SSBs and increase the likelihood of the formation of lethal DSBs. Mirin, an MRN complex inhibitor, works by blocking the exonuclease activity of MRE11 [53]. Inhibitors of RAD51, such as B02 [89] and RI-1 [21] have also been identified. Inhibitors of EGFR [127, 238] and tyrosine kinase BCR-Abl [33, 211] have been shown to inhibit the nuclear localization of Rad51 and BRCA1, which are key components of HR.

Cells do not normally rely on one pathway and parallel pathways allow for repair to proceed if one fails [94]. Interestingly, cancer cells tend to be deficient in one or more parallel repair pathways [80], a feature that is very crucial and can be exploited for therapy. Many therapies targeting cancer, either chemotherapy or radiotherapy, rely on creating DSBs to cause cell death, but also cause cytotoxicity in normal cells. Thus, understanding the extent of functional redundancy is crucial to optimize cancer therapy, which led to the birth of the concept of synthetic lethality (SL). SL is defined as the lethal synergistic effect that comes from blocking two different pathways that, if blocked separately, do not lead to death. SL in HR-mutant cells was first identified when PARP inhibitors were shown to sensitize BRCA-defective cells [20, 61]. The identification of defective HR could provide a foundation to stratify patients for PARP inhbitor treatment [94]. Subsequent studies showed the sensitizing effect of PARP inhibitors on cells defective in different HR genes. It was revealed that a deficiency in RAD51, RAD54, DSS1, RPA1, NBS1, signaling proteins ATR and ATM, CHK1, CHK2, FANCD2, FANCA, or FANCC sensitized the cell to PARP inhibition [153]. Many exceptions and challenges remain. For example, deficiency in RAD52 did not induce sensitivity to PARP inhibition, which was attributed to the fact that RAD52 is more involved in the SSA pathway rather than gene conversion, which is the main repair pathway involved in PARP inhibited DSBs [153]. SL will be covered in more detail in Chap. 9.

Since RAD51 is a key component of HR, it is currently being established as the mainstay marker for HR [163]. RAD51 foci are used in clinical trials to test the response of primary breast cancer cells to neoadjuvant chemotherapy. A study showed that patients that exhibited low RAD51 score, suggestive of defective HR, were likely to respond well to anthracycline-based chemotherapy [76]. Another study tested a BRCA1 and BRCA2 array comparative genomic hybridization (aCGH) classifier, the level of BRCA1 promoter methylation, BRCA1 mRNA expression, and EMSY amplification, collectively assessing BRCA dysfunction. The results showed that a BRCA2-like aCGH profile is a powerful predictor of chemotherapy response in ER-positive breast cancers [133]. However, BRCA1 abnormalities were

not predictive of better response in triple negative tumors [133]. Loss of heterozygosity can be used as a marker for defective HR genes through the generation of a DNA-based HR deficiency score and thus can be used to predict the sensitivity of tumors to PARP inhibitors [2].

Non-Homologous End Joining

During G0, G1, and early S phase of the cell cycle, NHEJ is regarded as the main mechanism for DNA repair and is responsible for nearly 85% of DSB repair processes during these phases [129]. HR was considered the only repair mechanism that is error-free and also the cell's first choice for repair. When HR is not possible, due to either lack of sister chromatid or defective HR genes, cells turn to the erroneous NHEJ pathway. However, recent studies suggest that it is NHEJ that is first called into action, and only when NHEJ is unsuccessful does HR take charge to proceed with an error-free repair [23]. It is now considered that NHEJ is, in fact, a complex process that is not, in the slightest, secondary to HR [80].

NHEJ utilizes microhomology (typically 1-4 nucleotides) on the two ends of the DSB to repair the break in a simple ligation process. This is known as classical NHEJ (C-NHEJ). However, sometimes the already exposed overhangs are not compatible with each other, and so, simple ligation is not an option. This is where alternative NHEJ (A-NHEJ) takes place. A-NHEJ makes use of what is known as end processing, resection and polymerization, to create compatible overhangs (typically 6-8 nucleotides) for ligation. A-NHEJ is therefore considered a mutagenic process as it leads to deletions and/or additions of DNA sequences. The C-NHEJ pathway consists of synapsis, then end-processing, and finally ligation [239]. Synapsis is important in NHEJ as it keeps the two ends in close proximity or in alignment with each other. Synapsis is thought to rely on the ring shape of the Ku heterodimer, which consists of Ku70 and Ku86. Moreover, DNA-PKcs -in addition to the binding ability of the nucleosome itself-is suggested to play a similar role in synapsis [129]. Next, Ku's binding to the broken DNA end facilitates the binding of DNA-PKcs. This binding activates the DNA-PKcs's kinase function, which is rather important for the phosphorylation of Artemis, a nuclease used in end resection, and DNA-PKcs' autophosphorylation allows for its dissociation from the DNA. Autophosphorylation defects of the DNA-PKcs do not allow the subsequent steps of NHEJ to take place [240]. This may then redirect the cell towards HR instead of NHEJ. As with resection, gap filling using Polu creates compatible ends for ligation. Finally, XRCC4 mediates the interaction of the DNA ends with LigIV.

In the absence of the Ku heterodimer, A-NHEJ takes over the repair process. The characteristic feature for A-NHEJ is its repair through microhomology. However, this microhomology between the DNA ends is created at the expense of DNA sequences that are deleted in the process. The exonucleases responsible for this are thought to be CtIP and Mre11 [23]. Once end resection to uncover microhomology is complete, LIGIII or, to a lesser extent, LIGI seals the two ends.

Non-Homologous End Joining and Cancer

Translocations, whether caused by C-NHEJ or A-NHEJ, are considered a hallmark of cancer. Most translocations do not actually lead to fusion genes of a neomorphic nature that drive the cell into a hyperproliferative state, and so most of the translocations do not contribute to tumorigenesis. However, increased frequency of translocations in a tumor is considered an indicator of bad prognosis [23]. Because of the yet poorly understood A-NHEJ mechanism, it is unclear how much microhomology is utilized by the classic and alternative mechanisms to allow us to differentiate between them. Not only this, inconsistent results have been published suggesting that C-NHEJ is, in fact, responsible for most translocations that occur. In other words, science is still lacking solid evidence as to which pathway is responsible for these translocations, partially due to the poor understanding of the mechanisms in question.

As with HR, NHEJ repair pathway can be exploited for cancer therapy. Inhibitors that target different proteins known to be involved in NHEJ are under investigation. This inhibition is thought to redirect the cell towards more faithful pathways, such as HR. Inhibitors of DNA-PKcs are studied as part of combination therapy with chemotherapeutic agents [233]. NU7026 and NU7441, inhibitors of DNA-PKcs, enhanced the sensitivity of poor prognosis B-cell chronic lymphocytic leukemia (B-CLL) to topoisomerase II poisons [243]. In another study, NU7026 rendered pancreatic ductal adenocarcinoma cells not only sensitive to IR but also triggered apoptosis via the expression of cleaved caspase-3 [128]. Similarly, knockdown of NHEJ genes using shRNAs targeting Ku70 and Ku80 in 8988T cells boosted the cells' sensitivity to IR [128]. Known PI3Ks inhibitors LY294002 and Wortmannin also enhanced the cytotoxicity of DSB-forming agents; the former enhances the cytotoxicity to IR [199] and the latter to both etoposide and IR [16]. Further, in 2012, SCR7, an inhibitor of LIGIV, was discovered and shown to inhibit NHEJ by interfering with the ligase's DNA binding ability and so activating apoptotic pathways. In addition, the same study showed that SCR7 suppressed tumor progression in vivo [214]. As a form of SL, DNA-PK inhibitors could also be used in combination with chemotherapy in ATM-mutant tumours [138]. In fact, in the SL approach targeting HR (e.g. BRCA-2 deficient cells) through PARP inhibitors, resistance might occur. This resistance is attributed to deletions occurring in BRCA genes that could be mediated by A-NHEJ repair [57]. Interestingly, this resistance could be restored through NHEJ inhibition, a process called synthetic viability [17].

Moreover, the inhibition of DNA-PKcs in BRCA-deficient cells has been proven to decrease genome instability [179]. In a recent study, it was shown that knockdown of LIGIII, the ligase used for the error-prone A-NHEJ, sensitized KRASmutated leukemic cells to chemotherapy [77].

Selective inhibition of NHEJ factors in BRCA-deficient cells could be useful to treat cancer; however, total inhibition of NHEJ pathway will subject cells to massive DSB accumulation, which will ultimately lead to cancer progression. NHEJ could favor global genomic integrity at the price of some deletions that could go unnoticed due to their location at sequences with no known coding function [48].

Fanconi Anemia: A Cancer Predisposition Syndrome

Clinical Aspect of Fanconi Anemia

FA is a rare recessive disorder characterized by developmental abnormalities, bone marrow failure (BMF), and an increased cancer risk.

a. Developmental Abnormalities More than two thirds of FA patients have a broad spectrum of congenital abnormalities such as short stature, radial ray abnormalities, microcephaly, microphthalmia, and genital malformations [224]. The congenital malformations could be due to inappropriate increase in the p53-dependent apoptosis because of the inability of the FA-deficient cells to repair DNA damage during embryogenesis [135].

b. Bone Marrow Failure Anaemia as a consequence of BMF is usually the first life-threatening symptom with which individuals with FA present. During the first decade of life, most FA patients develop BMF, ranging from mild to severe [27, 117]. The fact that all blood lineages eventually become deficient strongly implies hematopoietic stem cell (HSC) dysfunction. Knockdown of FANCA and FANCD2 in human embryonic stem cells (hESC) results in reduced numbers of HSC and progenitor cells, suggesting an important role for the FA pathway during embryonic hematopoiesis [229]. Murine Fance-/- mice show reduced numbers of fetal liver HSC and progenitor cell pool with decline in serial repopulating capacity [102].

The role of FA pathway in stem cell function was recently revealed by discovering that FA-deficient fibroblasts are refractory to reprogramming, (induced pluripotent stem cells (iPSC)). The reprogramming barrier of FA fibroblasts could be bypassed by either genetic complementation or reprogramming under hypoxic conditions [192, 164]. Recently, Yung et al. [254] have reported derivation of iPSC lines from a FANCC-deficient patient under normoxic conditions, but at much reduced efficiency. The FANCC-deficient patient-specific iPSC lines and FANCC-deficient hESC result in reduced number of clonogenic hematopoietic progenitors due to increased apoptosis in culture.

c. Cancer Predisposition FA patients bear a significant predisposition to develop cancer. Patients exhibit high risk of hematopoietic malignancies, including myelodysplastic syndrome and acute leukemia. Acute lymphoblastic leukemia is occasionally reported, but AML is the most common. Patients with FA have 800-fold higher risk of developing AML than the general population [197, 198]. FA patients have a common pattern of specific chromosomal abnormalities, frequent gains of the chromosomal regions (1q and 3q), and partial or complete loss of chromosome 7, which can be used as predictive markers [157]. FA patients also bear a significant predisposition to solid tumour formation. By the fourth decade of life, about onethird of FA patients will develop a solid tumor, but this may be an underestimate as many patients do not live long enough to develop a tumor [198]. Of the solid tumors diagnosed in FA patients, squamous cell carcinoma of the head and neck and gynecological tract occur at a higher rate. These tumors are also associated with human papilloma virus, but the relative contribution of the virus to squamous cell carcinoma in FA patients is not clear [196].

FA Pathway

FA is caused by biallelic mutations in any of the 16 genes (FANCA, -B, -C, -D1, -D2, -E, -F, -G, -I, -J, -L, -M, -N, -P, -O, or -Q) [15, 104]. FA proteins can be subdivided into three groups of proteins: core complex, ID complex, and downstream effectors proteins. FA core complex consists of eight proteins (FANCA, -B, -C, -E, -F, -G, -L, and -M) and five FA-associated proteins (FAAP20, FAAP24, FAAP100, MHF1, and MHF2). Approximately 90% of FA patients have mutations in the eight FA core complex proteins [7, 36, 105, 122, 132, 247, 248]. The FA core complex is essential for monoubiquitination of the ID complex (FANCD2 and -I) by the E3 ubiquitin ligase FANCL. UBE2T functions as the E2 ligase, but it has not been associated with FA [46, 142]. Most FA core complex proteins, apart from FANCL and FANCM, do not have functional domains and they only act as a scaffold.

Monoubiquitinated ID complex recruits the downstream effector proteins which have affinity for the ubiquitin. Four of the downstream FA proteins (FANCD1/ BRCA2, FANCJ/BRIP1, FANCN/PALB2, and FANCO/RAD51C), also known as breast and ovarian cancer susceptibility genes, are necessary for HR to re-establish collapsed replication forks as a result of DSB formation during ICLs processing [87, 123, 231, 236, 244]. FANCP/SLX4 interacts with multiple nucleases, and FANCQ/ ERCC4 is a structural endonuclease (discussed below). In addition, ubiquitin-specific peptidase 1 (USP1) and the USP1-associated protein (UAF1) are necessary for the completion of FA pathway by regulating the deubiquitination of ID complex [38, 107, 166, 174].

Replication-Dependent Interstrand Crosslink Repair

Despite the genetic and phenotypic heterogeneity of FA, the hallmark of all FA patient-derived cells is hypersensitivity to interstrand crosslinking agents such as mitomycin C, diepoxybutane, and cisplatin. Exposure of FA cells to interstrand crosslinking agents induces high levels of chromosomal aberrations, which are utilized as a diagnostic feature for FA [10, 11]. ICLs are highly toxic because they act as an absolute block to both DNA transcription and DNA replication, so they are widely used in anticancer therapies. ICLs can be repaired by replication dependent and by replication independent pathways [241]. FA pathway is specifically activated during S-phase, suggesting a role for FA pathway in the repair of ICLs during replication [5, 222]. In the following part, we will discuss the steps of ICL repair by FA pathway during S-phase.

a. Recognition Among FA proteins core complex, FANCM is of particular interest because it acts as a sensor of damage and mediator of checkpoint signaling [39, 40]. FANCM has an evolutionarily conserved helicase domain bearing ATP-dependent DNA translocase activity [69, 70, 156]. This ATPase activity is dispensable for core complex targeting and ID complex ubiquitination, but it is required for replication fork stability and efficient checkpoint response [14, 40, 91]. The ATP-independent DNA binding activity of FANCM is important for ID complex ubiquitination and for FA core complex recruitment to chromatin [106, 160, 245]. FANCM forms a complex with FAAP24 and MHF1/MHF2 with distinct DNA binding specificities; the former prefers ssDNA, whereas the latter prefers dsDNA [36, 210, 247].

When a replication fork encounters an ICL, polymerization is arrested. The FANCM–FAAP24–MHF1/2 complex recognizes the stalled replication fork and recruits the FA core complex, and the translocase activity of FANCM prevents the collapse of replication fork [36, 106, 210, 247]. Beside the FANCM–FAAP24–MHF1/2 complex, the MutS has a redundant role in activating the FA pathway and recognizing ICLs by enhancing the recruitment of the FA core complex [90, 237, 242].

b. Signaling Replication obstacles and stalled forks are signaled by ATR. ATR activation proceeds in two largely independent steps. First, a stalled DNA replication fork generates a stretch of ssDNA covered with RPA, which in turn recruits ATRIP-ATR. In parallel, the Rad9-Rad1-Hus1 (9-1-1) checkpoint clamp, loaded onto DNA by the RAD17-RFC clamp loader, recruits TOPBP1. Interaction of ATRIP-ATR with the TOPBP1 leads to S-phase checkpoint activation [26, 63]. Recently, FANCM was found to be required for the activation of ATR-mediated checkpoint signaling, and the translocase activity of FANCM is essential for this role [40, 91, 141, 203]. FANCM and FAAP24 interact with the checkpoint protein HCLK2, and the depletion of FANCM or FAAP24 results in a phenotype similar to cells lacking HCLK2, ATR, or CHK1, including a high rate of spontaneous DNA lesion, nuclear abnormalities and supernumerary centrosomes, and checkpoint defects in cells subjected to replication stress [40].

As we discussed above, the RPA covered ssDNA is critical for ATR activation, but ICL cannot accumulate a sufficient amount of a long stretch of ssDNA because the accumulation of the ssDNA requires the uncoupling of DNA polymerase and DNA helicase at the stalled replication fork, and both are blocked by the ICL, so how ICL induces ATR activation is puzzling. A recent study has answered this question by showing that FANCM-FAAP24 complex is required for the recruitment of RPA to ICL-stalled replication forks [91]. The activated ATR is essential for checkpoint signaling via CHK1 phosphorylation [26, 37, 63], and it is also required for the FA pathway activation through FANCI phosphorylation [96, 207]. Recently, two distinct ATR signaling subpathways following cellular exposure to interstrand crosslinking agents have been proposed. One is dependent on RAD17 and TOPBP1 in which the important substrate is CHK1. This pathway is also affected by the FANCM-FAAP24 complex. The second is dependent on FA core complex where

it functions in ATR activation by enhancing chromatin binding of ATRIP, and the important substrate of this pathway is FANCI phosphorylation which will then activate FANCD2 monoubiquination [226].

The role of FANCM in cell signaling following exposure to interstrand crosslinking agents is not completely understood. The FANCM and FAAP24 have been implicated in checkpoint signaling in the above studies. However, Wang et al. [237] have recently found that FANCM and FAAP24 are not fully epistatic in repairing ICLs such that FAAP24 induces ATR-mediated checkpoint activation, while FANCM participates in recombination-independent ICL repair by promoting the recruitment of lesion incision activities, which requires FANCM translocase activity. In addition, the above reports placed FANCM upstream of ATR, but Singh et al. [209] have recently proposed that ATR precedes FANCM, because ATR phosphorylates FANCM (S1045) in response to genotoxic stress. This event is essential for the role of FANCM in FA pathway integrity, recruitment of FANCM to the DNA damaged site, prevention of premature mitotic entry, and the activation of CHK1 and G2/M checkpoints.

Coordination of Other DNA Repair Pathways During Interstrand Cross Link Repair

During ICL repair, FA pathway coordinates many repair pathways; excision repair, TLS, and HR. The FA pathway is essential for these processes as the nuclease and TLS steps depend on FANCD2 and its ubiquitination [111, 137, 190].

a. Nucleolytic Incision Following ICL recognition, it is thought that dual incisions are made 5' and 3' to the lesion. Many nucleases including FAN1, XPF, MUS81, SLX1, and SNM1A have been repoted to be implicated in ICL repair [78, 110, 165, 212, 217, 235].

a.i. FAN1 Many reports have identified FAN1 to be required for ICL repair [115, 136, 143, 205, 212]. FAN1 possesses intrinsic 5'-3' exonuclease activity and endonuclease activity that cleaves nicked and branched structures [212]. FAN1 interacts specifically with the monoubiquitinated FANCD2 through its UBZ domain, and through this interaction it is recruited to chromatin [115, 136, 143, 205, 212].

Although the above reports support a role for FAN1 in the FA pathway, there is also strong evidence against such a role. Although FAN1-depleted cells are sensitive to interstrand crosslinking agents, ICL-dependent DSBs arise in these cells with normal frequency but persist longer than in wild type cells. This suggests that FAN1 may have nuclease functions following the ICL incision step [115, 143]. FAN1-null DT40 cells do not have elevated SCE frequencies, a feature of all other FA mutant DT40 cells. In addition, FA proteins and FAN1 appear to have nonepistatic functions because DT40 cells deficient in both FAN1 and FANCC or FAN1 and FANCJ,

have increased sensitivity to interstrand crosslinking agents compared to cells deficient in only FAN1. This suggests that FAN1 may participate in the processing of ICL independent from the classical FA pathway [251]. In support of an independent role for FAN1 in ICL repair from FA pathway, a study of four patients carrying a homozygous 15q13.3 microdeletion (which includes FAN1) showed that the patients had no characteristic symptoms of FA [228]. On the other hand, [256] identified mutations in FAN1 as a cause of karyomegalic interstitial nephritis, a disorder that serves as a model for renal fibrosis. Karyomegalic interstitial nephritis has none of the characteristic hallmarks of FA, but cells from patients with karyomegalic interstitial nephritis have significant sensitivity to interstrand crosslinking agents, which is complemented with wild-type FAN1.

a.ii. SLX4 SLX4/FANCP is a recently identified FA protein. Cells depleted of SLX4 are hypersensitive to interstrand crosslinking agents [43, 165]. Slx4-null mice have many key features of FA, including developmental defects, reduced fertility, and defect in the hematopoietic compartment [43]. SLX4 contains UBZ domain that is required for interaction with monoubiquitylated FANCD2 and for recruitment to DNA damage [246]. SLX4 functions as a scaffold for three structure-specific nucleases: XPF-ERCC1, MUS81-EME1, and SLX1. Genetic studies indicate that XPF-ERCC1 is the most relevant nuclease for ICL repair [12, 62, 109, 110, 165, 217]. The SLX4-dependent XPF-ERCC1 activity is critical for ICL repair but is dispensable for topoisomerase I poison-induced DNA damage repair. Conversely, SLX4-MUS81 is essential for topoisomerase I poison-induced DNA damage repair but is less critical for ICL repair. Mutation of SLX4 that inhibits interaction with SLX1 leads to partial resistance to interstrand crosslinking agents and topoisomerase I poison [110]. Recently, XPF was identified as FA protein (FANCO). Whole exome analysis of unclassified FA individuals revealed biallelic germline mutations in XPF encoding gene (ERCC4) [15]. Interestingly, another patient with defect in XPF encoding gene was reported to have clinical features of three different DNA-repair disorders-Cockayne syndrome, xeroderma pigmentosum, and FA [104]. SNM1A may digest the unhooked oligonucleotides to create a better substrate for the TLS. SNM1A-depleted cells are sensitive to ICL agents [235].

Translesion Synthesis

Following incision of ICL, TLS is required to bypass the lesion. Since ICL lesion involves the two strands of the DNA, both strands cannot be used as the template. TLS uses low-fidelity polymerases to bypass the bulky damaged lesions to generate an intact template for HR-mediated repair. Monoubiquitylated ID complex is required for TLS steps during ICL repair [111]. PCNA acts as scaffold to which TLS polymerases bind [8]. Cells with defect in the TLS polymerases, Rev1 and Rev3, are hypersensitive to ICL agents [172, 173, 208].

Homologous Recombination

Following TLS, HR uses the produced intact template to repair the DSB generated by incision. Cells deficient in HR proteins are sensitive to interstrand crosslinking agents [13, 114, 161]. FA pathway is involved in HR activation, and cells lacking FA proteins are deficient in activating HR [167, 212]. In DT40 cells, FANCC and XRCC2 were shown to be epistatic in ICL repair [172]. However, FANCD2 and FANCI do not regulate chromatin loading of the key HR enzyme RAD51 [137].

In addition to promoting HR, many studies proposed that FA proteins actively suppress NHEJ. As we discussed above, the cell can repair DSB by HR or NHEJ. The choice depends on the cell cycle phase, where the HR works on S-phase and the NHEJ works on G1 phase. In addition, HR requires extensive resection to create a long 3' overhang, but NHEJ requires little, if any, resection [92]. The FA pathway is thought to have a role in pathway choice by funneling the DSB created by ICL processing into HR. Inhibition of NHEJ components in FA-deficient cells suppresses hypersensitivity to interstrand crosslinking agents, diminishes chromosome breaks, and reverses defective HR [3, 176]. In contrast, deletion of 53BP1 or Ku80 increases the sensitivity of FANCD2-deficient cells to interstrand crosslinking agents than FANCD2 deficiency alone [22].

Another Replication-Dependent Pathway for Interstrand Crosslink Repair

Recently, a new mechanism for ICL repair during S-phase was revealed. HELQ, 3'-5' DNA helicase with strand displacement activity, was found to participate in ICL repair independent from FA. HELQ helicase-deficient mice exhibit subfertility, germcell attrition, ICL sensitivity, and tumour predisposition. HELQ interacts directly with the RAD51 paralog complex BCDX2 and functions in parallel to the FA pathway to promote efficient HR at damaged replication forks. HELQ may also be involved in ATR-mediated CHK1 activation but not in ATR-mediated FANCD2 monoubiquitination. Thus, HELQ has a critical role in replication-coupled ICL repair, germ cell maintenance, and tumour suppression in mammals [4, 219].

Fanconi Anemia Pathway, Bifunctional Alkylating agent, and Therapy-Related Myeloid Neoplasms

ICLs induced by bifunctional alkylating agents result in the formation of DSBs which may lead to chromosomal rearrangements and in turn t-MN. Since t-MN arises in patients who have already developed one type of cancer, the inherited cancer susceptibility mutations may be involved in tumorigenesis [31].

References

- 1. Abbotts R, Madhusudan S (2012) Targeting DNA base excision repair: a new strategy for personalised cancer therapy. Clin Med 12(6):42–46. doi:10.7861/clinmedicine.12-6-s42
- Abkevich V, Timms KM, Hennessy BT, Potter J, Carey MS, Meyer LA, Smith-McCune K, Broaddus R, Lu KH, Chen J, Tran TV, Williams D, Iliev D, Jammulapati S, FitzGerald LM, Krivak T, DeLoia JA, Gutin A, Mills GB, Lanchbury JS (2012) Patterns of genomic loss of heterozygosity predict homologous recombination repair defects in epithelial ovarian cancer. Br J Cancer 107(10):1776–1782. doi:10.1038/bjc.2012.451
- Adamo A, Collis SJ, Adelman CA, Silva N, Horejsi Z, Ward JD, Martinez-Perez E, Boulton SJ, La Volpe A (2010) Preventing nonhomologous end joining suppresses DNA repair defects of Fanconi anemia. Mol Cell 39(1):25–35. doi:10.1016/j.molcel.2010.06.026
- Adelman CA, Lolo RL, Birkbak NJ, Murina O, Matsuzaki K, Horejsi Z, Parmar K, Borel V, Skehel JM, Stamp G, D'Andrea A, Sartori AA, Swanton C, Boulton SJ (2013). HELQ promotes RAD51 paralogue-dependent repair to avert germ cell loss and tumorigenesis. Nature 502(7471):381–384. doi:10.1038/nature12565
- Akkari YM, Bateman RL, Reifsteck CA, D'Andrea AD, Olson SB, Grompe M (2001) The 4N cell cycle delay in Fanconi anemia reflects growth arrest in late S phase. Mol Genet Metab 74(4):403–412
- Alagoz M, Wells OS, El-Khamisy SF (2014) TDP1 deficiency sensitizes human cells to base damage via distinct topoisomerase I and PARP mechanisms with potential applications for cancer therapy. Nucleic Acids Res 42(5):3089–3103. doi:10.1093/nar/gkt1260
- Ali AM, Pradhan A, Singh TR, Du C, Li J, Wahengbam K, Grassman E, Auerbach AD, Pang Q, Meetei AR (2012) FAAP20: a novel ubiquitin-binding FA nuclear core-complex protein required for functional integrity of the FA-BRCA DNA repair pathway. Blood 119(14):3285–3294. doi:10.1182/blood-2011-10-385963
- Andersen PL, Xu F, Xiao W (2008) Eukaryotic DNA damage tolerance and translesion synthesis through covalent modifications of PCNA. Cell Res 18(1):162–173. doi:10.1038/ cr.2007.114
- Arlow T, Scott K, Wagenseller A, Gammie A (2013) Proteasome inhibition rescues clinically significant unstable variants of the mismatch repair protein Msh2. Proc Natl Acad Sci U S A 110(1):246–251. doi:10.1073/pnas.1215510110
- 10. Auerbach AD (1993) Fanconi anemia diagnosis and the diepoxybutane (DEB) test. Exp hematol 21(6):731–733
- 11. Auerbach AD (2009) Fanconi anemia and its diagnosis. Mutat Res 668:4-10
- Bhagwat N, Olsen AL, Wang AT, Hanada K, Stuckert P, Kanaar R, D'Andrea A, Niedernhofer LJ, McHugh PJ (2009) XPF-ERCC1 participates in the Fanconi anemia pathway of crosslink repair. Mol Cell Biol 29(24):6427–6437. doi:10.1128/MCB.00086-09
- Bhattacharyya A, Ear US, Koller BH, Weichselbaum RR, Bishop DK (2000) The breast cancer susceptibility gene BRCA1 is required for subnuclear assembly of Rad51 and survival following treatment with the DNA cross-linking agent cisplatin. J Biol Chem 275(31):23899–23903. doi:10.1074/jbc.C000276200
- Blackford AN, Schwab RA, Nieminuszczy J, Deans AJ, West SC, Niedzwiedz W (2012) The DNA translocase activity of FANCM protects stalled replication forks. Hum Mol Genet 21(9):2005–2016. doi:10.1093/hmg/dds013
- Bogliolo M, Schuster B, Stoepker C, Derkunt B, Su Y, Raams A, Trujillo JP, Minguillon J, Ramirez MJ, Pujol R, Casado JA, Banos R, Rio P, Knies K, Zuniga S, Benitez J, Bueren JA, Jaspers NG, Scharer OD, de Winter JP, Schindler D, Surralles J (2013) Mutations in ERCC4, encoding the DNA-repair endonuclease XPF, cause Fanconi anemia. Am J Hum Genet 92(5):800–806. doi:10.1016/j.ajhg.2013.04.002
- Boulton S, Kyle S, Durkacz BW (2000) Mechanisms of enhancement of cytotoxicity in etoposide and ionising radiation-treated cells by the protein kinase inhibitor wortmannin. Eur J Cancer 36(4):535–541

- 17. Bouwman P, Aly A, Escandell JM, Pieterse M, Bartkova J, van der Gulden H, Hiddingh S, Thanasoula M, Kulkarni A, Yang Q, Haffty BG, Tommiska J, Blomqvist C, Drapkin R, Adams DJ, Nevanlinna H, Bartek J, Tarsounas M, Ganesan S, Jonkers J (2010) 53BP1 loss rescues BRCA1 deficiency and is associated with triple-negative and BRCA-mutated breast cancers. Nat Struct Mol Biol 17(6):688–695. doi:10.1038/nsmb.1831
- Branch P, Aquilina G, Bignami M, Karran P (1993) Defective mismatch binding and a mutator phenotype in cells tolerant to DNA damage. Nature 362(6421):652–654. doi:10.1038/362652a0
- Bronner CE, Baker SM, Morrison PT, Warren G, Smith LG, Lescoe MK, Kane M, Earabino C, Lipford J, Lindblom A et al (1994) Mutation in the DNA mismatch repair gene homologue hMLH1 is associated with hereditary non-polyposis colon cancer. Nature 368(6468):258–261. doi:10.1038/368258a0
- Bryant HE, Schultz N, Thomas HD, Parker KM, Flower D, Lopez E, Kyle S, Meuth M, Curtin NJ, Helleday T (2005) Specific killing of BRCA2-deficient tumours with inhibitors of poly(ADP-ribose) polymerase. Nature 434(7035):913–917. doi:10.1038/nature03443
- Budke B, Logan HL, Kalin JH, Zelivianskaia AS, Cameron McGuire W, Miller LL, Stark JM, Kozikowski AP, Bishop DK, Connell PP (2012) RI-1: a chemical inhibitor of RAD51 that disrupts homologous recombination in human cells. Nucleic Acids Res 40(15):7347–7357. doi:10.1093/nar/gks353
- Bunting SF, Callen E, Kozak ML, Kim JM, Wong N, Lopez-Contreras AJ, Ludwig T, Baer R, Faryabi RB, Malhowski A, Chen HT, Fernandez-Capetillo O, D'Andrea A, Nussenzweig A (2012) BRCA1 functions independently of homologous recombination in DNA interstrand crosslink repair. Mol Cell 46(2):125–135. doi:10.1016/j.molcel.2012.02.015
- Bunting SF, Nussenzweig A (2013) End-joining, translocations and cancer. Nat Rev Cancer 13(7):443–454. doi:10.1038/nrc3537
- Burdett V, Baitinger C, Viswanathan M, Lovett ST, Modrich P (2001) In vivo requirement for RecJ, ExoVII, ExoI, and ExoX in methyl-directed mismatch repair. Proc Natl Acad Sci U S A 98(12):6765–6770. doi:10.1073/pnas.121183298
- Burgers PM (2009) Polymerase dynamics at the eukaryotic DNA replication fork. J Biol Chem 284(7):4041–4045. doi:10.1074/jbc.R800062200
- Burrows AE, Elledge SJ (2008) How ATR turns on: TopBP1 goes on ATRIP with ATR. Genes Dev 22(11):1416–1421. doi:10.1101/gad.1685108
- Butturini A, Gale RP, Verlander PC, Adler-Brecher B, Gillio AP, Auerbach AD (1994) Hematologic abnormalities in Fanconi anemia: an International Fanconi Anemia Registry study. Blood 84(5):1650–1655
- Caldecott KW (2007) Mammalian single-strand break repair: mechanisms and links with chromatin. DNA Repair 6(4):443–453. doi:10.1016/j.dnarep.2006.10.006
- 29. Caldecott KW (2008) Single-strand break repair and genetic disease. Nat Rev Genet 9(8):619–631. doi:10.1038/nrg2380
- Cannavo E, Marra G, Sabates-Bellver J, Menigatti M, Lipkin SM, Fischer F, Cejka P, Jiricny J (2005) Expression of the MutL homologue hMLH3 in human cells and its role in DNA mismatch repair. Cancer Res 65(23):10759–10766. doi:10.1158/0008-5472.CAN-05-2528
- Casorelli I, Bossa C, Bignami M (2012) DNA damage and repair in human cancer: molecular mechanisms and contribution to therapy-related leukemias. Int J Environ Res Public Health 9(8):2636–2657. doi:10.3390/ijerph9082636
- Chernikova SB, Game JC, Brown JM (2012) Inhibiting homologous recombination for cancer therapy. Cancer Biol Ther 13(2):61–68. doi:10.4161.cbt.13.2.18872, doi:10.4161/ cbt.13.2.18872
- Choudhury A, Zhao H, Jalali F, Al Rashid S, Ran J, Supiot S, Kiltie AE, Bristow RG (2009) Targeting homologous recombination using imatinib results in enhanced tumor cell chemosensitivity and radiosensitivity. Mol Cancer Ther 8(1):203–213. doi:10.1158/1535-7163. MCT-08-0959
- 34. Church DN, Briggs SE, Palles C, Domingo E, Kearsey SJ, Grimes JM, Gorman M, Martin L, Howarth KM, Hodgson SV, Kaur K, Taylor J, Tomlinson IP (2013) DNA polymerase

epsilon and delta exonuclease domain mutations in endometrial cancer. Hum Mol Genet 22(14):2820–2828. doi:10.1093/hmg/ddt131

- Ciccia A, Elledge SJ (2010) The DNA damage response: making it safe to play with knives. Mol Cell 40(2):179–204. doi:10.1016/j.molcel.2010.09.019
- Ciccia A, Ling C, Coulthard R, Yan Z, Xue Y, Meetei AR, Laghmani el H, Joenje H, McDonald N, de Winter JP, Wang W, West SC (2007) Identification of FAAP24, a Fanconi anemia core complex protein that interacts with FANCM. Mol Cell 25(3):331–343. doi:10.1016/j.molcel.2007.01.003
- Cimprich KA, Cortez D (2008) ATR: an essential regulator of genome integrity. Nat Rev Mol Cell Biol 9(8):616–627. doi:10.1038/nrm2450
- Cohn MA, Kowal P, Yang K, Haas W, Huang TT, Gygi SP, D'Andrea AD (2007) A UAF1containing multisubunit protein complex regulates the Fanconi anemia pathway. Mol Cell 28(5):786–797. doi:10.1016/j.molcel.2007.09.031
- Collis SJ, Barber LJ, Clark AJ, Martin JS, Ward JD, Boulton SJ (2007) HCLK2 is essential for the mammalian S-phase checkpoint and impacts on Chk1 stability. Nat Cell Biol 9(4):391–401. doi:10.1038/ncb1555
- Collis SJ, Ciccia A, Deans AJ, Horejsi Z, Martin JS, Maslen SL, Skehel JM, Elledge SJ, West SC, Boulton SJ (2008) FANCM and FAAP24 function in ATR-mediated checkpoint signaling independently of the Fanconi anemia core complex. Mol Cell 32(3):313–324. doi:10.1016/j. molcel.2008.10.014
- Constantin N, Dzantiev L, Kadyrov FA, Modrich P (2005) Human mismatch repair: reconstitution of a nick-directed bidirectional reaction. J Biol Chem 280(48):39752–39761. doi:10.1074/jbc.M509701200
- 42. Cooke MS, Evans MD, Dizdaroglu M, Lunec J (2003) Oxidative DNA damage: mechanisms, mutation, and disease. FASEB J 17(10):1195–1214. doi:10.1096/fj.02-0752rev
- 43. Crossan GP, van der Weyden L, Rosado IV, Langevin F, Gaillard PH, McIntyre RE, Gallagher F, Kettunen MI, Lewis DY, Brindle K, Arends MJ, Adams DJ, Patel KJ (2011) Disruption of mouse Slx4, a regulator of structure-specific nucleases, phenocopies Fanconi anemia. Nat Genet 43(2):147–152. doi:10.1038/ng.752
- Curtin NJ (2012) DNA repair dysregulation from cancer driver to therapeutic target. Nat Rev Cancer 12(12):801–817. doi:10.1038/nrc3399
- 45. de Wind N, Dekker M, Berns A, Radman M, te Riele H (1995) Inactivation of the mouse Msh2 gene results in mismatch repair deficiency, methylation tolerance, hyperrecombination, and predisposition to cancer. Cell 82(2):321–330
- de Winter JP, Joenje H (2009) The genetic and molecular basis of Fanconi anemia. Mutat Res 668(1–2):11–19. doi:10.1016/j.mrfmmm.2008.11.004
- Deem A, Keszthelyi A, Blackgrove T, Vayl A, Coffey B, Mathur R, Chabes A, Malkova A (2011) Break-induced replication is highly inaccurate. PLoS Biol 9(2):e1000594. doi:10.1371/ journal.pbio.1000594
- Deriano L, Roth DB (2013) Modernizing the nonhomologous end-joining repertoire: alternative and classical NHEJ share the stage. Annu Rev Genet 47:433–455. doi:10.1146/annurevgenet-110711-155540
- Dianov G, Price A, Lindahl T (1992) Generation of single-nucleotide repair patches following excision of uracil residues from DNA. Mol Cell Biol 12(4):1605–1612
- Drost M, Lutzen A, van Hees S, Ferreira D, Calleja F, Zonneveld JB, Nielsen FC, Rasmussen LJ, de Wind N (2013) Genetic screens to identify pathogenic gene variants in the common cancer predisposition Lynch syndrome. Proc Natl Acad Sci U S A 110(23):9403–9408. doi:10.1073/ pnas.1220537110
- Drummond JT, Genschel J, Wolf E, Modrich P (1997) DHFR/MSH3 amplification in methotrexate-resistant cells alters the hMutSalpha/hMutSbeta ratio and reduces the efficiency of base-base mismatch repair. Proc Natl Acad Sci U S A 94(19):10144–10149
- 52. Drummond JT, Li GM, Longley MJ, Modrich P (1995) Isolation of an hMSH2-p160 heterodimer that restores DNA mismatch repair to tumor cells. Sci 268(5219):1909–1912
- Dupre A, Boyer-Chatenet L, Sattler RM, Modi AP, Lee JH, Nicolette ML, Kopelovich L, Jasin M, Baer R, Paull TT, Gautier J (2008) A forward chemical genetic screen reveals an

inhibitor of the Mre11-Rad50-Nbs1 complex. Nat Chem Biol 4(2):119-125. doi:10.1038/ nchembio.63

- 54. Duval A, Hamelin R (2002a) Genetic instability in human mismatch repair deficient cancers. Ann Genet 45(2):71–75
- Duval A, Hamelin R (2002b) Mutations at coding repeat sequences in mismatch repairdeficient human cancers: toward a new concept of target genes for instability. Cancer Res 62(9):2447–2454
- Dzantiev L, Constantin N, Genschel J, Iyer RR, Burgers PM, Modrich P (2004) A defined human system that supports bidirectional mismatch-provoked excision. Mol Cell 15(1):31–41. doi:10.1016/j.molcel.2004.06.016
- 57. Edwards SL, Brough R, Lord CJ, Natrajan R, Vatcheva R, Levine DA, Boyd J, Reis-Filho JS, Ashworth A (2008) Resistance to therapy caused by intragenic deletion in BRCA2. Nature 451(7182):1111–1115. doi:10.1038/nature06548
- Eich M, Roos WP, Nikolova T, Kaina B (2013) Contribution of ATM and ATR to the resistance of glioblastoma and malignant melanoma cells to the methylating anticancer drug temozolomide. Mol Cancer Ther 12(11):2529–2540. doi:10.1158/1535-7163.MCT-13-0136
- El-Khamisy SF, Masutani M, Suzuki H, Caldecott KW (2003) A requirement for PARP-1 for the assembly or stability of XRCC1 nuclear foci at sites of oxidative DNA damage. Nucleic Acids Res 31(19):5526–5533
- Ellegren H (2004) Microsatellites: simple sequences with complex evolution. Nat Rev Genet 5(6):435–445. doi:10.1038/nrg1348
- Farmer H, McCabe N, Lord CJ, Tutt AN, Johnson DA, Richardson TB, Santarosa M, Dillon KJ, Hickson I, Knights C, Martin NM, Jackson SP, Smith GC, Ashworth A (2005) Targeting the DNA repair defect in BRCA mutant cells as a therapeutic strategy. Nature 434(7035):917– 921. doi:10.1038/nature03445
- Fekairi S, Scaglione S, Chahwan C, Taylor ER, Tissier A, Coulon S, Dong MQ, Ruse C, Yates JR, 3rd, Russell P, Fuchs RP, McGowan CH, Gaillard PH (2009) Human SLX4 is a Holliday junction resolvase subunit that binds multiple DNA repair/recombination endonucleases. Cell 138(1):78–89. doi:10.1016/j.cell.2009.06.029
- Flynn RL, Zou L (2011) ATR: a master conductor of cellular responses to DNA replication stress. Trends Biochem Sci 36(3):133–140. doi:10.1016/j.tibs.2010.09.005
- 64. Fokas E, Prevo R, Pollard JR, Reaper PM, Charlton PA, Cornelissen B, Vallis KA, Hammond EM, Olcina MM, Gillies McKennaW, Muschel RJ, Brunner TB (2012) Targeting ATR in vivo using the novel inhibitor VE-822 results in selective sensitization of pancreatic tumors to radiation. Cell Death Dis 3:e441. doi:10.1038/cddis.2012.181
- 65. Foksinski M, Kotzbach R, Szymanski W, Olinski R (2000) The level of typical biomarker of oxidative stress 8-hydroxy-2'-deoxyguanosine is higher in uterine myomas than in control tissues and correlates with the size of the tumor. Free Radic Biol Med 29(7):597–601
- Fortini P, Dogliotti E (2007) Base damage and single-strand break repair: mechanisms and functional significance of short- and long-patch repair subpathways. DNA Repair 6(4):398– 409. doi:10.1016/j.dnarep.2006.10.008
- 67. Fortini P, Pascucci B, Parlanti E, Sobol RW, Wilson SH, Dogliotti E (1998) Different DNA polymerases are involved in the short- and long-patch base excision repair in mammalian cells. Biochemistry 37(11):3575–3580. doi:10.1021/bi972999h
- Frosina G, Fortini P, Rossi O, Carrozzino F, Raspaglio G, Cox LS, Lane DP, Abbondandolo A, Dogliotti E (1996) Two pathways for base excision repair in mammalian cells. J Biol Chem 271(16):9573–9578
- Gari K, Decaillet C, Delannoy M, Wu L, Constantinou A (2008a) Remodeling of DNA replication structures by the branch point translocase FANCM. Proc Natl Acad Sci U S A 105(42):16107–16112. doi:10.1073/pnas.0804777105
- Gari K, Decaillet C, Stasiak AZ, Stasiak A, Constantinou A (2008b) The Fanconi anemia protein FANCM can promote branch migration of Holliday junctions and replication forks. Mol Cell 29(1):141–148. doi:10.1016/j.molcel.2007.11.032
- Gerson SL (2004) MGMT: its role in cancer aetiology and cancer therapeutics. Nat Rev Cancer 4(4):296–307. doi:10.1038/nrc1319
- Ghodgaonkar MM, Lazzaro F, Olivera-Pimentel M, Artola-Boran M, Cejka P, Reijns MA, Jackson AP, Plevani P, Muzi-Falconi M, Jiricny J (2013) Ribonucleotides misincorporated into DNA act as strand-discrimination signals in eukaryotic mismatch repair. Mol Cell 50(3):323–332. doi:10.1016/j.molcel.2013.03.019
- 73. Giglia-Mari G, Zotter A, Vermeulen W (2011) DNA damage response. Cold Spring Harb Perspect Biol 3(1):a000745. doi:10.1101/cshperspect.a000745
- Godley LA, Larson RA (2008) Therapy-related myeloid leukemia. Semin Oncol 35(4):418– 429. doi:10.1053/j.seminoncol.2008.04.012
- Goellner EM, Grimme B, Brown AR, Lin YC, Wang XH, Sugrue KF, Mitchell L, Trivedi RN, Tang JB, Sobol RW (2011) Overcoming temozolomide resistance in glioblastoma via dual inhibition of NAD+ biosynthesis and base excision repair. Cancer Res 71(6):2308–2317. doi:10.1158/0008-5472.CAN-10-3213
- Graeser M, McCarthy A, Lord CJ, Savage K, Hills M, Salter J, Orr N, Parton M, Smith IE, Reis-Filho JS, Dowsett M, Ashworth A, Turner NC (2010) A marker of homologous recombination predicts pathologic complete response to neoadjuvant chemotherapy in primary breast cancer. Clin Cancer Res 16(24):6159–6168. doi:10.1158/1078-0432.CCR-10-1027
- Hahnel PS, Enders B, Sasca D, Roos WP, Kaina B, Bullinger L, Theobald M, Kindler T (2014) Targeting components of the alternative NHEJ pathway sensitizes KRAS mutant leukemic cells to chemotherapy. Blood 123(15):2355–2366. doi:10.1182/blood-2013-01-477620
- Hanada K, Budzowska M, Modesti M, Maas A, Wyman C, Essers J, Kanaar R (2006) The structure-specific endonuclease Mus81-Eme1 promotes conversion of interstrand DNA crosslinks into double-strands breaks. EMBO J 25(20):4921–4932. doi:10.1038/sj.emboj.7601344
- Harrington JM, Kolodner RD (2007) Saccharomyces cerevisiae Msh2-Msh3 acts in repair of base-base mispairs. Mol Cell Bio 27(18):6546–6554. doi:10.1128/MCB.00855-07
- Helleday T (2010) Homologous recombination in cancer development, treatment and development of drug resistance. Carcinogenesis 31(6):955–960. doi:10.1093/carcin/bgq064
- Helleday T, Lo J, van Gent DC, Engelward BP (2007) DNA double-strand break repair: from mechanistic understanding to cancer treatment. DNA Repair 6(7):923–935. doi:10.1016/j. dnarep.2007.02.006
- Herman JG, Umar A, Polyak K, Graff JR, Ahuja N, Issa JP, Markowitz S, Willson JK, Hamilton SR, Kinzler KW, Kane MF, Kolodner RD, Vogelstein B, Kunkel TA, Baylin SB (1998) Incidence and functional consequences of hMLH1 promoter hypermethylation in colorectal carcinoma. Proc Natl Acad Sci U S A 95(12):6870–6875
- Hickman MJ, Samson LD (2004) Apoptotic signaling in response to a single type of DNA lesion, O(6)-methylguanine. Mol Cell 14(1):105–116
- Hiller B, Achleitner M, Glage S, Naumann R, Behrendt R, Roers A (2012) Mammalian RNase H2 removes ribonucleotides from DNA to maintain genome integrity. J Exp Med 209(8):1419–1426. doi:10.1084/jem.20120876
- Hoeijmakers JH (2009) DNA damage, aging, and cancer. N Engl J Med 361(15):1475–1485. doi:10.1056/NEJMra0804615
- Holmes J Jr, Clark S, Modrich P (1990) Strand-specific mismatch correction in nuclear extracts of human and Drosophila melanogaster cell lines. Proc Natl Acad Sci U S A 87(15):5837–5841
- Howlett NG, Taniguchi T, Olson S, Cox B, Waisfisz Q, De Die-Smulders C, Persky N, Grompe M, Joenje H, Pals G, Ikeda H, Fox EA, D'Andrea AD (2002) Biallelic inactivation of BRCA2 in Fanconi anemia. Science 297(5581):606–609. doi:10.1126/science.1073834
- Hsieh P, Yamane K (2008) DNA mismatch repair: molecular mechanism, cancer, and ageing. Mech Ageing Dev 129(7–8):391–407. doi:10.1016/j.mad.2008.02.012
- Huang F, Motlekar NA, Burgwin CM, Napper AD, Diamond SL, Mazin AV (2011a) Identification of specific inhibitors of human RAD51 recombinase using high-throughput screening. ACS Chem Biol 6(6):628–635. doi:10.1021/cb100428c
- Huang M, Kennedy R, Ali AM, Moreau LA, Meetei AR, D'Andrea AD, Chen CC (2011b) Human MutS and FANCM complexes function as redundant DNA damage sensors in the Fanconi Anemia pathway. DNA Repair 10(12):1203–1212. doi:10.1016/j.dnarep.2011.09.006

- Huang M, Kim JM, Shiotani B, Yang K, Zou L, D'Andrea AD (2010) The FANCM/ FAAP24 complex is required for the DNA interstrand crosslink-induced checkpoint response. Mol Cell 39(2):259–268. doi:10.1016/j.molcel.2010.07.005
- 92. Huertas P (2010) DNA resection in eukaryotes: deciding how to fix the break. Nat Struct Mol Bio 17(1):11–16. doi:10.1038/nsmb.1710
- Huertas P, Cortes-Ledesma F, Sartori AA, Aguilera A, Jackson SP (2008) CDK targets Sae2 to control DNA-end resection and homologous recombination. Nature 455(7213):689–692. doi:10.1038/nature07215
- Huhn D, Bolck HA, Sartori AA (2013) Targeting DNA double-strand break signalling and repair: recent advances in cancer therapy. Swiss Med Wkly 143:w13837. doi:10.4414/ smw.2013.13837
- 95. Isaacs RJ, Spielmann HP (2004) A model for initial DNA lesion recognition by NER and MMR based on local conformational flexibility. DNA Repair 3(5):455–464
- 96. Ishiai M, Kitao H, Smogorzewska A, Tomida J, Kinomura A, Uchida E, Saberi A, Kinoshita E, Kinoshita-Kikuta E, Koike T, Tashiro S, Elledge SJ, Takata M (2008) FANCI phosphorylation functions as a molecular switch to turn on the Fanconi anemia pathway. Nat Struct Mol Bio 15(11):1138–1146. doi:10.1038/nsmb.1504
- 97. Jiricny J (1994) Colon cancer and DNA repair: have mismatches met their match? Trends Genet (TIG) 10(5):164–168
- Jiricny J (2006) The multifaceted mismatch-repair system. Nat Rev Mol Cell Bio 7(5): 335–346. doi:10.1038/nrm1907
- Jung B, Doctolero RT, Tajima A, Nguyen AK, Keku T, Sandler RS, Carethers JM (2004) Loss of activin receptor type 2 protein expression in microsatellite unstable colon cancers. Gastroenterology 126(3):654–659
- Kadyrov FA, Dzantiev L, Constantin N, Modrich P (2006) Endonucleolytic function of Mut-Lalpha in human mismatch repair. Cell 126(2):297–308. doi:10.1016/j.cell.2006.05.039
- 101. Kadyrov FA, Holmes SF, Arana ME, Lukianova OA, O'Donnell M, Kunkel TA, Modrich P (2007) Saccharomyces cerevisiae MutLalpha is a mismatch repair endonuclease. J Biol Chem 282(51):37181–37190. doi:10.1074/jbc.M707617200
- Kamimae-Lanning AN, Goloviznina NA, Kurre P (2013) Fetal origins of hematopoietic failure in a murine model of Fanconi anemia. Blood 121(11):2008–2012. doi:10.1182/ blood-2012-06-439679
- Kantake N, Sugiyama T, Kolodner RD, Kowalczykowski SC (2003) The recombinationdeficient mutant RPA (rfa1-t11) is displaced slowly from single-stranded DNA by Rad51 protein. J Biol Chem 278(26):23410–23417. doi:10.1074/jbc.M302995200
- 104. Kashiyama K, Nakazawa Y, Pilz DT, Guo C, Shimada M, Sasaki K, Fawcett H, Wing JF, Lewin SO, Carr L, Li TS, Yoshiura K, Utani A, Hirano A, Yamashita S, Greenblatt D, Nardo T, Stefanini M, McGibbon D, Sarkany R, Fassihi H, Takahashi Y, Nagayama Y, Mitsutake N, Lehmann AR, Ogi T (2013) Malfunction of nuclease ERCC1-XPF results in diverse clinical manifestations and causes Cockayne syndrome, xeroderma pigmentosum, and Fanconi anemia. Am J Hum Genet 92(5):807–819. doi:10.1016/j.ajhg.2013.04.007
- Kim H, Yang K, Dejsuphong D, D'Andrea AD (2012) Regulation of Rev1 by the Fanconi anemia core complex. Nat Struct Mol Bio 19(2):164–170. doi:10.1038/nsmb.2222
- Kim JM, Kee Y, Gurtan A, D'Andrea AD (2008) Cell cycle-dependent chromatin loading of the Fanconi anemia core complex by FANCM/FAAP24. Blood 111(10):5215–5222. doi:10.1182/blood-2007-09-113092
- Kim JM, Parmar K, Huang M, Weinstock DM, Ruit CA, Kutok JL, D'Andrea AD (2009) Inactivation of murine Usp1 results in genomic instability and a Fanconi anemia phenotype. Dev Cell 16(2):314–320. doi:10.1016/j.devcel.2009.01.001
- Kim TM, Laird PW, Park PJ (2013a) The landscape of microsatellite instability in colorectal and endometrial cancer genomes. Cell 155(4):858–868. doi:10.1016/j.cell.2013.10.015
- Kim Y, Lach FP, Desetty R, Hanenberg H, Auerbach AD, Smogorzewska A (2011) Mutations of the SLX4 gene in Fanconi anemia. Nat Genet 43(2):142–146. doi:10.1038/ng.750

- Kim Y, Spitz GS, Veturi U, Lach FP, Auerbach AD, Smogorzewska A (2013b) Regulation of multiple DNA repair pathways by the Fanconi anemia protein SLX4. Blood 121(1): 54–63. doi:10.1182/blood-2012-07-441212
- 111. Knipscheer P, Raschle M, Smogorzewska A, Enoiu M, Ho TV, Scharer OD, Elledge SJ, Walter JC (2009) The Fanconi anemia pathway promotes replication-dependent DNA interstrand cross-link repair. Science 326(5960):1698–1701. doi:10.1126/science.1182372
- 112. Kolodner R (1996) Biochemistry and genetics of eukaryotic mismatch repair. Genes Dev 10(12):1433–1442
- 113. Kondo N, Takahashi A, Ono K, Ohnishi T (2010) DNA damage induced by alkylating agents and repair pathways. J Nucleic Acids 2010:543531. doi:10.4061/2010/543531
- 114. Kraakman-van der Zwet M, Overkamp WJ, van Lange RE, Essers J, van Duijn-Goedhart A, Wiggers I, Swaminathan S, van Buul PP, Errami A, Tan RT, Jaspers NG, Sharan SK, Kanaar R, Zdzienicka MZ (2002) Brca2 (XRCC11) deficiency results in radioresistant DNA synthesis and a higher frequency of spontaneous deletions. Mol Cell Bio 22(2):669–679
- 115. Kratz K, Schopf B, Kaden S, Sendoel A, Eberhard R, Lademann C, Cannavo E, Sartori AA, Hengartner MO, Jiricny J (2010) Deficiency of FANCD2-associated nuclease KIAA1018/ FAN1 sensitizes cells to interstrand crosslinking agents. Cell 142(1):77–88. doi:10.1016/j. cell.2010.06.022
- Kunkel TA, Erie DA (2005) DNA mismatch repair. Annu Rev Biochem 74:681–710. doi:10.1146/annurev.biochem.74.082803.133243
- 117. Kutler DI, Singh B, Satagopan J, Batish SD, Berwick M, Giampietro PF, Hanenberg H, Auerbach AD (2003) A 20-year perspective on the International Fanconi Anemia Registry (IFAR). Blood 101(4):1249–1256. doi:10.1182/blood-2002-07-2170
- Leach FS, Nicolaides NC, Papadopoulos N, Liu B, Jen J, Parsons R, Peltomaki P, Sistonen P, Aaltonen LA, Nystrom-Lahti M et al (1993) Mutations of a mutS homolog in hereditary nonpolyposis colorectal cancer. Cell 75(6):1215–1225
- Lee JH, Paull TT (2004) Direct activation of the ATM protein kinase by the Mre11/Rad50/ Nbs1 complex. Science 304(5667):93–96. doi:10.1126/science.1091496
- Lee JH, Paull TT (2005) ATM activation by DNA double-strand breaks through the Mre11-Rad50-Nbs1 complex. Science 308(5721):551–554. doi:10.1126/science.1108297
- 121. Lee JY, Chang J, Joseph N, Ghirlando R, Rao DN, Yang W (2005) MutH complexed with hemi- and unmethylated DNAs: coupling base recognition and DNA cleavage. Mol Cell 20(1):155–166. doi:10.1016/j.molcel.2005.08.019
- 122. Leung JW, Wang Y, Fong KW, Huen MS, Li L, Chen J (2012) Fanconi anemia (FA) binding protein FAAP20 stabilizes FA complementation group A (FANCA) and participates in interstrand cross-link repair. Proc Natl Acad Sci U S A 109(12):4491–4496. doi:10.1073/ pnas.1118720109
- 123. Levitus M, Waisfisz Q, Godthelp BC, de Vries Y, Hussain S, Wiegant WW, Elghalbzouri-Maghrani E, Steltenpool J, Rooimans MA, Pals G, Arwert F, Mathew CG, Zdzienicka MZ, Hiom K, De Winter JP, Joenje H (2005) The DNA helicase BRIP1 is defective in Fanconi anemia complementation group J. Nat Genet 37(9):934–935. doi:10.1038/ng1625
- 124. Li F, Mao G, Tong D, Huang J, Gu L, Yang W, Li GM (2013) The histone mark H3K-36me3 regulates human DNA mismatch repair through its interaction with MutSalpha. Cell 153(3):590–600. doi:10.1016/j.cell.2013.03.025
- 125. Li F, Tian L, Gu L, Li GM (2009) Evidence that nucleosomes inhibit mismatch repair in eukaryotic cells. J Biol Chem 284(48):33056–33061. doi:10.1074/jbc.M109.049874
- Li GM (2008) Mechanisms and functions of DNA mismatch repair. Cell Res 18(1):85–98. doi:10.1038/cr.2007.115
- 127. Li L, Wang H, Yang ES, Arteaga CL, Xia F (2008) Erlotinib attenuates homologous recombinational repair of chromosomal breaks in human breast cancer cells. Cancer Res 68(22):9141–9146. doi:10.1158/0008-5472.CAN-08-1127
- Li YH, Wang X, Pan Y, Lee DH, Chowdhury D, Kimmelman AC (2012) Inhibition of nonhomologous end joining repair impairs pancreatic cancer growth and enhances radiation response. PloS ONE 7(6):e39588. doi:10.1371/journal.pone.0039588

- Lieber MR, Ma Y, Pannicke U, Schwarz K (2003) Mechanism and regulation of human non-homologous DNA end-joining. Nat Rev Mol Cell Biol 4(9):712–720. doi:10.1038/ nrm1202
- 130. Lin DP, Wang Y, Scherer SJ, Clark AB, Yang K, Avdievich E, Jin B, Werling U, Parris T, Kurihara N, Umar A, Kucherlapati R, Lipkin M, Kunkel TA, Edelmann W (2004) An Msh2 point mutation uncouples DNA mismatch repair and apoptosis. Cancer Res 64(2):517–522
- 131. Lindahl T (1993) Instability and decay of the primary structure of DNA. Nature 362(6422):709-715. doi:10.1038/362709a0
- 132. Ling C, Ishiai M, Ali AM, Medhurst AL, Neveling K, Kalb R, Yan Z, Xue Y, Oostra AB, Auerbach AD, Hoatlin ME, Schindler D, Joenje H, de Winter JP, Takata M, Meetei AR, Wang W (2007) FAAP100 is essential for activation of the Fanconi anemia-associated DNA damage response pathway. EMBO J 26(8):2104–2114. doi:10.1038/sj.emboj.7601666
- 133. Lips EH, Mulder L, Hannemann J, Laddach N, Vrancken Peeters MT, van de Vijver MJ, Wesseling J, Nederlof PM, Rodenhuis S (2011) Indicators of homologous recombination deficiency in breast cancer and association with response to neoadjuvant chemotherapy. Annals Oncol 22(4):870–876. doi:10.1093/annonc/mdq468
- Liu T, Ghosal G, Yuan J, Chen J, Huang J (2010a) FAN1 acts with FANCI-FANCD2 to promote DNA interstrand cross-link repair. Science 329(5992):693–696. doi:10.1126/science.1192656
- 135. Liu TX, Howlett NG, Deng M, Langenau DM, Hsu K, Rhodes J, Kanki JP, D'Andrea AD, Look AT (2003) Knockdown of zebrafish Fancd2 causes developmental abnormalities via p53-dependent apoptosis. Dev Cell 5(6):903–914
- Liu Y, Fang Y, Shao H, Lindsey-Boltz L, Sancar A, Modrich P (2010b) Interactions of human mismatch repair proteins MutSalpha and MutLalpha with proteins of the ATR-Chk1 pathway. J Biol Chem 285(8):5974–5982. doi:10.1074/jbc.M109.076109
- 137. Long DT, Raschle M, Joukov V, Walter JC (2011) Mechanism of RAD51-dependent DNA interstrand cross-link repair. Science 333(6038):84–87. doi:10.1126/science.1204258
- Lord CJ, Ashworth A (2012) The DNA damage response and cancer therapy. Nature 481(7381):287–294. doi:10.1038/nature10760
- Lujan SA, Williams JS, Clausen AR, Clark AB, Kunkel TA (2013) Ribonucleotides are signals for mismatch repair of leading-strand replication errors. Mol Cell 50(3):437–443. doi:10.1016/j.molcel.2013.03.017
- Lujan SA, Williams JS, Pursell ZF, Abdulovic-Cui AA, Clark AB, Nick McElhinny SA, Kunkel TA (2012) Mismatch repair balances leading and lagging strand DNA replication fidelity. PLoS Genet 8(10):e1003016. doi:10.1371/journal.pgen.1003016
- Luke-Glaser S, Luke B, Grossi S, Constantinou A (2010) FANCM regulates DNA chain elongation and is stabilized by S-phase checkpoint signalling. EMBO J 29(4):795–805. doi:10.1038/emboj.2009.371
- 142. Machida YJ, Machida Y, Chen Y, Gurtan AM, Kupfer GM, D'Andrea AD, Dutta A (2006) UBE2T is the E2 in the Fanconi anemia pathway and undergoes negative autoregulation. Mol Cell 23(4):589–596. doi:10.1016/j.molcel.2006.06.024
- 143. MacKay C, Declais AC, Lundin C, Agostinho A, Deans AJ, MacArtney TJ, Hofmann K, Gartner A, West SC, Helleday T, Lilley DM, Rouse J (2010) Identification of KIAA1018/ FAN1, a DNA repair nuclease recruited to DNA damage by monoubiquitinated FANCD2. Cell 142(1):65–76. doi:10.1016/j.cell.2010.06.021
- Maher RL, Branagan AM, Morrical SW (2011) Coordination of DNA replication and recombination activities in the maintenance of genome stability. J Cell Biochem 112(10): 2672–2682. doi:10.1002/jcb.23211
- Margison GP, Povey AC, Kaina B, Santibanez Koref MF (2003) Variability and regulation of O6-alkylguanine-DNA alkyltransferase. Carcinogenesis 24(4):625–635
- 146. Markowitz S, Wang J, Myeroff L, Parsons R, Sun L, Lutterbaugh J, Fan RS, Zborowska E, Kinzler KW, Vogelstein B et al (1995) Inactivation of the type II TGF-beta receptor in colon cancer cells with microsatellite instability. Science 268(5215):1336–1338

- 147. Marra G, Iaccarino I, Lettieri T, Roscilli G, Delmastro P, Jiricny J (1998) Mismatch repair deficiency associated with overexpression of the MSH3 gene. Proc Natl Acad Sci U S A 95(15):8568–8573
- Marsischky GT, Filosi N, Kane MF, Kolodner R (1996) Redundancy of Saccharomyces cerevisiae MSH3 and MSH6 in MSH2-dependent mismatch repair. Genes Dev 10(4):407– 420
- 149. Masson M, Niedergang C, Schreiber V, Muller S, Menissier-de Murcia J, de Murcia G (1998) XRCC1 is specifically associated with poly(ADP-ribose) polymerase and negatively regulates its activity following DNA damage. Mol Cell Biol 18(6):3563–3571
- 150. Matson SW, Robertson AB (2006) The UvrD helicase and its modulation by the mismatch repair protein MutL. Nucleic Acids Res 34(15):4089–4097. doi:10.1093/nar/gkl450
- Matsumoto Y, Kim K (1995) Excision of deoxyribose phosphate residues by DNA polymerase beta during DNA repair. Science 269(5224):699–702
- Maynard S, Schurman SH, Harboe C, de Souza-Pinto NC, Bohr VA (2009) Base excision repair of oxidative DNA damage and association with cancer and aging. Carcinogenesis 30(1):2–10. doi:10.1093/carcin/bgn250
- 153. McCabe N, Turner NC, Lord CJ, Kluzek K, Bialkowska A, Swift S, Giavara S, O'Connor MJ, Tutt AN, Zdzienicka MZ, Smith GC, Ashworth A (2006) Deficiency in the repair of DNA damage by homologous recombination and sensitivity to poly(ADP-ribose) polymerase inhibition. Cancer Res 66(16):8109–8115. doi:10.1158/0008-5472.CAN-06-0140
- McNally R, Bowman GD, Goedken ER, O'Donnell M, Kuriyan J (2010) Analysis of the role of PCNA-DNA contacts during clamp loading. BMC Struct Biol 10:3. doi:10.1186/1472-6807-10-3
- Medina-Arana V, Delgado L, Bravo A, Martin J, Fernandez-Peralta AM, Gonzalez-Aguilera JJ (2012) Tumor spectrum in lynch syndrome, DNA mismatch repair system and endogenous carcinogens. J Surg Oncol 106(1):10–16. doi:10.1002/jso.23054
- 156. Meetei AR, Medhurst AL, Ling C, Xue Y, Singh TR, Bier P, Steltenpool J, Stone S, Dokal I, Mathew CG, Hoatlin M, Joenje H, de Winter JP, Wang W (2005) A human ortholog of archaeal DNA repair protein Hef is defective in Fanconi anemia complementation group M. Nat Genet 37(9):958–963. doi:10.1038/ng1626
- 157. Meyer S, Neitzel H, Tonnies H (2012) Chromosomal aberrations associated with clonal evolution and leukemic transformation in fanconi anemia: clinical and biological implications. Anemia 2012:349837. doi:10.1155/2012/349837
- Mojas N, Lopes M, Jiricny J (2007) Mismatch repair-dependent processing of methylation damage gives rise to persistent single-stranded gaps in newly replicated DNA. Gen Dev 21(24):3342–3355. doi:10.1101/gad.455407
- 159. Moreno V, Gemignani F, Landi S, Gioia-Patricola L, Chabrier A, Blanco I, Gonzalez S, Guino E, Capella G, Canzian F (2006) Polymorphisms in genes of nucleotide and base excision repair: risk and prognosis of colorectal cancer. Clin Cancer Res 12(7 Pt 1):2101– 2108. doi:10.1158/1078-0432.CCR-05-1363
- Mosedale G, Niedzwiedz W, Alpi A, Perrina F, Pereira-Leal JB, Johnson M, Langevin F, Pace P, Patel KJ (2005) The vertebrate Hef ortholog is a component of the Fanconi anemia tumor-suppressor pathway. Nat Struct Mol Biol 12(9):763–771. doi:10.1038/nsmb981
- Moynahan ME, Cui TY, Jasin M (2001) Homology-directed dna repair, mitomycin-c resistance, and chromosome stability is restored with correction of a Brca1 mutation. Cancer Res 61(12):4842–4850
- Mueser TC, Williams KJ (2011) Alkylation chemotherapy: Mechanistic potential and pitfalls. In: Advances in Genetics Research (Urbano KV, ed.), Nova Science Publishers, pp 179–196
- 163. Mukhopadhyay A, Elattar A, Cerbinskaite A, Wilkinson SJ, Drew Y, Kyle S, Los G, Hostomsky Z, Edmondson RJ, Curtin NJ (2010) Development of a functional assay for homologous recombination status in primary cultures of epithelial ovarian tumor and correlation with sensitivity to poly(ADP-ribose) polymerase inhibitors. Clin Cancer Res 16(8):2344–2351. doi:10.1158/1078-0432.CCR-09-2758

- 164. Muller LU, Milsom MD, Harris CE, Vyas R, Brumme KM, Parmar K, Moreau LA, Schambach A, Park IH, London WB, Strait K, Schlaeger T, Devine AL, Grassman E, D'Andrea A, Daley GQ, Williams DA (2012) Overcoming reprogramming resistance of Fanconi anemia cells. Blood 119(23):5449–5457. doi:10.1182/blood-2012-02-408674
- 165. Munoz IM, Hain K, Declais AC, Gardiner M, Toh GW, Sanchez-Pulido L, Heuckmann JM, Toth R, Macartney T, Eppink B, Kanaar R, Ponting CP, Lilley DM, Rouse J (2009) Coordination of structure-specific nucleases by human SLX4/BTBD12 is required for DNA repair. Mol Cell 35(1):116–127. doi:10.1016/j.molcel.2009.06.020
- 166. Murai J, Yang K, Dejsuphong D, Hirota K, Takeda S, D'Andrea AD (2011) The USP1/ UAF1 complex promotes double-strand break repair through homologous recombination. Mol Cell Biol 31(12):2462–2469. doi:10.1128/MCB.05058-11
- 167. Nakanishi K, Yang YG, Pierce AJ, Taniguchi T, Digweed M, D'Andrea AD, Wang ZQ, Jasin M (2005) Human Fanconi anemia monoubiquitination pathway promotes homologous DNA repair. Proc Natl Acad Sci U S A 102(4):1110–1115. doi:10.1073/pnas.0407796102
- Nick McElhinny SA, Gordenin DA, Stith CM, Burgers PM, Kunkel TA (2008) Division of labor at the eukaryotic replication fork. Mol Cell 30(2):137–144. doi:10.1016/j.molcel.2008.02.022
- 169. Nick McElhinny SA, Kissling GE, Kunkel TA (2010a) Differential correction of laggingstrand replication errors made by DNA polymerases {alpha} and {delta}. Proc Natl Acad Sci U S A 107(49):21070–21075. doi:10.1073/pnas.1013048107
- Nick McElhinny SA, Watts BE, Kumar D, Watt DL, Lundstrom EB, Burgers PM, Johansson E, Chabes A, Kunkel TA (2010b) Abundant ribonucleotide incorporation into DNA by yeast replicative polymerases. Proc Natl Acad Sci U S A 107(11):4949–4954. doi:10.1073/pnas.0914857107
- 171. Nick McElhinny SA, Kumar D, Clark AB, Watt DL, Watts BE, Lundstrom EB, Johansson E, Chabes A, Kunkel TA (2010c) Genome instability due to ribonucleotide incorporation into DNA. Nat Chem Biol 6:774–781. doi:10.1038/nchembio.424
- 172. Niedzwiedz W, Mosedale G, Johnson M, Ong CY, Pace P, Patel KJ (2004) The Fanconi anaemia gene FANCC promotes homologous recombination and error-prone DNA repair. Mol Cell 15(4):607–620. doi:10.1016/j.molcel.2004.08.009
- 173. Nojima K, Hochegger H, Saberi A, Fukushima T, Kikuchi K, Yoshimura M, Orelli BJ, Bishop DK, Hirano S, Ohzeki M, Ishiai M, Yamamoto K, Takata M, Arakawa H, Buerstedde JM, Yamazoe M, Kawamoto T, Araki K, Takahashi JA, Hashimoto N, Takeda S, Sonoda E (2005) Multiple repair pathways mediate tolerance to chemotherapeutic cross-linking agents in vertebrate cells. Cancer Res 65(24):11704–11711. doi:10.1158/0008-5472.CAN-05-1214
- 174. Oestergaard VH, Langevin F, Kuiken HJ, Pace P, Niedzwiedz W, Simpson LJ, Ohzeki M, Takata M, Sale JE, Patel KJ (2007) Deubiquitination of FANCD2 is required for DNA crosslink repair. Mol Cell 28(5):798–809. doi:10.1016/j.molcel.2007.09.020
- 175. Pabla N, Ma Z, McIlhatton MA, Fishel R, Dong Z (2011) hMSH2 recruits ATR to DNA damage sites for activation during DNA damage-induced apoptosis. J Biol Chem 286(12):10411–10418. doi:10.1074/jbc.M110.210989
- Pace P, Mosedale G, Hodskinson MR, Rosado IV, Sivasubramaniam M, Patel KJ (2010) Ku70 corrupts DNA repair in the absence of the Fanconi anemia pathway. Science 329(5988):219–223. doi:10.1126/science.1192277
- 177. Palles C, Cazier JB, Howarth KM, Domingo E, Jones AM, Broderick P, Kemp Z, Spain SL, Guarino E, Salguero I, Sherborne A, Chubb D, Carvajal-Carmona LG, Ma Y, Kaur K, Dobbins S, Barclay E, Gorman M, Martin L, Kovac MB, Humphray S, Lucassen A, Holmes CC, Bentley D, Donnelly P, Taylor J, Petridis C, Roylance R, Sawyer EJ, Kerr DJ, Clark S, Grimes J, Kearsey SE, Thomas HJ, McVean G, Houlston RS, Tomlinson I (2013) Germline mutations affecting the proofreading domains of POLE and POLD1 predispose to colorectal adenomas and carcinomas. Nat Genet 45(2):136–144. doi:10.1038/ng.2503
- 178. Pascual AM, Tellez N, Bosca I, Mallada J, Belenguer A, Abellan I, Sempere AP, Fernandez P, Magraner MJ, Coret F, Sanz MA, Montalban X, Casanova B (2009) Revision of the risk

of secondary leukaemia after mitoxantrone in multiple sclerosis populations is required. Mult Scler 15(11):1303–1310. doi:10.1177/1352458509107015

- 179. Patel AG, Sarkaria JN, Kaufmann SH (2011) Nonhomologous end joining drives poly(ADPribose) polymerase (PARP) inhibitor lethality in homologous recombination-deficient cells. Proc Natl Acad Sci U S A 108(8):3406–3411. doi:10.1073/pnas.1013715108
- 180. Peasland A, Wang LZ, Rowling E, Kyle S, Chen T, Hopkins A, Cliby WA, Sarkaria J, Beale G, Edmondson RJ, Curtin NJ (2011) Identification and evaluation of a potent novel ATR inhibitor, NU6027, in breast and ovarian cancer cell lines. Br J Cancer 105(3):372–381. doi:10.1038/bjc.2011.243
- 181. Pedersen-Bjergaard J, Rowley JD (1994) The balanced and the unbalanced chromosome aberrations of acute myeloid leukemia may develop in different ways and may contribute differently to malignant transformation. Blood 83(10):2780–2786
- Pena-Diaz J, Jiricny J (2010) PCNA and MutLalpha: partners in crime in triplet repeat expansion? Proc Natl Acad Sci U S A 107(38):16409–16410. doi:10.1073/pnas.1011692107
- Pena-Diaz J, Jiricny J (2012) Mammalian mismatch repair: error-free or error-prone? Trends Biochem Sci 37(5):206–214. doi:10.1016/j.tibs.2012.03.001
- Preston BD, Albertson TM, Herr AJ (2010) DNA replication fidelity and cancer. Semin Cancer Bio 20(5):281–293. doi:10.1016/j.semcancer.2010.10.009
- 185. Pluciennik A, Dzantiev L, Iyer RR, Constantin N, Kadyrov FA, Modrich P (2010) PCNA function in the activation and strand direction of MutLalpha endonuclease in mismatch repair. Proc Natl Acad Sci U S A 107(37):16066–16071. doi:10.1073/pnas.1010662107
- Podlutsky AJ, Dianova II, Podust VN, Bohr VA, Dianov GL (2001) Human DNA polymerase beta initiates DNA synthesis during long-patch repair of reduced AP sites in DNA. EMBO J 20(6):1477–1482. doi:10.1093/emboj/20.6.1477
- 187. Pursell ZF, Isoz I, Lundstrom EB, Johansson E, Kunkel TA (2007) Yeast DNA polymerase epsilon participates in leading-strand DNA replication. Science 317(5834):127–130. doi:10.1126/science.1144067
- Quennet V, Beucher A, Barton O, Takeda S, Lobrich M (2011) CtIP and MRN promote non-homologous end-joining of etoposide-induced DNA double-strand breaks in G1. Nucleic Acids Res 39(6):2144–2152. doi:10.1093/nar/gkq1175
- Rampino N, Yamamoto H, Ionov Y, Li Y, Sawai H, Reed JC, Perucho M (1997) Somatic frameshift mutations in the BAX gene in colon cancers of the microsatellite mutator phenotype. Science 275(5302):967–969
- Raschle M, Knipscheer P, Enoiu M, Angelov T, Sun J, Griffith JD, Ellenberger TE, Scharer OD, Walter JC (2008) Mechanism of replication-coupled DNA interstrand crosslink repair. Cell 134(6):969–980. doi:10.1016/j.cell.2008.08.030
- 191. Rasmussen LJ, Heinen CD, Royer-Pokora B, Drost M, Tavtigian S, Hofstra RM, de Wind N (2012) Pathological assessment of mismatch repair gene variants in Lynch syndrome: past, present, and future. Hum Mutat 33(12):1617–1625. doi:10.1002/humu.22168
- 192. Raya A, Rodriguez-Piza I, Guenechea G, Vassena R, Navarro S, Barrero MJ, Consiglio A, Castella M, Rio P, Sleep E, Gonzalez F, Tiscornia G, Garreta E, Aasen T, Veiga A, Verma IM, Surralles J, Bueren J, Izpisua Belmonte JC (2009) Disease-corrected haematopoiet-ic progenitors from Fanconi anaemia induced pluripotent stem cells. Nature 460(7251): 53–59. doi:10.1038/nature08129
- 193. Reijns MA, Rabe B, Rigby RE, Mill P, Astell KR, Lettice LA, Boyle S, Leitch A, Keighren M, Kilanowski F, Devenney PS, Sexton D, Grimes G, Holt IJ, Hill RE, Taylor MS, Lawson KA, Dorin JR, Jackson AP (2012) Enzymatic removal of ribonucleotides from DNA is essential for mammalian genome integrity and development. Cell 149(5):1008–1022. doi:10.1016/j.cell.2012.04.011
- Robertson AB, Pattishall SR, Gibbons EA, Matson SW (2006) MutL-catalyzed ATP hydrolysis is required at a post-UvrD loading step in methyl-directed mismatch repair. J Biol Chem 281(29):19949–19959. doi:10.1074/jbc.M601604200
- Romanova NV, Crouse GF (2013) Different roles of eukaryotic MutS and MutL complexes in repair of small insertion and deletion loops in yeast. PLoS Genet 9(10):e1003920. doi:10.1371/journal.pgen.1003920

- 196. Romick-Rosendale LE, Lui VW, Grandis JR, Wells SI (2013) The Fanconi anemia pathway: repairing the link between DNA damage and squamous cell carcinoma. Mutat Res 743–744:78–88. doi:10.1016/j.mrfmmm.2013.01.001
- 197. Rosenberg PS, Alter BP, Ebell W (2008) Cancer risks in Fanconi anemia: findings from the German Fanconi Anemia Registry. Haematol 93(4):511–517. doi:10.3324/haematol.12234
- Rosenberg PS, Greene MH, Alter BP (2003) Cancer incidence in persons with Fanconi anemia. Blood 101(3):822–826. doi:10.1182/blood-2002-05-1498
- 199. Rosenzweig KE, Youmell MB, Palayoor ST, Price BD (1997) Radiosensitization of human tumor cells by the phosphatidylinositol3-kinase inhibitors wortmannin and LY294002 correlates with inhibition of DNA-dependent protein kinase and prolonged G2-M delay. Clinical cancer research: an official journal of the American Association for. Cancer Res 3(7):1149–1156
- Saini N, Ramakrishnan S, Elango R, Ayyar S, Zhang Y, Deem A, Ira G, Haber JE, Lobachev KS, Malkova A (2013) Migrating bubble during break-induced replication drives conservative DNA synthesis. Nature 502(7471):389–392. doi:10.1038/nature12584
- 201. Sakumi K, Tominaga Y, Furuichi M, Xu P, Tsuzuki T, Sekiguchi M, Nakabeppu Y (2003) Ogg1 knockout-associated lung tumorigenesis and its suppression by Mth1 gene disruption. Cancer Res 63(5):902–905
- Schopf B, Bregenhorn S, Quivy JP, Kadyrov FA, Almouzni G, Jiricny J (2012) Interplay between mismatch repair and chromatin assembly. Proc Natl Acad Sci U S A 109(6): 1895–1900. doi:10.1073/pnas.1106696109
- Schwab RA, Blackford AN, Niedzwiedz W (2010) ATR activation and replication fork restart are defective in FANCM-deficient cells. EMBO J 29(4):806–818. doi:10.1038/emboj.2009.385
- Sharma S, Salehi F, Scheithauer BW, Rotondo F, Syro LV, Kovacs K (2009) Role of MGMT in tumor development, progression, diagnosis, treatment and prognosis. Anticancer Res 29(10):3759–3768
- 205. Shereda RD, Machida Y, Machida YJ (2010) Human KIAA1018/FAN1 localizes to stalled replication forks via its ubiquitin-binding domain. Cell Cycle 9(19):3977–3983
- 206. Shibutani S, Takeshita M, Grollman AP (1991) Insertion of specific bases during DNA synthesis past the oxidation-damaged base 8-oxodG. Nature 349(6308):431–434. doi:10.1038/349431a0
- 207. Shigechi T, Tomida J, Sato K, Kobayashi M, Eykelenboom JK, Pessina F, Zhang Y, Uchida E, Ishiai M, Lowndes NF, Yamamoto K, Kurumizaka H, Maehara Y, Takata M (2012) ATR-ATRIP kinase complex triggers activation of the Fanconi anemia DNA repair pathway. Cancer Res 72(5):1149–1156. doi:10.1158/0008-5472.CAN-11-2904
- Simpson LJ, Sale JE (2003) Rev1 is essential for DNA damage tolerance and non-templated immunoglobulin gene mutation in a vertebrate cell line. EMBO J 22(7):1654–1664. doi:10.1093/emboj/cdg161
- 209. Singh TR, Ali AM, Paramasivam M, Pradhan A, Wahengbam K, Seidman MM, Meetei AR (2013) ATR-dependent phosphorylation of FANCM at serine 1045 is essential for FANCM functions. Cancer Res 73(14):4300–4310. doi:10.1158/0008-5472.CAN-12-3976
- Singh TR, Saro D, Ali AM, Zheng XF, Du CH, Killen MW, Sachpatzidis A, Wahengbam K, Pierce AJ, Xiong Y, Sung P, Meetei AR (2010) MHF1-MHF2, a histone-fold-containing protein complex, participates in the Fanconi anemia pathway via FANCM. Mol Cell 37(6):879–886. doi:10.1016/j.molcel.2010.01.036
- Slupianek A, Dasgupta Y, Ren SY, Gurdek E, Donlin M, Nieborowska-Skorska M, Fleury F, Skorski T (2011) Targeting RAD51 phosphotyrosine-315 to prevent unfaithful recombination repair in BCR-ABL1 leukemia. Blood 118(4):1062–1068. doi:10.1182/blood-2010-09-307256
- 212. Smogorzewska A, Desetty R, Saito TT, Schlabach M, Lach FP, Sowa ME, Clark AB, Kunkel TA, Harper JW, Colaiacovo MP, Elledge SJ (2010) A genetic screen identifies FAN1, a Fanconi anemia-associated nuclease necessary for DNA interstrand crosslink repair. Mol Cell 39(1):36–47. doi:10.1016/j.molcel.2010.06.023

- Spratt TE, Levy DE (1997) Structure of the hydrogen bonding complex of O6-methylguanine with cytosine and thymine during DNA replication. Nucleic Acids Res 25(16): 3354–3361
- 214. Srivastava M, Nambiar M, Sharma S, Karki SS, Goldsmith G, Hegde M, Kumar S, Pandey M, Singh RK, Ray P, Natarajan R, Kelkar M, De A, Choudhary B, Raghavan SC (2012) An inhibitor of nonhomologous end-joining abrogates double-strand break repair and impedes cancer progression. Cell 151(7):1474–1487. doi:10.1016/j.cell.2012.11.054
- Sugo N, Aratani Y, Nagashima Y, Kubota Y, Koyama H (2000) Neonatal lethality with abnormal neurogenesis in mice deficient in DNA polymerase beta. EMBO J 19(6): 1397–1404. doi:10.1093/emboj/19.6.1397
- Sung P, Krejci L, Van Komen S, Sehorn MG (2003) Rad51 recombinase and recombination mediators. J Biol Chem 278(44):42729–42732. doi:10.1074/jbc.R300027200
- 217. Svendsen JM, Smogorzewska A, Sowa ME, O'Connell BC, Gygi SP, Elledge SJ, Harper JW (2009) Mammalian BTBD12/SLX4 assembles a Holliday junction resolvase and is required for DNA repair. Cell 138(1):63–77. doi:10.1016/j.cell.2009.06.030
- Swerdlow SH, Campo E, Harris NL, Jaffe ES, Pileri SA, Stein H, Thiele J, Vardiman JW (2008) WHO classification of tumours of haematopoietic and lymphoid tissues, 4th ed. IARC, France
- Takata K, Reh S, Tomida J, Person MD, Wood RD (2013) Human DNA helicase HELQ participates in DNA interstrand crosslink tolerance with ATR and RAD51 paralogs. Nat Commu 4:2338. doi:10.1038/ncomms3338
- Takeda S, Nakamura K, Taniguchi Y, Paull TT (2007) Ctp1/CtIP and the MRN complex collaborate in the initial steps of homologous recombination. Mol Cell 28(3):351–352. doi:10.1016/j.molcel.2007.10.016
- 221. Tang JB, Goellner EM, Wang XH, Trivedi RN, Croix CM St, Jelezcova E, Svilar D, Brown AR, Sobol RW (2010) Bioenergetic metabolites regulate base excision repair-dependent cell death in response to DNA damage. Mol Cancer Res (MCR) 8(1):67–79. doi:10.1158/1541-7786.MCR-09-0411
- 222. Taniguchi T, Garcia-Higuera I, Andreassen PR, Gregory RC, Grompe M, D'Andrea AD (2002) S-phase-specific interaction of the Fanconi anemia protein, FANCD2, with BRCA1 and RAD51. Blood 100(7):2414–2420. doi:10.1182/blood-2002-01-0278
- 223. Thomas DC, Roberts JD, Kunkel TA (1991) Heteroduplex repair in extracts of human HeLa cells. J Biol Chem 266(6):3744–3751
- 224. Tischkowitz MD, Hodgson SV (2003) Fanconi anaemia. J Med Genet 40(1):1-10
- 225. Toledo LI, Murga M, Zur R, Soria R, Rodriguez A, Martinez S, Oyarzabal J, Pastor J, Bischoff JR, Fernandez-Capetillo O (2011) A cell-based screen identifies ATR inhibitors with synthetic lethal properties for cancer-associated mutations. Nat Struct Mol Biol 18 (6):721–727. doi:10.1038/nsmb.2076
- 226. Tomida J, Itaya A, Shigechi T, Unno J, Uchida E, Ikura M, Masuda Y, Matsuda S, Adachi J, Kobayashi M, Meetei AR, Maehara Y, Yamamoto K, Kamiya K, Matsuura A, Matsuda T, Ikura T, Ishiai M, Takata M (2013) A novel interplay between the Fanconi anemia core complex and ATR-ATRIP kinase during DNA cross-link repair. Nucleic Acids Res 41(14):6930–6941. doi:10.1093/nar/gkt467
- 227. Trivedi RN, Wang XH, Jelezcova E, Goellner EM, Tang JB, Sobol RW (2008) Human methyl purine DNA glycosylase and DNA polymerase beta expression collectively predict sensitivity to temozolomide. Mol Pharmacol 74(2):505–516. doi:10.1124/mol.108.045112
- Trujillo JP, Mina LB, Pujol R, Bogliolo M, Andrieux J, Holder M, Schuster B, Schindler D, Surralles J (2012) On the role of FAN1 in Fanconi anemia. Blood 120(1):86–89. doi:10.1182/blood-2012-04-420604
- 229. Tulpule A, Lensch MW, Miller JD, Austin K, D'Andrea A, Schlaeger TM, Shimamura A, Daley GQ (2010) Knockdown of Fanconi anemia genes in human embryonic stem cells reveals early developmental defects in the hematopoietic lineage. Blood 115(17):3453–3462. doi:10.1182/blood-2009-10-246694

- Ueda K, Oka J, Naruniya S, Miyakawa N, Hayaishi O (1972) Poly ADP-ribose glycohydrolase from rat liver nuclei, a novel enzyme degrading the polymer. Biochem Biophys Res Commun 46(2):516–523
- 231. Vaz F, Hanenberg H, Schuster B, Barker K, Wiek C, Erven V, Neveling K, Endt D, Kesterton I, Autore F, Fraternali F, Freund M, Hartmann L, Grimwade D, Roberts RG, Schaal H, Mohammed S, Rahman N, Schindler D, Mathew CG (2010) Mutation of the RAD51C gene in a Fanconi anemia-like disorder. Nat Genet 42(5):406–409. doi:10.1038/ng.570
- 232. Veigl ML, Kasturi L, Olechnowicz J, Ma AH, Lutterbaugh JD, Periyasamy S, Li GM, Drummond J, Modrich PL, Sedwick WD, Markowitz SD (1998) Biallelic inactivation of hMLH1 by epigenetic gene silencing, a novel mechanism causing human MSI cancers. Proc Natl Acad Sci U S A 95(15):8698–8702
- 233. Veuger SJ, Curtin NJ, Richardson CJ, Smith GC, Durkacz BW (2003) Radiosensitization and DNA repair inhibition by the combined use of novel inhibitors of DNA-dependent protein kinase and poly(ADP-ribose) polymerase-1. Cancer Res 63(18):6008–6015
- Wallace SS, Murphy DL, Sweasy JB (2012) Base excision repair and cancer. Cancer Lett 327(1–2):73–89. doi:10.1016/j.canlet.2011.12.038
- 235. Wang AT, Sengerova B, Cattell E, Inagawa T, Hartley JM, Kiakos K, Burgess-Brown NA, Swift LP, Enzlin JH, Schofield CJ, Gileadi O, Hartley JA, McHugh PJ (2011) Human SN-M1A and XPF-ERCC1 collaborate to initiate DNA interstrand cross-link repair. Genes Dev 25(17):1859–1870. doi:10.1101/gad.15699211
- Wang W (2007) Emergence of a DNA-damage response network consisting of Fanconi anaemia and BRCA proteins. Nat Rev Genet 8(10):735–748. doi:10.1038/nrg2159
- 237. Wang Y, Leung JW, Jiang Y, Lowery MG, Do H, Vasquez KM, Chen J, Wang W, Li L (2013a) FANCM and FAAP24 maintain genome stability via cooperative as well as unique functions. Mol Cell 49(5):997–1009. doi:10.1016/j.molcel.2012.12.010
- 238. Wang Y, Yuan JL, Zhang YT, Ma JJ, Xu P, Shi CH, Zhang W, Li YM, Fu Q, Zhu GF, Xue W, Lei YH, Gao JY, Wang JY, Shao C, Yi CG, Wang H (2013b) Inhibition of both EGFR and IGF1R sensitized prostate cancer cells to radiation by synergistic suppression of DNA homologous recombination repair. PloS ONE 8(8):e68784. doi:10.1371/journal. pone.0068784
- Weterings E, Chen DJ (2008) The endless tale of non-homologous end-joining. Cell Res 18(1):114–124. doi:10.1038/cr.2008.3
- 240. Weterings E, van Gent DC (2004) The mechanism of non-homologous end-joining: a synopsis of synapsis. DNA Repair 3(11):1425–1435. doi:10.1016/j.dnarep.2004.06.003
- Williams HL, Gottesman ME, Gautier J (2013) The differences between ICL repair during and outside of S phase. Trends Biochem Sci 38(8):386–393. doi:10.1016/j.tibs.2013.05.004
- Williams SA, Wilson JB, Clark AP, Mitson-Salazar A, Tomashevski A, Ananth S, Glazer PM, Semmes OJ, Bale AE, Jones NJ, Kupfer GM (2011) Functional and physical interaction between the mismatch repair and FA-BRCA pathways. Hum Mol Genet 20(22):4395– 4410. doi:10.1093/hmg/ddr366
- 243. Willmore E, de Caux S, Sunter NJ, Tilby MJ, Jackson GH, Austin CA, Durkacz BW (2004) A novel DNA-dependent protein kinase inhibitor, NU7026, potentiates the cytotoxicity of topoisomerase II poisons used in the treatment of leukemia. Blood 103(12):4659–4665. doi:10.1182/blood-2003-07-2527
- 244. Xia B, Dorsman JC, Ameziane N, de Vries Y, Rooimans MA, Sheng Q, Pals G, Errami A, Gluckman E, Llera J, Wang W, Livingston DM, Joenje H, de Winter JP (2007) Fanconi anemia is associated with a defect in the BRCA2 partner PALB2. Nat Genet 39(2):159–161. doi:10.1038/ng1942
- 245. Xue Y, Li Y, Guo R, Ling C, Wang W (2008) FANCM of the Fanconi anemia core complex is required for both monoubiquitination and DNA repair. Hum Mol Genet 17(11): 1641–1652. doi:10.1093/hmg/ddn054
- 246. Yamamoto KN, Kobayashi S, Tsuda M, Kurumizaka H, Takata M, Kono K, Jiricny J, Takeda S, Hirota K (2011) Involvement of SLX4 in interstrand cross-link repair is regulated by the Fanconi anemia pathway. Proc Natl Acad Sci U S A 108(16):6492–6496. doi:10.1073/pnas.1018487108

- 247. Yan Z, Delannoy M, Ling C, Daee D, Osman F, Muniandy PA, Shen X, Oostra AB, Du H, Steltenpool J, Lin T, Schuster B, Decaillet C, Stasiak A, Stasiak AZ, Stone S, Hoatlin ME, Schindler D, Woodcock CL, Joenje H, Sen R, de Winter JP, Li L, Seidman MM, Whitby MC, Myung K, Constantinou A, Wang W (2010) A histone-fold complex and FANCM form a conserved DNA-remodeling complex to maintain genome stability. Mol Cell 37(6): 865–878. doi:10.1016/j.molcel.2010.01.039
- 248. Yan Z, Guo R, Paramasivam M, Shen W, Ling C, Fox D 3rd, Wang Y, Oostra AB, Kuehl J, Lee DY, Takata M, Hoatlin ME, Schindler D, Joenje H, de Winter JP, Li L, Seidman MM, Wang W (2012) A ubiquitin-binding protein, FAAP20, links RNF8-mediated ubiquitination to the Fanconi anemia DNA repair network. Mol Cell 47(1):61–75. doi:10.1016/j. molcel.2012.05.026
- Yang G, Scherer SJ, Shell SS, Yang K, Kim M, Lipkin M, Kucherlapati R, Kolodner RD, Edelmann W (2004) Dominant effects of an Msh6 missense mutation on DNA repair and cancer susceptibility. Cancer Cell 6(2):139–150. doi:10.1016/j.ccr.2004.06.024
- York SJ, Modrich P (2006) Mismatch repair-dependent iterative excision at irreparable O6-methylguanine lesions in human nuclear extracts. J Biol Chem 281(32):22674–22683. doi:10.1074/jbc.M603667200
- 251. Yoshikiyo K, Kratz K, Hirota K, Nishihara K, Takata M, Kurumizaka H, Horimoto S, Takeda S, Jiricny J (2010) KIAA1018/FAN1 nuclease protects cells against genomic instability induced by interstrand cross-linking agents. Proc Natl Acad Sci U S A 107(50):21553–21557. doi:10.1073/pnas.1011081107
- Yoshioka K, Yoshioka Y, Hsieh P (2006) ATR kinase activation mediated by MutSalpha and MutLalpha in response to cytotoxic O6-methylguanine adducts. Mol Cell 22(4):501–510. doi:10.1016/j.molcel.2006.04.023
- 253. You Z, Shi LZ, Zhu Q, Wu P, Zhang YW, Basilio A, Tonnu N, Verma IM, Berns MW, Hunter T (2009) CtIP links DNA double-strand break sensing to resection. Mol Cell 36(6): 954–969. doi:10.1016/j.molcel.2009.12.002
- 254. Yung SK, Tilgner K, Ledran MH, Habibollah S, Neganova I, Singhapol C, Saretzki G, Stojkovic M, Armstrong L, Przyborski S, Lako M (2013) Brief report: human pluripotent stem cell models of fanconi anemia deficiency reveal an important role for fanconi anemia proteins in cellular reprogramming and survival of hematopoietic progenitors. Stem Cells 31(5):1022–1029. doi:10.1002/stem.1308
- Zhang Y, Yuan F, Presnell SR, Tian K, Gao Y, Tomkinson AE, Gu L, Li GM (2005) Reconstitution of 5'-directed human mismatch repair in a purified system. Cell 122(5):693–705. doi:10.1016/j.cell.2005.06.027
- 256. Zhou W, Otto EA, Cluckey A, Airik R, Hurd TW, Chaki M, Diaz K, Lach FP, Bennett GR, Gee HY, Ghosh AK, Natarajan S, Thongthip S, Veturi U, Allen SJ, Janssen S, Ramaswami G, Dixon J, Burkhalter F, Spoendlin M, Moch H, Mihatsch MJ, Verine J, Reade R, Soliman H, Godin M, Kiss D, Monga G, Mazzucco G, Amann K, Artunc F, Newland RC, Wiech T, Zschiedrich S, Huber TB, Friedl A, Slaats GG, Joles JA, Goldschmeding R, Washburn J, Giles RH, Levy S, Smogorzewska A, Hildebrandt F (2012) FAN1 mutations cause karyomegalic interstitial nephritis, linking chronic kidney failure to defective DNA damage repair. Nat Genet 44(8):910–915. doi:10.1038/ng.23

Chapter 4 Mathematical Modeling for DNA Repair, Carcinogenesis and Cancer Detection

Jonathan Tang, Walter Georgescu, Thomas Deschamps, Steven M. Yannone and Sylvain V. Costes

Abstract The constant damage of DNA in human cells is considered the main cause of aging and cancer. In this review, we discuss the most lethal form of DNA damage, the DNA double strand break (DSB), and how it relates to cancer. DSB sensor proteins in the nucleus detect DNA breaks within minutes following damage. These proteins are now routinely labeled by immunocytochemistry, and access to high throughput fluorescence microscopy and robotics open the door to rapid measurement of DSB levels in individuals. This method, often referred as the DSB foci assay, leads to images showing small bright spots at the site of each damage in the nucleus. We first discuss how energy consumption in the cell leads to detectable baseline levels of foci per cell measured in peripheral blood lymphocytes. Mathematical kinetics are then described to infer both genetic defects in DNA repair and environmental factors influencing these levels. We emphasize ionizing radiation. which is the principal environmental factor that increases DSB levels. Mathematical models associating a mutation probability for each DSB have been used to explain the dose dependence of cancer incidence observed after exposure to high doses of radiation. The main assumption in these models is that high mutation frequency can eventually lead to tumor suppressor gene deletion or oncogene amplification. We conclude by suggesting that the growing stream of genetic and phenotypic measurements related to DNA repair and DNA damage will lead to more accurate predictive tools for cancer risk and individualized cancer prevention.

Keywords Radiation Induced Foci \cdot DNA double strand breaks \cdot DNA repair kinetics \cdot baseline DNA damage \cdot cancer model \cdot cancer risk \cdot multi-stage clonal expansion model \cdot cancer detection

S. V. Costes $(\boxtimes) \cdot W$. Georgescu \cdot S. M. Yannone

Life Sciences Division, Lawrence Berkeley National Laboratory, MS:977, Berkeley, CA 94720, USA e-mail: svcostes@lbl.gov

J. Tang · T. Deschamps · S. M. Yannone · S. V. Costes Exogen Biotechnology Inc., Berkeley, CA, USA e-mail: svcostes@lbl.gov

[©] Springer International Publishing Switzerland 2015

C. Maxwell, C. Roskelley (eds.), Genomic Instability and Cancer Metastasis,

Cancer Metastasis - Biology and Treatment 20, DOI 10.1007/978-3-319-12136-9_4

Abbreviations

ABM	Agent-based models
AT	Ataxia telangiectasia
DNA	Deoxyribonucleic acid
DSB	Double strand break
FA	Fanconi anemia
LNT	Linear-no-threshold hypothesis
IR	Ionizing radiation
PBL	Peripheral blood lymphocytes
RIF	Radiation-induced foci
ROS	Reactive oxygen species
SSB	Single strand breaks
TSCE	Two-stage clonal expansion
XP	Xeroderma pigmentosum

Introduction

Our DNA is vulnerable to injury by agents generated from within the cell as well as from external sources. Every day, each of the 10 trillion cells in the human body receives tens of thousands of DNA lesions that threaten the stability of our genome. These lesions include base modifications, DNA adducts, crosslinks, and singlestranded and double-stranded breaks. The efficiency of the repair of DNA damage is influenced by individual genetics, aging, and metabolism. Mammals have developed remarkably complex and in some cases redundant repair machinery to maintain genetic integrity for decades.

In this chapter, we will focus our attention on one specific class of DNA damage and how it impacts cancer risk. Namely, the DNA double-strand break (DSB) is unique in that both strands of the double helix are severed, thereby disrupting genome continuity. Because DSBs result in unconnected DNA termini, DSBs often lead to loss of genetic information in the forms of deletions, mutations, and/or translocations, all of which can be contributing factors to genome instability and the development of cancers [1-3].

Sources of DNA DSBs

Endogenous DNA damage can give rise to DSBs through a number of cellular processes. One source of DSBs is via reactive oxygen species (ROS), including superoxide, hydroxyl radicals, singlet oxygen, and hydrogen peroxide [4, 5]. ROS production is an inevitable byproduct of mitochondrial energy production and these molecules can diffuse from the mitochondria to the nuclei and cause DNA damage. ROS can also be generated from exogenous sources like metal ions and phorbol esters and other reactive molecules [6]. Regardless of the source, ROS have the potential to induce DSBs by directly reacting with DNA to generate DSBs, but more often, unrepaired ROS-damage leads to DSBs through a variety of cellular processes. Different types of ROS-induced DNA damage can be transmuted into DSBs through aberrant processing by DNA repair enzymes, DNA replication machinery, and other transactions with damaged DNA [7]. For example, during base-excision repair of single chemically damaged base, several intermediates are formed that can lead to the formation of DSBs [8]. Likewise, two independent single-stranded breaks (SSBs) on opposite strands, but separated by less than 10 base pairs can result in a DSB. SSBs can also be converted to DSBs when encountered by the DNA replication machinery [9]. Additionally, repair of base mismatches in the DNA sequence generates SSBs as intermediates, which can result in DSB formation [10]. All of these mechanisms, and others, can give rise to DSBs in healthy normal cells and pose a significant challenge to genome integrity.

In addition to endogenous DNA damage and repair processes, many organisms initiate programmed recombinational events by purposely introducing DSBs in specific genomic contexts. An example in single-celled organisms is mating type switching in yeast, a programmed recombination initiated by DSBs introduced into the genome by the regulated activity of an endonuclease (reviewed in [11]). In mammals, DSBs are introduced to initiate V(D)J recombination and class switch recombination events in immune cells. These processes give rise to immunoglobulin and T-cell receptor immunodiversity and are essential for B and T cell maturation [12, 13]. Defects in these processes can lead to severe organismal consequences including immunodeficiency and cancer [14–16]. Adding to these specific endogenous DSBs, errors in other cellular processes, like DNA replication, incomplete dissociation before mitotic segregation and other errors in DNA transactions can cause sporadic DSBs in genomes.

Among exogenous agents causing DSBs, ionizing radiation (IR) is the most extensively studied and broadly applied in medical procedures (e.g. X-rays). IR causes DSBs by direct collision of charged particles, γ -rays or X-rays, with the DNA double helix. Secondarily, IR also interacts with water to cause the formation of a variety of ROS, which contribute significantly to radiation toxicity. For example, chromosomal aberrations resulting from radiation induced DSB can be reduced by 60% when OH radicals are being scavenged [17]. Importantly, many, if not most, cancer therapies are based on overwhelming the DSB repair capacity of cells, and toxicity is inherently biased towards rapidly dividing cells, such as the malignancy, immune cells, hair follicles, and cells lining the intestine. The cellular toxicity of DSBs has motivated the development of a number of radio-mimetic drugs that imitate radiation in that they cause large numbers of DSBs in living cells [18]. Radiation, and many chemotherapeutic drugs commonly used for cancer treatments, relies on the same mode of action, namely, causing overwhelming numbers of DSBs in cellular DNA. Ironically, cancer treatments based on inducing toxic levels of DSBs also cause highly clastogenic DNA lesions that can lead to cancer. The complexity of factors giving rise to DSBs, both from within cells and the

environment, together with a diverse set of biological pathways involved in DNA repair, demand consideration when designing mathematical models of this complex and medically relevant phenomena.

DNA Double Strand Breaks Baseline Levels

Before being able to interpret the impact of genotoxic treatment on DNA using DNA double strand break assays, one needs to establish baseline levels of DNA damage. This limits the kind of assay one can use. Classic DSB assays, such as pulse field gel electrophoresis [19] or the neutral comet assay [20], are direct measures of DNA fragments but they typically require high levels of DNA damage, have limited sensitivity for detecting low levels of DSBs, and can be difficult to replicate between independent laboratories. The seminal discovery that the histone variant H2AX was specifically modified only at sites of DSBs [21] gave rise to immunofluorescent techniques and a quantitative surrogate marker for radiation-induced DSBs in eukaryotic cells [22]. In contrast to the direct measurement of fragmented DNA (i.e. comet assays), quantifying phosphorylated H2AX (γ H2AX) [21] or p53 binding protein 1 (53BP1) localized to DSB sites [23] is relatively simple. Importantly, quantification using this method is unambiguous and DSBs quantified as sub-nuclear spots, called "foci", have proven to be more reliable in detecting minor DNA repair defects in human cells [24].

Nuclear foci assays have been used extensively in laboratory settings, for quantifying DSBs in humans undergoing radiotherapy [25], and as a biomarker for aging and disease (reviewed in [26]). Remarkably, within seconds to minutes following IR, repair and checkpoint proteins are localized and/or modified at DSB sites, leading to the formation of radiation-induced foci (RIF). These rapid DNA damage responses typically reach a maximum at 0.5-1 h post-IR and diminish as DNA repair proceeds [21, 27–32]. The rapid appearance of γ H2AX foci in cells after damage (seconds) has led to several mechanistic hypotheses to account for the speed and discrete localization of foci. A recent study suggests that chromatin structure instantly changes upon disruption of genome continuity, thereby immediately and locally activating kinases and dictating yH2AX foci size and location [33]. Regardless of the mechanism, the majority of studies quantifying DNA double strand breaks using the RIF assay have been based on manual scoring [34–39], which has led to statistical uncertainty [40], subjective bias, and lack of reproducibility. Recent high throughput approaches using true 3D automatic foci detection software [31] has led to reproducible RIF results that can be compared across different research laboratories [41]. In this latter work, we showed that spontaneous damage is occurring in a random manner leading to a number of DSB/cell following a Poisson distribution [41], and quantification matches human counts from individuals trained in foci recognition (illustrated in Fig. 4.1).

Assuming DSB are produced at a constant rate P and foci are resolved at a rate k, one can write the following mathematical law:



Fig. 4.1 Standardization and calibration of DSB detection using RIF assay. Images illustrate human breast cells exposed to IR and immunostained for 53BP1 as previously described [41]. Increasing doses of IR can be used to generate an expected linear number of DSB and can be quantified automatically using state of the art wavelet imaging tools. When comparing to gold standard (average human visual inspection done blindly by two individuals), the algorithm performs as well. The capacity of quantifying thousands of nuclei within minutes in a reproducible manner, has allowed standardization and high throughput quantification of DSB

$$\frac{dC}{dt} = P - kC \Longrightarrow C(t) = \frac{P}{k} \left(1 - e^{-kt} \right)$$
(4.1)

Where, C is the average number of DSB at time t in one nucleus and P is the constant rate of DSB being produced in the same cell under physiological conditions. Assuming the tissue of interest is at steady state for endogenous damage production, C should be constant. Therefore, we can resolve the spontaneous DSB rate at steady state:

$$\frac{dC}{dt} = 0 \Rightarrow C(t) = \frac{P}{k} = C_0 \Rightarrow P = kC_0$$
(4.2)

To approximate the spontaneous DSB production, let us use the full range of reported values both for k, the repair rate, and C_0 from the literature. It has been shown that repair kinetics can vary, depending on the type of lesion with half-lives as fast as ~5 min or as slow as 3 h [42, 43]. Studies on Italian children have reported spontaneous foci in peripheral blood lymphocytes (PBL) as low as 0.004 foci/ cell [44]. In contrast, levels as high as 0.5–1 foci/cell in adult healthy donors PBL have been reported [45, 46]. This leads to a minimum spontaneous damage rate of 9.2E–4 DSB/h and a maximum of 5.2 DSB/h. In other words, during a 24 h time course, data suggest PBL may have to repair as many as 125 DSB or as few as 0.02. This is a large range of variation that should be reduced once the foci assay has been standardized across labs, by using validated imaging tools for spot quantification and by characterizing the same lymphocyte cell subtype to reduce cell specific bias [47]. Preliminary data on PBL, using our standardized approach, suggest 0.5 foci/ cell (data not shown), suggesting the ambient level of DSB is ~10–50 DSB/day.

Age affects DSB levels, which have been shown to double or triple over a life time in healthy subjects [46, 48, 49]. These studies concluded that aging may cause accumulation of DSB and/or that repair capacity is reduced with age, with a known higher genomic instability for older groups. However, if one looks at healthy older individuals (age >70), no significant increase compared to the youngest age group is observed [46], suggesting life style or genetics may be a determining factor for this aging process. It is also important to note in this study that telomere shortening was ruled out as a mechanism for increase DNA damage, contradicting previous studies [50, 51].

Evaluating DSB Repair Kinetics with the RIF Assay

Repair kinetics studies done with RIF use discrete time points in fixed specimens after the induction of damage with IR. However, foci formation at a DSB is not immediate and seems to occur asynchronously [41]. As such, there is a delay between DSB production and its detection via the foci assay. Such delay adds an important confounding factor when computing repair kinetics and we previously offered a mathematical correction for it [41]. We illustrated this issue by irradiating live breast cells expressing 53BP1 GFP and showing a cell with total of three RIF being produced over a 4-h time course (cumulative counts), but with a maximum of only two RIF at any time point (observable counts). In order to interpret RIF kinetics in an unbiased manner, we therefore had to introduce a mathematical formalism describing RIF formation as shown below:

$$\begin{cases} \frac{dC_0}{dt} = -k_1 C_0 \\ \frac{dC_1}{dt} = k_1 C_0 - k_2 C_1 \end{cases} \begin{cases} C_0(t) = C_0(0) \cdot e^{-k_1 t} \\ \frac{dC_1}{dt} = k_1 C_0(0) \cdot e^{-k_1 t} - k_2 C_1 \\ C_0(t) = \alpha D \cdot e^{-k_1 t} \\ C_1(t) = \frac{\alpha D k_1}{k_2 - k_1} \left(e^{-k_1 t} - e^{-k_2 t} \right) \end{cases}$$
(4.3)

where C_0 and C_1 are the average number of DSB and RIF per nucleus at time t, respectively and where RIF are formed at rate k_1 and they are resolved at rate k_2 and where α is the number of naked DSB/Gy before formation of RIF and D is the dose delivered to the cell. In this model, we assume that such chemical process, $DSB \rightarrow RIF \rightarrow RIF_{resolved}$, is irreversible. $C_1(t)$ in equation 4.3 can be used to fit the number of RIF at a given time (static measure). However, using time lapse imaging one can also measure the total number of RIF that have been produced since t=0 (cumulated measure). This can be described mathematically as:



$$C_{c}(t) = \alpha D(1 - e^{-k_{1}t})$$
(4.4)

Equation 4.4 is derived simply by setting $k_2 = 0$ and using the same formalism as in equation 4.3.

Figure 4.2 illustrates the model.

Another model has also been proposed by Foray et al., where the model associates a repair probability unique to each DSB [52]:

$$C_{1}(t) = \frac{C_{1}(0)}{\left(1 + \beta \cdot t\right)^{(1+\alpha)}}$$
(4.5)

where α and β are the shape and location parameters for the Gamma probability distribution function. This approach yields more accurate repair rates and repair probabilities for the population of DSBs using Euler's Gamma function. However, this method does not predict the cumulated number of RIF throughout the whole kinetic, and can only fit the monotonically decreasing portion of the observable RIFs (see our previous review [39]).

From DNA Damage to Cancer

The link between DNA damage repair defects and cancer was first recognized with the study of the hereditary cancer-prone disease xeroderma pigmentosum (XP) when cells from XP individuals were found to be defective in repairing DNA SSBs induced by UV light [53]. Since that seminal discovery, a number of DNA repair genes and repair processes have been linked to many forms of cancer, aging, and disease (reviewed in part in [14, 54–56] and in Chap. 3). The γ H2AX assay has been used to successfully identify repair defects in cells from individuals with genetic diseases like ataxia telangiectasia (AT), Nijmegen breakage syndrome [57],

XP [58], Fanconi anemia (FA) [59], and radiation sensitive severe combined immunodeficiency [24] among others. Immunostaining for γ H2AX and flow cytometry detection was successfully used to identify individuals heterozygous for ataxia telangiectasia mutations [60] where flow cytometry screening and subsequent quantification of γ H2AX foci with microscopy was utilized to assess DSB repair proficiency [61]. Investigation of γ H2AX as a marker of inflammation-induced DNA damage [62], and as a potential marker of viral infection [63] that potentially suppresses DSB repair [64] are also reported. Of particular relevance was the discovery of breast cancer susceptibility proteins type 1 and 2 (BRCA1 and BRCA2) which are clearly linked to cancer and play essential roles in DNA DSB repair [65]. The sizeable body of work linking DNA repair defects to cancer risks effectively establish the simple paradigm that defective DNA repair leads to genomic instability, which in turn can lead to cancer.

In addition to this somewhat oversimplified paradigm, the effects of defective DNA repair can manifest as a number of pathologies. While a comprehensive review of all of the links between DNA repair and disease is beyond the scope of this chapter, here we point to a few intriguing associations. Neurological disorders are linked to DNA repair defective syndromes like AT and XP [66] and the DSB repair defective Ligase IV syndrome [67]. In addition, oxidative stress (including oxidative DNA damage) is associated with the neurodegenerative pathologies Alzheimer's and Parkinson's [68, 69]. Both cancer predisposition and early onset of aging pathologies are caused by defects in a highly conserved group of RecQ DNA helicases [70]. Defects in one such helicase, the WRN protein, cause a predisposition for a broad spectrum of cancers and an early onset of many aging pathologies. The WRN protein interacts with DSB repair proteins [71] and a number of other DNA repair pathways [72]. While the precise mechanistic links are still being discovered, it is clear that DNA repair defects impact genome stability and cancer predisposition in addition to a number of other important pathologies.

Epidemiological Study: The A-Bomb Survivors

The most documented argument favoring the correlation between DSB and cancer is the epidemiologic data from the Atomic Bomb survivors [73, 74]. These analyses were based on 17,448 first primary cancers (including non-melanoma skin cancer) diagnosed from 1958 through 1998 among 105,427 individuals. IR is known to induce DSB via the production of reactive oxygen species and direct DNA ionization. Therefore, increasing radiation doses increases in a linear manner the amount of DSBs. Similarly, cancer incidence from the A-bomb data show an increase with dose, but the linearity remains a very controversial topic [41, 73] and only significant doses of IR (>0.1 Gy), known to induce several DSB (at least 3), are currently required to establish a clear cancer risk. These epidemiological data most clearly demonstrate the impact of elevated DSBs on cancer risk and occurrence in a large human population.

Modeling Cancer Using DNA Damage

Modeling Radiation-Induced Carcinogenesis

When it comes to cancer risk from IR, the current risk model assumes the linear-nothreshold hypothesis (LNT), which implies that any amount of IR exposure is harmful. LNT is used to set dose limits for radiation occupational workers or the general public. The LNT is based mainly on data from the Japanese A-bomb survivors and secondarily on arguments involving the dose-response of surrogate endpoints. Gene mutations that are caused primarily by DSB are thought to be the initiating events of cancer (see Fig. 4.3). One poorly repaired DSB can lead to a point mutation while misrejoining of two DSBs leads to more complicated genome rearrangements. Physical laws suggest DSB frequencies are proportional to dose. Therefore, it is well accepted that point mutations are linear with dose since it requires only one DSB, whereas DSB misrejoinings are dependent to the dose squared [75]. In the dose-range of radiation cancer epidemiology, the quadratic term is almost always negligible, especially at low dose rates, as the first lesion is probably repaired before the second mutation occurs [76]. However, we have shown that DSB move into regularly spaced nuclear domains of 1.55 µm interval [41, 77]. Therefore, as the dose of IR increases, the probability of having two DSBs in the same repair center increases non-linearly, increasing further the risk of DSB misrejoining. Therefore, extrapolating risk linearly from high dose, as done with the LNT, could lead to overestimation of cancer risk at low doses.

Carcinogenesis is thought to occur in four interdependent stages as depicted in Fig. 4.3. The first stage is *Initiation* and is typically caused by chemical, physical, or biological agents, which irreversibly and heritably alter cellular genomes resulting in an enhanced growth potential. Mutation of tumor suppressor genes have been



Fig. 4.3 The four stages of carcinogenesis and the role of DSB and mutation in this process. *Red circles* represent mutation affecting genes in the cancer process. (i.e. Tumor Suppressor Genes or Oncogenes)

thought to be the most likely initiating events, as originally introduced by Knudson [78]. This neoplastic potential is only realized, however, if the cell later undergoes *Promotion*, the second stage of carcinogenesis. *Promotion* is often thought to be the rate-limiting step in carcinogenesis since it has been shown that initiation alone is not sufficient to induce cancer [79]. Further genomic modifications due to genomic instability can lead to Transformation, which is often characterized by cell immortalization and oncogenic activation. Once a cell is transformed, additional proliferation leads to a full tumor. This last phase of carcinogenesis is defined as Progression.

In order to account for the observed power of age dependence in radiation-induced carcinomas, a multi-stage theory of carcinogenesis was first introduced [80, 81]. However, this model suggested 5 to 7 rate-limiting stages, which contradicted biological data. This discrepancy was resolved by the two-stage clonal expansion model (TSCE), which assumed a cell leads to a tumor by two separate mutations (Initiation and Transformation, with probability μ_1 and μ_2 respectively) and clonal expansion (Promotion and Progression) [82–84] (See Fig. 4.3). Promotion is characterized with a proliferation rate α , while Progression is modeled by a lag time (t_{lag}) required to reach a full tumor. DNA damage occurring spontaneously or via genotoxic stress can influence the rate of initiation, increasing the probability of developing cancer.

However, TSCE neglects the influence of intercellular and extracellular interactions in the tumor growth and predicts a final tumor that is unrealistic in that its cells are clonally identical. This is inconsistent with what is observed where tumors are heterogeneous and cell-cell and cell-extracellular matrix (ECM) interactions have been shown to influence tumor progression. In fact, a new paradigm for carcinogenesis has been gaining some credibility over the past two decades emphasizing a key role of cell-cell and cell-ECM interactions in the maintenance of tissue organization [85]. For example, during angiogenesis a tumor manages to communicate with its microenvironment to elicit the proliferation of endothelial cells to form blood vessels that will supply the tumor with oxygen. Another illustration of this paradigm is the existence of cancer susceptibility genes whose mutation affect genomic stability but associate with cancer only in certain tissues (e.g. BRCA1 for breast cancer, APC for colon cancer). This would suggest that the cellular and tissue context itself plays a role in causing the initiated cell to start proliferating.

Recent work on colon cancer introduced genomic instability in the TSCE model in order to better fit the data [86]. Fits were excellent but also suggested that radiation only played a small role in initiating genomic destabilization. The idea that non-mutational radiation effects play a critical role in destabilizing the genome is supported by the literature describing radiation-induced genomic instability as a non-targeted effect [87–89]. These types of multi-cellular interactions typically lack mathematical formalism due to the impossibility of reducing them to single entities such as cells. A new kind of formalism is required to adequately represent and model these multi-cellular interactions in a system.

Advances in computer science have engendered new approaches to model biological systems in ways that can formalize a system of interacting components. One such approach is agent-based models (ABM), which naturally describe complex adaptive systems as the results of interactive software objects in various contexts [90]. Agents are non-deterministic codes originally developed for artificial intelligence. Each agent behaves individually in response to its situation on the basis of a set of contextual rules. In the case of representing cells within a tissue, agents may execute various behaviors appropriate for the system, such as proliferation, differentiation or death. A modeling framework that permits the integration of the different factors leading to homeostasis might include cooperation and competition, spatial organization, physical and molecular interactions between cells and their microenvironment, biological pathways, or genomic status. By modeling an irradiated tissue/organ/organism as a system of interacting cells that integrates these many factors, cancer can more realistically be analyzed as an emergent phenomenon of a perturbed system [91].

Our group has used ABM extensively to understand the emergent biological behaviors resulting from these complex multi-cellular interactions. For example, using ABM we showed that radiation could induce premature senescence of normal mammary epithelial cells, allowing for an accelerated outgrowth and selection of preneoplastic cells in culture [92]. We also used ABM to model the complex interplay between apoptosis, proliferation, and polarization needed to maintain a normal 3D structure of mammary epithelial cells [93]. Surprisingly, our simulations revealed a synergism between polarization and apoptosis in achieving growth arrest necessary to achieve a normal mammary acinar morphology. More recently, we used a multi-scale ABM to evaluate the effects of radiation on stem cell kinetics in the mammary gland during development [39]. The model made the prediction that irradiation during puberty, but not during adulthood, induces stem cell self-renewal and subsequent mammary stem cell enrichment, which in turn leads to estrogen receptor negative breast cancer. This prediction was validated functionally using *in vitro* and in vivo experiments, and may help explain why there is an elevated breast cancer risk after exposure to IR in young girls, but not adult women over the age of 40.

ABM hold great potential for modeling the increasing complexity of unraveling mechanism in carcinogenesis. Figure 4.3 depicts how one could integrate the impact of the microenvironment and inflammation in an ABM to predict tumor incidence *in silico*. One could assume that tumors arise via successive DNA mutation and that promotion are influenced by non-targeted effects (i.e. persistent effects on the microenvironment, modifying cellular behavior and baseline DNA damage level, modifying the frequency of both initiation and conversion (μ_1 and μ_2 terms respectively)). One can also generalize TSCE to more stages before a tumor arises, if necessary (i.e. Multi Stage Clonal Expansion).

Using Foci Assay as Biomarkers for Cancer Risk and Cancer Detection

Several studies have suggested that γ H2AX quantification may be a powerful diagnostic tool for cancer risk and development and many other diseases [25, 26, 44, 94]. Elevated levels of γ H2AX in premalignant lesions can induce cell cycle arrest and cellular senescence, thus serving a tumor-suppressive function [95–99]. These findings suggested the use of the γ H2AX assay may be an effective bioassay for early stage cancers [94]. Analysis of 30 human biopsies from colon, breast, ovary, liver and kidney cancers support the use of γ H2AX in detecting early stage of tumor development [94] while other reports use γ H2AX foci as a marker of cancer progression and treatment [100, 101]. Interestingly, sporadic breast cancer, gliomas, and lymphomas have been associated with genetic variations in the H2AX gene or changes in its promoter region [102–104].

Repair kinetics has also been explored as a way to predict radiation sensitivity and cancer risk. For example, primary lymphocytes from radiation sensitive mice with known defects in DNA repair (e.g. Balb/C and SCID mice) have RIF repair kinetics 1.5 to 6 times slower than resistant strains such as C57Bl/6 J (repair rates inferred from [105]). Similarly, ATM defects that result in persistent RIF [41, 106] are associated with radiation-induced carcinogenesis in mice [107], and increased toxicity from radiotherapy in ATM heterozygous patients [108, 109]. Additionally, DNA repair deficiencies are considered risk factors for both acute radiation toxicity and cancer, independently of the type of radiation [110–114]. Importantly, functional RIF kinetic assays may be sensitive enough to detect subtle genetic differences between individuals that would be hard to detect with more classic methodologies, such as single nucleotide polymorphism arrays. For example, recent studies on primary human fibroblasts, derived from 25 apparently healthy individuals and ten patients with DNA repair-defective syndromes have shown a wide variation in RIF levels and kinetics between individuals [115]. Our mathematical interpretation of RIF kinetics establishes novel metrics that can characterize such subtle differences. This approach may help identify genes that have not been previously implicated in DNA repair and individuals that are sensitive or cancer-prone in response to radiation exposure.

Some researchers have looked at DSB levels or DSB repair in lymphocytes as a potential indicator for cancer risk. The use of lymphocytes is ideal as they can be obtained in a relatively non-invasive manner. A recent study showed that baseline γ H2AX foci levels in lymphocytes alone might be sufficient to detect individuals with breast cancer prior to treatment [116]. Elevated DNA DSB levels in lymphocytes of breast cancer patients have been observed previously, but using the comet assay [117, 118]. Others have additionally observed defective DNA DSB repair in lymphocytes in untreated patients with bladder [119], esophageal [120], and lung cancer [121]. These studies show great promise for the use of the γ H2AX foci assay on lymphocytes for the detection and possible prevention of many cancers.

The development and broad application of the DSB foci assays brings major advantages to mathematical modeling of DNA damage and repair. First, this technology allows quantitative scoring of one specific type of DNA damage on a 'per genome' basis. Second, the assay is based on the cellular recognition of DSBs, not an estimate or average of broken molecules in a population of cells. These advantages permit accurate and precise measurements to be taken and compared between laboratories, samples, tissues, and individuals. This breakthrough technology will enhance the speed and accuracy of developing and testing mathematical models that quantitatively describe DNA damage processes critical to cancer, aging, and other pathologies that are the focus of modern medicine.

Disclaimer This manuscript has been authored by an author at Lawrence Berkeley National Laboratory under Contract No. DE-AC02–05CH11231 with the U.S. Department of Energy. The U.S. Government, and the publisher, by accepting the article for publication, acknowledges, that the U.S. Government retains a nonexclusive, paid-up, irrevocable, world-wide license to publish or reproduce the published form of this manuscript, or allow others to do so, for U.S. Government purposes.

Acknowledgements SVC, JT are supported by NASA Specialized Center for Research in Radiation Health Effects [NNJ09HC64I]. SVC, WG are supported by the Low Dose Scientific Focus Area, United States Department of Energy [DE-AC02–05CH11231]. SVC, TD and SY are supported by the Lawrence Berkeley National Laboratory Innovation grant [DE-AC02–05CH11231].

References

- Schar P (2001) Spontaneous DNA damage, genome instability, and cancer-when DNA replication escapes control. Cell 104:329–332
- 2. Maslov AY, Vijg J (2009) Genome instability, cancer and aging. Biochim Biophys Acta 1790:963–969
- Storchova Z, Pellman D (2004) From polyploidy to aneuploidy, genome instability and cancer. Nat Rev Mol Cell Biol 5:45–54
- Cadet J, Wagner JR (2013) DNA base damage by reactive oxygen species, oxidizing agents, and UV radiation. Cold Spring Harbor Perspect Biol 5(2)
- Ewing D (1987) Radiation sensitization of E. coli B/r by mixtures of oxygen and nitrous oxide. Int J Radiat Biol Relat Stud Phys Chem Med 52:921–933
- Klaunig JE, Kamendulis LM (2004) The role of oxidative stress in carcinogenesis. Annu Rev Pharmacol Toxicol 44:239–267
- Donley N, Thayer MJ (2013) Replication timing DNA, genome stability and cancer: late and/ or delayed DNA replication timing is associated with increased genomic instability. Semin Cancer Biol 23:80–89
- 8. Wilson DM 3rd, Bohr VA (2007) The mechanics of base excision repair, and its relationship to aging and disease. DNA Repair 6:544–559
- Kuzminov A (2001) Single-strand interruptions in replicating chromosomes cause doublestrand breaks. Proc Natl Acad Sci U S A 98:8241–8246
- Nowosielska A, Marinus MG DNA mismatch repair-induced double-strand breaks. DNA Repair 7:48–56
- 11. Haber JE (2012) Mating-type genes and MAT switching in Saccharomyces cerevisiae. Genetics 191:33–64
- 12. Edry E, Melamed D (2007) Class switch recombination: a friend and a foe. Clin Immunol 123:244–251
- 13. Soulas-Sprauel P, Rivera-Munoz P, Malivert L, Le Guyader G, Abramowski V, Revy P, Villartay JP de (2007) V(D)J and immunoglobulin class switch recombinations: a paradigm to study the regulation of DNA end-joining. Oncogene 26:7780–7791
- 14. Risinger MA, Groden J (2004) Crosslinks and crosstalk: human cancer syndromes and DNA repair defects. Cancer cell 6:539–545

- Cowan MJ, Neven B, Cavazanna-Calvo M, Fischer A, Puck J, (2008) Hematopoietic stem cell transplantation for severe combined immunodeficiency diseases. Biol Blood Marrow Transplant 14:73–75
- Roddam PL, Rollinson S, O'Driscoll M, Jeggo PA, Jack A, Morgan GJ (2002) Genetic variants of NHEJ DNA ligase IV can affect the risk of developing multiple myeloma, a tumour characterised by aberrant class switch recombination. J Med Genet 39:900–905
- Sasaki MS, Matsubara S (1977) Free radical scavenging in protection of human lymphocytes against chromosome aberration formation by gamma-ray irradiation. Int J Radiat Biol Relat Stud Phys Chem Med 32:439–445
- 18. Povirk LF (1996) DNA damage and mutagenesis by radiomimetic DNA-cleaving agents: bleomycin, neocarzinostatin and other enediynes. Mutat Res 355:71–89
- Stenerlow B, Karlsson KH, Cooper B, Rydberg B (2003) Measurement of prompt DNA double-strand breaks in mammalian cells without including heat-labile sites: results for cells deficient in nonhomologous end joining. Radiat Res 159:502–510
- Olive PL, Banath JP (1995) Radiation-induced DNA double-strand breaks produced in histone-depleted tumor-cell nuclei measured using the neutral comet assay. Radiat Res 142:144–152
- Rogakou EP, Pilch DR, Orr AH, Ivanova VS, Bonner WM (1998) DNA double-stranded breaks induce histone H2AX phosphorylation on serine 139. J Biol Chem 273:5858–5868
- Nelms BE, Maser RS, MacKay JF, Lagally MG, Petrini JH (1998) In situ visualization of DNA double-strand break repair in human fibroblasts. Science 280:590–592
- Schultz LB, Chehab NH, Malikzay A, Halazonetis TD (2000) p53 binding protein 1 (53BP1) is an early participant in the cellular response to DNA double-strand breaks. J Cell Biol 151:1381–1390
- Wang J, Pluth JM, Cooper PK, Cowan MJ, Chen DJ, Yannone SM (2005) Artemis deficiency confers a DNA double-strand break repair defect and Artemis phosphorylation status is altered by DNA damage and cell cycle progression. DNA Repair 4:556–570
- 25. Sak A, Stuschke M (2010) Use of gammaH2AX and other biomarkers of double-strand breaks during radiotherapy. Semin Radiat Oncol 20:223–231
- Mah LJ, El-Osta A, Karagiannis TC (2010) GammaH2AX as a molecular marker of aging and disease. Epigenetics 5:129–136
- Anderson L, Henderson C, Adachi Y (2001) Phosphorylation and rapid relocalization of 53BP1 to nuclear foci upon DNA damage. Mol Cell Biol 21:1719–1729
- Rothkamm K, Kruger I, Thompson LH, Lobrich M (2003) Pathways of DNA double-strand break repair during the mammalian cell cycle. Mol Cell Biol 23:5706–5715
- 29. Costes SV, Boissiere A, Ravani S, Romano R, Parvin B, Barcellos-Hoff MH (2006) Imaging features that discriminate between foci induced by high- and low-LET radiation in human fibroblasts. Radiat Res 165:505–515
- Leatherbarrow EL, Harper JV, Cucinotta FA, O'Neill P (2006) Induction and quantification of gamma-H2AX foci following low and high LET-irradiation. Int J Radiat Biol 82:111–118
- Costes SV, Ponomarev A, Chen JL, Nguyen D, Cucinotta FA, Barcellos-Hoff MH (2007) Image-based modeling reveals dynamic redistribution of DNA damage into nuclear sub-domains. PLoS Comput Biol 3:e155
- Markova E, Schultz N, Belyaev IY (2007) Kinetics and dose-response of residual 53BP1/ gamma-H2AX foci: co-localization, relationship with DSB repair and clonogenic survival. Int J Radiat Biol 83:319–329
- 33. Munoz DP, Kawahara M, Yannone SM ((2013) An autonomous chromatin/DNA-PK mechanism for localized DNA damage signaling in mammalian cells. Nucleic Acids Res 41:2894–2906
- Sedelnikova OA, Pilch DR, Redon C, Bonner WM (2003) Histone H2AX in DNA damage and repair. Cancer Biol Ther 2:233–235
- Kinner A, Wu W, Staudt C, Iliakis G (2008) Gamma-H2AX in recognition and signaling of DNA double-strand breaks in the context of chromatin. Nucleic Acids Res 36:5678–5694

- Costes SV, Chiolo I, Pluth JM, Barcellos-Hoff MH, Jakob B (2010) Spatiotemporal characterization of ionizing radiation induced DNA damage foci and their relation to chromatin organization. Mutat Res 704:78–87
- 37. Falk M, Lukasova E, Kozubek S (2010) Higher-order chromatin structure in DSB induction, repair and misrepair. Mutat Res 704:88–100
- Lobrich M, Shibata A, Beucher A, Fisher A, Ensminger M, Goodarzi AA, Barton O, Jeggo PA (2010) GammaH2AX foci analysis for monitoring DNA double-strand break repair: strengths, limitations and optimization. Cell Cycle 9:662–669
- Chiolo I, Tang J, Georgescu W, Costes SV (2013) Nuclear dynamics of radiation-induced foci in euchromatin and heterochromatin. Mutat Res 750:56–66
- 40. Rothkamm K, Lobrich M (2003) Evidence for a lack of DNA double-strand break repair in human cells exposed to very low x-ray doses. Proc Natl Acad Sci U S A 100:5057–5062
- 41. Neumaier T, Swenson J, Pham C, Polyzos A, Lo AT, Yang P, Dyball J, Asaithamby A, Chen DJ, Bissell MJ, Thalhammer S, Costes SV (2012) Evidence for formation of DNA repair centers and dose-response nonlinearity in human cells. Proc Natl Acad Sci U S A 109:443–448
- 42. Wang H, Zeng ZC, Bui TA, Sonoda E, Takata M, Takeda S, Iliakis G (2001) Efficient rejoining of radiation-induced DNA double-strand breaks in vertebrate cells deficient in genes of the RAD52 epistasis group. Oncogene 20:2212–2224
- Karlsson KH, Radulescu I, Rydberg B, Stenerlow B (2008) Repair of radiation-induced heatlabile sites is independent of DNA-PKcs, XRCC1 and PARP. Radiat Res 169:506–512
- 44. Scarpato R, Verola C, Fabiani B, Bianchi V, Saggese G, Federico G (2011) Nuclear damage in peripheral lymphocytes of obese and overweight Italian children as evaluated by the gamma-H2AX focus assay and micronucleus test. Faseb J 25:685–693
- 45. Markova E, Hillert L, Malmgren L, Persson BR, Belyaev IY (2005) Microwaves from GSM mobile telephones affect 53BP1 and gamma-H2AX foci in human lymphocytes from hypersensitive and healthy persons. Environ Health Perspect 113:1172–1177
- 46. Rube CE, Fricke A, Widmann TA, Furst T, Madry H, Pfreundschuh M, Rube C (2011) Accumulation of DNA damage in hematopoietic stem and progenitor cells during human aging. Plos ONE 6:e17487
- 47. Lacoste S, Bhatia R, Bhatia S, O'Connor TR (2014) Granulocytes affect double-strand break repair assays in primary human lymphocytes. Plos ONE 9:e93185
- Sedelnikova OA, Horikawa I, Redon C, Nakamura A, Zimonjic DB, Popescu NC, Bonner WM (2008) Delayed kinetics of DNA double-strand break processing in normal and pathological aging. Aging Cell 7:89–100
- 49. Joyce NC, Harris DL, Zhu CC (2011) Age-related gene response of human corneal endothelium to oxidative stress and DNA damage. Invest Ophthalmol Vis Sci 52:1641–1649
- Nakamura AJ, Chiang YJ, Hathcock KS, Horikawa I, Sedelnikova OA, Hodes RJ, Bonner WM (2008) Both telomeric and non-telomeric DNA damage are determinants of mammalian cellular senescence. Epigenetics Chromatin 1:6
- Noren Hooten N, Abdelmohsen K, Gorospe M, Ejiogu N, Zonderman AB, Evans MK (2010) microRNA expression patterns reveal differential expression of target genes with age. Plos ONE 5:e10724
- Foray N, Charvet AM, Duchemin D, Favaudon V, Lavalette D (2005) The repair rate of radiation-induced DNA damage: a stochastic interpretation based on the gamma function. J Theor Biol 236:448–458
- Cleaver JE (1968) Defective repair replication of DNA in xeroderma pigmentosum. Nature 218:652–656
- 54. Cha HJ, Yim H (2013) The accumulation of DNA repair defects is the molecular origin of carcinogenesis. Tumour Biol 34:3293–3302
- Knoch J, Kamenisch Y, Kubisch C, Berneburg M (2012) Rare hereditary diseases with defects in DNA-repair. Eur J Dermatol 22:443–455
- Nahas SA, Gatti RA (2009) DNA double strand break repair defects, primary immunodeficiency disorders, and 'radiosensitivity'. Curr Opin Allergy Clin Immunol 9:510–516

- 57. Porcedda P, Turinetto V, Lantelme E, Fontanella E, Chrzanowska K, Ragona R, De Marchi M, Delia D, Giachino C (2006) Impaired elimination of DNA double-strand break-containing lymphocytes in ataxia telangiectasia and Nijmegen breakage syndrome. DNA repair 5:904–913
- Abbaszadeh F, Clingen PH, Arlett CF, Plowman PN, Bourton EC, Themis M, Makarov EM, Newbold RF, Green MH, Parris CN (2010) A novel splice variant of the DNA-PKcs gene is associated with clinical and cellular radiosensitivity in a patient with xeroderma pigmentosum. J Med Genet 47:176–181
- Leskovac A, Vujic D, Guc-Scekic M, Petrovic S, Joksic I, Slijepcevic P, Joksic G (2010) Fanconi anemia is characterized by delayed repair kinetics of DNA double-strand breaks. Tohoku J Exp Med 221:69–76
- Nahas SA, Butch AW, Du L, Gatti RA (2009) Rapid flow cytometry-based structural maintenance of chromosomes 1 (SMC1) phosphorylation assay for identification of ataxia-telangiectasia homozygotes and heterozygotes. Clin Chem 55:463–472
- Raynaud CM, Hernandez J, Llorca FP, Nuciforo P, Mathieu MC, Commo F, Delaloge S, Sabatier L, Andre F, Soria JC (2010) DNA damage repair and telomere length in normal breast, preneoplastic lesions, and invasive cancer. Am J Clin Oncol 33:341–345
- 62. Risques RA, Lai LA, Brentnall TA, Li L, Feng ZD, Gallaher J, Mandelson MT, Potter JD, Bronner MP, Rabinovitch PS (2008) Ulcerative colitis is a disease of accelerated colon aging: evidence from telomere attrition and DNA damage. Gastroenterology 135:410–418
- 63. Nichols GJ, Schaack J, Ornelles DA (2009) Widespread phosphorylation of histone H2AX by species C adenovirus infection requires viral DNA replication. J Virol 83:5987–5998
- 64. Hart LS, Yannone SM, Naczki C, Orlando JS, Waters SB, Akman SA, Chen DJ, Ornelles D, Koumenis C (2005) The adenovirus E4orf6 protein inhibits DNA double strand break repair and radiosensitizes human tumor cells in an E1B-55 K-independent manner. J Biol Chem 280:1474–1481
- O'Donovan PJ, Livingston DM (2010) BRCA1 and BRCA2: breast/ovarian cancer susceptibility gene products and participants in DNA double-strand break repair. Carcinogenesis 31:961–967
- Kulkarni A, Wilson DM 3rd (2008) The involvement of DNA-damage and -repair defects in neurological dysfunction. Am J Hum Genet 82:539–566
- 67. Chistiakov DA (2010) Ligase IV syndrome. Adv Exp Med Biol 685:175-185
- Moreira PI, Nunomura A, Nakamura M, Takeda A, Shenk JC, Aliev G, Smith MA, Perry G (2008) Nucleic acid oxidation in Alzheimer disease. Free Radic Biol Med 44:1493–1505
- 69. Nakabeppu Y, Tsuchimoto D, Yamaguchi H, Sakumi K (2007) Oxidative damage in nucleic acids and Parkinson's disease. J Neurosci Res 85:919–934
- Bernstein KA, Gangloff S, Rothstein R (2010) The RecQ DNA helicases in DNA repair. Annu Rev Genet 44:393–417
- Yannone SM, Roy S, Chan DW, Murphy MB, Huang S, Campisi J, Chen DJ (2001) Werner syndrome protein is regulated and phosphorylated by DNA-dependent protein kinase. J Biol Chem 276:38242–38248
- 72. Kusumoto R, Muftuoglu M, Bohr VA (2007) The role of WRN in DNA repair is affected by post-translational modifications. Mech Ageing Dev 128:50–57
- 73. Brenner DJ, Doll R, Goodhead DT, Hall EJ, Land CE, Little JB, Lubin JH, Preston DL, Preston RJ, Puskin JS, Ron E, Sachs RK, Samet JM, Setlow RB, Zaider M (2003) Cancer risks attributable to low doses of ionizing radiation: assessing what we really know. Proc Natl Acad Sci U S A 100:13761–13766
- Preston DL, Ron E, Tokuoka S, Funamoto S, Nishi N, Soda M, Mabuchi K, Kodama K (2007) Solid cancer incidence in atomic bomb survivors: 1958–1998. Radiat Res 168:1–64
- Costes S, Sachs R, Hlatky L, Vannais D, Waldren C, Fouladi B (2001) Large-mutation spectra induced at hemizygous loci by low-LET radiation: evidence for intrachromosomal proximity effects. Radiat Res 156:545–557
- 76. Brenner DJ, Sachs RK (2006) Estimating radiation-induced cancer risks at very low doses: rationale for using a linear no-threshold approach. Radiat Environ Biophys 44:253–256

- 77. Vadhavkar N, Pham C, Georgescu W, Deschamps T, Heuskin AC, Tang J, Costes SV (2014) Combinatorial DNA damage pairing model based on X-ray-induced foci predicts the dose and LET dependence of cell death in human breast cells. Radiat Res 182:273–281
- Knudson AG Jr.(1971) Mutation and cancer statistical study of retinoblastoma. Proc Natl Acad Sci U S A 68:820–823
- 79. Berenblum I, Shubik P (1949) The persistence of latent tumour cells induced in the mouse's skin by a single application of 9:10-dimethyl-1:2-benzanthracene. Br J Cancer 3:384–386
- Armitage P, Doll R (1954) The age distribution of cancer and a multi-stage theory of carcinogenesis. Br J Cancer 8:1–12
- Armitage P, Doll R (1957) A two-stage theory of carcinogenesis in relation to the age distribution of human cancer. Br J Cancer 11:161–169
- 82. Moolgavkar SH, Dewanji A, Venzon DJ (1988) A stochastic two-stage model for cancer risk assessment I: the hazard function and the probability of tumor. Risk Anal 8:383–392
- Moolgavkar SH, Knudson AG Jr. (1981) Mutation and cancer a model for human carcinogenesis. J Natl Cancer Inst 66:1037–1052
- Moolgavkar SH, Luebeck G (1990) Two-event model for carcinogenesis: biological, mathematical, and statistical considerations. Risk Anal 10:323–341
- Bissell MJ, Radisky D, Rizki A, Weaver VM, Petersen OW (2002) The organizing principle: microenvironmental influences in the normal and malignant bresat. Differentiation 70:537–546
- Little JB (2003) Genomic instability and bystander effects: a historical perspective. Oncogene 22:6978–6987
- 87. Wright EG (2000) Inducible genomic instability: new insights into the biological effects of ionizing radiation. Med Confl Surviv 16:117–130
- Morgan WF (2003) Non-targeted and delayed effects of exposure to ionizing radiation: II. radiation-induced genomic instability and bystander effects in vivo, clastogenic factors and transgenerational effects. Radiat Res 159:581–596
- Morgan WF (2003) Non-targeted and delayed effects of exposure to ionizing radiation: I. Radiation-induced genomic instability and bystander effects in vitro. Radiat Res 159:567–580
- Bonabeau E (2002) Agent-based modeling: methods and techniques for simulating human systems. Proc Natl Acad Sci U S A 99(Suppl 3):7280–7287
- 91. Barcellos-Hoff MH (2007) Cancer as an emergent phenomenon in systems radiation biology. Radiat Env Biophys 47:33–38
- 92. Mukhopadhyay R, Costes SV, Bazarov AV, Hines WC, Barcellos-Hoff MH, Yaswen P (2010) Promotion of variant human mammary epithelial cell outgrowth by ionizing radiation: an agent-based model supported by in vitro studies. Breast Cancer Res 12:R11
- Tang J, Enderling H, Becker-Weimann S, Pham C, Polyzos A, Chen CY, Costes SV (2011) Phenotypic transition maps of 3D breast acini obtained by imaging-guided agent-based modeling. Integr Biol 3:408–421
- 94. Sedelnikova OA, Bonner WM (2006) gamma H2AX in cancer cells Cell Cycle 5:2909-2913
- Taneja N, Davis M, Choy JS, Beckett MA, Singh R, Kron SJ, Weichselbaum RR (2004) Histone H2AX phosphorylation as a predictor of radiosensitivity and target for radiotherapy. J Biol Chem 279:2273–2280
- 96. Gorgoulis VG, Vassiliou LV, Karakaidos P, Zacharatos P, Kotsinas A, Liloglou T, Venere M, Ditullio RA Jr, Kastrinakis NG, Levy B, Kletsas D, Yoneta A, Herlyn M, Kittas C, Halazonetis TD (2005) Activation of the DNA damage checkpoint and genomic instability in human precancerous lesions. Nature 434:907–913
- Bartkova J, Horejsi Z, Koed K, Kramer A, Tort F, Zieger K, Guldberg P, Sehested M, Nesland JM, Lukas C, Orntoft T, Lukas J, Bartek J (2005) DNA damage response as a candidate anticancer barrier in early human tumorigenesis. Nature 434:864–870
- Yu T, MacPhail SH, Banath JP, Klokov D, Olive PL (2006) Endogenous expression of phosphorylated histone H2AX in tumors in relation to DNA double-strand breaks and genomic instability. DNA Repair 5:935–946

- Collado M, Gil J, Efeyan A, Guerra C, Schuhmacher AJ, Barradas M, Benguria A, Zaballos A, Flores JM, Barbacid M, Beach D, Serrano M (2005) Tumour biology: senescence in premalignant tumours. Nature 436:642
- Sak A, Grehl S, Engelhard M, Wierlemann A, Kaelberlah HP, Erichsen P, Pottgen C, Groneberg M, Stuschke M (2009) Long-term in vivo effects of cisplatin on gamma-H2AX foci signaling in peripheral lymphocytes of tumor patients after irradiation. Clin Cancer Res 15:2927–2934
- Podhorecka M, Skladanowski A, Bozko P (2010) H2AX P: its role in DNA damage response and cancer therapy. J Nucleic Acids 2010
- 102. Novik KL, Spinelli JJ, Macarthur AC, Shumansky K, Sipahimalani P, Leach S, Lai A, Connors JM, Gascoyne RD, Gallagher RP, Brooks-Wilson AR (2007) Genetic variation in H2AFX contributes to risk of non-Hodgkin lymphoma. Cancer Epidemiol Biomarkers Prev 16:1098–1106
- Downs JA (2007) Chromatin structure and DNA double-strand break responses in cancer progression and therapy. Oncogene 26:7765–7772
- 104. Fan WW, Zhou KK, Zhao YJ, Wu WT, Chen HY, Jin L, Chen G, Shi JL, Wei QY, Zhang TB, Du GH, Mao Y, Lu DR, Zhou LF (2011) Possible association between genetic variants in the H2AFX promoter region and risk of adult glioma in a Chinese Han population. J Neuro-Oncol 105:211–218
- Rube CE, Grudzenski S, Kuhne M, Dong X, Rief N, Lobrich M, Rube C (2008) DNA double-strand break repair of blood lymphocytes and normal tissues analysed in a preclinical mouse model: implications for radiosensitivity testing. Clin Cancer Res 14:6546–6555
- Goodarzi AA, Noon AT, Deckbar D, Ziv Y, Shiloh Y, Lobrich M, Jeggo PA (2008) ATM signaling facilitates repair of DNA double-strand breaks associated with heterochromatin. Mol Cell 31:167–177
- Smilenov LB, Brenner DJ, Hall EJ (2001) Modest increased sensitivity to radiation oncogenesis in ATM heterozygous versus wild-type mammalian cells. Cancer Res 61:5710–5713
- Varghese S, Schmidt-Ullrich RK, Dritschilo A, Jung M (1999) Enhanced radiation late effects and cellular radiation sensitivity in an ATM heterozygous breast cancer patient. Radiat Oncol Invest 7:231–237
- Broeks A, Urbanus JH, Floore AN, Dahler EC, Klijn JG, Rutgers EJ, Devilee P, Russell NS, Leeuwen FE van, Veer LJ van't (2000) ATM-heterozygous germline mutations contribute to breast cancer-susceptibility. Am J Hum Genet 66:494–500
- 110. Twardella D, Popanda O, Helmbold I, Ebbeler R, Benner A, Fournier D von, Haase W, Sautter-Bihl ML, Wenz F, Schmezer P, Chang-Claude J (2003) Personal characteristics, therapy modalities and individual DNA repair capacity as predictive factors of acute skin toxicity in an unselected cohort of breast cancer patients receiving radiotherapy. Radiother Oncol 69:145–153
- 111. Gabelova A, Farkasova T, Gurska S, Machackova Z, Lukacko P, Witkovsky V (2008) Radiosensitivity of peripheral blood lymphocytes from healthy donors and cervical cancer patients; the correspondence of in vitro data with the clinical outcome. Neoplasma 55:182–191
- 112. Sterpone S, Cornetta T, Padua L, Mastellone V, Giammarino D, Testa A, Tirindelli D, Cozzi R, Donato V (2010) DNA repair capacity and acute radiotherapy adverse effects in Italian breast cancer patients. Mutat Res 684:43–48
- 113. Bourton EC, Plowman PN, Smith D, Arlett CF, Parris CN (2011) Prolonged expression of the gamma-H2AX DNA repair biomarker correlates with excess acute and chronic toxicity from radiotherapy treatment. Int J Cancer 129:2928–2934
- 114. Goutham HV, Mumbrekar KD, Vadhiraja BM, Fernandes DJ, Sharan K, Kanive Parashiva G, Kapaettu S, Bola Sadashiva SR (2012) DNA double-strand break analysis by gamma-H2AX foci: a useful method for determining the overreactors to radiation-induced acute reactions among head-and-neck cancer patients. Int J Radiat Oncol Biol Phys 84:e607–e612
- 115. Wilson PF, Nham PB, Urbin SS, Hinz JM, Jones IM, Thompson LH (2010) Inter-individual variation in DNA double-strand break repair in human fibroblasts before and after exposure to low doses of ionizing radiation. Mutat Res 683:91–97

- 116. Mumbrekar KD, Fernandes DJ, Goutham HV, Sharan K, Vadhiraja BM, Satyamoorthy K, Bola Sadashiva SR (2014) Influence of double-strand break repair on radiation therapy-induced acute skin reactions in breast cancer patients. Int J Radiat Oncol Biol Phys 88:671–676
- 117. Shahidi M, Mozdarani H, Bryant PE (2007) Radiation sensitivity of leukocytes from healthy individuals and breast cancer patients as measured by the alkaline and neutral comet assay. Cancer Lett 257:263–273
- Sanchez P, Penarroja R, Gallegos F, Bravo JL, Rojas E, Benitez-Bribiesca L (2004) DNA damage in peripheral lymphocytes of untreated breast cancer patients. Arch Med Res 35:480–483
- 119. Fernandez MI, Gong YL, Ye YQ, Lin J, Chang D, Wu XF (2013) Gamma-H2AX level in peripheral blood lymphocytes as a risk predictor for bladder cancer. Cancer Res 73
- Xu E, Gong Y, Gu J, Jie L, Ajani JA, Wu X (2013) Risk assessment of esophageal adenocarcinoma using gamma-H2AX assay. Cancer Epidemiol Biomarkers Prev 22:1797–1804
- 121. He YG, Gong YL, Lin J, Chang DW, Gu J, Roth JA, Wu XF (2013) Ionizing radiationinduced gamma-H2AX activity in whole blood culture and the risk of lung cancer. Cancer Epidem Biomar 22:443–451

Chapter 5 Animal Models of Metastasis

Dawn R Cochrane, Dong Lin, Graham Dellaire, Elizabeth C Halvorsen, Jason N Berman, Yuzhou Wang, David G Huntsman and Kevin L Bennewith

Abstract Tumour metastasis is a complex and dynamic process that is estimated to be associated with over 90% of cancer-related deaths. The various stages of the metastatic cascade are made up of interactions between metastatic tumour cells, the solid tumour microenvironment, host normal cells, and host tissue. While our understanding of tumour cell migration and invasion has been greatly improved using a variety of *in vitro* cell culture systems, the inherent complexity of the entire metastatic process is best studied using animal models. There are several in vivo models that are amenable to studying tumour metastasis, and this Chapter will focus on some of the more notable animal models that are commonly used in the metastasis field. We will first consider the zebrafish as a burgeoning model for tumour cell migration and invasion that is particularly well-suited for intravital imaging of labeled tumour cells to study mechanisms underlying metastatic tumour cell dissemination. We then discuss selected genetically engineered mouse models of metastatic cancer, which are powerful tools to elucidate the oncogenic potential of genes of interest. Murine tumour xenografts are also considered as the traditional "gold standard" models for metastasis research. We have included descriptions of strategies to artificially seed murine tissues with tumour cells to study the proliferation

K. L. Bennewith (🖂) · E. C. Halvorsen

E. C. Halvorsen e-mail: lhalvorsen@bccrc.ca

D. R. Cochrane · D. G. Huntsman Centre for Translational and Applied Genomics, British Columbia Cancer Agency, Vancouver, BC, Canada

D. Lin · Y. Wang Experimental Therapeutics, British Columbia Cancer Agency, Vancouver, BC, Canada

G. Dellaire · J. N. Berman Department of Pathology, Dalhousie University, Halifax, NS, Canada

J. N. Berman

Department of Microbiology & Immunology, Dalhousie University, Halifax, NS, Canada Department of Pediatrics, IWK Health Centre and Dalhousie University, Halifax, NS, Canada

© Springer International Publishing Switzerland 2015 C. Maxwell, C. Roskelley (eds.), *Genomic Instability and Cancer Metastasis*, Cancer Metastasis - Biology and Treatment 20, DOI 10.1007/978-3-319-12136-9_5

Integrative Oncology, British Columbia Cancer Agency, Vancouver, BC, Canada e-mail: kbennewi@bccrc.ca

and survival of these cells in host tissues, and we provide an overview of quantitative methods to study the development of spontaneous metastases from solid tumour xenografts. We conclude with the use of patient-derived xenografts (PDXs), which are becoming increasingly common tools to validate experimental findings established in other model systems in an effort to enhance the clinical applicability of *in vivo* data.

Keywords Metastasis • Zebrafish • Genetically engineered mouse models (GEMMs) • Murine tumour xenografts • Patient-derived xenografts (PDX) • Biomarkers • Non-invasive imaging modalities • Personalized cancer therapeutics

Abbreviations

18FDG	18F labeled deoxyglucose
AR	Androgen response
СТ	Computed tomography
EMT	Epithelial to mesenchymal transition
GEMM	Genetically engineered mouse model
MMTV	Mouse mammary tumour virus
LN	Lymph node
PB	Probasin
PDX	Patient-derived xenograft
PET	Positron emission tomography
PIN	Prostatic intraepithelial neoplasia
PyVmT	Polyoma virus middle T
SRC	Subrenal capsule
TAM	Tumour associated macrophage
TIC	Tumour initiating cell
TRAMP	Transgenic adenocarcinoma of the mouse prostate
WAP	Whey acidic protein
XT	Xenotransplantation

Introduction

Tumour metastasis is a complex, multi-step process [1] that is remarkably inefficient, with <0.01% of disseminated tumours cells estimated to eventually form macroscopic metastatic tumours. The process of metastasis can include local invasion of tumour cells into adjacent tissue, however the severe morbidity and mortality associated with metastatic disease is typically due to the development of metastatic tumour foci in tissues that are distant from the primary tumour. Tumour cells can metastasize through the lymphatic system to lymph nodes and/or through the bloodstream to a variety of distant tissues including the lungs, liver, brain, and bone marrow. A great deal of work has been devoted to understanding the genetic

96

and phenotypic characteristics of metastatic tumour cells [2], and tumour cells must have (or develop) the ability to migrate and invade in order to disseminate from a primary tumour. Metastasis begins with dissemination of tumour cells away from the primary tumour, which is a process that can involve tumour cell migration, invasion through extracellular matrix, and intravasation of tumour cells into the circulation. Most tumour cells that enter the bloodstream from a primary tumour mass die in the circulation before the cells can invade distant organs [3, 4]. Tumour cells that enter metastatic target organs can die or lie dormant for extended periods of time [5]. with only a small proportion surviving and proliferating to form micrometastatic tumour foci. The continued growth of micrometastatic tumour foci into life-threatening macrometastatic tumours requires a switch to active angiogenesis [6–8] that is dependent upon active collaboration between the metastatic tumour cells and the surrounding host tissue. The role of host tissue in allowing (or promoting) metastatic tumour growth was first postulated by Stephen Paget in the late 1800s, with his 'seed-and-soil' hypothesis [9] suggesting that metastatic tumour cells (seeds) must enter suitable host tissues (soil) in order to grow into tumour metastases. Indeed, different tumour cell types preferentially form metastases in specific organs, and the site-specificity of metastatic growth seems to reflect the inherent (or acquired) properties of the disseminated tumour cells and the metastatic target organ itself. For example, expression of stromal-derived factor-1 (SDF1) by specific host tissues is thought to facilitate the invasion of tumour cells expressing CXCR4 [10], and the accumulation of bone marrow-derived cells in "pre-metastatic niches" have been shown to increase metastatic tumour growth in murine xenograft systems [11-13].

In order to study the complexity of the metastatic process and the role of the host in metastatic colonization, it is imperative to utilize animal models and there are several *in vivo* models that have been used to discover and study genes that are associated with metastasis. This Chapter will focus on some of the more notable animal models that are in common use in the metastasis field. The first section of this Chapter will focus on the zebrafish as a burgeoning model for tumour cell migration and invasion that is particularly well-suited for the intravital imaging of mechanisms underlying metastasis. This section is followed by a section describing selected genetically engineered mouse models of metastatic cancer and murine xenograft models, both of which have traditionally been considered the "gold standard" models for metastasis research. We will discuss methods to artificially seed tissues with tumour cells, and provide an overview of strategies to study spontaneous metastasis from tumour xenografts. The use of patient-derived xenografts that recapitulate clinical metastatic disease will also be considered.

Zebrafish as a Model for Metastasis

Zebrafish (*Danio rerio*) embryos develop rapidly and are transparent in the early stages of development, which has made them an attractive vertebrate model organism for developmental biologists modeling human diseases [14–16]. More recently,

scientists have used zebrafish as a model organism due to the ease at which their genomes can be manipulated, the inherent capacity to perform forward genetic screening and the convenience of conducting rapid loss of function studies using gene silencing morpholino antisense oligonucleotides. Over the last decade, cancer biologists have also begun to take advantage of these features to use zebrafish to model human malignancies, including as a model of invasion and metastasis by employing human tumour xenografts in the fish.

History of Human Tumour Xenotransplantation (XT) in Zebrafish

For xenotransplantation (XT) studies, cultured cells or primary tumours are fluorescently labeled and injected into zebrafish embryos. Since the developing larvae are transparent, the overall tumour burden and the location of the transplanted cells can be tracked using fluorescent microscopy. Fluorescent labeling is usually performed by either making stable cell lines that express fluorescent proteins or using a membrane stain (such as, CM-DiI or PKH2). Several anatomic sites within the embryo have been tested for injection location, including the duct of Cuvier and the fourth ventricle of the central nervous system, but the volk sac has emerged as the preferential site of injection for xenotransplantation. Similarly, different zebrafish developmental stages have been assaved, with 24-48 h being the most common time point used for injections [17-19]. Importantly, zebrafish embryos are immunopermissive at early developmental stages, which facilitates xenograft work without requiring the induction of immunosuppression in the fish. While used less commonly, human cells can also be transplanted into the peritoneal cavity of adult zebrafish, however to avoid rejection by the immune system, the fish must undergo immune ablation with sub-lethal irradiation or dexamethasone treatment, prior to engraftment [20]. In the zebrafish model system, there are more limited possibilities for orthotopic injection of human cells to more closely mimic tumour interaction with an appropriate microenvironment, however glioblastoma cells have been injected into the hindbrain ventricle of the central nervous system [18]. Studies using primary human cell engraftment are limited, and include leukemic cells [21], as well as pancreatic, breast and prostate tumours [19, 22, 23]. A factor in the maintenance of zebrafish embryos bearing xenografts is the different temperature requirements of the embryos and the cancer cells. Human cancer cell lines are maintained at 37 °C. while zebrafish are maintained at 28 °C [24]. Empirical studies have found that incubation 34-35 °C enables the zebrafish to develop normally and the cancer cells continue to proliferate [17].

One of the main advantages of zebrafish XT is that the fish embryos are transparent and thus amenable to visualization of both tumours and host physiology using fluorescent microscopy of living embryos. As such, several transgenic lines of fish have been created which are particularly useful for direct observation of tumour-host microenvironment interactions. The Tg(fli1-eGFP) line expresses GFP from the fli1 promoter for expression in the endothelial cells to highlight the fish vasculature; a powerful tool for studying tumour-associated angiogenic events [25] as well as tumour cell intravasation (or extravasation) into (or from) the vasculature [26, 27]. Other transgenic lines have been made to label immune cells, such as Tg(mpx:GFP) for neutrophils [28, 29], Tg(mpeg1:eGFP) for macrophages [30] and Tg(cd41:eGFP) for platelets and hematopoietic stem cells [31], which may be useful for investigating the contribution of the innate immune response to tumour invasion and metastasis.

A Tool for Mechanistic Metastasis Studies

Studies have been performed to evaluate the utility of zebrafish xenografts as models for metastasis. The first report of the successful xenotransplantation of human cancer cells in zebrafish occurred in 2005, where metastatic melanoma cells were injected into blastula stage embryos at 5h post fertilization. It was found that the melanoma cells retain a dedifferentiated state, migrate and divide in the developing fish [32]. In another early study, two pancreatic cancer cell lines derived from the same patient, one highly invasive and the other poorly invasive, were transplanted into zebrafish embryos. Two days after injection, the highly invasive line was found to have migrated throughout the embryo, whereas the poorly invasive line remained in one location [19]. Breast cancer cells were injected into zebrafish embryonic duct of Cuvier (directly into the circulation) and the pattern of invasion and metastasis mimicked what had been previously seen in mouse metastasis models [33]. Recently, a more extensive evaluation of this model system was performed using the transplantation of breast, prostate, colon and pancreatic cell lines of known invasive potential [22, 27]. These studies reveal that the zebrafish model reliably recapitulates the invasion data from in vitro studies and mouse metastasis models [22]. Taken together, these studies have demonstrated that the zebrafish xenograft model is faithful to the features of invasive, metastatic cancer.

Movement into and out of vasculature are essential steps in the metastatic cascade, but can be difficult to study in murine models, generally necessitating histological evaluation at discrete time points. Using transgenic fish that have fluorescent vasculature Tg(fli1-eGFP) and high resolution imaging techniques, the interaction between tumour cells and vasculature can be observed in real time. Furthermore, studies in zebrafish xenografts show direct links between the ability of tumour cells to induce host angiogenesis and their ability to metastasize. Breast cancer cells with low metastatic potential injected into zebrafish embryos promoted angiogenesis and became highly invasive under hypoxic conditions. This hypoxia-induced pathogenic angiogenesis was promoted by VEGF secreted by the tumour cells [34–36]. In an early study, breast cancer cells were injected into the peritoneal cavity of adult zebrafish (immunosuppressed with dexamethasone treatment) and had the ability to

invade in a RhoC dependent manner. Live imaging of the intravasation process showed that the tumour cells underwent cytoskeletal remodeling to form protrusive structures that invaded into newly formed vasculature [26]. High resolution microscopy has shown that tumour cells induce active remodeling of vessels at the sites of extravasation, rather than damage or induce vascular leakage, as has been previously thought. The remodeling is characterized by a clustering of endothelial cells around the tumour cells and a change in the architecture of endothelial cell-cell junctions. Furthermore, the expression of Twist, a transcription factor involved in epithelial to mesenchymal transition (EMT), promoted extravasation [34]. Similarly, E-cadherin (CDH1) knock-down in 4T1 murine mammary tumour cells was shown to increase dissemination throughout the zebrafish embryo, consistent with the promotion of EMT by loss of E-cadherin expression [27]. Live imaging has indicated that metastatic cells extravasate and colonize in areas high in neutrophils, and that the neutrophils actively remodel the collagen matrix to assist in the process. It was shown that the inhibition of VEGF signaling decreased tumour growth and angiogenesis, but paradoxically increased neutrophil migration and formation of micrometastases [29].

The zebrafish model system has also been used to dissect pathways relevant for metastasis, highlighting potential areas for therapeutic intervention. In particular, as compared to in vitro assays, the zebrafish provides host vasculature, innate immune cells and stroma that provide some of the host-tumour interactions that may play a role in tumour cell invasion and metastasis. Cell intrinsic factors that may promote metastasis and their interaction with the host tumour environment can also be explored. For example, several studies have demonstrated the importance of matrix metalloproteinases in invasion and metastasis. Galectin-3, which controls PAR-1 and MMP-1, was found to be increased in metastatic gastric cancer. Metastasis of gastric cancer cells in the zebrafish XT model system was inhibited by knockdown of galectin-3, Par-1 or MMP-1 [37]. Similarly, manipulation of another galectin, Gal-4, in pancreatic cancer cells causes changes in metastatic potential [38]. SMYD3 is overexpressed via epigenetic mechanisms in many cancers and is associated with increased MMP-9 expression. Knockdown of either SMYD3 or MMP-9 is capable of decreasing metastatic spread in zebrafish XT model [39]. Another factor, TGF- β , which has long been known to play a role in metastasis, has also been evaluated using zebrafish XT. Specifically, inhibition of TGF- β signaling, or downstream effectors, resulted in decreased metastatic potential in the zebrafish xenograft metastasis model [33].

The zebrafish model has also been leveraged to perform studies of cancer stem cells or tumour initiating cells (TICs) and their role in cancer progression and metastasis. For example, glioma cells or cancer stem cells derived from a glioma cell line were transplanted into zebrafish embryos and metastatic potential assessed. The stem cell population displayed higher metastatic potential, a process which was dependent on MMP-9 and dampened with an MMP-9 inhibitor [40]. TICs isolated from several prostate cell lines, as well as from primary prostate tumours, showed very high metastatic ability in the zebrafish xenograft model, even with very few transplanted cells [23].
Advantages and Disadvantages of Zebrafish XT as A Metastasis Model

Mouse models have been the gold standard for metastasis studies, although the zebrafish XT model system does offer some advantages. First, zebrafish XT does not require immune compromised animals. While human tumour xenograft studies have to be performed in mouse strains that lack a competent immune system, the zebrafish immune system does not completely mature until 28 days post-fertilization [41], negating the need for immunocompromised strains. Secondly, the direct visualization and live imaging of metastasis within zebrafish embryos can be readily performed due to their transparency. Third, their diminutive size and extraordinary fecundity (a single clutch can number in the hundreds of animals) can facilitate medium throughput screening of hundreds of embryos distributed into 96-well plates; a feat that is not possible with any other vertebrate animal model. The number of cells injected into zebrafish embryos is also low compared to murine xenografts, typically 50-100 cells [43-45], which may provide an advantage in cases where there are limited materials, such as with primary tissue transplants (e.g. from needle biopsy). Finally, automated or semi-automated whole animal imaging, combined with image analysis algorithms, enables quantification of metastatic spread and dissemination kinetics [27, 42].

Despite the advantages discussed above, zebrafish also present several possible disadvantages that should be considered. One of the chief concerns, as alluded to briefly above, is the possible lack of appropriate orthotopic sites for tumour-host interactions within the fish. Zebrafish lack lungs, prostate and mammary tissue and as such no orthotopic sites for tumours derived from these tissues exist in the fish, and thus conclusions drawn from xenograft studies using these tumours should be confirmed using other assays or models, such as mice. In addition, appropriate growth factors for the faithful proliferation of certain tumours may also be lacking (or in limited quantities) to sustain growth of tumour within fish. However, this challenge can be partly mitigated by supplementing growth factors directly in the embryo water of engrafted fish. Finally, zebrafish, despite sharing >70% genetic identify with humans, nonetheless lack several genes (e.g. the p14/ARF (CDKN2A) and PML tumour suppressors) and harbor at times multiple paralogs of human genes (e.g. there four Twist paralogs in zebrafish), which must be considered in interpreting results from zebrafish cancer models [46].

Future Studies

The zebrafish XT model system will continue to be used for mechanistic studies that will complement other model systems. However, to improve and expand the use of zebrafish XT some of the challenges regarding lack of appropriate orthotopic sites and genetic dissimilarities between fish and humans need to be addressed through new approaches. For example, co-injection of "helper-cells" or stroma expressing

appropriate tissue-specific factors may aid in engraftment of disparate tumour types for which the appropriate orthotopic tissue is lacking in the fish.

A second major area of future work using zebrafish XT will be the employment of human primary tumour xenografts. While the vast majority of experiments in zebrafish xenografts have used cultured cells, there is movement towards the injection of tumour cells, similar to patient-derived xenograft (PDX) models used in mice. The first such study used labeled primary pancreatic tumours that were transplanted into zebrafish embryos and found to be metastatic, whereas non-cancerous tissue showed no signs of invasion or metastasis [19]. In another study, a subpopulation of cells, the putative TIC population, was isolated from prostate tumours and transplanted into zebrafish [23]. In order to characterize invasive properties prior to xenografts, primary lung tumour cells were maintained in short term cell culture (5–10 days). The amount of invasion and micrometastases found in the zebrafish xenografts of the primary cultures mirrored the invasive potential found *in vitro* [22]. This approach has recently been pioneered for primary patient bone marrow-derived diagnostic leukemia samples, with the opportunity to personalize therapy by providing real time *in vivo* response data to specific targeted agents (Bentley et al., submitted).

Another burgeoning area employing the zebrafish XT model is chemosensitivity assays to define tumour-drug interactions and possibly even provide an avenue for personalization of therapy using PDX. Finally, the mechanistic insights gleaned from the metastasis studies using zebrafish XT will highlight which pathways can be targeted for anti-metastasis therapy. Therefore, future studies are also likely to leverage both the metastasis and chemosensitivity applications of the model system to evaluate potential novel anti-metastasis therapies.

Genetically Engineered Mouse Models of Metastatic Cancer

Genetically engineered mouse models (GEMMs) have been generated in an attempt to model spontaneous tumour formation and progression. As a tool for metastasis research, it is important the models mimic the progression from localized cancer to metastatic disease and that the metastatic lesions occur in the appropriate locations. While there are metastatic GEMMs for many cancers, including bladder, thyroid, lung, pancreas, and skin, this section will focus on genetic models of breast and prostate cancer. These models have been invaluable for study of the metastatic cascade and dissection of the factors which contribute to metastasis.

Breast Cancer

The most common sites for breast cancer metastasis are liver, brain, lung and bone. To induce mammary tumours, oncogenes such as Her2 (human epidermal growth factor receptor-2), MET (hepatocyte growth factor receptor), Wnt-1 (wingless-type

mouse mammary tumour virus integration site family member-1) can be overexpressed specifically in mammary epithelial cells. The mammary specific promoters used most often to drive these oncogenes to establish transgenic mice are the mouse mammary tumour virus (MMTV) or whey acidic protein (WAP). When Her2/neu was constitutively overexpressed in mammary epithelial cells under the control of the MMTV promoter, tumours arose with an average onset of 4 months and lung metastases were detected at high frequency [47]. Invasion and metastasis can be accelerated in the Her2/neu transgenic mice by the additional activation of Akt/ PKB (phosphokinase B) [48]. Activating MET mutations cause mammary tumours and spontaneous metastasis to the lung, lymph nodes (LN), kidney and heart [49]. Metastasis in MMTV-Wnt-1 mice are rare at the time of initial tumour detection, however, after the primary tumour is removed, frequent lung and LN metastases are observed [50].

PyVmT (polyoma virus middle T) mice recapitulate breast tumour progression in humans from a pre-malignant state through to the development of metastatic mammary tumours [51]. PyVmT mice develop poorly differentiated mammary tumours with high penetrance of pulmonary metastases, and transplantation of spontaneous-ly-derived PyVmT tumours into cleared mammary fat pads result in transplantable metastatic mammary tumours [52]. This model has been useful in studying various factors involved in breast tumour metastasis, including the importance of Akt/PKB signaling in tumour development and metastatic progression [53]. The role of host cells in promoting PyVmT metastasis has also been studied extensively, including the role of colony-stimulating factor-1 (CSF-1) induced tumour associated macrophages (TAMs) on primary tumour growth and metastasis [54, 55].

Mouse models with inducible, rather than constitutive, oncogene overexpression provide a means to study tumour progression and metastatic growth with tighter control over the induction of the oncogene. For example, transgenic mice that express the Her2 transgene under control of the tetracycline promoter create inducible overexpression of Her2/neu in the mammary gland upon treatment of the mice with doxycycline. Conditional activation of Her2/neu with doxycycline results in rapid onset of mammary tumours and pulmonary metastases. Both the primary and metastatic lesions continue to rely on Her2/neu activation as withdrawal of doxycycline results in regression [56]. Similarly, when Wnt-1 is conditionally overexpressed, tumours metastasize and subsequently regress upon removal of doxycycline [57]. However, recurrence of the regressed tumours differs in the two model systems, with tumour recurrence being much more frequent with the *Her2/neu* oncogene than with *Wnt-1* [58]. In the Wnt-1 inducible system, loss of a p53 allele decreases the number of tumours that regress while also increasing the frequency of recurrent tumours [57].

Mammary specific deletion of tumour suppressor genes has also been used to model breast cancer metastases. For example, p53-deficient mice generated using the WAP-Cre system resulted in estrogen receptor positive tumours that metastasized to the lungs and liver with a long latency [59]. Similarly, loss of BRCA1 (breast cancer-1, early onset) alone in the mammary gland produces tumours, albeit at low frequency and a long latency [60], and tumours that formed in this model demonstrated alterations in p53 expression. A combination of BRCA1 loss and loss of one p53 allele produces tumours that metastasize to the liver and lungs with a shorter latency than with p53 deficiency alone [61]. Loss of both p53 and E-cadherin using the WAP-Cre system results in lobular tumours which metastasize to skin, lungs, liver, spleen, pancreas, GI tract and peritoneal cavity, with bone metastases also observed at low frequency [62].

Loss of PTEN (phosphatase and tensin homolog) and Her2 overexpression are genetic events that often co-occur in breast cancers, and GEMMs with Her2 overexpression results in tumours with the basal-like subtype that metastasize to the lungs [63]. In a similar model, control of Her2 expression and Cre recombinase were put under the control of the MMTV promoter in a bi-cistronic transcript to generate the PTEN-deficient "NIC" mice. The tumours in these mice were of the luminal sub-type of breast cancer and there was an increase in metastatic lung lesions in these PTEN-deficient mice compared to the PTEN heterozygotes or the parental strain [64]. There was increased angiogenesis observed at the primary tumour site, which is postulated to contribute to the higher rates of metastasis [64].

Overall, there are several elegant mouse models that spontaneously, or after induction, form mammary tumours that can metastasize, and these models have provided important insights into the role of various oncogenes and proto-oncogenes in mammary tumour development. It should be noted that mice have ten mammary fat pads, all of which can develop mammary tumours in these models. Thus, experiments designed to study the effect of surgical primary tumour resection on the growth and development of residual metastatic disease is technically challenging in GEMMs, and murine xenograft models are better suited for these types of studies as will be discussed in sections 4.4–4.5 below.

Prostate Cancer

Prostate cancers metastasize primarily to bone and lymph nodes; however, there are no GEMMs that consistently and reproducibly metastasize to bone. Prostate specific expression of oncogenes is often achieved using the probasin (PB) promoter or the modified ARR₂PB promoter, which increases prostate gland specificity by adding two androgen response (AR) elements to the PB promoter. Valkenburg and Williams have authored an extensive review describing various mouse models of prostate cancer, including discussion of how closely each model recapitulates human prostate cancer [65].

PB driven expression of the SV40 T antigen is used to generate the TRAMP (transgenic adenocarcinoma of the mouse prostate) model. The TRAMP model follows progression of the human disease from prostatic intraepithelial neoplasia (PIN), to multifocal adenocarcinoma, followed by invasive metastases in lung and LN (rarely bone, kidney and adrenal glands) [66–68]. The Lady model is a modified version of the TRAMP model, and was created in 1998 using a longer PB promoter fragment linked to a deletion mutant of SV40 that only expresses the large T antigen (and not the small T antigen). Prostate tumours generated in Lady

mice are invasive and display features of neuroendocrine differentiation, but do not metastasize. Additional genetic alterations have been made to Lady and TRAMP models in an attempt to discover factors that contribute to altered metastatic potential in the two models. For example, overexpression of hepsin (cell surface serine protease often overexpressed in prostate cancer) in the Lady model has no impact on primary tumour formation, but promotes metastatic dissemination to lung, liver, and bone [69]. Similarly, expression of a dominant negative TGF- β receptor type II, thereby blocking TGF- β signaling, produces primary tumours of similar sizes, but with increased metastases compared to mice expressing T-antigen alone [70]. Prostate specific deletion of fibroblast growth factor receptor-1 (FGFR1) in the TRAMP background results in smaller tumours and less metastases, indicating its importance in prostate cancer progression [71].

Other transgenic models that mimic the progression of prostate cancer from PIN to metastasis include the PSP-KIMAP model, where the SV40 T antigen is knocked into PSP94 (prostate secretory protein 94), which is highly expressed in prostate cells. Metastases are observed in the lymph nodes, lung and liver [72]. Since PTEN is often lost in prostate cancer and in 40% of metastatic lesions [73], mice with prostate specific deletion of PTEN (using ARR₂PB-Cre) have been created. Prostate-specific deletion of PTEN results in PIN and eventually invasive prostate cancer with metastases in lymph nodes and lungs [74]. Similarly, SMAD4 is downregulated in high grade PIN and prostate cancer compared to benign prostatic potential [76]. Conditional inactivation of either p53 or retinoblastoma protein (Rb) individually in the prostate results in PIN and neoplasia, although combined loss of both tumour suppressors causes a more aggressive phenotype with neuroendocrine differentiation and distal metastases to the lymph nodes, liver and lung.

As with the mammary tumour GEMMs discussed above, the generation of prostate cancer GEMMs has greatly increased our understanding of oncogenes and tumour suppressor genes that are involved in prostate tumourigenesis and prostate cancer progression. Prostate cancer GEMMs that consistently metastasize to bone have not been reported and this is an important consideration since bone is the primary site of prostate cancer metastasis. The study of prostate tumour metastasis is therefore restricted to the use of prostate tumour xenografts as will be discussed in Sects. 4.4–4.5.

Future Directions

Current GEMMs offer a good framework for assessing the role of particular oncogenes, proto-oncogenes, and tumour suppressor genes in tumour development and progression, with several models also allowing study of the metastatic process. One important caveat to the use of GEMMs is that tumours result from ubiquitous expression of transgenes within the cells of a tissue, which does not mimic the state of human cancers where a minority of cancer cells arises in a milieu of untransformed cells. There are also no currently available models for spontaneous brain cancer metastases, and the kinetics of GEMM tumour progression and metastasis is difficult to compare to patients in the clinic. Primary tumour volumes can limit the study of metastases, since mice can succumb to a large primary tumour burden prior to the detection of metastases in distant tissues and surgical resection of spontaneous primary tumours is often not possible. In these instances, tumour xenograft models allow researchers to more easily distinguish the role of a particular gene on growth of the primary tumour xenograft compared to the development and growth of tumour metastases.

Murine Xenograft Models

Murine xenografts have been used for decades to study primary tumour development, response to therapy, and the metastatic process. With the widespread use of tumour xenografts to study metastatic disease in mice, this section will focus on two important considerations for designing experiments to determine the effect of a particular gene on the metastatic process. We will first discuss different tumour cell implantation methods that have been employed to study the colonization of tissues with tumour cells, and will subsequently consider important methodologies for quantifying the development and growth of metastatic tumours in mice.

Implantation Sites and Artificial Models of Metastasis

Xenografted tumours were originally grown by implanting tumour cell suspensions in subcutaneous sites, although efforts to more closely model the natural tumour environment has led researchers to increasingly implant tumour xenografts in orthotopic sites. While orthotopic implantation of many tumour types can require surgical expertise (e.g., pancreatic tumours), there are several advantages inherent with implanting tumour xenografts in the tissue of origin. Murine skin is relatively poorly vascularized relative to most other organs, and there can be profound differences between subcutaneous and orthotopic tumour xenografts. The site of primary tumour implantation can have a profound effect on response of the tumour to chemotherapy [77], and transcription factors such as hypoxia-inducible factor-1 (HIF-1) can produce opposing effects on the growth of tumours implanted in subcutaneous or orthotopic sites [78]. Similarly, subcutaneously implanted tumours can be more poorly oxygenated than orthotopically implanted tumours of the same cell type [79], with orthotopic tumours thought to more accurately reflect the tumour microenvironment found in patients. Metastatic dissemination of tumour cells from a primary tumour can also be profoundly influenced by xenograft implantation site [80] with obvious implications for studying metastasis in pre-clinical models. In general, the use of murine tumour xenografts to study metastasis is appealing because the entire process of metastasis (from tumour cell dissemination to target organ colonization) is required to produce a measureable metastatic tumour.

5 Animal Models of Metastasis

Surgical resection of primary tumour xenografts can be performed in order to study the growth of spontaneous tumour metastases in the absence of the primary tumour, and to derive increasingly aggressive metastatic variants of tumour cell lines. Orthotopic breast tumours or melanomas are the most amenable to surgical resection, and removal of the primary tumour allows growth of disseminated tumour cells in metastatic tissues for periods of time that would otherwise be restricted by primary tumour volume. To generate clinically relevant metastatic variants of orthotopically implanted tumour cells, metastatic tumour foci are isolated from tissues after surgical resection of the primary tumour and re-implanted in the orthotopic site for repeated generations. This methodology has been used to generate derivatives of the original tumour cell line that spontaneously metastasize to the tissue of interest [81, 82]. Importantly, these spontaneously derived metastatic tumour cell variants have been used to assess fundamental differences between chemotherapeutic response of primary tumours and tumour metastases, and between pre-clinical tumour xenograft models and patients in the clinic [83].

In order to study the process of tumour cell colonization in greater detail, many groups rely on artificial distribution of tumour cells to metastatic target organs. Intravenous injection of tumour cells into the lateral tail vein has been used to directly seed lung tissue with tumour cells. Similarly, intrasplenic injections have been used to seed the liver with tumour cells, intraperitoneal injection of ovarian cancer cells is used to generate ascites and multiple metastatic nodules in the peritoneal cavity, and intracardiac injection of tumour cells are typically used to seed tissues throughout the mouse including the brain and bone. It is important to note that while each of these methods may produce tumours in target tissues (depending on the capacity of the injected tumour cells to colonize the tissue), none of these methods recapitulate (or require) early events of the metastatic process such as tumour cell migration, invasion, and intravasation. Seeding tissues with tumour cells has produced important contributions to our understanding of the genes involved in promoting metastatic growth. Systemic injection of tumour cells facilitates intravital imaging of tumour cell extravasation in tissues at defined timepoints after tumour cell delivery into the circulation. Also, tumour cell foci that grow in tissues after intravenous or intracardiac injection of tumour cells can be harvested and reinjected several times to select for tumour cells that preferentially colonize a particular tissue. Expression analyses of the resultant tumour cell lines have been used to identify many genes that mediate metastatic growth in tissues such as the lungs [84] or brain [85]. These models differ from the spontaneously metastasizing orthotopic tumour cell line derivatives discussed above, but have nevertheless provided important insights into molecular mediators of tissue colonization.

Regardless of the choice of murine xenograft model, the quality of data generated to describe the role of a particular gene in the process of metastasis is entirely dependent on the method selected to quantify metastatic growth. Common methods to measure experimental metastases range from enumeration of macroscopic metastases on the surface of a tissue to highly sensitive single cell analyses of disaggregated tissues that contain metastatic tumour cells.

Quantification of Metastatic Disease

The crudest methods to quantify metastases involve weighing tissues from tumourbearing animals to determine the overall mass of the metastatic tumour burden, and counting macroscopic metastatic foci on the surface of the excised tissue. While these methods may be appropriate for quantifying large differences in metastatic tumour burden between experimental groups, these sorts of measurements are unable to determine how a particular gene is influencing the different steps of metastasis and would typically require continuation of experiments beyond humane endpoints for the animals unless the primary tumour is surgically resected. Non-invasive imaging modalities are far better alternatives for studying the development of macroscopic metastases, with the distinct advantage that metastatic tumour growth can be quantified in the same animal over time.

Positron emission tomography (PET) combined with computed tomography (CT) allows researchers to monitor metastatic tumour burden in the same mouse over time, and quantification of 18F labeled deoxyglucose (18FDG; an analogue of glucose) by PET is the most common method to detect metabolically active tumour metastases in animal models and the clinic. Detection of experimental radio-labeled antibodies against tumour cell specific markers is also facilitated by PET imaging, and the development of novel PET agents has the potential to provide improvements over 18FDG in terms of tumour specificity and the detection of tumour sub-types. Radio-labeled PET tracers emit positrons that interact with electrons to produce a pair of gamma photons traveling in opposite directions, and the detection of these gamma photons provides a means to localize PET signals. The resolution of PET imaging is on the order of a few millimeters, depending on the energy of the emitted positron (and hence the radioisotope used in the tracer) and the instrumentation used for imaging [86]. Overall, PET/CT imaging is an excellent modality to detect metastatic disease and is applicable to analyzing the metastatic growth of spontaneously derived tumours, which represents a distinct advantage over imaging modalities that are based on tumour cells that have been genetically modified to express fluorescent markers or bioluminescent enzymes (see below). However, the detection sensitivity of PET/CT is not amenable to quantifying microscopic metastatic disease, and PET/CT imaging requires access to the imaging equipment, appropriate expertise for image analyses, and a cyclotron to produce the radiolabeled PET tracers. The imaging and data analysis can also be labour intensive relative to non-invasive imaging modalities based on the detection and quantification of bioluminescent or fluorescent tumour cells.

Bioluminescent imaging is an alternative to PET/CT imaging that is applicable to model systems where the tumour cells can be genetically manipulated. Tumour cells are engineered to stably express a luciferase enzyme prior to introduction into mice, and intraperitoneal injection of the enzyme substrate (e.g., D-luciferin for firefly luciferase) causes the production of bioluminescent light from the tumour cells. Bioluminescence is detected and quantified by a high sensitivity camera coupled to an image analysis system, and the bioluminescent signal intensity can be monitored in the same mouse over time. Most systems allow imaging of five mice simultaneously, and image analysis software to quantify photon emission from tumour cells is relatively straightforward to use. However, in the case of a mouse with metastatic disease, it is difficult to determine the source of bioluminescenceproducing tumour cells *in situ*, and confirmation of the source of metastatic disease requires combination of bioluminescent imaging with CT scans or post-mortem comparative analyses. Large primary tumours tend to mask light emitted from smaller metastatic tumour foci, and emitted light can be quenched by tissues containing high levels of hemoglobin (e.g., liver). Regardless, bioluminescent imaging is increasingly being used to quantify metastatic disease in xenograft models of metastasis. While non-invasive imaging modalities allow researchers to monitor and quantify metastatic tumour growth in the same animal as a function of time, these methods necessarily require the development of macroscopic tumours dependent on the detection sensitivity of the imaging modality. Genes that affect earlier steps in the metastatic process or metastatic tumour cells that do not form large metastatic tumour foci are more readily studied using complimentary microscopic or single cell-based analyses.

Tissue sections stained with haematoxylin and eosin allow observation of metastatic tumour cell foci that can be enumerated and/or processed by image analysis software to determine overall metastatic tumour area in a given tissue section. Microscopic analyses of tissues provides important information about the relative size and number of metastatic tumour cell foci in a tissue, provided appropriate consideration is given to analyze multiple sections from increasing depth in the tissue in order to get an accurate (3-dimensional) indication of metastatic growth. Identification of smaller metastatic tumour cell foci can be challenging, and is often facilitated by staining serial sections for tumour cell specific markers (e.g., cytokeratins) or through the use of fluorescently-tagged tumour cells. Fluorescentlytagged tumour cells are also amenable to analyzing metastatic tumour cell content in disaggregated tissue by flow cytometry. However, it should be noted that some immunocompetent mouse strains (e.g., Balb/c mice) have been reported to mount an immune response against tumour cells expressing fluorescent proteins [87–89] and therefore the use of fluorescently-tagged tumour cells in a particular model system should be carefully considered [90]. The most accurate cell-based method to quantify metastatic tumour cells is to plate bulk cells derived from disaggregated tissues into ex vivo clonogenic assays. Clonogenic assays allow detection of single clonogenic tumour cells in an entire tissue provided the plating efficiency of the tumour cell lines ex vivo is taken into account. Transfection of tumour cells with drug-resistance markers prior to implantation in the mouse allows quantification of clonogenic metastatic tumour cells in disaggregated tissue without the staining background generated by fibroblasts and immune cells present in the disaggregated cell suspension.

Taken together, there are a variety of methods available to quantify tumour metastases in spontaneous and transplantable tumour xenografts. Traditional human cancer cell line-based xenograft models have provided insights into the pathogenesis and metastasis of cancer, resulting in the development of a handful of therapeutic agents for treatment of the disease. However, most of the data obtained on potential cancer therapeutics with such in vivo models failed to translate into improvements in the management of cancer patients [91, 92]. Only about 5% of new potential anti-cancer drugs, that successfully passed preclinical in vivo tests, have significant efficacy in clinical trials and are approved for clinical usage by the U.S. Food & Drug Administration [93]. This discrepancy appears to stem from a lack of ability of established cancer cell lines to accurately predict the efficacy of anticancer agents in the clinic. This is thought to be due to increased cellular homogeneity after long-term culturing of the cell lines, contrasting with the heterogeneity of the parental cancers. Furthermore, cell line-based xenografts rarely possess the tissue architecture of the original cancer specimens from which the cell lines were derived and, consequently, do not properly represent the micro-environment as found in the original malignancies. There is therefore an urgent need for experimental cancer models that better represent the heterogeneity of human tumours and accurately predict clinical drug efficacies. In view of this, PDX models, generated by transplanting freshly resected patients' tumour tissue into immunodeficient murine hosts without an intermediate *in vitro* culture step, represent a major advancement.

Patient-Derived Xenografts

Implantation Sites

There are currently three common PDX graft sites used in immunodeficient mice to establish PDX models, namely the subcutaneous, orthotopic and subrenal capsule (SRC) sites. The subcutaneous graft site is the most commonly used site and has various advantages, including easy tissue implantation and monitoring of the developing tumour [94]. However, this site is known for a generally low engraftment rate, particularly in the case of some types of cancer [95]. This may be due to its well-known lack of vascularization and hence potentially inadequate nutrient supply that can lead to loss of cancer subpopulations. Also, the metastatic rates of subcutaneous xenografts are low, even for grafts derived from tumours that were highly metastatic in the patient. The orthotopic graft site provides a microenvironment similar to that of the original cancer and is theoretically the ideal graft site for testing spontaneous metastatic ability of cancer tissue. However, this site can have a limited xenograft carrying capacity depending on the tumour type, which can restrict the use of orthotopic implantation sites for establishment of models of some types of cancer. Successful engraftment at the orthotopic site was also found to be limited to highly advanced cancers, as found for the subcutaneous site [96]. PDX grafting in the SRC was first proposed for tissue implantation by Bogden [97]. A major advantage of the SRC graft site is the availability of an instant blood supply due to the high vascularization of the kidney [98, 99]. This provides high graft perfusion and the abundant supply of nutrients, hormones, growth factors and oxygen to the transplanted tissue is instrumental to the success of engraftment [100-104]. In addition, measuring the area of graft growing into the kidney parenchyma allows quantitative examination of the local invasion ability of implanted tumours. As found with orthotopic grafting, the surgical procedure involved in SRC grafting is more difficult than subcutaneous implantation of tumour fragments, but the relatively high PDX take rate make SRC grafting a preferred site for PDX development.

Metastatic PDX Models of Various Types of Cancer

The study of mechanisms of cancer metastasis is hampered by limited access to clinical metastatic samples and the scarcity of *in vivo* models that show spontaneous, clinically relevant metastasis [105]. Since the first report of successful xenografting of a patient's tumour into nude mice in 1969, a number of PDX models derived from various types of primary and metastatic clinical samples have been established [94]. Some of the xenografts showed spontaneous, clinically relevant metastatic ability and provided unlimited sources of metastatic tissue for investigation of cancer metastasis mechanisms and identification of new therapeutic targets.

There is a long history of success with the establishment of colorectal PDX models [106, 107]. Recently, characterization of a large panel of colorectal PDX models showed that 13 out of 41 models (32%), established via orthotopic grafting, gave rise to metastases in mesenteric lymph nodes, liver or lung. Interestingly, the sites of metastatic dissemination of cancer cells in these models are similar to those commonly seen in the clinic [108]. PDX models showing similar spontaneous metastasis were also reported by other groups [109]. Orthotopic PDX models of pancreatic cancer have been also reported to retain stromal components and develop regional and distant metastases. Spontaneous metastases to liver and lung and development of malignant ascites have been observed in a number of studies [110–112].

Historically, the lowest success rates and metastatic abilities of PDX models have been observed in hormone-dependent mammary and prostatic carcinomas [113, 114]. However, recent efforts, involving orthotopic implantation of human breast cancer tissue with supplementation of estrogen, have resulted in improved engraftment efficiency and generated a number of improved PDX models [115, 116]. These models represent the diverse cancer subtypes observed in the clinic and, importantly, maintain essential features of the original tumours, especially their ability to metastasize to specific sites [115]. In the case of prostate cancer, a few PDX models derived from subcutaneous implantation of metastatic and/or castration-resistant cancer tissues have been described to show spontaneous metastasis to lymph node, liver and lung [113, 117, 118]. Recently, a new panel of PDX models developed via SRC grafting from needle biopsies, as well as primary and metastatic tissues, have shown a much higher engraftment rate [119-122]. Importantly, these models, including multiple models derived from biopsies of the same patient, showed various spontaneous metastatic abilities and provided valuable tools for studying intratumoural heterogeneity and underlying mechanisms of metastasis. Furthermore, recent studies of both breast and prostate cancers showed that the rate of engraftment could be used as a prognostic factor for patient survival time even for individuals with newly diagnosed disease who did not have detectable

metastases at the time of surgery. Thus, the PDX take rate in mice could potentially be used as a surrogate prognostic indicator of aggressiveness and risk of disease progression.

Advantages and Major Applications of PDX Models

There are a number of general advantages of PDX models compared to traditional cell line xenograft models. First, the PDX models retain histologic characteristics and certain intratumoural cellular heterogeneity seen in the parental tumours [115, 123–126]. Secondly, these PDX models, especially initially, maintain the human stromal component, and therefore are thought to more accurately represent the complex biochemical and physical interactions between the cancer cells and various components of their microenvironment as found in the original malignancies [94, 127]. Third, the PDX models preserve molecular characteristics of the original cancer, including chromosomal copy number variants [115, 128, 129], single-nucleotide polymorphisms [116, 130] and gene expression profiles [116, 126, 128, 131, 132]. Fourth, PDX models show responses to anti-cancer therapy that are more representative of patient tumour response in the clinic [133–135].

PDX models provide valuable tools for (i) addressing a variety of questions regarding the cause, progression, and therapy of cancer (e.g., for investigating the mechanisms of cancer metastasis and identifying new therapeutic targets), (ii) translational research (e.g., for efficacy and toxicity testing of potential and established anticancer therapeutic approaches and biomarker discovery) and (iii) personalized cancer therapy.

Application of PDX Models to the Study of Cancer Metastasis

PDX models retain most aspects of the human tumour microenvironment at an early passage, and they therefore provide opportunity to study the role of the solid tumour microenvironment in promoting metastatic growth that can complement work done in other tumour xenograft models. For example, the poorly oxygenated (hypoxic) microenvironment in solid tumours is known to be associated with treatment resistance, increased metastatic potential, and poor outcome. David Hedley and colleagues recently developed a series of orthotopically grown PDX models of pancreatic cancer and identified significant associations between hypoxia and aggressive growth or development of metastases in these models [112]. Furthermore, gene expression analysis showed increased expression of genes involved in cell survival and proliferation in the hypoxic models. This study indicated that hypoxia is a major adverse prognostic factor in pancreatic cancer patients and supports the development of hypoxia-targeting therapy.

Identification of metastasis-driving genes by a comparative gene expression profile analysis of primary and secondary clinical samples forms a major challenge, since such samples are not homogeneous, but contain sub-populations of cancer cells that vary widely in metastatic abilities. To overcome this hurdle, a number of paired transplantable metastatic and non-metastatic prostate cancer PDX models have been successfully developed from individual patients' primary cancer tissues, such as the paired metastatic PCa1-met and non-metastatic PCa2 [120, 121], the LTL220M and LTL220N [122] and the LTL313 series PDX models [126]. Comparative gene expression analyses and bioinformatics and network analyses of these PDXs have led to identification of a number of cancer metastasis-associated genes [121, 136]. Similarly, differentially expressed miRNAs in a pair of metastatic and non-metastatic prostate cancer PDXs, LTL313B and LTL313H, have been identified, which likely include potential biomarkers for prostate cancer metastasis [137].

PDXs for Drug and Biomarker Discovery

In the era of targeted cancer therapy, it is important to evaluate drug efficacies using experimental models showing clinically relevant expression of molecular targets. Preservation of a patient's tumour genomic profile and tumour microenvironment in PDXs gives the opportunity to use primary patient tumour grafts as a model to improve the translation of preclinical therapeutic strategies in oncology. One preclinical setting in which PDXs can be extremely relevant is presented by evaluation of new potential drugs for cancer treatment. New drugs can be tested using a panel of PDXs which faithfully represent the heterogeneity of a specific cancer type, potentially leading to identification of the best treatment regimen for the particular cancer. To date, a large number of approved and pre-clinical anticancer drugs have been tested using PDX models, as recently reviewed [94, 138].

PDX models are also used for biomarker discovery. For example, taking advantage of a panel of metastatic colorectal PDX models, Bertotti et al. identified HER2 as a predictor of resistance to anti-epidermal growth factor receptor Cetuximab antibodies and revealed that the combined inhibition of HER2 and EGFR induced overt, long-lasting tumour regression [131]. The response to Cetuximab was also investigated using PDXs generated from various types of cancer, leading to identification of MET activation as a mechanism for drug resistance [139]. In addition, John J. Tentler et al. have identified activation of the Wnt pathway as a biomarker predicting resistance to a mitogen-activated protein kinase kinase MEK1/2 inhibitor, AZD6244, in Kras mutant colorectal cancer PDTX models, indicating a potential combination therapy strategy for the disease [140].

Personalized Cancer Therapy

Since each patient's cancer is unique, cancer therapy should ideally be tailored to individual patients. Choosing the most effective, least toxic regimen for a patient is

one of the major challenges faced by oncologists today [141]. In view of this, early generation PDX models of the patient's own malignancy may be useful for predictive drug efficacy testing for personalized cancer therapy. One attempt to use PDX models aiding in the selection of appropriate chemotherapeutic agents in personalized cancer therapy is the use of "mouse Avatars" [134, 135, 138, 142]. In a pilot study, Hidalgo et al. implanted various advanced solid tumours resected from 14 patients in immunodeficient mice, and treated them with 63 drugs in 232 treatment regimens. Overall, a significant correlation was observed between drug activity in the model and clinical outcome, both in terms of resistance and sensitivity. Importantly, using the treatments selected for each individual patient based on the results obtained with the corresponding PDX, 11 of 14 patients achieved a partial response. It should be noted that the treatments selected were not obvious and would not have been the first choice for a conventional second or third line treatment. This suggests that PDX models can be used to select effective personalized treatments. Recently, the same group integrated next-generation sequencing with the Avatar mouse model, identified tumour-specific mutation and copy-number alterations, and treated the PDX models with targeted drugs. It was reported that 6 out of 13 patients who received a personalized treatment achieved durable partial remissions, demonstrating that analysis of somatic genetic alterations combined with use of PDX models can be performed in the clinical setting and facilitate selection of specific treatments for personalized cancer therapy. A similar concordance between early generation engrafted tumours and original patient tumours has been established by different groups as well [128, 132, 143]. Although at present the sample size is too small to conclude if this approach is better than the standard-of-care approach for selecting a particular therapy, the robustness and accuracy of these systems in predicting anti-cancer drug efficacy in an individual patient will likely be validated in the near future.

Caveats and Future Prospects

At present, PDX models likely provide the most clinically relevant models of human cancer. However, like other models, PDXs have their inherent limitations and deficiencies. First, the most commonly documented limitation of all the xenograft models is the requirement to use immunodeficient mice which lack the human immune system. To circumvent this limitation, more sophisticated PDX models (humanized models) should be developed via cografting of tumour tissue along with bone marrow stem cells of the same patient. This combination may reconstitute components of the human immune system in mice and allow investigation of the role of the immune system in cancer metastasis and efficacy of immune-based therapies. Secondly, although PDX models are relatively stable for several generations, increasing histopathological and molecular differences between patient tumours and PDXs are foreseeable with continual passaging in the animals. It is therefore prudent to establish cryopreservation of PDXs at an early generation, ensuring preservation of the cellular and molecular characteristics of the original tumour and a relatively robust supply of a particular patient's tumour. The maintenance of a successful PDX program requires ample financial resources and specialized expertise to reliably and reproducibly perform the PDX implants, which is a major factor hampering the widespread usage of PDX models compared to traditional cell line-based xenograft systems.

In summary, PDX models closely mimic the original cancers in terms of histopathology, tumour heterogeneity, chromosomal aberrations, gene expression profiles, tumour aggressiveness, and response to therapy. PDX models continue to be refined in order to increase take rate and to further enhance the already high clinical relevance of the system.

Concluding Remarks

While there are a variety of animal models that can be applied to the study of solid tumour metastasis, the choice of model system depends largely on the research question. Investigators seeking high-throughput analyses of genes or compounds that affect tumour cell migration may find that the zebrafish model is suitable for their purposes. Experiments designed to test the oncogenic potential of a particular gene may be suitable for combination with established genetically engineered mouse models, while dissecting the contribution of a particular gene to individual steps of the metastatic process is well-suited to the application of murine xenograft models. In order to maximize the clinical applicability of experimental findings, researchers are increasingly turning to patient-derived xenografts in order to validate experimental findings established in other model systems and in an effort to personalize cancer therapy.

References

- Fidler IJ (2003) The pathogenesis of cancer metastasis: the 'seed and soil' hypothesis revisited. Nat Rev Cancer 3(6):453–458
- Nguyen DX, Massague J (2007) Genetic determinants of cancer metastasis. Nat Rev Genet 8(5):341–352
- 3. Gupta GP, Massague J (2006) Cancer metastasis: building a framework. Cell 127(4):679-695
- Steeg PS (2006) Tumor metastasis: mechanistic insights and clinical challenges. Nat Med 12(8):895–904
- Townson JL, Chambers AF (2006) Dormancy of solitary metastatic cells. Cell Cycle 5(16):1744–1750
- Gao D, Nolan DJ, Mellick AS, Bambino K, McDonnell K, Mittal V (2008) Endothelial progenitor cells control the angiogenic switch in mouse lung metastasis. Science 319(5860):195– 198
- Holmgren L, O'Reilly MS, Folkman J (1995) Dormancy of micrometastases: balanced proliferation and apoptosis in the presence of angiogenesis suppression. Nat Med 1(2):149–153
- Naumov GN, Akslen LA, Folkman J (2006) Role of angiogenesis in human tumor dormancy: animal models of the angiogenic switch. Cell Cycle 5(16):1779–1787

- 9. Paget S (1989) The distribution of secondary growths in cancer of the breast. Cancer Metastasis Rev 8(2):98–101
- Muller A, Homey B, Soto H, Ge N, Catron D, Buchanan ME, McClanahan T, Murphy E, Yuan W, Wagner SN et al (2001) Involvement of chemokine receptors in breast cancer metastasis. Nature 410(6824):50–56
- 11. Bennewith KL, Erler JT, Giaccia AJ (2011) Pre-metastatic niches. In: Siemann DW (ed) Tumor microenvironment, 1st edn. Wiley, West Sussex, pp 161–182
- 12. Erler JT, Bennewith KL, Cox TR, Lang G, Bird D, Koong A, Le QT, Giaccia AJ (2009) Hypoxia-induced lysyl oxidase is a critical mediator of bone marrow cell recruitment to form the premetastatic niche. Cancer Cell 15(1):35–44
- Kaplan RN, Riba RD, Zacharoulis S, Bramley AH, Vincent L, Costa C, MacDonald DD, Jin DK, Shido K, Kerns SA et al (2005) VEGFR1-positive haematopoietic bone marrow progenitors initiate the pre-metastatic niche. Nature 438(7069):820–827
- 14. Dodd A, Curtis PM, Williams LC, Love DR (2000) Zebrafish: bridging the gap between development and disease. Hum Mol Genet 9(16):2443–2449
- Detrich HW, 3rd, Westerfield M, Zon LI (1999) Overview of the Zebrafish system. Methods Cell Bio 59:3–10
- Espinola-Zavaleta N, Munoz-Castellanos L, Herrera MG, Keirns C (2006) Understanding left ventricular outflow obstruction: anatomoechocardiographic correlation. Congenit Heart Dis 1(4):161–168
- Haldi M, Ton C, Seng WL, McGrath P (2006) Human melanoma cells transplanted into zebrafish proliferate, migrate, produce melanin, form masses and stimulate angiogenesis in zebrafish. Angiogenesis 9(3):139–151
- Geiger GA, Fu W, Kao GD (2008) Temozolomide-mediated radiosensitization of human glioma cells in a zebrafish embryonic system. Cancer Res 68(9):3396–3404
- Marques IJ, Weiss FU, Vlecken DH, Nitsche C, Bakkers J, Lagendijk AK, Partecke LI, Heidecke CD, Lerch MM, Bagowski CP (2009) Metastatic behaviour of primary human tumours in a zebrafish xenotransplantation model. BMC Cancer 9:128
- Traver D, Paw BH, Poss KD, Penberthy WT, Lin S, Zon LI (2003) Transplantation and in vivo imaging of multilineage engraftment in zebrafish bloodless mutants. Nat Immunol 4(12):1238–1246
- Pruvot B, Jacquel A, Droin N, Auberger P, Bouscary D, Tamburini J, Muller M, Fontenay M, Chluba J, Solary E (2011) Leukemic cell xenograft in zebrafish embryo for investigating drug efficacy. Haematologica 96(4):612–616
- 22. Teng Y, Xie X, Walker S, White DT, Mumm JS, Cowell JK (2013) Evaluating human cancer cell metastasis in zebrafish. BMC Cancer 13:453
- Bansal N, Davis S, Tereshchenko I, Budak-Alpdogan T, Zhong H, Stein MN, Kim IY, Dipaola RS, Bertino JR, Sabaawy HE (2014) Enrichment of human prostate cancer cells with tumor initiating properties in mouse and zebrafish xenografts by differential adhesion. Prostate 74(2):187–200
- 24. Detrich HW, 3rd, Westerfield M, Zon LI (2010) The zebrafish: cellular and developmental biology, part A. Preface. Methods Cell Biol 100:xiii
- Lawson ND, Weinstein BM (2002) In vivo imaging of embryonic vascular development using transgenic zebrafish. Dev Bio 248(2):307–318
- Stoletov K, Kato H, Zardouzian E, Kelber J, Yang J, Shattil S, Klemke R (2010) Visualizing extravasation dynamics of metastatic tumor cells. J Cell Sci 123(Pt 13):2332–2341
- Ghotra VP, He S, de Bont H, van der Ent W, Spaink HP, van de Water B, Snaar-Jagalska BE, Danen EH (2012) Automated whole animal bio-imaging assay for human cancer dissemination. PloS One 7(2):e31281
- Renshaw SA, Loynes CA, Trushell DM, Elworthy S, Ingham PW, Whyte MK (2006) A transgenic zebrafish model of neutrophilic inflammation. Blood 108(13):3976–3978

- 5 Animal Models of Metastasis
- 29. He S, Lamers GE, Beenakker JW, Cui C, Ghotra VP, Danen EH, Meijer AH, Spaink HP, Snaar-Jagalska BE (2012) Neutrophil-mediated experimental metastasis is enhanced by VEGFR inhibition in a zebrafish xenograft model. J Pathol 227(4):431–445
- Ellett F, Pase L, Hayman JW, Andrianopoulos A, Lieschke GJ (2011) mpeg1 promoter transgenes direct macrophage-lineage expression in zebrafish. Blood 117(4):e49–56
- Lin HF, Traver D, Zhu H, Dooley K, Paw BH, Zon LI, Handin RI (2005) Analysis of thrombocyte development in CD41-GFP transgenic zebrafish. Blood 106(12):3803–3810
- Lee LM, Seftor EA, Bonde G, Cornell RA, Hendrix MJ (2005) The fate of human malignant melanoma cells transplanted into zebrafish embryos: assessment of migration and cell division in the absence of tumor formation. Dev Dyn 233(4):1560–1570
- Drabsch Y, He S, Zhang L, Snaar-Jagalska BE, Ten Dijke P (2013) Transforming growth factor-beta signalling controls human breast cancer metastasis in a zebrafish xenograft model. Breast Cancer Res 15(6):R106
- 34. Lee SL, Rouhi P, Dahl Jensen L, Zhang D, Ji H, Hauptmann G, Ingham P, Cao Y (2009) Hypoxia-induced pathological angiogenesis mediates tumor cell dissemination, invasion, and metastasis in a zebrafish tumor model. Proc Natl Acad Sci U S A 106(46):19485–19490
- Rouhi P, Jensen LD, Cao Z, Hosaka K, Lanne T, Wahlberg E, Steffensen JF, Cao Y (2010) Hypoxia-induced metastasis model in embryonic zebrafish. Nat Protoc 5(12):1911–1918
- Rouhi P, Lee SL, Cao Z, Hedlund EM, Jensen LD, Cao Y (2010) Pathological angiogenesis facilitates tumor cell dissemination and metastasis. Cell Cycle 9(5):913–917
- Kim SJ, Shin JY, Lee KD, Bae YK, Choi IJ, Park SH, Chun KH (2011) Galectin-3 facilitates cell motility in gastric cancer by up-regulating protease-activated receptor-1 (PAR-1) and matrix metalloproteinase-1 (MMP-1). PloS One 6(9):e25103
- 38. Belo AI, van der Sar AM, Tefsen B, van Die I (2013) Galectin-4 reduces migration and metastasis formation of pancreatic cancer cells. PloS One 8(6):e65957
- Cock-Rada AM, Medjkane S, Janski N, Yousfi N, Perichon M, Chaussepied M, Chluba J, Langsley G, Weitzman JB (2012) SMYD3 promotes cancer invasion by epigenetic upregulation of the metalloproteinase MMP-9. Cancer Res 72(3):810–820
- Yang XJ, Cui W, Gu A, Xu C, Yu SC, Li TT, Cui YH, Zhang X, Bian XW (2013) A novel zebrafish xenotransplantation model for study of glioma stem cell invasion. PloS One 8(4):e61801
- Lam SH, Chua HL, Gong Z, Lam TJ, Sin YM (2004) Development and maturation of the immune system in zebrafish, Danio rerio: a gene expression profiling, in situ hybridization and immunological study. Dev Comp Immunol 28(1):9–28
- Annila T, Lihavainen E, Marques IJ, Williams DR, Yli-Harja O, Ribeiro A (2013) ZebIAT, an image analysis tool for registering zebrafish embryos and quantifying cancer metastasis. BMC Bioinform 14(Suppl 10):S5
- Smith AC, Raimondi AR, Salthouse CD, Ignatius MS, Blackburn JS, Mizgirev IV, Storer NY, de Jong JL, Chen AT, Zhou Y et al (2010) High-throughput cell transplantation establishes that tumor-initiating cells are abundant in zebrafish T-cell acute lymphoblastic leukemia. Blood 115(16):3296–3303
- Corkery DP, Dellaire G, Berman JN (2011) Leukaemia xenotransplantation in zebrafish-chemotherapy response assay in vivo. Br J Haematol 153(6):786–789
- Eguiara A, Holgado O, Beloqui I, Abalde L, Sanchez Y, Callol C, Martin AG (2011) Xenografts in zebrafish embryos as a rapid functional assay for breast cancer stem-like cell identification. Cell Cycle 10(21):3751–3757
- 46. Howe K, Clark MD, Torroja CF, Torrance J, Berthelot C, Muffato M, Collins JE, Humphray S, McLaren K, Matthews L et al (2013) The zebrafish reference genome sequence and its relationship to the human genome. Nature 496(7446):498–503
- Guy CT, Webster MA, Schaller M, Parsons TJ, Cardiff RD, Muller WJ (1992) Expression of the neu protooncogene in the mammary epithelium of transgenic mice induces metastatic disease. Proc Natl Acad Sci U S A 89(22):10578–10582

- Hutchinson JN, Jin J, Cardiff RD, Woodgett JR, Muller WJ (2004) Activation of Akt-1 (PKBalpha) can accelerate ErbB-2-mediated mammary tumorigenesis but suppresses tumor invasion. Cancer Res 64(9):3171–3178
- Jeffers M, Fiscella M, Webb CP, Anver M, Koochekpour S, Vande Woude GF (1998) The mutationally activated Met receptor mediates motility and metastasis. Proc Natl Acad Sci U S A 95(24):14417–14422
- Li Y, Hively WP, Varmus HE (2000) Use of MMTV-Wnt-1 transgenic mice for studying the genetic basis of breast cancer. Oncogene 19(8):1002–1009
- Lin EY, Jones JG, Li P, Zhu L, Whitney KD, Muller WJ, Pollard JW (2003) Progression to malignancy in the polyoma middle T oncoprotein mouse breast cancer model provides a reliable model for human diseases. Am J Pathol 163(5):2113–2126
- Maglione JE, Moghanaki D, Young LJ, Manner CK, Ellies LG, Joseph SO, Nicholson B, Cardiff RD, MacLeod CL (2001) Transgenic Polyoma middle-T mice model premalignant mammary disease. Cancer Res 61(22):8298–8305
- Maroulakou IG, Oemler W, Naber SP, Tsichlis PN (2007) Akt1 ablation inhibits, whereas Akt2 ablation accelerates, the development of mammary adenocarcinomas in mouse mammary tumor virus (MMTV)-ErbB2/neu and MMTV-polyoma middle T transgenic mice. Cancer Res 67(1):167–177
- Lin EY, Nguyen AV, Russell RG, Pollard JW (2001) Colony-stimulating factor 1 promotes progression of mammary tumors to malignancy. J Exp Med 193(6):727–740
- 55. Hernandez L, Smirnova T, Kedrin D, Wyckoff J, Zhu L, Stanley ER, Cox D, Muller WJ, Pollard JW, Van Rooijen N et al (2009) The EGF/CSF-1 paracrine invasion loop can be triggered by heregulin beta1 and CXCL12. Cancer Res 69(7):3221–3227
- Moody SE, Sarkisian CJ, Hahn KT, Gunther EJ, Pickup S, Dugan KD, Innocent N, Cardiff RD, Schnall MD, Chodosh LA (2002) Conditional activation of Neu in the mammary epithelium of transgenic mice results in reversible pulmonary metastasis. Cancer Cell 2(6):451–461
- Gunther EJ, Moody SE, Belka GK, Hahn KT, Innocent N, Dugan KD, Cardiff RD, Chodosh LA (2003) Impact of p53 loss on reversal and recurrence of conditional Wnt-induced tumorigenesis. Genes Dev 17(4):488–501
- Tilli MT, Furth PA (2003) Conditional mouse models demonstrate oncogene-dependent differences in tumor maintenance and recurrence. BCR 5(4):202–205
- 59. Lin SC, Lee KF, Nikitin AY, Hilsenbeck SG, Cardiff RD, Li A, Kang KW, Frank SA, Lee WH, Lee EY (2004) Somatic mutation of p53 leads to estrogen receptor alpha-positive and -negative mouse mammary tumors with high frequency of metastasis. Cancer Res 64(10):3525–3532
- Xu X, Wagner KU, Larson D, Weaver Z, Li C, Ried T, Hennighausen L, Wynshaw-Boris A, Deng CX (1999) Conditional mutation of Brca1 in mammary epithelial cells results in blunted ductal morphogenesis and tumour formation. Nat Genet 22(1):37–43
- Brodie SG, Xu X, Qiao W, Li WM, Cao L, Deng CX (2001) Multiple genetic changes are associated with mammary tumorigenesis in Brca1 conditional knockout mice. Oncogene 20(51):7514–7523
- 62. Derksen PW, Braumuller TM, van der Burg E, Hornsveld M, Mesman E, Wesseling J, Krimpenfort P, Jonkers J (2011) Mammary-specific inactivation of E-cadherin and p53 impairs functional gland development and leads to pleomorphic invasive lobular carcinoma in mice. Dis Models Mech 4(3):347–358
- Dourdin N, Schade B, Lesurf R, Hallett M, Munn RJ, Cardiff RD, Muller WJ (2008) Phosphatase and tensin homologue deleted on chromosome 10 deficiency accelerates tumor induction in a mouse model of ErbB-2 mammary tumorigenesis. Cancer Res 68(7):2122–2131
- Schade B, Rao T, Dourdin N, Lesurf R, Hallett M, Cardiff RD, Muller WJ (2009) PTEN deficiency in a luminal ErbB-2 mouse model results in dramatic acceleration of mammary tumorigenesis and metastasis. J Biol Chem 284(28):19018–19026
- 65. Valkenburg KC, Williams BO (2011) Mouse models of prostate cancer. Prostate cancer 2011:895238

- 5 Animal Models of Metastasis
- Greenberg NM, DeMayo F, Finegold MJ, Medina D, Tilley WD, Aspinall JO, Cunha GR, Donjacour AA, Matusik RJ, Rosen JM (1995) Prostate cancer in a transgenic mouse. Proc Natl Acad Sci U S A 92(8):3439–3443
- Gingrich JR, Barrios RJ, Morton RA, Boyce BF, DeMayo FJ, Finegold MJ, Angelopoulou R, Rosen JM, Greenberg NM (1996) Metastatic prostate cancer in a transgenic mouse. Cancer Res 56(18):4096–4102
- Foster BA, Gingrich JR, Kwon ED, Madias C, Greenberg NM (1997) Characterization of prostatic epithelial cell lines derived from transgenic adenocarcinoma of the mouse prostate (TRAMP) model. Cancer Res 57(16):3325–3330
- Klezovitch O, Chevillet J, Mirosevich J, Roberts RL, Matusik RJ, Vasioukhin V (2004) Hepsin promotes prostate cancer progression and metastasis. Cancer Cell 6(2):185–195
- Tu WH, Thomas TZ, Masumori N, Bhowmick NA, Gorska AE, Shyr Y, Kasper S, Case T, Roberts RL, Shappell SB et al (2003) The loss of TGF-beta signaling promotes prostate cancer metastasis. Neoplasia 5(3):267–277
- Yang F, Zhang Y, Ressler SJ, Ittmann MM, Ayala GE, Dang TD, Wang F, Rowley DR (2013) FGFR1 is essential for prostate cancer progression and metastasis. Cancer Res 73(12):3716– 3724
- Duan W, Gabril MY, Moussa M, Chan FL, Sakai H, Fong G, Xuan JW (2005) Knockin of SV40 Tag oncogene in a mouse adenocarcinoma of the prostate model demonstrates advantageous features over the transgenic model. Oncogene 24(9):1510–1524
- Gray IC, Stewart LM, Phillips SM, Hamilton JA, Gray NE, Watson GJ, Spurr NK, Snary D (1998) Mutation and expression analysis of the putative prostate tumour-suppressor gene PTEN. Bri J Cancer 78(10):1296–1300
- 74. Wang S, Gao J, Lei Q, Rozengurt N, Pritchard C, Jiao J, Thomas GV, Li G, Roy-Burman P, Nelson PS et al (2003) Prostate-specific deletion of the murine Pten tumor suppressor gene leads to metastatic prostate cancer. Cancer Cell 4(3):209–221
- Zeng L, Rowland RG, Lele SM, Kyprianou N (2004) Apoptosis incidence and protein expression of p53, TGF-beta receptor II, p27Kip1, and Smad4 in benign, premalignant, and malignant human prostate. Hum Pathol 35(3):290–297
- Ding Z, Wu CJ, Chu GC, Xiao Y, Ho D, Zhang J, Perry SR, Labrot ES, Wu X, Lis R et al (2011) SMAD4-dependent barrier constrains prostate cancer growth and metastatic progression. Nature 470(7333):269–273
- Wilmanns C, Fan D, O'Brian CA, Bucana CD, Fidler IJ (1992) Orthotopic and ectopic organ environments differentially influence the sensitivity of murine colon carcinoma cells to doxorubicin and 5-fluorouracil. Int J Cancer 52(1):98–104
- Blouw B, Song H, Tihan T, Bosze J, Ferrara N, Gerber HP, Johnson RS, Bergers G (2003) The hypoxic response of tumors is dependent on their microenvironment. Cancer Cell 4(2):133–146
- Graves EE, Vilalta M, Cecic IK, Erler JT, Tran PT, Felsher D, Sayles L, Sweet-Cordero A, Le QT, Giaccia AJ (2010) Hypoxia in models of lung cancer: implications for targeted therapeutics. Clin Cancer Res 16(19):4843–4852
- 80. Fidler IJ (2006) Models for spontaneous metastasis. Cancer Res 66(19):9787
- 81. Cruz-Munoz W, Man S, Xu P, Kerbel RS (2008) Development of a preclinical model of spontaneous human melanoma central nervous system metastasis. Cancer Res 68(12):4500–4505
- Munoz R, Man S, Shaked Y, Lee CR, Wong J, Francia G, Kerbel RS (2006) Highly efficacious nontoxic preclinical treatment for advanced metastatic breast cancer using combination oral UFT-cyclophosphamide metronomic chemotherapy. Cancer Res 66(7):3386–3391
- Francia G, Cruz-Munoz W, Man S, Xu P, Kerbel RS (2011) Mouse models of advanced spontaneous metastasis for experimental therapeutics. Nat Rev Cancer 11(2):135–141
- Minn AJ, Gupta GP, Siegel PM, Bos PD, Shu W, Giri DD, Viale A, Olshen AB, Gerald WL, Massague J (2005) Genes that mediate breast cancer metastasis to lung. Nature 436(7050):518–524

- Bos PD, Zhang XH, Nadal C, Shu W, Gomis RR, Nguyen DX, Minn AJ, van de Vijver MJ, Gerald WL, Foekens JA et al (2009) Genes that mediate breast cancer metastasis to the brain. Nature 459(7249):1005–1009
- Moses WW (2011) Fundamental Limits of Spatial Resolution in PET. Nucl Instrum Methods Phys Res A 648(Suppl 1):1S236–S240
- Bosiljcic M, Hamilton MJ, Banath JP, Lepard NE, McDougal DC, Jia JX, Krystal G, Bennewith KL (2011) Myeloid suppressor cells regulate the lung environment-letter. Cancer Res 71(14):5050–5051; author reply 5052–5053
- Gambotto A, Dworacki G, Cicinnati V, Kenniston T, Steitz J, Tuting T, Robbins PD, De-Leo AB (2000) Immunogenicity of enhanced green fluorescent protein (EGFP) in BALB/c mice: identification of an H2-Kd-restricted CTL epitope. Gene Ther 7(23):2036–2040
- Stripecke R, Carmen Villacres M, Skelton D, Satake N, Halene S, Kohn D (1999) Immune response to green fluorescent protein: implications for gene therapy. Gene Ther 6(7):1305– 1312
- 90. Steinbauer M, Guba M, Cernaianu G, Kohl G, Cetto M, Kunz-Schughart LA, Geissler EK, Falk W, Jauch KW (2003) GFP-transfected tumor cells are useful in examining early metastasis in vivo, but immune reaction precludes long-term tumor development studies in immunocompetent mice. Clin Exp Metastasis 20(2):135–141
- Williams SA, Anderson WC, Santaguida MT, Dylla SJ (2013) Patient-derived xenografts, the cancer stem cell paradigm, and cancer pathobiology in the 21st century. Lab Invest 93(9):970–982
- Von Hoff DD (1998) There are no bad anticancer agents, only bad clinical trial designstwenty-first Richard and Hinda Rosenthal Foundation Award Lecture. Clin Cancer Res 4(5):1079–1086
- Gutman S, Kessler LG (2006) The US Food and Drug Administration perspective on cancer biomarker development. Nat Rev Cancer 6(7):565–571
- Tentler JJ, Tan AC, Weekes CD, Jimeno A, Leong S, Pitts TM, Arcaroli JJ, Messersmith WA, Eckhardt SG (2012) Patient-derived tumour xenografts as models for oncology drug development. Nat Rev Clin Oncol 9(6):338–350
- 95. van Weerden WM, de Ridder CM, Verdaasdonk CL, Romijn JC, van der Kwast TH, Schroder FH, van Steenbrugge GJ (1996) Development of seven new human prostate tumor xenograft models and their histopathological characterization. Am J Pathol 149(3):1055– 1062
- Lin D, Xue H, Wang Y, Wu R, Watahiki A, Dong X, Cheng H, Wyatt AW, Collins CC, Gout PW (2014) Next generation patient-derived prostate cancer xenograft models. Asian J Androl 16(3):407–412
- Bogden AE, Haskell PM, LePage DJ, Kelton DE, Cobb WR, Esber HJ (1979) Growth of human tumor xenografts implanted under the renal capsule of normal immunocompetent mice. Exp Cell Biol 47(4):281–293
- Ott CE, Knox FG (1976) Tissue pressures and fluid dynamics in the kidney. Fed Proc 35(8):1872–1875
- 99. Pinter G (1988) Renal lymph: vital for the kidney and valuable for the physiologist. Physiology 3(5):189–193
- Tunstead JR, Thomas M, Hornsby PJ (1999) Early events in the formation of a tissue structure from dispersed bovine adrenocortical cells following transplantation into scid mice. J Mol Med (Berl) 77(9):666–676
- Cunha GR (1976) Epithelial-stromal interactions in development of the urogenital tract. Int Rev Cytol 47:137–194
- Cunha GR, Lung B, Kato K (1977) Role of the epithelial-stromal interaction during the development and expression of ovary-independent vaginal hyperplasia. Dev Bio 56(1):52–67
- 103. Bogden AE, Griffin W, Reich SD, Costanza ME, Cobb WR (1984) Predictive testing with the subrenal capsule assay. Cancer Treat Rev 11(Suppl A):113–124
- 104. Griffin TW, Bogden AE, Reich SD, Antonelli D, Hunter RE, Ward A, Yu DT, Greene HL, Costanza ME (1983) Initial clinical trials of the subrenal capsule assay as a predictor of tumor response to chemotherapy. Cancer 52(12):2185–2192

5 Animal Models of Metastasis

- 105. Cheon DJ, Orsulic S (2011) Mouse models of cancer. Annu Rev Pathol 6:95-119
- 106. Guenot D, Guerin E, Aguillon-Romain S, Pencreach E, Schneider A, Neuville A, Chenard MP, Duluc I, Du Manoir S, Brigand C et al (2006) Primary tumour genetic alterations and intra-tumoral heterogeneity are maintained in xenografts of human colon cancers showing chromosome instability. J Pathol 208(5):643–652
- Fichtner I, Slisow W, Gill J, Becker M, Elbe B, Hillebrand T, Bibby M (2004) Anticancer drug response and expression of molecular markers in early-passage xenotransplanted colon carcinomas. Eur J Cancer 40(2):298–307
- Julien S, Merino-Trigo A, Lacroix L, Pocard M, Goere D, Mariani P, Landron S, Bigot L, Nemati F, Dartigues P et al (2012) Characterization of a large panel of patient-derived tumor xenografts representing the clinical heterogeneity of human colorectal cancer. Clin Cancer Res 18(19):5314–5328
- Puig I, Chicote I, Tenbaum SP, Arques O, Herance JR, Gispert JD, Jimenez J, Landolfi S, Caci K, Allende H et al (2013) A personalized preclinical model to evaluate the metastatic potential of patient-derived colon cancer initiating cells. Clin Cancer Res 19(24):6787– 6801
- Kim MP, Evans DB, Wang H, Abbruzzese JL, Fleming JB, Gallick GE (2009) Generation of orthotopic and heterotopic human pancreatic cancer xenografts in immunodeficient mice. Nat Protoc 4(11):1670–1680
- 111. Fu X, Guadagni F, Hoffman RM (1992) A metastatic nude-mouse model of human pancreatic cancer constructed orthotopically with histologically intact patient specimens. Proc Natl Acad Sci U S A 89(12):5645–5649
- 112. Chang Q, Jurisica I, Do T, Hedley DW (2011) Hypoxia predicts aggressive growth and spontaneous metastasis formation from orthotopically grown primary xenografts of human pancreatic cancer. Cancer Res 71(8):3110–3120
- 113. van Weerden WM, Romijn JC (2000) Use of nude mouse xenograft models in prostate cancer research. Prostate 43(4):263–271
- Fiebig HH, Burger AM (2001) Human tumor xenografts and explants. In: Teicher BA (ed) Tumor models in cancer research. Humana Press, Totowa pp 113–137
- 115. DeRose YS, Wang G, Lin YC, Bernard PS, Buys SS, Ebbert MT, Factor R, Matsen C, Milash BA, Nelson E et al (2011) Tumor grafts derived from women with breast cancer authentically reflect tumor pathology, growth, metastasis and disease outcomes. Nat Med 17(11):1514–1520
- 116. Zhang X, Claerhout S, Prat A, Dobrolecki LE, Petrovic I, Lai Q, Landis MD, Wiechmann L, Schiff R, Giuliano M et al (2013) A renewable tissue resource of phenotypically stable, biologically and ethnically diverse, patient-derived human breast cancer xenograft models. Cancer Res 73(15):4885–4897
- 117. Corey E, Quinn JE, Buhler KR, Nelson PS, Macoska JA, True LD, Vessella RL (2003) LuCaP 35: a new model of prostate cancer progression to androgen independence. Prostate 55(4):239–246
- 118. Ellis WJ, Vessella RL, Buhler KR, Bladou F, True LD, Bigler SA, Curtis D, Lange PH (1996) Characterization of a novel androgen-sensitive, prostate-specific antigen-producing prostatic carcinoma xenograft: LuCaP 23. Clin Cancer Res 2(6):1039–1048
- Dong Lin AW, Hui Xue, Yuwei Wang, Xin Dong, et al (2014) High fidelity patient-derived xenografts for accelerating prostate cancer discovery and drug development. Cancer Res 74(4):1272–1283
- 120. Wang Y, Xue H, Cutz JC, Bayani J, Mawji NR, Chen WG, Goetz LJ, Hayward SW, Sadar MD, Gilks CB et al (2005) An orthotopic metastatic prostate cancer model in SCID mice via grafting of a transplantable human prostate tumor line. Lab Invest 85(11):1392–1404
- 121. Lin D, Watahiki A, Bayani J, Zhang F, Liu L, Ling V, Sadar MD, English J, Fazli L, So A et al (2008) ASAP1, a gene at 8q24, is associated with prostate cancer metastasis. Cancer Res 68(11):4352–4359
- 122. Lin D, Bayani J, Wang Y, Sadar MD, Yoshimoto M, Gout PW, Squire JA (2010) Development of metastatic and non-metastatic tumor lines from a patient's prostate cancer

specimen-identification of a small subpopulation with metastatic potential in the primary tumor. Prostate 70(15):1636–1644

- 123. Loukopoulos P, Kanetaka K, Takamura M, Shibata T, Sakamoto M, Hirohashi S (2004) Orthotopic transplantation models of pancreatic adenocarcinoma derived from cell lines and primary tumors and displaying varying metastatic activity. Pancreas 29(3):193–203
- 124. Cutz JC, Guan J, Bayani J, Yoshimoto M, Xue H, Sutcliffe M, English J, Flint J, LeRiche J, Yee J et al (2006) Establishment in severe combined immunodeficiency mice of subrenal capsule xenografts and transplantable tumor lines from a variety of primary human lung cancers: potential models for studying tumor progression-related changes. Clin Cancer Res 12(13):4043–4054
- Lee CH, Xue H, Sutcliffe M, Gout PW, Huntsman DG, Miller DM, Gilks CB, Wang YZ (2005) Establishment of subrenal capsule xenografts of primary human ovarian tumors in SCID mice: potential models. Gynecol Oncol 96(1):48–55
- 126. Lin D, Wyatt AW, Xue H, Wang Y, Dong X, Haegert A, Wu R, Brahmbhatt S, Mo F, Jong L et al (2014) High fidelity patient-derived xenografts for accelerating prostate cancer discovery and drug development. Cancer Res 74(4):1272–1283
- 127. Garber K (2009) From human to mouse and back: 'tumorgraft' models surge in popularity. J Natl Cancer Inst 101(1):6–8
- 128. Daniel VC, Marchionni L, Hierman JS, Rhodes JT, Devereux WL, Rudin CM, Yung R, Parmigiani G, Dorsch M, Peacock CD et al (2009) A primary xenograft model of small-cell lung cancer reveals irreversible changes in gene expression imposed by culture in vitro. Cancer Res 69(8):3364–3373
- 129. Reyal F, Guyader C, Decraene C, Lucchesi C, Auger N, Assayag F, De Plater L, Gentien D, Poupon MF, Cottu P et al (2012) Molecular profiling of patient-derived breast cancer xenografts. BCR 14(1):R11
- McEvoy J, Ulyanov A, Brennan R, Wu G, Pounds S, Zhang J, Dyer MA (2012) Analysis of MDM2 and MDM4 single nucleotide polymorphisms, mRNA splicing and protein expression in retinoblastoma. PloS One 7(8):e42739
- 131. Bertotti A, Migliardi G, Galimi F, Sassi F, Torti D, Isella C, Cora D, Di Nicolantonio F, Buscarino M, Petti C et al (2011) A molecularly annotated platform of patient-derived xenografts ("xenopatients") identifies HER2 as an effective therapeutic target in cetuximabresistant colorectal cancer. Cancer Discov 1(6):508–523
- 132. Fichtner I, Rolff J, Soong R, Hoffmann J, Hammer S, Sommer A, Becker M, Merk J (2008) Establishment of patient-derived non-small cell lung cancer xenografts as models for the identification of predictive biomarkers. Clin Cancer Res 14(20):6456–6468
- Ciardiello F, Normanno N (2011) HER2 signaling and resistance to the anti-EGFR monoclonal antibody cetuximab: a further step toward personalized medicine for patients with colorectal cancer. Cancer Discov 1(6):472–474
- Hidalgo M, Bruckheimer E, Rajeshkumar NV, Garrido-Laguna I, De Oliveira E, Rubio-Viqueira B, Strawn S, Wick MJ, Martell J, Sidransky D (2011) A pilot clinical study of treatment guided by personalized tumorgrafts in patients with advanced cancer. Mol Cancer Ther 10(8):1311–1316
- 135. Garralda E, Paz K, Lopez-Casas PP, Jones S, Katz A, Kann LM, Lopez-Rios F, Sarno F, Al-Shahrour F, Vasquez D et al (2014) Integrated next-generation sequencing and avatar mouse models for personalized cancer treatment. Clin Cancer Res 20(9):2476–2484
- Chiang YT, Wang K, Fazli L, Qi RZ, Gleave ME, Collins CC, Gout PW, Wang Y (2014) GATA2 as a potential metastasis-driving gene in prostate cancer. Oncotarget 5(2):451–461.
- Watahiki A, Wang Y, Morris J, Dennis K, O'Dwyer HM, Gleave M, Gout PW (2011) MicroRNAs associated with metastatic prostate cancer. PloS One 6(9):e24950
- 138. Malaney P, Nicosia SV, Dave V (2014) One mouse, one patient paradigm: new avatars of personalized cancer therapy. Cancer Lett 344(1):1–12
- 139. Krumbach R, Schuler J, Hofmann M, Giesemann T, Fiebig HH, Beckers T (2011) Primary resistance to cetuximab in a panel of patient-derived tumour xenograft models: activation of MET as one mechanism for drug resistance. Eur J Cancer 47(8):1231–1243

5 Animal Models of Metastasis

- 140. Tentler JJ, Nallapareddy S, Tan AC, Spreafico A, Pitts TM, Morelli MP, Selby HM, Kachaeva MI, Flanigan SA, Kulikowski GN et al (2010) Identification of predictive markers of response to the MEK1/2 inhibitor selumetinib (AZD6244) in K-ras-mutated colorectal cancer. Mol Cancer Ther 9(12):3351–3362
- Gout PW, Wang Y (2012) Drug sensitivity testing for personalized lung cancer therapy. J Thorac Dis 4(1):17–18
- 142. Morelli MP, Calvo E, Ordonez E, Wick MJ, Viqueira BR, Lopez-Casas PP, Bruckheimer E, Calles-Blanco A, Sidransky D, Hidalgo M (2012) Prioritizing phase I treatment options through preclinical testing on personalized tumorgraft. J Clin Oncol 30(4):e45–48
- 143. Dong X, Guan J, English JC, Flint J, Yee J, Evans K, Murray N, Macaulay C, Ng RT, Gout PW et al (2010) Patient-derived first generation xenografts of non-small cell lung cancers: promising tools for predicting drug responses for personalized chemotherapy. Clin Cancer Res 16(5):1442–1451

Chapter 6 Microenvironmental Control of Metastatic Progression

Calvin D. Roskelley

Abstract As solid tumors progress, the surrounding microenvironment is altered dramatically. This microenvironment contains stromal and immune cells, some resident and some newly recruited, that are often activated due to factors released by the tumor cells themselves. These activated cells then release soluble factors that feed forward on the tumour cells in a symbiotic manner. Activated cells also alter the deposition and processing of extracellular matrix molecules in the microenvironment which further affects both the genotype and the phenotype of tumor cells. More specifically, these microenvironmental alterations can have profound effects on the genome and epigenome of tumor cells as well as their signal transduction pathways, both biochemical and mechanical. All of these effects contribute to the invasion and progression of the metastatic tumor organ.

Keywords Microenvironment · Tumor progression · Stroma · Extracellular matrix

Abbreviations

CAFs	Cancer-associated fibroblasts
CIMP	CpG island hypermethylation phenotype
DNA	Deoxyribonucleic acid
ECM	Extracellular matrix
EMT	Epithelial to mesenchymal transition
HIF1α	Hypoxia inducible factor 1alpha
MET	Mesenchymal to epithelial transition
MSI	Microsatellite instability
TAMs	Tumor-associated macrophages
TGF-β	Transforming growth factor beta
VEGF	Vascular endothelial growth factor

C. D. Roskelley (🖂)

Department of Cellular and Physiological Sciences, University of British Columbia, 2350 Health Sciences Mall, Vancouver, BC V6T 1Z3, Canada e-mail: roskelly@mail.ubc.ca

[©] Springer International Publishing Switzerland 2015

C. Maxwell, C. Roskelley (eds.), *Genomic Instability and Cancer Metastasis*, Cancer Metastasis - Biology and Treatment 20, DOI 10.1007/978-3-319-12136-9_6

Introduction

Given that the overwhelming majority of cancer victims succumb to the formation and expansion of metastatic lesions, an overarching goal of modern cancer research is to determine how changes in defined cohorts of oncogenes and tumor suppressors facilitate the emergence of the malignant phenotype within individual tumor cells and their progeny. As a result, identifying actionable targets that are the products of such dysregulated genes has become the cornerstone of rational cancer therapeutics. However, even when they are directed against bona fide targets, such therapies are often only partially effective and they are invariably susceptible to acquired resistance as the tumor evolves towards full blown malignancy. The recognition that many non-tumor cell autonomous events contribute to this progressive evolution, together with data generated by genome-wide profiling both at the transcriptomic and epigenomic levels, are starting to give us a systems-based picture of why this is the case. What is becoming increasingly clear from such studies is that the tumor microenvironment is a major multifactorial contributor to this progressive evolution [23].

In early *in situ* lesions that have been initiated by mutagenic insult, the microenvironment still closely resembles the normal tissue that the lesion arises in. In many cases, notably the breast [48], the prostate [56, 60], and the thyroid gland [25], this near 'normal' tissue microenvironment actually acts to suppress the further expansion of microcarcinomatous lesions. However, as the tumor progresses, the surrounding tissue microenvironment is replaced by an ever-changing milieu that is itself abnormal. Importantly, this abnormal microenvironment co-evolves with the tumor cells as part of the malignant tumor 'organ'. Rather than being suppressive, the microenvironment of the malignant tumor organ instead functions to promote the invasion and metastatic spread of the expanding lesion [5].

In addition to cellular differences, the microenvironment of the malignant tumor organ also differs in its extracellular components compared to the near normal microenvironment that surrounds early tumorigenic lesions. For example, factors released into the extracellular milieu by the tumor cells themselves can act to 'activate' nearby stromal cells in the surrounding microenvironment. The responding stromal cells can consist of those that already reside within the tissue affected or they can be recruited from other sites, most notably the bone marrow. The latter cells often have an ability to expand due to their progenitor characteristics which can further expand the activated stromal cell pool. This expansion and activation has been best documented in the case of cancer-associated fibroblasts (CAF's; [24]), although many other stromal cell types can also be 'activated' by factors in the tumor microenvironment.

CAF's are capable of modifying the acellular architecture of the tumor microenvironment by altering its insoluble extracellular matrix (ECM). This is achieved through changes in the production and deposition of matrix molecules as well as the alteration of the structure and interaction of those matrix molecules already present in the ECM. Examples include CAF-mediated changes in the deposition of collagens, their extracellular processing by metalloproteases and their cross-linking by lysyl oxidases [11, 31]. CAF's also release soluble factors into the tumor microenvironment that act in a feed-forward way to further stimulate the growth of, or alter the phenotype of, tumor cells and to further recruit and activate more stromal cells. Tumor cell proliferation and survival can be facilitated by the CAF-mediated release of factors such as IGF-1, EGF, HGF and IL6, while a major tumor cell phenotype modifier that can be released by CAFs is transforming growth factor beta (TGF- β) (see epithelial-mesenchymal transformation section, below; [64]). Stromal cell recruitment and their feed forward activation can be initiated by the CAF-mediated release of cytokines such as CCL5 and SDF-1 [32, 35]. Importantly, the cytokines and growth factors that are released into the primary tumor microenvironment can act as short range paracrine factors as described above or they can also travel to distant sites and sow the microenvironmental soil to facilitate the expansion and local invasion of secondary metastatic lesions [76].

Acting in concert, soluble factors released by both the tumor cells and the activated stromal cells in the microenvironment can act over considerable distances to recruit immune cells to the lesion site. This immune infiltration, which has many of the hallmarks of a chronic inflammatory state, consists of varying numbers and ratios of lymphoid and myeloid cells, the precise nature of which depends on the tumor site involved and the malignant state of the lesion [59]. These infiltrates produce their own growth factors and cytokines that then further influence nearby and tumor and stromal cells in the manner described above. They also secrete proteolytic enzymes that can remodel the ECM [42]. Such remodeling can have profound effects on the tissue microenvironment as it can release and/or activate soluble factors that are normally sequestered within the insoluble portion of the ECM.

Soluble factors and ECM fragments released into the tumor microenvironment by proteolytic degradation can also act in a paracrine manner on nearby vascular endothelial cells and their surrounding pericyte stem-like population to initiate an angiogenic response. A critical factor that helps drive this angiogenic response is vascular endothelial growth factor (VEGF); [61], although VEGF-independent factors come to the fore as the malignant tumor organ evolves which has been confounding to targeted anti-angiogenic therapy development [63]. Regardless, the resulting formation of new blood vessels leads to an increase in blood supply that is critical to the malignant tumor organ's survival. Concurrently, the angiogenic process itself further alters the landscape of the tumor microenvironment because the resulting newly formed blood vessels are often tortuous, porous and 'leaky' which increases the hydrostatic pressure within the tissue. In addition, tumor cells are subjected to widely varying oxygen tensions depending how far they are from these new vessels [7, 19]. In areas of low oxygen, the resulting hypoxia induces metabolic alterations and a suite of gene expression changes in both the tumor and the stromal cells that lead to further microenvironmental changes in soluble factor production and ECM production and modulation that can affect a number of tumor characteristics including genome stability (see below). It is becoming increasingly clear that a wide variety of responses to metabolic alterations within the malignant tumor microenvironment are also mediated by the activated stromal cells. For example, in response to the release of reactive oxygen species by tumor cell, nearby CAFs upregulate the expression of glycolytic enzymes which leads to their increased production and secretion of pyruvate and lactic acid that are then secreted and utilized by the surrounding tumor cells as a critical alternative energy source for their continued rapid proliferation [55].

While the cellular and acellular aspects of the microenvironment broadly influence tumor and stromal cell phenotypes, they also impinge upon specific processes that are critical for malignant progression. These include alterations to genomic stability, the epigenome, non-coding RNAs, immunomodulation, mesenchymal transformation and mechanotransduction. We provide specific examples below and expand on individual processes in the chapters that follow.

Hypoxia-Induced Genomic Instability

In normal tissues, the microenvironmental oxygen tension can be as high as 10%. In contrast, the oxygen tension within rapidly expanding locally advanced solid tumors is often less than 1% due in large part to their high metabolic rate [71]. The resulting hypoxia leads to the increased production of factors by both tumor and stromal cells that are transcriptional regulators, the prototype of which is hypoxia inducible factor 1alpha (HIF1 α). These regulators both stimulate and suppress the expression of a wide variety of genes whose products modify the microenvironment to facilitate metastatic progression. These include the afore-mentioned angiogenic factor VEGF as well as osteopontin, an ECM protein that facilitates tumor cell invasion. The latter factor facilitates the movement of tumor cell cohorts even farther away from blood vessels which further exacerbates the hypoxic state locally within the lesion [43].

Hypoxia also acts to suppress the expression of a number of homologous recombination (HR) genes involved in repairing the DNA double strand breaks caused by ionizing radiation and radiomimetic drugs [44]. In some contexts this suppression is HIF1 α -dependent [35] while in others it is not [4]. Interestingly, in some cases these two means of hypoxia-induced suppression can act on the same HR gene. This is the case, for example, with the suppression of BRCA1. Hypoxia also suppresses the expression of genes involved in DNA mismatch repair. This leads to an increase in spontaneous mutations that are associated with microsatellite instability (MSI) during experimental colorectal carcinoma progression [17, 62]. Furthermore, in human colorectal tumors HIF1 α expression, which is used as an indication of hypoxia, and MSI are associated are associated with poor outcome/progression [21].

There is some evidence that hypoxia can also affect chromosome segregation during mitosis. Mechanistically, it appears that this occurs due to an alteration of the centrosome that leads to a defect in mitotic spindle formation [46]. Taken together, these and many other observations [43] indicate that hypoxia-mediated defects in DNA repair and chromosome segregation accelerate the genomic instability that is already intrinsic to the growing tumor organ, thereby facilitating the continued evolution of malignant progression [9].

Epigenetic Dysregulation

The most widely studied epigenomic change that is correlated with tumorigenesis is the CpG island hypermethylation phenotype (CIMP). Essentially, this is a broad measure of suppressive promoter methylation that has been observed in bladder, breast, endometrial, gastric, colorectal, hepatocellular, lung, ovarian, pancreatic, renal cell and prostate carcinomas as well as leukemias, melanomas and gliomas. Drivers of this phenotype, including mutations in the isocitrade dehydrogenase-1 gene that result in the accumulation of the hypermethylating oncometabolite 2HG [70], are now being identified. However, while it is clear that the CIMP phenotype contributes to tumor formation, its role in tumor progression is less clear. For example, in colorectal cancer CIMP functionally contributes to the initial tumor formation but not tumor progression. Instead, a subsequent trend towards hypomethylation becomes more prominent as lesions progress from adenomas to invasive cancers [67]. In addition to contributing to a general increase in genomic instability, this hypomethylation has been shown to specifically trigger the production and release of soluble growth factors and modulators including insulin-like growth factor-2 (IGF2) and IGF2 binding protein-3 into the tumor microenvironment [29, 41]. These factors act in an autocrine fashion to increase tumor cell proliferation and invasion and they act in a paracrine fashion to activate stromal cells which, as was described above, have feed forward effects on tumour cells that contribute to malignant progression.

LINE-1 hypomethylation within the long interspersed nucleotide element-1 (LINE-1) often occurs in metastatic prostate [74] and metastatic endocrine pancreatic [10] carcinomas. While this epigenetic mark is often used as a general indicator of hypomethylation, it is also known to be functionally significant in tumor progression in a number of ways. Specifically, it can lead to the activation of adjacent genes as well as an increase in chromosomal instability [18, 65, 72] as well as genomic instability [1, 50].

Immunomodulation

Cytokines such as interleukin-4 and -13, produced by malignant and stromal cells within the tumor microenvironment in a manner that mimics the end stages of wound healing, cause an immune suppression that is tumor promoting. This is initiated, in large part, by the cytokine-mediated recruitment of monocytes to the lesion. These new recruits then differentiate into alternatively activated tumor-associated macro-phages (TAMs) that skew towards an 'M2' phenotype that is immunosuppressive

[47, 59]. More specifically, alternatively activated TAMs do not exhibit the cytotoxicity of typical macrophages [54]. Instead, they release paracrine-acting factors such as the chemokine CCL22 [12] and they generate nitrogen species, particularly under hypoxic microenvironmental conditions [14], that suppress the infiltration and proliferation of T-lymphocytes into the tumor microenvironment. Immunosuppressive TAM's also secrete VEGF-A [40] which augments the hypoxia-induced increase in angiogenesis within the tumor microenvironment described above. Alternatively activated TAM's are an attractive anti-metastatic therapeutic target given their profound ability to facilitate tumor progression by contributing to an escape from immune surveillance while simultaneously promoting angiogenesis. Experimentally, TAMs can be targeted by blocking the cytokine CSF-1 [13, 45], which is required for the proliferation and differentiation all macrophage populations. Unfortunately, this approach is a very broad one and is likely to have long term side effects in patients. A more specific approach would be the reprogramming of TAM's into more conventional 'antigen-presenting' immune response-promoting macrophages that are known to have anti-tumor effect. Experimentally, this has been achieved using histidine-rich glycoprotein [57].

Interestingly, some cytotoxic drugs (eg. paclitaxel) can suppress the M2 TAM phenotype and skew it towards a pro-inflammatory M1 phenotype that is then antitumoral [34]. Thus, a goal of the field has been to identify mediators that drive this proinflammatory M1 TAM skewing in a predictable manner. One such mediator is the microRNA miR-511–3p [66]. Thus, non-coding RNA's are capable of modulating effectors in the microenvironment that play a critical role in tumor progression.

Immunomodulatory changes during tumor progression are discussed in more detail by Gregor Reid in Chap. 8 of this volume.

Mesenchymal Transformation

A major driver of tumor cell invasion is the epithelial-to-mesenchymal transformation (EMT; [69]). During the EMT process there is a breakdown in apical-basal polarity followed by the loss of adhesive junctions between stationary epithelial cells. The resulting individual cells acquire an anterior-posterior polarity and they become motile and mesenchymal [22]. Therefore, classical markers of this transformation are the loss of the epithelial cell-cell adhesion molecule E-Cadherin and the upregulation the mesenchymal intermediate filament protein vimentin. These changes, particularly at expanding tumor fronts, are often used as an indicator of invasive progression and the onset of the metastatic process [30].

During normal development, EMTs contribute to organogenesis and the formation of the body plan in a manner that is tightly regulated, both spatially and temporally [26]. For example, this occurs during gastrulation where a precisely controlled EMT leads to the production of invasive mesenchymal cells that move into the embryo and later re-aggregate to form the mesodermal condensations during

primary germ layer formation [39]. While microenvironmental organizing centers (eg. the primitive knot, Spemann's organizer) that regulate the position and timing of developmental EMT's have been identified, the precise nature of the instructive paracrine factors they release have still not been well characterized. In contrast, the core transcriptional machinery that acts to initiate the gene expression changes that initiate an EMT has been determined. This includes the Snail, Zeb and Twist transcription factors which act on the E-cadherin promoter to inhibit the gene's expression as well as stimulate the expression of secreted factors that further stimulate an EMT [52]. An example of the latter is platelet-derived growth factor (PDGF) which itself stimulates the localized production and activation of metalloproteases that degrade the ECM in the microenvironment to facilitate the migration and invasion of mesenchymal cells produced by the EMT [16].

During malignant tumor progression the precise spatial and temporal control of EMT is disrupted, most often because of an inappropriate accumulation and/ or activation of EMT-inducing factors within the tumor microenvironment. One such factor is TGF-B which, when it acts on epithelial-derived tumor cells (but not normal epithelial cells) stimulates the activity of the core EMT transcription factor complex [33]. TGF-B is often produced and secreted by cancer-associated fibroblasts in the microenvironment [8]. Interestingly, TGF-B that is produced by platelets and released into the microenvironment due to the leakiness of new vessels formed by angiogenesis can also contribute to transient tumor cell EMT at sites of thrombosis [36]. This has important implications for the movement of tumor cells from the stroma into the vasculature by a process known as intravasation. Additionally, it may also help explain why circulating tumor cells themselves can be mesenchymal [75]. The latter point is not trivial in terms of survival in the circulation as mesenchymally-transformed cells tend to be resistant to the suspensionmediated apoptosis that normally occurs when epithelial cells are detached from the ECM [27].

Other factors found in the tumor microenvironment that can act to stimulate the core EMT transcriptional machinery including the afore-mentioned PDGF produced by CAF's as well as WNT or WNT-like factors produced by recruited mesenchymal stem cells and interleukin-6 (IL-6) produced by TAMs [49]. Importantly, the removal of such factors can shift the tumor cell phenotype from the mesenchymal back to the epithelial in a process known as mesenchymal-epithelial transformation (MET). This often occurs during the later stages of metastatic progression where an MET is proposed to contribute to the colonization of distant sites after tumor cells have left the vasculature by extravasation. Such transient shifts between epithelial and mesenchymal phenotypes can also be regulated by the oxygen tension in the microenvironment given that hypoxia upregulates the core EMT transcriptional complex via the actions of HIF-1 α [73]. Thus, the mesenchymal phenotype is often plastic, unless mutations within the E-cadherin gene and/or stable, epigeneticallydriven changes in E-cadherin expression occur. As such, there are varying tumor microenvironment-driven metastable and stable states of mesenchymal transformation within tumor lesions that have important implications for therapeutic treatment strategies bent on reversing the process [68].

Mechanotransduction

Once they have acquired the ability to become invasive, either by varying degrees of mesenchymal transformation or by other means that can include either collective or single-celled amoeboid migration [20], tumor cells move through the tumor microenvironment by interacting with the ECM, the components of which are highly modified due to changes in component deposition, molecular cross-linking, and proteolytic processing within that microenvironment [58]. Ultimately, the molecular composition of the ECM greatly contributes to changes in motile phenotype of the invading tumor cells based on, for example, the soluble factors it sequesters and the specific nature of the cell surface integrins that it engages [28]. However, it is becoming increasingly clear that the mechanical properties of the ECM also play an important role in regulating the phenotype of the invading tumor cells. In this case, physical changes in the ECM can dramatically alter mechanical signals within tumor cells that influence proliferation, survival and the invasive phenotype itself [15]. Experimentally, this can be achieved by artificially crosslinking ECM components, particularly collagens, to increase the stiffness of the matrix which increases intracellular tension and integrin-mediated biochemical signaling within the tumor cell [38]. Thus, mechanical cues in the ECM are translated intracellularly by the cytoskeleton and signaling moieties that are modulated by tension applied through integrin-containing adhesion complexes. Such collagen cross-linking can be achieved by the actions lysyl-oxidase which is released into the tissue microenvironment by CAFs. In yet another example of a feed forward mechanism, lysyl oxidase-dependent collagen crosslinking will further activate CAF's themselves in an integrin signaling-dependent manner [3] and this effect can be so strong that it can facilitate tumor invasion and metastatic progression even when TGF-B is removed [53].

Importantly, collagen crosslinking-dependent increases in radiologically observable mammographic 'density' is a major risk factor for breast carcinoma formation and progression [6]. The latter effect may be facilitated by the fact that ECM stiffness-mediated mechanotransduction augments the ability of soluble factors sequestered within the tumor microenvironment to efficiently induce an EMT [37].

Mechanotransduction events that contribute to tumor progression are discussed in more detail by Celeste Nelson's group in Chap. 7 of this volume.

Summary

It is now clear that the microenvironment that a tumor cell finds itself within can greatly affect its phenotype regardless its genotype. These microenvironmental effects are mediated by surrounding stromal cells, soluble factors, and the extracellular matrix all of which act together, with tumor cells, to form the tumor organ. Therefore, given the molecular and cellular complexity within the tumor organ, it is very difficult to predict the response of any one component of the organ to a particular therapeutic treatment when that component is viewed in isolation. While this complexity can be extremely problematic when viewed from a reductionist point of view, particularly when it contributes to the failure of agents targeted against specific tumor cell-intrinsic oncogenes or tumor suppressors, it also provides myriad new therapeutic opportunities to halt the emergence of those microenvironment-dependent tumour progression phenotypes that contribute to the overwhelming majority of cancer deaths due to metastasis.

References

- Baba Y, Huttenhower C, Nosho K, Tanaka N, Shima K, Hazra A, Schernhammer ES, Hunter DJ, Giovannucci EL, Fuchs CS, Ogino S (2010a) Epigenomic diversity of colorectal cancer indicated by LINE-1 methylation in a database of 869 tumors. Mol Cancer 9:125–134
- Baba Y, Nosho K, Shima K, Huttenhower C, Tanaka N, Hazra A, Giovannucci EL, Fuchs CS, Ogino S (2010b) Hypomethylation of the IGF2 DMR in colorectal tumors, detected by bisulfite pyrosequencing, is associated with poor prognosis. Gastroenterology 139(6):1855–1864
- Barker HE, Bird D, Lang G, Erler JT (2013) Tumor-secreted LOXL2 activates fibroblasts through FAK signaling. Mol Cancer Res 11(11):1425–1436
- 4. Bindra RS, Schaffer PJ, Meng A, Woo J, Maseide K, Roth ME, Lizardi P, Hedley DW, Bristow RG, Glazer PM (2005) Alterations in DNA repair gene expression under hypoxia: elucidating the mechanisms of hypoxia-induced genetic instability. Ann N Y Acad Sci 4:184–195
- Bissell MJ, Hines WC (2011) Why don't we get more cancer? A proposed role of the microenvironment in restraining cancer progression. Nat Med 17(3):320–329
- 6. Boyd NF, Martin LJ, Yaffe MJ, Minkin S (2011) Mammographic density and breast cancer risk: current understanding and future prospects. Breast Cancer Res 13(6):223
- Carmeliet P, Jain RK (2011) Molecular mechanisms and clinical applications of angiogenesis. Nature 473(7347):298–307
- Chaffer CL, Weinberg RA (2011) A perspective on cancer cell metastasis. Science 331(6024):1559–1564
- 9. Chan N, Koch CJ, Bristow RG (2009) Tumor hypoxia as a modifier of DNA strand break and cross-link repair. Curr Mol Med 4:401–410
- Choi IS, Estecio MR, Nagano Y, Kim do H, White JA, Yao JC, Issa JP, Rashid A (2007) Hypomethylation of LINE-1 and Alu in well-differentiated neuroendocrine tumors (pancreatic endocrine tumors and carcinoid tumors). Mod Pathol 20:802–810
- 11. Čirri P, Chiarugi P (2012) Cancer-associated-fibroblasts and tumour cells: a diabolic liaison driving cancer progression. Cancer Metastasis Rev 31(1–2):195–208
- Curiel TJ, Coukos G, Zou L, Alvarez X, Cheng P, Mottram P, Evdemon-Hogan M, Conejo-Garcia JR, Zhang L, Burow M, Zhu Y, Wei S, Kryczek I, Daniel B, Gordon A, Myers L, Lackner A, Disis ML, Knutson KL, Chen L, Zou W (2004) Specific recruitment of regulatory T cells in ovarian carcinoma fosters immune privilege and predicts reduced survival. Nat Med 10(9):942–949
- DeNardo DG, Brennan DJ, Rexhepaj E, Ruffell B, Shiao SL, Madden SF, Gallagher WM, Wadhwani N, Keil SD, Junaid SA, Rugo HS, Hwang ES, Jirström K, West BL, Coussens LM (2011) Leukocyte complexity predicts breast cancer survival and functionally regulates response to chemotherapy. Cancer Discov 1(1):54–67
- Doedens AL1, Stockmann C, Rubinstein MP, Liao D, Zhang N, DeNardo DG, Coussens LM, Karin M, Goldrath AW, Johnson RS (2010) Macrophage expression of hypoxia-inducible factor-1 alpha suppresses T-cell function and promotes tumor progression. Cancer Res 70(19):7465–7475

- DuFort CC, Paszek MJ, Weaver VM (2011) Balancing forces: architectural control of mechanotransduction. Nat Rev Mol Cell Biol 12:308–319
- 16. Eckert MA, Lwin TM, Chang AT, Kim J, Danis E, Ohno-Machado L, Yang J (2011) Twistlinduced invadopodia formation promotes tumor metastasis. Cancer Cell 19:372–386
- Edwards RA, Witherspoon M, Wang K, Afrasiabi K, Pham T, Birnbaumer L, Lipkin SM (2009) Epigenetic repression of DNA mismatch repair by inflammation and hypoxia in inflammatory bowel disease-associated colorectal cancer. Cancer Res 4:6423–6429
- Faulkner GJ, Kimura Y, Daub CO, Wani S, Plessy C, Irvine KM, Schroder K, Cloonan N, Steptoe AL, Lassmann T, Waki K, Hornig N, Arakawa T, Takahashi H, Kawai J, Forrest AR, Suzuki H, Hayashizaki Y, Hume DA, Orlando V, Grimmond SM, Carninci P (2009) The regulated retrotransposon transcriptome of mammalian cells. Nat Genet 41:563–571
- 19. Folkman J, Watson K, Ingber D, Hanahan D (1989) Induction of angiogenesis during the transition from hyperplasia to neoplasia. Nature 339(6219):58–61
- Friedl P1, Locker J, Sahai E, Segall JE (2012) Classifying collective cancer cell invasion. Nat Cell Biol 14(8):777–783
- Furlan D, Sahnane N, Carnevali I, Cerutti R, Bertoni F, Kwee I, Uccella S, Bertolini V, Chiaravalli AM, Capella C (2008) Up-regulation of the hypoxia-inducible factor-1 transcriptional pathway in colorectal carcinomas. Hum Pathol 4:1483–1494
- 22. Greenburg G, Hay E D (1982). Epithelia suspended in collagen gels can lose polarity and express characteristics of migrating mesenchymal cells. J Cell Biol 95:333–339
- 23. Hanahan D, Coussens LM (2012) Accessories to the crime: functions of cells recruited to the tumor microenvironment. Cancer Cell 21:309–322
- 24. Hanahan D, Weinberg RA (2011) Hallmarks of cancer: the next generation. Cell 144(5):646–674
- 25. Harach HR, Franssila KO, Wasenius VM (1985) Occult papillary carcinoma of the thyroid. A "normal" finding in Finland. A systematic autopsy study. Cancer 56:531–538
- 26. Hay E D (2005) The mesenchymal cell, its role in the embryo, and the remarkable signaling mechanisms that create it. Dev Dyn 233:706–720
- 27. Huang RY, Wong MK, Tan TZ, Kuay KT, Ng AH, Chung VY, Chu YS, Matsumura N, Lai HC, Lee YF, Sim WJ, Chai C, Pietschmann E, Mori S, Low JJ, Choolani M, Thiery JP (2013) An EMT spectrum defines an anoikis-resistant and spheroidogenic intermediate mesen-chymal state that is sensitive to e-cadherin restoration by a src-kinase inhibitor, saracatinib (AZD0530). Cell Death Dis 4:e915
- 28. Hynes RO (2009) The extracellular matrix: not just pretty fibrils. Science 326:1216-1219
- 29. Ito Y1, Koessler T, Ibrahim AE, Rai S, Vowler SL, Abu-Amero S, Silva AL, Maia AT, Huddleston JE, Uribe-Lewis S, Woodfine K, Jagodic M, Nativio R, Dunning A, Moore G, Klenova E, Bingham S, Pharoah PD, Brenton JD, Beck S, Sandhu MS, Murrell A (2008) Somatically acquired hypomethylation of IGF2 in breast and colorectal cancer. Hum Mol Genet 17(17):2633–2643
- Kalluri R, Weinberg RA (2009) The basics of epithelial-mesenchymal transition. J Clin Invest 119(6):1420–1428
- 31. Kalluri R, Zeisberg M (2006) Fibroblasts in cancer. Nat Rev Cancer 6(5):392-401
- Karnoub AE, Dash AB, Vo AP, Sullivan A, Brooks MW, Bell GW, Richardson AL, Polyak K, Tubo R, Weinberg RA (2007) Mesenchymal stem cells within tumour stroma promote breast cancer metastasis. Nature 449(7162):557–563
- Katsuno Y, Lamouille S, Derynck R (2013) TGF-β signaling and EMT in cancer progression. Curr Opin Oncol 25(1):76–84
- Kodumudi KN, Woan K, Gilvary DL, Sahakian E, Wei S, Djeu JY (2010) A novel chemoimmunomodulating property of docetaxel: suppression of myeloid-derived suppressor cells in tumor bearers. Clin Cancer Res 16(18):4583–4594
- 35. Kojima Y, Acar A, Eaton EN, Mellody KT, Scheel C, Ben-Porath I, Onder TT, Wang ZC, Richardson AL, Weinberg RA, Orimo A (2010) Autocrine TGF-beta and stromal cell-derived factor-1 (SDF-1) signaling drives the evolution of tumor-promoting mammary stromal myofibroblasts. Proc Natl Acad Sci U S A 107(46):20009–20014

- 6 Microenvironmental Control of Metastatic Progression
- Labelle M, Begum S, Hynes RO (2011) Direct signaling between platelets and cancer cells induces an epithelial-mesenchymal-like transition and promotes metastasis. Cancer Cell 20(5):576–590
- Lee K, Chen QK, Lui C, Cichon MA, Radisky DC, Nelson CM (2012) Matrix compliance regulates Rac1b localization, NADPH oxidase assembly, and epithelial-mesenchymal transition. Mol Biol Cell 23(20):4097–4108
- Levental KR, Yu H, Kass L, Lakins JN, Egeblad M, Erler JT, Fong SF, Csiszar K, Giaccia A, Weninger W, Yamauchi M, Gasser DL, Weaver VM (2009) Matrix crosslinking forces tumor progression by enhancing integrin signaling. Cell 139(5):891–906
- Lim J, Thiery JP (2012) Epithelial-mesenchymal transitions: insights from development. Development 139(19):3471–3486
- Lin EY, Li JF, Bricard G, Wang W, Deng Y, Sellers R, Porcelli SA, Pollard JW (2007) Vascular endothelial growth factor restores delayed tumor progression in tumors depleted of macrophages. Mol Oncol 1(3):288–302
- 41. Lochhead P, Imamura Y, Morikawa T, Kuchiba A, Yamauchi M, Liao X, Qian ZR, Nishihara R, Wu K, Meyerhardt JA, Fuchs CS, Ogino S (2012) Insulin-like growth factor 2 messenger RNA binding protein 3 (IGF2BP3) is a marker of unfavourable prognosis in colorectal cancer. Eur J Cancer 48(18):3405–3413
- Lu P, Weaver VM, Werb Z (2012) The extracellular matrix: a dynamic niche in cancer progression. J Cell Biol 196(4):395–406
- 43. Luoto KR, Kumareswaran R, Bristow RG (2013) Tumor hypoxia as a driving force in genetic instability. Genome Integr 4(1):5
- Meng AX, Jalali F, Cuddihy A, Chan N, Bindra RS, Glazer PM, Bristow RG (2005) Hypoxia down-regulates DNA double strand break repair gene expression in prostate cancer cells. Radiother Oncol 76:168–176
- 45. Mitchem JB, Brennan DJ, Knolhoff BL, Belt BA, Zhu Y, Sanford DE, Belaygorod L, Carpenter D, Collins L, Piwnica-Worms D, Hewitt S, Udupi GM, Gallagher WM, Wegner C, West BL, Wang-Gillam A, Goedegebuure P, Linehan DC, DeNardo DG (2013) Targeting tumor-infiltrating macrophages decreases tumor-initiating cells, relieves immunosuppression, and improves chemotherapeutic responses. Cancer Res 73(3):1128–1141
- Moser SC, Bensaddek D, Ortmann B, Maure JF, Mudie S, Blow JJ, Lamond AI, Swedlow JR, Rocha S (2013) PHD1 Links cell-cycle progression to oxygen sensing through hydroxylation of the centrosomal protein Cep192. Dev Cell 4:381–392
- 47. Movahedi K, Laoui D, Gysemans C, Baeten M, Stangé G, Van den Bossche J, Mack M, Pipeleers D, In't Veld P, De Baetselier P, Van Ginderachter JA (2010) Different tumor microenvironments contain functionally distinct subsets of macrophages derived from Ly6C(high) monocytes. Cancer Res 70(14):5728–5739
- Nielsen M, Thomsen JL, Primdahl S, Dyreborg U, Andersen JA (1987) Breast cancer and atypia among young and middle-aged women: a study of 110 medicolegal autopsies. Br J Cancer 56:814–819
- Nieto MA (2011) The ins and outs of the epithelial to mesenchymal transition in health and disease. Annu Rev Cell Dev Biol 27:347–376
- Ogino S, Nosho K, Irahara N, Shima K, Baba Y, Kirkner GJ, Meyerhardt JA, Fuchs CS (2009) Prognostic significance and molecular associations of 18q loss of heterozygosity: a cohort study of microsatellite stable colorectal cancers. J Clin Oncol 27(27):4591–4598
- Ogino S, Nosho K, Irahara N, Shima K, Baba Y, Kirkner GJ, Meyerhard JA, Fuchs CS (2009) Prognostic significance and molecular associations of 18q loss of heterozygosity: a cohort study of microsatellite stable colorectal cancers. J Clin Oncol 26:5713–5720
- 52. Peinado H, Olmeda D, Cano A (2007) Snail, Zeb and bHLH factors in tumour progression: an alliance against the epithelial phenotype? Nat Rev Cancer 7:415–428
- 53. Pickup MW, Laklai H, Acerbi I, Owens P, Gorska AE, Chytil A, Aakre M, Weaver VM, Moses HL (2013) Stromally derived lysyl oxidase promotes metastasis of transforming growth factor-β-deficient mouse mammary carcinomas. Cancer Res 73(17):5336–5346

- Qian BZ1, Pollard JW (2010) Macrophage diversity enhances tumor progression and metastasis. Cell 141(1):39–51
- 55. Rattigan YI1, Patel BB, Ackerstaff E, Sukenick G, Koutcher JA, Glod JW, Banerjee D (2012) Lactate is a mediator of metabolic cooperation between stromal carcinoma associated fibroblasts and glycolytic tumor cells in the tumor microenvironment. Exp Cell Res 318(4):326–335
- Rich AR (2007) On the frequency of occurrence of occult carcinoma of the prostrate. Int J Epidemiol 36:274–277
- 57. Rolny C, Mazzone M, Tugues S, Laoui D, Johansson I, Coulon C, Squadrito ML, Segura I, Li X, Knevels E, Costa S, Vinckier S, Dresselaer T, Åkerud P, De Mol M, Salomäki H, Phillipson M, Wyns S, Larsson E, Buysschaert I, Botling J, Himmelreich U, Van Ginderachter JA, De Palma M, Dewerchin M, Claesson-Welsh L, Carmeliet P (2011) HRG inhibits tumor growth and metastasis by inducing macrophage polarization and vessel normalization through down-regulation of PIGF. Cancer Cell 19(1):31–44
- Rubashkin MG, Ou G, Weaver VM (2014) Deconstructing signaling in three dimensions. Biochemistry 53(13):2078–2090
- Ruffell B, Au A, Rugo HS, Esserman LJ, Hwang ES, Coussens LM (2012) Leukocyte composition of human breast cancer. Proc Natl Acad Sci U S A 109(8):2796–2801
- Sakr WA, Haas GP, Cassin BF, Pontes JE, Crissman JD (1993) The frequency of carcinoma and intraepithelial neoplasia of the prostate in young male patients. J Urol 150:379–385
- Senger DR, Van de Water L, Brown LF, Nagy JA, Yeo KT, Yeo TK, Berse B, Jackman RW, Dvorak AM, Dvorak HF (1993) Vascular permeability factor (VPF, VEGF) in tumor biology. Cancer Metastasis Rev 12:303–324
- 62. Shahrzad S, Quayle L, Stone C, Plumb C, Shirasawa S, Rak JW, Coomber BL (2005) Ischemia-induced K-ras mutations in human colorectal cancer cells: role of microenvironmental regulation of MSH2 expression. Cancer Res 4:8134–8141
- 63. Sitohy B, Nagy JA, Dvorak HF (2012) Anti-VEGF/VEGFR therapy for cancer: reassessing the target. Cancer Res 72(8):1909–1914
- Spaeth EL, Dembinski JL, Sasser AK, Watson K, Klopp A, Hall B, Andreeff M, Marini F (2009) Mesenchymal stem cell transition to tumor-associated fibroblasts contributes to fibrovascular network expansion and tumor progression. PLoS ONE 4(4):e4992
- 65. Speek M (2001) Antisense promoter of human L1 retrotransposon drives transcription of adjacent cellular genes. Mol Cell Biol 21:1973–1985
- Squadrito ML, Pucci F, Magri L, Moi D, Gilfillan GD, Ranghetti A, Casazza A, Mazzone M, Lyle R, Naldini L, De Palma M (2012) miR-511-3p modulates genetic programs of tumorassociated macrophages. Cell Rep 1(2):141–154
- 67. Sunami E, de Maat M, Vu A, Turner RR, Hoon DS (2011) LINE-1 hypomethylation during primary colon cancer progression. PLoS ONE 6(4):e18884
- 68. Tam WL, Weinberg RA (2013) The epigenetics of epithelial-mesenchymal plasticity in cancer. Nat Med 19(11):1438–1449
- Thiery JP, Acloque H, Huang RY, Nieto MA (2009) Epithelial-mesenchymal transitions in development and disease. Cell 139(5):871–890
- Turcan S, Fabius AW, Borodovsky A, Pedraza A, Brennan C, Huse J, Viale A, Riggins GJ, Chan TA (2013) Efficient inductino of differentiation and growth inhibition in IDH1 mutant glioma cells by the DNMT inhibitor decitabine. Oncotarget 4:1729–338
- Vaupel P, Harrison L (2004) Tumor hypoxia: causative factors, compensatory mechanisms, and cellular response. Oncologist 4(Suppl 5):4–9
- 72. Yamada Y, Jackson-Grusby L, Linhart H, Meissner A, Eden A, Lin H, Jaenish R (2005) Opposing effects of DNA hypomethylation on intestinal and liver carcinogenesis. Proc Natl Acad Sci U S A 102:13580–13585
- 73. Yang MH, Wu MZ, Chiou SH, Chen PM, Chang SY, Liu CJ, Teng SC, Wu KJ (2008) Direct regulation of TWIST by HIF-1alpha promotes metastasis. Nat Cell Biol 10(3):295–305

- 6 Microenvironmental Control of Metastatic Progression
- 74. Yegnasubramanian S, Haffner MC, Zhang Y Gurel B, Cornish TC, Wu Z, Irizarry RA, Morgan J, Hicks J, DeWeese TL, Isaacs WB, Bova GS, DeMarzo AM, Nelson WG (2008) DNA hypomethylation arises later in prostate cancer progression than CpG island hypermethylation and contributes to metastatic tumor heterogeneity. Cancer Res 68:8954–8967
- 75. Yu M1, Bardia A, Wittner BS, Stott SL, Smas ME, Ting DT, Isakoff SJ, Ciciliano JC, Wells MN, Shah AM, Concannon KF, Donaldson MC, Sequist LV, Brachtel E, Sgroi D, Baselga J, Ramaswamy S, Toner M, Haber DA, Maheswaran S (2013) Circulating breast tumor cells exhibit dynamic changes in epithelial and mesenchymal composition. Science 339(6119):580–584
- Zhang XH, Jin X, Malladi S, Zou Y, Wen YH, Brogi E, Smid M, Foekens JA, Massagué J (2013) Selection of bone metastasis seeds by mesenchymal signals in the primary tumor stroma. Cell 154(5):1060–1073
Chapter 7 Mechanotransduction, Metastasis and Genomic Instability

Allison K. Simi, Alexandra S. Piotrowski and Celeste M. Nelson

Abstract Cells translate mechanical forces in the environment into biochemical signals in a process called mechanotransduction. In this way, mechanical forces direct cell behavior, including motility, proliferation, and differentiation, and become important in physiological processes such as development and wound healing. Abnormalities in mechanotransduction can lead to aberrant cell behavior and disease, including cancer. Changes in extracellular mechanical forces or defects in mechanosensors can result in misregulation of signaling pathways inside the cell, and ultimately lead to malignancy. Here, we discuss the ways in which physical attributes of the tumor microenvironment can promote metastasis and genomic instability, two hallmark features of cancer.

Keywords Mechanical stress · EMT · Stiffness

Abbreviations

2D	Two-dimensional
3D	Three-dimensional
αSMA	α -smooth muscle actin
bFGF	Basic fibroblast growth factor
ECM	Extracellular matrix
EGF	Epidermal growth factor
EMT	Epithelial-mesenchymal transition
ERK	Extracellular-signal-regulated kinase
FAK	Focal adhesion kinase
FGF	Fibroblast growth factor
GIN	Genomic instability

Authors Allison Simi and Alexandra Piotrowski have contributed equally to this work.

e-mail: celesten@princeton.edu

© Springer International Publishing Switzerland 2015

C. Maxwell, C. Roskelley (eds.), *Genomic Instability and Cancer Metastasis*, Cancer Metastasis - Biology and Treatment 20, DOI 10.1007/978-3-319-12136-9_7

C. M. Nelson (🖂) · A. K. Simi · A. S. Piotrowski

Department of Chemical & Biological Engineering, Princeton University, 303 Hoyt Laboratory, William Street, Princeton, NJ 08544, USA

C. M. Nelson

Department of Molecular Biology, Princeton University, 303 Hoyt Laboratory, William Street, Princeton, NJ 08544, USA

IFP	Interstitial fluid pressure
ILK	Integrin-linked kinase
MET	Mechanoelectrical transduction
MLC	Myosin light chain
MMP	Matrix metalloproteinase
PDGF	Platelet-derived growth factor
PI3K	Phosphoinositide 3-kinase
PTEN	Phosphatase and tensin homolog
ROCK	Rho-associated kinase
ROS	Reactive oxygen species
RTK	Receptor tyrosine kinase
TAZ	transcriptional co-activator with PDZ-binding motif
TGF-β	Transforming growth factor β
VEGF	Vascular endothelial growth factor
YAP	Yes-associated protein

Introduction

Over a decade ago, Hanahan and Weinberg defined several features of cancer that they considered essential for the acquisition of a malignant phenotype, including replicative immortality, evasion of growth suppressors, evasion of apoptosis, stimulation of angiogenesis, stimulation of proliferation, and invasion and metastasis [1]. Since then, a flood of cancer research has led to modification and expansion of the proposed hallmarks; metastasis and genomic instability are two that persist [2]. Cancer is widely regarded as a disease of the cell, and cell behavior is directed by both biochemical and physical cues, which can work independently or synergistically [3]. Accordingly, the tumor microenvironment has been shown to affect tumor progression [4, 5]. This chapter focuses on the physical factors and mechanical forces that tumor cells encounter in the tumor microenvironment, which can in turn alter their behavior. Cells convert the physical signals they receive into biological responses via a process known as mechanotransduction [6].

Mechanotransduction involves both the external environment and internal signaling [7]. The transmission of external forces to intracellular signaling is centered on proteins that are activated by force, such as integrins [8, 9] and T-cell receptors [10]. Many cellular phenotypes, including morphology, motility, and proliferation, are governed by external mechanical forces [11–13]. Thus, mechanotransduction is central to a variety of physiologically normal processes, including embryonic development, differentiation, wound healing, and angiogenesis [14, 15]. Defects in mechanotransduction are known to be involved in several diseases, including cancer [16]. Understanding how defects in mechanotransduction affect tumor progression will add to our fundamental knowledge of cancer biology and may suggest new approaches for treatment.

How Mechanotransduction Regulates Normal Cell Behavior

Extracellular Factors Affecting Mechanotransduction in Normal Cells

Most cells are anchorage dependent: they need to adhere to a substratum to prevent apoptosis and promote cell cycle progression [17]. Thus, the mechanical microenvironment is important for cell survival. Cells sense their environment via conformational changes in mechanically responsive proteins, known as mechanosensors. Physical forces induce these conformational changes, which result in downstream signaling inside the cell [18, 14]. Forces can originate from a variety of features, including the rigidity of the extracellular matrix (ECM), static or dynamic fluid flow, and tissue growth [6]. These forces are further classified into specific types of loads that cells can detect. For example, forces incurred by blood flow include hydrodynamic pressure, shear stress, and cyclic strain, and all of these help regulate endothelial cell behaviors [19] such as cell reorientation [20].

Cells can also respond to mechanical loads by secreting biochemical factors, some of which result in subsequent ECM remodeling. Growth factors comprise one class of proteins that are important in this respect. Transforming growth factor β (TGF- β) is sequestered in the ECM, and is released when internal contractility of myofibroblasts is balanced externally by a stiff matrix, causing conformational changes in protein complexes embedded in the ECM. Free TGF- β starts a feed-forward loop, causing increased deposition of ECM proteins and additional (increased) expression of TGF- β [21]. Various other growth factors increase activity as a result of mechanical load, as evidenced by endothelial secretion of basic fibroblast growth factor (bFGF) in response to shear stress and hydrostatic pressure [22, 23]. Mechanical forces also regulate the expression of matrix remodeling proteins such matrix metalloproteinases (MMPs). This is seen in human monocytes/macrophages, which have been shown to increase expression of MMPs under cyclic strain, and thus contribute to ECM degradation [24].

Intracellular Factors Affecting Mechanotransduction in Normal Cells

There are several intracellular components involved in receiving mechanical signals and eliciting a response (Fig. 7.1). A feature that is particularly important to mechanical sensing is contractility; all cells have a network of cytoskeletal proteins (actin, microtubules, intermediate filaments) that aid in cell structure and mobility [17]. Cytoskeletal contractility creates a balance between intra- and extracellular forces acting on the cell, and thus is important for cells to be able to respond to forces in the surrounding microenvironment [25]. This balance exists so that



Fig. 7.1 Schematic of intracellular mechanotransduction pathways connecting the *ECM* to the *cytoplasm* and *nucleus*.

mechanical forces in the microenvironment and internal cellular tension can work together to regulate cell behavior, evident, for example, in changes in fibroblast proliferation when matrix stiffness and actomyosin contractility are decoupled [26]. Moreover, external mechanical stimuli help define the state of the cytoskeletal components through various pathways. For example, it has been shown that tensile forces regulate the expression of α -smooth muscle actin (α SMA), a gene important for cytoskeletal contractility, in osteoblasts [27], and that cytoskeletal tension in fibroblasts changes to match the stiffness of the substratum [28].

Communication between ECM and the cytoskeleton is mediated by mechanosensors, proteins or structures that can sense physical changes in the microenvironment and translate these into chemical signals inside the cell [15]. Mechanosensors are diverse and exist everywhere in the body, from ears to kidneys: mechanoelectrical transduction (MET) channels in cochlear hair cells respond to sound vibrations to induce the signaling necessary in auditory sensation [29], and primary cilia in renal epithelia respond to fluid flow to maintain homeostasis [30]. Yet the sensing mechanisms of many mechanosensors remain poorly understood.

The most well-studied mechanosensors are integrins, which contain extracellular, transmembrane, and cytoplasmic domains [31]. Integrins are composed of α - and β -subunits that form heterodimers [32]. Different types of integrins can bind to various ligands present in the ECM and induce signaling to regulate a variety of processes including attachment, migration, proliferation, and differentiation [33]. Through detection of external mechanical stresses, integrins promote changes in cytoskeletal structure and can activate signal transduction cascades [34–36]. Integrin activity is also essential for the formation of focal adhesions, which act as centers of mechanotransduction [37]. Focal adhesions are protein complexes localized at the plasma membrane that link the ECM to the actin cytoskeleton. In addition to integrins, focal adhesions include hundreds of proteins, the most well-characterized of which are talin, paxillin, vinculin, focal adhesion kinase (FAK) and Src family kinases, which act as signaling molecules [38]. The formation of focal adhesions is regulated by both external forces and cytoskeletal contractility [39].

Other intracellular components involved in mechanotransduction include G proteins, receptor tyrosine kinases (RTKs), extracellular-signal-regulated kinases (ERKs), and stretch-activated ion channels [6].

G proteins are localized at focal adhesion sites and can undergo conformational changes induced by mechanical stress to promote cell growth. G proteins are activated in cardiac fibroblasts in response to stretch, as well as in endothelial cells and osteocytes in response to shear stress [40–42].

RTKs are transmembrane proteins that dimerize to become activated, and are involved in integrin-mediated mechanotransduction downstream of G proteins. Dimerization is triggered by binding of the receptor to extracellular ligands such as epidermal growth factor (EGF) and platelet-derived growth factor (PDGF), leading to further signaling [43]. RTKs can also activate ERKs, which are important for gene expression and protein synthesis [44].

ERKs are kinases that play an important role in intracellular signaling, such as the activation of cytoplasmic and nuclear regulatory proteins. These kinases can be activated in response to mechanical stimuli. Shear stress and stretch have been shown to activate ERKs in aortic endothelial cells and pulmonary epithelial cells, respectively [45, 46].

Stretch-activated ion channels allow ions such as Ca^{2+} to move in and out of cells, which regulates several cellular processes. Cell stretching has been shown to increase intracellular levels of Ca^{2+} in several cell types [47, 48]. Intracellular Ca^{2+} levels are also important for the activation of other proteins in the mechanotransduction signaling cascade, such as ERKs [49].

Mechanotransduction and Metastasis

The invasion of primary tumors into their surrounding tissue and subsequent metastatic spread to other organs are among the largest obstacles to cancer treatment, and metastasis is the main cause of cancer-related deaths [50]. Metastasis relies on the ability of tumor cells to migrate from the primary tumor and form new lesions at distant locations [51]. Invasion and metastasis require physical interactions between malignant cells and the microenvironment, a process that inherently involves mechanosensing and mechanotransduction [16]. Both extracellular factors in the physical tumor microenvironment and intracellular factors within cancer cells contribute to mechanotransduction during invasion and metastasis. Identifying how mechanotransduction becomes abnormally regulated in cancer cells is necessary to understand the mechanisms that underlie invasion and metastasis.

Extracellular Factors Affecting Mechanotransduction in Tumors

The physical microenvironment within a solid tumor differs from that of normal tissue in several ways (Fig. 7.2): uncontrolled proliferation results in increased mechanical compression in a spatially restricted environment [52]; there is an increase in the production of ECM components (of which collagen is the most prevalent structural protein), which exhibit increased alignment, crosslinking, bundling, and stiffening [53, 54]; poorly formed blood vessels and the absence of functional lymphatics lead to increased interstitial fluid pressure (IFP) [155]. These changes in the extracellular environment can alter the behavior of tumor cells via mechanotransduction pathways, which are important for both invasion and metastasis. For example, mechanical compression can promote invasion and metastasis [55]. Compression has been shown to enhance cell-substratum adhesion in two-dimensional (2D) cell culture compression assays [52]. Moreover, compression can facilitate invasion by increasing the release and activation of ECM-degrading MMPs [56]. Mechanical loading can also alter cell shape and motility through compression-dependent changes in cytoskeletal dynamics [57].

The ECM is the framework for intercellular crosstalk, adhesion, and migration [58]. Solid tumors exhibit increased ECM stiffness and crosslinking, and changes in the structural components and mechanical properties of the ECM can promote an invasive phenotype in cancer cells [7, 16, 59]. For example, the mode by which tumor cells migrate is strongly dependent on the physical properties of the ECM [60].



Fig. 7.2 Cartoon illustrating the physical changes in the tumor microenvironment compared to that of normal tissue. **a** Normal tissue microenvironment. The microenvironment in normal tissues contains linearized blood vessels that perfuse the tissue. *Lymphatic vessels* are present to drain excess fluids and maintain fluid homeostasis. *ECM* proteins make up the loose connective framework. **b** Tumor microenvironment. Poorly formed blood vessels *leak fluid* and *plasma macromolecules* into the interstitium. Many solid tumors lack a functioning lymphatic system. There are larger amounts of *ECM* proteins that are highly aligned, crosslinked, bundled, and stiffened. In addition, uncontrolled proliferation of cells in a confined space results in mechanical compression.

Changes in ECM composition and architecture also affect the distribution and activation of soluble factors (e.g., growth factors, cytokines, MMPs) that are themselves involved in cell behavioral changes and mechanotransduction [61]. ECM stiffness can promote the malignant behavior of tumor cells by increasing the expression and activity of adhesion receptors, thereby also activating mechanotransduction pathways [12]. For example, force has been shown to influence the development of focal adhesions since maturation of these complexes requires mechanical tension [62].

Increased ECM stiffness also directs cell behavior by increasing external resistance forces experienced by the cell [63]. Links to the ECM via integrins and focal adhesions can relay these stresses to the cytoskeleton, alter the balance of intracellular forces, and stimulate signal transduction cascades that influence cell behavior [7]. Moreover, increased ECM stiffness can disrupt epithelial polarity and induce migration and metastasis [64]. Cells have also been shown to migrate preferentially to regions of increased ECM stiffness via mechanotaxis/durotaxis [65, 66]. Finally, the crosslinking of ECM by lysyl oxidase, which can also stiffen the matrix and induce fibrosis, can promote tumorigenesis via enhanced integrin signaling [58].

ECM remodeling by tumor and stromal cells is important for both invasion and metastasis. For example, migrating tumor cells exhibit pericellular proteolytic degradation to make room for further migration [67]. Proteases such as MMPs are recruited to integrin assemblies and other adhesion receptors at the leading edge of a migrating cell to model and degrade the ECM [68]. Cancer cells have also been shown to realign their surrounding ECM perpendicular to the tumor boundary, altering its architecture for improved adhesion and migration, creating diverse routes for dissemination [69]. Migration is mediated by several types of proteolytic structures enriched with F-actin, β 1-integrins, and MMPs, which are key players in mechanotransduction [70]. Single cell migration can also occur without proteolytic degradation under the mode of amoeboid migration [71]. The microscale architecture of the ECM, including the alignment of fibers and the location and size of pores, dictates the mechanisms of invasion and metastasis applied by cancer cells [72].

IFP and interstitial fluid flow have also been shown to affect the migratory and invasive behaviors of tumor cells [73, 156, 157]. In a three-dimensional (3D) culture model in which single tumor cells were suspended in ECM, fluid flow was shown to increase the percentage of migratory cells as well as their speed [73]. In a similar study, interstitial fluid flow was shown to result in the upstream migration of cancer cells as a result of asymmetry in matrix adhesion stresses needed to balance drag from fluid flow [74]. The stresses induced by flow created a gradient of integrin activation across the cells. Components of focal adhesions, including FAK, paxillin, and vinculin, localized at the upstream side of the migrating cells.

Intracellular Factors Affecting Mechanotransduction in Tumors

It is well known that changes in mechanotransduction promote invasion and metastasis [75]. The intracellular factors affecting mechanotransduction pathways in tumor cells may be altered in response to changes in the tumor microenvironment,

or to genetic mutations and changes in gene expression within the tumor cells. Intracellular mechanotransduction can, in turn, lead to changes in gene expression to promote invasion and metastasis.

Cytoskeletal reorganization is important for changes in cell shape and motility, and therefore migration and metastasis [16]. Cytoskeletal tension is primarily regulated by ERKs and the Rho family of small GTPases. One effector of Rho is Rho-associated kinase (ROCK), which regulates actin cytoskeletal contractility via myosin light chain (MLC) phosphorylation [64]. Rho activity has been shown to be elevated in some tumors, though decreases in its activity have also been reported [76, 77]. Cytoskeletal tension is also affected by the mechanical properties of the ECM, such as stiffness and crosslinking [7]. Increased matrix stiffness promotes the clustering of integrins and the formation of focal adhesions, in addition to increasing activation of FAK and ERK, and enhancing ROCK-mediated cytoskeletal contractility [64]. ROCK is also involved in the disruption of adherens junctions and moving the tail end of the cell behind the leading edge to assist in cell locomotion [78–80]. Moreover, cell migration involves the extension of membrane protrusions resulting from the cycling of actin polymerization and depolymerization, which are regulated by Rho GTPases via the cofilin pathway [81, 82].

ECM crosslinking has also been shown to result in the aggregation and clustering of integrins as well as enhanced signaling via phosphoinositide 3-kinase (PI3K) to induce invasion [58, 64]. Other components of focal adhesions have also been implicated in tumor progression, including Src, the activity of which has been shown to influence proliferation, invasion and metastasis [83, 84]. Src activation is required for ECM degradation during migration [85]. In 3D culture studies of breast tumor cells, Src activity increases the strength of cellular forces on the ECM as well as the duration and length of cell membrane protrusions [86].

Whereas some cells in the tumor become stiffer, metastatic cells are more deformable and exhibit reduced cytoskeletal stiffness [87]. Lower levels of integrin expression along with decreased adhesion to the ECM have been associated with oncogenic transformation [88, 60]. This increased deformability is correlated with enhanced metastatic potential. For example, enhanced deformability enables metastatic cells to move through tight spaces, such as between endothelial cells, during intravasation and extravasation [89].

In addition to regulating the cytoskeleton and associated proteins, mechanotransduction can lead to gene expression changes that promote invasion and metastasis. Cancer cells undergo a variety of genetic mutations and gene expression changes during tumor progression, which can affect their interactions with the microenvironment and subsequent mechanotransduction. Mechanotransduction itself is one source of changes in gene expression in cancer cells. A major way that mechanotransduction can affect gene expression is via the epithelial-mesenchymal transition (EMT). EMT, in which epithelial genes are downregulated and mesenchymal genes are upregulated, is thought to be an important mechanism in both invasion and metastasis [90, 91]. ECM stiffness has been shown to promote EMT, through which cancer cells acquire a migratory phenotype via a variety of pathways, some of which include key players in mechanotransduction, such as RTKs [92]. In one pathway, EMT results from stiffness-mediated localization and signaling of Rac GTPases downstream of MMPs [93]. Mechanical stress and matrix rigidity can also induce EMT downstream of TGF- β [94, 95]. Furthermore, the activation of Rho GTPases is thought to contribute to EMT via the loss of adherens junctions between cells and the gain of mesenchymal characteristics [96].

Induction of EMT in tumor cells, which affects cytoskeletal organization and cell-cell and cell-matrix adhesions, can also alter how the cells sense exogenous forces, and therefore their responses to those forces [97, 98]. The downregulation of epithelial keratins results in reduced cytoskeletal stiffness and greater cell deformability, directly influencing the metastatic potential of tumor cells [99]. In addition to being more deformable than non-metastatic cells, metastatic cells also lose their anchorage dependence [100, 101]. Anoikis, or apoptosis induced by the loss of adhesion to the ECM, is suppressed in metastatic cells, allowing them to migrate and traverse through the bloodstream to distant organs [102, 103]. Anoikis is believed to be mediated by integrin signaling [104]. The activation of integrins and their associated proteins, including FAK and integrin-linked kinase (ILK), can suppress anoikis, indicating that mechanotransduction and apoptotic pathways are linked [105]. EMT can also suppress anoikis [106]. In particular, the downregulation of E-cadherin can protect cells against anoikis [107]. It is clear that several extracellular and intracellular components of mechanotransduction are altered in tumors, which promotes progression to invasive disease. Mechanotransduction, it seems, is another mechanism that can be hijacked to support malignant transformation.

Mechanotransduction and Genomic Instability

The term genomic instability (GIN) broadly describes the inability of a cell to pass on a copy of its DNA with fidelity. GIN can manifest itself in several ways, each the result of replicative stress caused by errors in DNA replication or the DNA damage response [108]. Microsatellite instability is the expansion or contraction of oligonucleotide repeats and results from mutations in mismatch repair genes [109, 110]; nucleotide excision-repair-related instability results from an impaired ability of the cell to remove and replace damaged nucleotides [111]; and chromosomal instability is a change in the structure or number of chromosomes, which typically occurs as a result of errors in DNA replication or mitosis [112, 113].

GIN is a defining feature of cancers, and is believed to be the driving force behind tumor progression. Various errors in DNA replication or repair processes lead to an abnormal genotype that continues to change with each generation of cells. As a result of GIN, tumors that originate from the same tissue and cell type can have wildly varying genetic profiles [114]. This intertumor heterogeneity, as well as subclonal heterogeneity within a single tumor, has been largely attributed to the Darwinian characteristics of cancer; that is, the evolution and adaptation of a cancer clone in response to external selective pressures [115]. Ultimately, this results in the acquisition of survival-enhancing features that allow a cancer to develop. The local microenvironment is one source of pressure that results in GIN [116] and increased survival. Mouse embryonic stem cells exposed to radiation develop a high frequency of mutation *in vivo* but not in culture, suggesting that the microenvironment of the cells contributed to their development [117]. More specifically, both physical features of the tumor microenvironment as well as onslaughts by external agents have been shown to increase the frequency of mutation, thus increasing the chances that one of these mutations will affect maintenance of genomic integrity. Hypoxia is one hallmark characteristic of the tumor microenvironment known to play a role in promoting GIN. Hypoxia induces an elevated frequency of mutation in tumorigenic mammalian cell lines [118]. Similarly, exposure to heat and serumstarvation increases mutations in mouse mammary carcinoma cells [119]. Little is known about how GIN may arise from mechanical aspects of the microenvironment; the following describes a body of work that supports this idea.

Mechanical Forces Affect Mitosis and Cell Cycle Progression

One risk factor for the development of GIN is an increase in cellular proliferation, and hence the chance for DNA copy errors to arise. Recently, the mechanical properties of the microenvironment have been considered a major factor in its influence on cell behavior, specifically the regulation of cell cycle progression and mitosis and subsequent maintenance of the genome. Several studies have shown that modulating mechanical forces acting on cells can affect proliferation: mechanical stretch can reduce proliferation of podocytes [120], enhance differentiation and reduce proliferation of preadipocytes [121], and in endothelial cells, directed mechanical forces (specifically, shear and stretch) promote homeostasis but non-uniform forces can result in sustained pro-inflammatory and proliferative signaling [122]. These effects can be mediated by cell-cell contact, such as through VE-cadherin in endothelial cells [123].

The adhesion of a cell to its surroundings can alone induce changes in proliferation. Micropatterning techniques have been used to isolate the effects of cell spreading and cell-cell junctions from the effects of substratum adhesion on cell behavior. Such studies have revealed that E-cadherin is sufficient to induce epithelial cell proliferation via Rac1 signaling, and both proteins are required for cell-cell contact-dependent proliferation [124]. Similar findings hold for endothelial or smooth muscle cells via PI3K signaling [125]. Cytoskeletal structure and associated signaling have also proven to be important in cell-cell adhesion-mediated proliferation, based on studies regarding the role of VE-cadherin in vascular endothelial cells [126]. Additionally, simply varying the nature of the substratum also affects proliferative behavior. The basement membrane interacts differently with normal or cancerous epithelial cell lines, affecting growth and differentiation [127].

There is also evidence that mechanotransduction can influence various aspects of mitosis, and thus the segregation of the genome into daughter cells. Physical features of the microenvironment are one avenue of mechanical influence on mitosis. For example, in HeLa cells (human cervical cancer cells), retraction fibers, which bind mitotic cells to the substratum, exert forces on the cell that dictate the orientation of the spindle during mitosis. This is mediated by regulation of the subcortical actin network [128]. Another study in HeLa cells similarly showed that the spatial distribution of ECM proteins helps determine the axis of division by regulating actin dynamics [129].

It would follow from these studies that mechanosensors and other intracellular mechanotransduction machinery are involved in the regulation of mitosis, and indeed this has been shown. Integrin-mediated adhesion is required for the cells to reorient the mitotic spindle parallel to the substratum [130]. Here again, cytoskeletal components are key communicators. G proteins and the motor protein dynein, both important in transmitting mechanical force, are also known to direct orientation of the spindle in development [131]. One can imagine that abnormal mechanical signaling, common to many diseases including cancer, could disrupt mitosis in a cell and thus generate genomically unstable progeny.

Mechanotransduction Regulates Biochemical Cues That Promote GIN

One way that mechanical stimuli ultimately promote changes in cell behavior is through intracellular signaling pathways that conclude with control of gene transcription. Genes regulated by mechanotransduction can affect a myriad of both normal and pathological processes in the body [14]. In the context of cancer, recent studies have suggested that important molecular targets of mechanotransduction include mitotic checkpoint genes and other cell-cycle regulators, which have long been associated with maintaining genomic stability [112, 132].

To discover mechanically-regulated genes associated with GIN, several studies have used polyacrylamide gels of varying stiffness to mimic the mechanical properties of the ECM, and thus determine the effects of substratum stiffness on cell behavior in culture [133]. Recent findings from these experiments show that the transcription factors YAP (Yes-associated protein) and TAZ (transcriptional coactivator with PDZ-binding motif), which have implications in growth, proliferation, and differentiation, become activated in response to cytoskeletal tension and cell spreading induced by a stiff substratum [134]. In human mammary epithelial cells, expression of tumor suppressor phosphatase and tensin homolog (PTEN) is reduced in the presence of microRNA miR-18a, which is modulated by ECM stiffness [135]. PTEN is antagonistic to PI3K, a protein involved in many pathways important for cell growth and survival that promotes cancer when misregulated [136]. Polyacrylamide gels were also used to show that matrix rigidity induces integrin clustering in mammary epithelial cells, which induces the formation of focal adhesions and generates cytoskeletal tension. This in turn activates ERK and enhances EGFdependent pathways that activate ERK, which is known for its involvement in cell cycle regulation [137, 64].

Other cell cycle-regulators are activated by adhesion to or disruption of the substratum. The protein p38 is best known for its role as a tumor suppressor, but also regulates mitotic entry and the spindle assembly checkpoint [138], and negatively regulates cell proliferation through a reactive oxygen species (ROS)-mediated response to stress [139]. When mammary epithelial cells lose adhesion to the substratum, p38 is activated and can induce apoptosis [140]. NM23-H1 is another protein associated with growth arrest, and this function was shown to be correlated with basement membrane assembly in human breast cancer cells [141].

Aside from the effects of mechanically-regulated gene transcription, cytokines and other signaling factors that contribute to cancer progression are often triggered by mechanical forces, and can induce GIN. For example, lung cancer cells show an increased production of ROS in response to shear stress [142]. ROS are well known to promote genetic mutations and cancer progression [143]. Furthermore, a xenograft of human skin overexpressing bFGF (in a cocktail with stem cell factor and endothelin-3) causes replication stress [144], the major source of GIN [108]. As previously described, bFGF is regulated by shear stress and hydrostatic pressure [23, 22].

Restructuring of the Stroma Results in GIN

In addition to signaling mediated by mechanosensors, cells can communicate with the microenvironment through various soluble factors that serve to restructure the surrounding stroma. In cancer, misregulation of these proteins has been linked to GIN. MMPs make up one class of proteins that remodel the ECM. Overexpression of MMPs can induce cell cycle progression, activate genotoxic pathways, and inhibit cytokinesis [145]. Furthermore, cells overexpressing MMPs often exhibit patterns of genomic irregularities [146]. The stroma is also heavily remodeled during the formation of new vasculature. Both cyclic and constant static stretch of endothelial cells increase the expression of vascular endothelial growth factor (VEGF) receptor and promote VEGF-induced proliferation, vasculogenesis, and angiogenesis [147]. VEGF has been shown to regulate the axis of division in endothelial cells, potentiating GIN [148]. Thus, through restructuring of the stroma, in addition to control of the cell cycle and associated proteins and cytokines by external forces, GIN is mediated by mechanotransduction in cancer cells.

Synopsis and Outlook

Aberrant mechanotransduction is a major contributor to tumor progression, metastasis, and GIN. Both mechanosensing and subsequent intracellular signaling alter properties of the cell that can lead to malignant transformation in cancer. Mechanotransduction is therefore important to study in order to understand the progression of this disease. Developing improved 2D and 3D cell culture models to mimic the tumor microenvironment will enable us to determine the effects of abnormal mechanotransduction in cancer progression. Beyond experimental models, computational models can characterize the effects of mechanical stretch on cell behavior [121]. Others begin to account for intratumor heterogeneity when predicting therapeutic response [149]. However, current computational models cannot cope with mutational frequency of cancer cells, and thus there is a disconnect between investigations of the causes and consequences of this feature.

Although many of the proteins involved in mechanotransduction are known (e.g. integrins, cytoskeleton, myosins, kinases), the precise mechanisms by which a cell perceives the mechanical information of its environment remain unclear [150]. In addition, mechanical forces in the microenvironment are known to affect the cell cycle, and abnormal expression of cell-cycle regulators can result in GIN [132]; however, a clear mechanotransduction pathway linking these two events has not been elucidated. Similarly, current knowledge on the mechanosensing capabilities of stem cells is limited; verifying which forces, molecular pathways, and mechanosensing proteins are most important in directing construction of the stem cell niche and stem cells [151, 152].

Components of mechanotransduction pathways are starting to be considered as potential therapeutic targets. For example, it has been shown that the disruption of Rho or ERK signaling results in a reduction of cytoskeletal tension that leads to a decrease in tumor cell proliferation and the repression of malignant progression [16, 64]. Targeting Src activity could reduce proliferation, invasion, and metastasis [153]. Restoring anoikis response might curb metastasis [154], and the inhibition of collagen crosslinking and integrin signaling might reduce invasion. In addition, the mechanical properties of isolated metastatic cancer cells could be diagnostic indicators for prognosis. As we broaden our current understanding of mechanotransduction as it relates to both normal cell functions and disease, we will be able to integrate this knowledge into a synergistic treatment strategy for cancer.

References

- 1. Hanahan D, Weinberg RA (2000) The hallmarks of cancer. Cell 100(1):57-70
- Hanahan D, Weinberg RA (2011) Hallmarks of cancer: the next generation. Cell 144(5):646–674
 Ingber DE (2008) Tensegrity-based mechanosensing from macro to micro. Prog Biophys Mol
- Biol 97(2–3):163–179 4. Fang H, Declerck YA (2013) Targeting the tumor microenvironment: from understanding path-
- ways to effective clinical trials. Cancer Res 73(16):4965–4977
 5. Fidler IJ, Poste G (2008) The "seed and soil" hypothesis revisited. Lancet Oncol 9(8):808
- Wang JH, Thampatty BP (2006) An introductory review of cell mechanobiology. Biomech Model Mechanobiol 5(1):1–16
- Huang S, Ingber DE (2005) Cell tension, matrix mechanics, and cancer development. Cancer Cell 8(3):175–176
- Friedland JC, Lee MH, Boettiger D (2009) Mechanically activated integrin switch controls alpha5beta1 function. Science 323(5914):642–644

- Paszek MJ, Boettiger D, Weaver VM, Hammer DA (2009) Integrin clustering is driven by mechanical resistance from the glycocalyx and the substrate. PLoS Comput Biol 5(12):e1000604
- 10. Ma Z, Finkel TH (2010) T cell receptor triggering by force. Trends Immunol 31(1):1-6
- 11. Pelham RJ Jr, Wang Y (1997) Cell locomotion and focal adhesions are regulated by substrate flexibility. Proc Natl Acad Sci U S A 94(25):13661–13665
- Yeung T, Georges PC, Flanagan LA, Marg B, Ortiz M, Funaki M, Zahir N, Ming W, Weaver V, Janmey PA (2005) Effects of substrate stiffness on cell morphology, cytoskeletal structure, and adhesion. Cell Motil Cytoskeleton 60(1):24–34
- Assoian RK, Klein EA (2008) Growth control by intracellular tension and extracellular stiffness. Trends Cell Biol 18(7):347–352
- 14. Orr AW, Helmke BP, Blackman BR, Schwartz MA (2006) Mechanisms of mechanotransduction. Dev Cell 10(1):11–20
- Vogel V, Sheetz M (2006) Local force and geometry sensing regulate cell functions. Nat Rev Mol Cell Biol 7(4):265–275
- Jaalouk DE, Lammerding J (2009) Mechanotransduction gone awry. Nat Rev Mol Cell Biol 10(1):63–73
- 17. Discher DE, Janmey P, Wang YL (2005) Tissue cells feel and respond to the stiffness of their substrate. Science 310(5751):1139–1143
- Ingber DE (2006) Cellular mechanotransduction: putting all the pieces together again. FASEB J 20(7):811–827
- Resnick N, Yahav H, Shay-Salit A, Shushy M, Schubert S, Zilberman LC, Wofovitz E (2003) Fluid shear stress and the vascular endothelium: for better and for worse. Prog Biophys Mol Biol 81(3):177–199
- 20. Wang JH, Goldschmidt-Clermont P, Wille J, Yin FC (2001) Specificity of endothelial cell reorientation in response to cyclic mechanical stretching. J Biomech 34(12):1563–1572
- 21. Wells RG, Discher DE (2008) Matrix elasticity, cytoskeletal tension, and TGF-beta: the insoluble and soluble meet. Sci Signal 1(10):pe13
- 22. Gloe T, Sohn HY, Meininger GA, Pohl U (2002) Shear stress-induced release of basic fibroblast growth factor from endothelial cells is mediated by matrix interaction via integrin alpha(v)beta3. J Biol Chem 277(26):23453–23458
- Acevedo AD, Bowser SS, Gerritsen ME, Bizios R (1993) Morphological and proliferative responses of endothelial cells to hydrostatic pressure: role of fibroblast growth factor. J Cell Physiol 157(3):603–614
- Yang JH, Sakamoto H, Xu EC, Lee RT (2000) Biomechanical regulation of human monocyte/macrophage molecular function. Am J Pathol 156(5):1797–1804
- Ingber DE (1997) Tensegrity: the architectural basis of cellular mechanotransduction. Annu Rev Physiol 59: 575–599
- 26. Mih JD, Marinkovic A, Liu F, Sharif AS, Tschumperlin DJ (2012) Matrix stiffness reverses the effect of actomyosin tension on cell proliferation. J Cell Sci 125(Pt 24):5974–5983
- Wang J, Su M, Fan J, Seth A, McCulloch CA (2002) Transcriptional regulation of a contractile gene by mechanical forces applied through integrins in osteoblasts. J Biol Chem 277(25):22889–22895
- Solon J, Levental I, Sengupta K, Georges PC, Janmey PA (2007) Fibroblast adaptation and stiffness matching to soft elastic substrates. Biophys J 93(12):4453–4461
- Fettiplace R, Hackney CM (2006) The sensory and motor roles of auditory hair cells. Nat Rev Neurosci 7(1):19–29
- Praetorius HA, Spring KR (2005) A physiological view of the primary cilium. Annu Rev Physiol 67:515–529
- Juliano RL, Haskill S (1993) Signal transduction from the extracellular matrix. J Cell Biol 120(3):577–585
- Guo W, Giancotti FG (2004) Integrin signalling during tumour progression. Nat Rev Mol Cell Biol 5(10):816–826
- Ross TD, Coon BG, Yun S, Baeyens N, Tanaka K, Ouyang M, Schwartz MA (2013) Integrins in mechanotransduction. Curr Opin Cell Biol 25(5):613–618

- 7 Mechanotransduction, Metastasis and Genomic Instability
- Schmidt CE, Horwitz AF, Lauffenburger DA, Sheetz MP (1993) Integrin-cytoskeletal interactions in migrating fibroblasts are dynamic, asymmetric, and regulated. J Cell Biol 123(4):977–991
- 35. Urbich C, Dernbach E, Reissner A, Vasa M, Zeiher AM, Dimmeler S (2002) Shear stressinduced endothelial cell migration involves integrin signaling via the fibronectin receptor subunits alpha(5) and beta(1). Arterioscler Thromb Vasc Biol 22(1):69–75
- Wang N, Butler JP, Ingber DE (1993) Mechanotransduction across the cell surface and through the cytoskeleton. Science 260(5111):1124–1127
- Burridge K, Chrzanowska-Wodnicka M (1996) Focal adhesions, contractility, and signaling. Annu Rev Cell Dev Biol 12:463–518
- Seong J, Wang N, Wang Y (2013) Mechanotransduction at focal adhesions: from physiology to cancer development. J Cell Mol Med 17(5):597–604
- Goldmann WH (2012) Mechanotransduction and focal adhesions. Cell Biol Int 36(7):649– 652
- Govey PM, Jacobs JM, Tilton SC, Loiselle AE, Zhang Y, Freeman WM, Waters KM, Karin NJ, Donahue HJ (2014) Integrative transcriptomic and proteomic analysis of osteocytic cells exposed to fluid flow reveals novel mechano-sensitive signaling pathways. J Biomech 47(8):1838–1845
- Gudi SR, Clark CB, Frangos JA (1996) Fluid flow rapidly activates g proteins in human endothelial cells. Involvement of G proteins in mechanochemical signal transduction. Circ Res 79(4):834–839
- 42. Gudi SR, Lee AA, Clark CB, Frangos JA (1998) Equibiaxial strain and strain rate stimulate early activation of G proteins in cardiac fibroblasts. Am J Physiol 274(5 Pt 1):C1424–1428
- 43. Ullrich A, Schlessinger J (1990) Signal transduction by receptors with tyrosine kinase activity. Cell 61(2):203–212
- Cobb MH, Robbins DJ, Boulton TG (1991) Erks, extracellular signal-regulated map-2 kinases. Curr Opin Cell Biol 3(6):1025–1032
- Chess PR, Toia L, Finkelstein JN (2000) Mechanical strain-induced proliferation and signaling in pulmonary epithelial h441 cells. Am J Physiol Lung Cell Mol Physiol 279(1):L43–L51
- 46. Jo H, Sipos K, Go YM, Law R, Rong J, McDonald JM (1997) Differential effect of shear stress on extracellular signal-regulated kinase and n-terminal jun kinase in endothelial cells. Gi2- and gbeta/gamma-dependent signaling pathways. J Biol Chem 272(2):1395–1401
- Pommerenke H, Schreiber E, Durr F, Nebe B, Hahnel C, Moller W, Rychly J (1996) Stimulation of integrin receptors using a magnetic drag force device induces an intracellular free calcium response. Eur J Cell Biol 70(2):157–164
- Shen J, Luscinskas FW, Connolly A, Dewey CF, Jr, Gimbrone MA Jr (1992) Fluid shear stress modulates cytosolic free calcium in vascular endothelial cells. Am J Physiol 262(2 Pt 1):C384–390
- Iwasaki H, Eguchi S, Ueno H, Marumo F, Hirata Y (2000) Mechanical stretch stimulates growth of vascular smooth muscle cells via epidermal growth factor receptor. Am J Physiol Heart Circ Physiol 278(2):H521–529
- Geiger TR, Peeper DS (2009) Metastasis mechanisms. Biochim Biophys Acta 1796(2): 293–308
- Mareel M, Leroy A (2003) Clinical, cellular, and molecular aspects of cancer invasion. Physiol Rev 83(2):337–376
- Tse JM, Cheng G, Tyrrell JA, Wilcox-Adelman SA, Boucher Y, Jain RK, Munn LL (2012) Mechanical compression drives cancer cells toward invasive phenotype. Proc Natl Acad Sci U S A 109(3):911–916
- Provenzano PP, Inman DR, Eliceiri KW, Knittel JG, Yan L, Rueden CT, White JG, Keely PJ (2008) Collagen density promotes mammary tumor initiation and progression. BMC Med 6:11
- Ronnov-Jessen L, Petersen OW, Bissell MJ (1996) Cellular changes involved in conversion of normal to malignant breast: importance of the stromal reaction. Physiol Rev 76(1):69–125
- 55. Paszek MJ, Weaver VM (2004) The tension mounts: mechanics meets morphogenesis and malignancy. J Mammary Gland Biol Neoplasia 9(4):325–342

- Reno F, Grazianetti P, Stella M, Magliacani G, Pezzuto C, Cannas M (2002) Release and activation of matrix metalloproteinase-9 during in vitro mechanical compression in hypertrophic scars. Arch Dermatol 138(4):475–478
- 57. Joshi HC, Chu D, Buxbaum RE, Heidemann SR (1985) Tension and compression in the cytoskeleton of pc 12 neurites. J Cell Biol 101(3):697–705
- Levental KR, Yu H, Kass L, Lakins JN, Egeblad M, Erler JT, Fong SF, Csiszar K, Giaccia A, Weninger W, Yamauchi M, Gasser DL, Weaver VM (2009) Matrix crosslinking forces tumor progression by enhancing integrin signaling. Cell 139(5):891–906
- 59. Suresh S (2007) Biomechanics and biophysics of cancer cells. Acta Biomater 3(4):413-438
- Petrie RJ, Yamada KM (2012) At the leading edge of three-dimensional cell migration. J Cell Sci 125(Pt 24):5917–5926
- McCawley LJ, Matrisian LM (2001) Matrix metalloproteinases: they're not just for matrix anymore! Curr Opin Cell Biol 13(5):534–540
- Balaban NQ, Schwarz US, Riveline D, Goichberg P, Tzur G, Sabanay I, Mahalu D, Safran S, Bershadsky A, Addadi L, Geiger B (2001) Force and focal adhesion assembly: a close relationship studied using elastic micropatterned substrates. Nat Cell Biol 3(5):466–472
- 63. Krouskop TA, Wheeler TM, Kallel F, Garra BS, Hall T (1998) Elastic moduli of breast and prostate tissues under compression. Ultrason Imaging 20(4):260–274
- Paszek MJ, Zahir N, Johnson KR, Lakins JN, Rozenberg GI, Gefen A, Reinhart-King CA, Margulies SS, Dembo M, Boettiger D, Hammer DA, Weaver VM (2005) Tensional homeostasis and the malignant phenotype. Cancer Cell 8(3):241–254
- 65. Gray DS, Tien J, Chen CS (2003) Repositioning of cells by mechanotaxis on surfaces with micropatterned young's modulus. J Biomed Mater Res A 66(3):605–614
- Lo CM, Wang HB, Dembo M, Wang YL (2000) Cell movement is guided by the rigidity of the substrate. Biophys J 79(1):144–152
- Wolf K, Wu YI, Liu Y, Geiger J, Tam E, Overall C, Stack MS, Friedl P (2007) Multi-step pericellular proteolysis controls the transition from individual to collective cancer cell invasion. Nat Cell Biol 9(8):893–904
- Friedl P, Wolf K (2003) Tumour-cell invasion and migration: diversity and escape mechanisms. Nat Rev Cancer 3(5):362–374
- Provenzano PP, Eliceiri KW, Campbell JM, Inman DR, White JG, Keely PJ (2006) Collagen reorganization at the tumor-stromal interface facilitates local invasion. BMC Med 4(1):38
- Wolf K, Friedl P (2009) Mapping proteolytic cancer cell-extracellular matrix interfaces. Clin Exp Metastasis 26(4):289–298
- Wolf K, Mazo I, Leung H, Engelke K, von Andrian UH, Deryugina EI, Strongin AY, Brocker EB, Friedl P (2003) Compensation mechanism in tumor cell migration: mesenchymal-amoeboid transition after blocking of pericellular proteolysis. J Cell Biol 160(2):267–277
- 72. Friedl P, Wolf K (2010) Plasticity of cell migration: a multiscale tuning model. J Cell Biol 188(1):11–19
- 73. Haessler U, Teo JC, Foretay D, Renaud P, Swartz MA (2012) Migration dynamics of breast cancer cells in a tunable 3D interstitial flow chamber. Integr Biol (Camb) 4(4):401–409
- Polacheck WJ, German AE, Mammoto A, Ingber DE, Kamm RD (2014) Mechanotransduction of fluid stresses governs 3D cell migration. Proc Natl Acad Sci U S A 111(7):2447–2452
- 75. Hebner C, Weaver VM, Debnath J (2008) Modeling morphogenesis and oncogenesis in three-dimensional breast epithelial cultures. Annu Rev Pathol 3:313–339
- Horiuchi A, Imai T, Wang C, Ohira S, Feng Y, Nikaido T, Konishi I (2003) Up-regulation of small GTPases, RhoA and RhoC, is associated with tumor progression in ovarian carcinoma. Lab Invest 83(6):861–870
- 77. Sahai E, Marshall CJ (2003) Differing modes of tumour cell invasion have distinct requirements for Rho/rock signalling and extracellular proteolysis. Nat Cell Biol 5 (8):711–719
- 78. Ridley AJ (2001) Rho GTPases and cell migration. J Cell Sci 114(Pt 15):2713–2722
- 79. Sahai E, Marshall CJ (2002) Rho-GTPases and cancer. Nat Rev Cancer 2(2):133-142
- Lozano E, Betson M, Braga VM (2003) Tumor progression: small GTPases and loss of cellcell adhesion. Bioessays 25 (5):452–463

- 7 Mechanotransduction, Metastasis and Genomic Instability
 - Pollard TD, Borisy GG (2003) Cellular motility driven by assembly and disassembly of actin filaments. Cell 112(4):453–465
 - Wang W, Eddy R, Condeelis J (2007) The cofilin pathway in breast cancer invasion and metastasis. Nat Rev Cancer 7(6):429–440
 - Parsons SJ, Parsons JT (2004) Src family kinases, key regulators of signal transduction. Oncogene 23(48):7906–7909
 - Thomas SM, Brugge JS (1997) Cellular functions regulated by Src family kinases. Annu Rev Cell Dev Biol 13:513–609
 - Kelley LC, Ammer AG, Hayes KE, Martin KH, Machida K, Jia L, Mayer BJ, Weed SA (2010) Oncogenic Src requires a wild-type counterpart to regulate invadopodia maturation. J Cell Sci 123(Pt 22):3923–3932
 - Polackwich RJ, Koch D, Arevalo R, Miermont AM, Jee KJ, Lazar J, Urbach J, Mueller SC, McAllister RG (2013) A novel 3d fibril force assay implicates Src in tumor cell force generation in collagen networks. PLoS ONE 8(3):e58138
 - Guck J, Schinkinger S, Lincoln B, Wottawah F, Ebert S, Romeyke M, Lenz D, Erickson HM, Ananthakrishnan R, Mitchell D, Kas J, Ulvick S, Bilby C (2005) Optical deformability as an inherent cell marker for testing malignant transformation and metastatic competence. Biophys J 88(5):3689–3698
 - Plantefaber LC, Hynes RO (1989) Changes in integrin receptors on oncogenically transformed cells. Cell 56(2):281–290
 - Ochalek T, Nordt FJ, Tullberg K, Burger MM (1988) Correlation between cell deformability and metastatic potential in b16-f1 melanoma cell variants. Cancer Res 48(18):5124– 5128
 - Kalluri R, Weinberg RA (2009) The basics of epithelial-mesenchymal transition. J Clin Invest 119(6):1420–1428
 - 91. Thiery JP (2002) Epithelial-mesenchymal transitions in tumour progression. Nat Rev Cancer 2(6):442–454
 - 92. Huber MA, Kraut N, Beug H (2005) Molecular requirements for epithelial-mesenchymal transition during tumor progression. Curr Opin Cell Biol 17(5):548–558
 - Lee K, Chen QK, Lui C, Cichon MA, Radisky DC, Nelson CM (2012) Matrix compliance regulates Rac1b localization, NADPH oxidase assembly, and epithelial-mesenchymal transition. Mol Biol Cell 23(20):4097–4108
 - Gomez EW, Chen QK, Gjorevski N, Nelson CM (2010) Tissue geometry patterns epithelial-mesenchymal transition via intercellular mechanotransduction. J Cell Biochem 110(1):44–51
 - Leight JL, Wozniak MA, Chen S, Lynch ML, Chen CS (2012) Matrix rigidity regulates a switch between tgf-beta1-induced apoptosis and epithelial-mesenchymal transition. Mol Biol Cell 23(5):781–791
 - 96. Bhowmick NA, Ghiassi M, Bakin A, Aakre M, Lundquist CA, Engel ME, Arteaga CL, Moses HL (2001) Transforming growth factor-beta1 mediates epithelial to mesenchymal transdifferentiation through a RhoA-dependent mechanism. Mol Biol Cell 12(1):27–36
 - 97. Huang H, Kamm RD, Lee RT (2004) Cell mechanics and mechanotransduction: pathways, probes, and physiology. Am J Physiol Cell Physiol 287(1):C1–11
 - Janmey PA, Weitz DA (2004) Dealing with mechanics: mechanisms of force transduction in cells. Trends Biochem Sci 29(7):364–370
- Seltmann K, Fritsch AW, Kas JA, Magin TM (2013) Keratins significantly contribute to cell stiffness and impact invasive behavior. Proc Natl Acad Sci U S A 110(46):18507–18512
- Huang S, Ingber DE (1999) The structural and mechanical complexity of cell-growth control. Nat Cell Biol 1(5):E131–E138
- Wang HB, Dembo M, Wang YL (2000) Substrate flexibility regulates growth and apoptosis of normal but not transformed cells. Am J Physiol Cell Physiol 279(5):C1345–C1350
- Geiger TR, Peeper DS (2005) The neurotrophic receptor trkb in anoikis resistance and metastasis: a perspective. Cancer Res 65(16):7033–7036

- Zhu Z, Sanchez-Sweatman O, Huang X, Wiltrout R, Khokha R, Zhao Q, Gorelik E (2001) Anoikis and metastatic potential of cloudman s91 melanoma cells. Cancer Res 61(4):1707–1716
- 104. Frisch SM, Ruoslahti E (1997) Integrins and anoikis. Curr Opin Cell Biol 9(5):701-706
- Attwell S, Roskelley C, Dedhar S (2000) The integrin-linked kinase (ilk) suppresses anoikis. Oncogene 19(33):3811–3815
- Frisch SM, Schaller M, Cieply B (2013) Mechanisms that link the oncogenic epithelialmesenchymal transition to suppression of anoikis. J Cell Sci 126(Pt 1):21–29
- 107. Derksen PW, Liu X, Saridin F, van der Gulden H, Zevenhoven J, Evers B, van Beijnum JR, Griffioen AW, Vink J, Krimpenfort P, Peterse JL, Cardiff RD, Berns A, Jonkers J (2006) Somatic inactivation of e-cadherin and p53 in mice leads to metastatic lobular mammary carcinoma through induction of anoikis resistance and angiogenesis. Cancer Cell 10(5):437–449
- 108. Aguilera A, Garcia-Muse T (2013) Causes of genome instability. Annu Rev Genet 47:1-32
- Fishel R, Lescoe MK, Rao MR, Copeland NG, Jenkins NA, Garber J, Kane M, Kolodner R (1993) The human mutator gene homolog msh2 and its association with hereditary nonpolyposis colon cancer. Cell 75(5):1027–1038
- Thibodeau SN, Bren G, Schaid D (1993) Microsatellite instability in cancer of the proximal colon. Science 260(5109):816–819
- 111. Al-Tassan N, Chmiel NH, Maynard J, Fleming N, Livingston AL, Williams GT, Hodges AK, Davies DR, David SS, Sampson JR, Cheadle JP (2002) Inherited variants of myh associated with somatic G:C → T:A mutations in colorectal tumors. Nat Genet 30(2):227–232
- Coschi CH, Dick FA (2012) Chromosome instability and deregulated proliferation: an unavoidable duo. Cell Mol Life Sci 69(12):2009–2024
- Negrini S, Gorgoulis VG, Halazonetis TD (2010) Genomic instability—an evolving hallmark of cancer. Nat Rev Mol Cell Biol 11(3):220–228
- Burrell RA, McGranahan N, Bartek J, Swanton C (2013) The causes and consequences of genetic heterogeneity in cancer evolution. Nature 501(7467):338–345
- 115. Greaves M, Maley CC (2012) Clonal evolution in cancer. Nature 481(7381):306–313
- Bindra RS, Glazer PM (2005) Genetic instability and the tumor microenvironment: towards the concept of microenvironment-induced mutagenesis. Mutat Res 569(1–2):75–85
- 117. Paquette B, Little JB (1994) In vivo enhancement of genomic instability in minisatellite sequences of mouse c3h/10t1/2 cells transformed in vitro by x-rays. Cancer Res 54(12):3173–3178
- Reynolds TY, Rockwell S, Glazer PM (1996) Genetic instability induced by the tumor microenvironment. Cancer Res 56(24):5754–5757
- Li CY, Little JB, Hu K, Zhang W, Zhang L, Dewhirst MW, Huang Q (2001) Persistent genetic instability in cancer cells induced by non-DNA-damaging stress exposures. Cancer Res 61(2):428–432
- Petermann AT, Hiromura K, Blonski M, Pippin J, Monkawa T, Durvasula R, Couser WG, Shankland SJ (2002) Mechanical stress reduces podocyte proliferation in vitro. Kidney Int 61(1):40–50
- Shoham N, Gefen A (2012) The influence of mechanical stretching on mitosis, growth, and adipose conversion in adipocyte cultures. Biomech Model Mechanobiol 11(7):1029–1045
- 122. Chien S (2007) Mechanotransduction and endothelial cell homeostasis: the wisdom of the cell. Am J Physiol Heart Circ Physiol 292(3):H1209–1224
- Liu WF, Nelson CM, Tan JL, Chen CS (2007) Cadherins, RhoA, and Rac1 are differentially required for stretch-mediated proliferation in endothelial versus smooth muscle cells. Circ Res 101(5):e44–52
- Liu WF, Nelson CM, Pirone DM, Chen CS (2006) E-cadherin engagement stimulates proliferation via Rac1. J Cell Biol 173(3):431–441
- Nelson CM, Chen CS (2002) Cell-cell signaling by direct contact increases cell proliferation via a PI3K-dependent signal. FEBS Lett 514(2–3):238–242
- Nelson CM, Chen CS (2003) Ve-cadherin simultaneously stimulates and inhibits cell proliferation by altering cytoskeletal structure and tension. J Cell Sci 116(Pt 17):3571–3581

- 7 Mechanotransduction, Metastasis and Genomic Instability
- 127. Petersen OW, Ronnov-Jessen L, Howlett AR, Bissell MJ (1992) Interaction with basement membrane serves to rapidly distinguish growth and differentiation pattern of normal and malignant human breast epithelial cells. Proc Natl Acad Sci U S A 89(19):9064–9068
- Fink J, Carpi N, Betz T, Betard A, Chebah M, Azioune A, Bornens M, Sykes C, Fetler L, Cuvelier D, Piel M (2011) External forces control mitotic spindle positioning. Nat Cell Biol 13(7):771–778
- Thery M, Racine V, Pepin A, Piel M, Chen Y, Sibarita JB, Bornens M (2005) The extracellular matrix guides the orientation of the cell division axis. Nat Cell Biol 7(10):947–953
- Toyoshima F, Nishida E (2007) Integrin-mediated adhesion orients the spindle parallel to the substratum in an eb1- and myosin x-dependent manner. EMBO J 26(6):1487–1498
- Ahringer J (2003) Control of cell polarity and mitotic spindle positioning in animal cells. Curr Opin Cell Biol 15(1):73–81
- Cahill DP, Lengauer C, Yu J, Riggins GJ, Willson JK, Markowitz SD, Kinzler KW, Vogelstein B (1998) Mutations of mitotic checkpoint genes in human cancers. Nature 392(6673):300–303
- Pelham RJ, Jr., Wang YL (1998) Cell locomotion and focal adhesions are regulated by the mechanical properties of the substrate. Biol Bull 194(3):348–349. (discussion 349–350)
- Dupont S, Morsut L, Aragona M, Enzo E, Giulitti S, Cordenonsi M, Zanconato F, Le Digabel J, Forcato M, Bicciato S, Elvassore N, Piccolo S (2011) Role of yap/taz in mechanotransduction. Nature 474(7350):179–183
- 135. Mouw JK, Yui Y, Damiano L, Bainer RO, Lakins JN, Acerbi I, Ou G, Wijekoon AC, Levental KR, Gilbert PM, Hwang ES, Chen YY, Weaver VM (2014) Tissue mechanics modulate microrna-dependent pten expression to regulate malignant progression. Nat Med 20(4):360–367
- Krasilnikov MA (2000) Phosphatidylinositol-3 kinase dependent pathways: the role in control of cell growth, survival, and malignant transformation. Biochemistry (Mosc) 65(1):59–67
- 137. McCubrey JA, Steelman LS, Chappell WH, Abrams SL, Wong EW, Chang F, Lehmann B, Terrian DM, Milella M, Tafuri A, Stivala F, Libra M, Basecke J, Evangelisti C, Martelli AM, Franklin RA (2007) Roles of the raf/mek/erk pathway in cell growth, malignant transformation and drug resistance. Biochim Biophys Acta 1773(8):1263–1284
- 138. Takenaka K, Moriguchi T, Nishida E (1998) Activation of the protein kinase p38 in the spindle assembly checkpoint and mitotic arrest. Science 280(5363):599–602
- Avivar-Valderas A, Wen HC, Aguirre-Ghiso JA (2014) Stress signaling and the shaping of the mammary tissue in development and cancer. Oncogene. doi:10.1038/onc.2013.554
- Wen HC, Avivar-Valderas A, Sosa MS, Girnius N, Farias EF, Davis RJ, Aguirre-Ghiso JA (2011) P38alpha signaling induces anoikis and lumen formation during mammary morphogenesis. Sci Signal 4(174):ra34
- 141. Howlett AR, Petersen OW, Steeg PS, Bissell MJ (1994) A novel function for the nm23-h1 gene: overexpression in human breast carcinoma cells leads to the formation of basement membrane and growth arrest. J Natl Cancer Inst 86(24):1838–1844
- Lo KY, Zhu Y, Tsai HF, Sun YS (2013) Effects of shear stresses and antioxidant concentrations on the production of reactive oxygen species in lung cancer cells. Biomicrofluidics 7(6):64108
- 143. Waris G, Ahsan H (2006) Reactive oxygen species: role in the development of cancer and various chronic conditions. J Carcinog 5:14
- 144. Gorgoulis VG, Vassiliou LV, Karakaidos P, Zacharatos P, Kotsinas A, Liloglou T, Venere M, Ditullio RA Jr, Kastrinakis NG, Levy B, Kletsas D, Yoneta A, Herlyn M, Kittas C, Halazonetis TD (2005) Activation of the DNA damage checkpoint and genomic instability in human precancerous lesions. Nature 434(7035):907–913
- Radisky DC, Bissell MJ (2006) Matrix metalloproteinase-induced genomic instability. Curr Opin Genet Dev 16(1):45–50
- 146. Radisky DC, Levy DD, Littlepage LE, Liu H, Nelson CM, Fata JE, Leake D, Godden EL, Albertson DG, Nieto MA, Werb Z, Bissell MJ (2005) Rac1b and reactive oxygen species mediate mmp-3-induced emt and genomic instability. Nature 436(7047):123–127

- 147. Zheng W, Christensen LP, Tomanek RJ (2008) Differential effects of cyclic and static stretch on coronary microvascular endothelial cell receptors and vasculogenic/angiogenic responses. Am J Physiol Heart Circ Physiol 295(2):H794–800
- Zeng G, Taylor SM, McColm JR, Kappas NC, Kearney JB, Williams LH, Hartnett ME, Bautch VL (2007) Orientation of endothelial cell division is regulated by VEGF signaling during blood vessel formation. Blood 109(4):1345–1352
- 149. Fedele C, Tothill RW, McArthur GA (2014) Navigating the challenge of tumor heterogeneity in cancer therapy. Cancer Discov 4(2):146–148
- Janmey PA, Miller RT (2011) Mechanisms of mechanical signaling in development and disease. J Cell Sci 124(Pt 1):9–18
- Liu YS, Lee OK (2014) In search of the pivot point of mechanotransduction: mechanosensing of stem cells. Cell Transplant 23(1):1–11
- 152. Hao J, Zhang Y, Ye R, Zheng Y, Zhao Z, Li J (2013) Mechanotransduction in cancer stem cells. Cell Biol Int 37(9):888–891
- 153. Gnoni A, Marech I, Silvestris N, Vacca A, Lorusso V (2011) Dasatinib: an anti-tumour agent via Src inhibition. Curr Drug Targets 12(4):563–578
- 154. Tan K, Goldstein D, Crowe P, Yang JL (2013) Uncovering a key to the process of metastasis in human cancers: a review of critical regulators of anoikis. J Cancer Res Clin Oncol 139(11):1795–1805
- 155. Chauhan et al. (2011) Delivery of molecular and nanoscale medicine to tumors: transport barriers and strategies. Annu Rev Chem Biomol Eng
- 156. Polacheck et al. (2011) Interstitial flow influences direction of tumor cell migration through competing mechanisms. PNAS
- 157. Tien et al. (2012) Modulation of invasive phenotype by interstitial pressure-driven convection in aggregates of human breast cancer cells. Plos One

Chapter 8 Immunomodulation and Genomic Instability

Gregor Reid

Abstract The interaction between a developing tumor and the immune system is complex and dynamic, and comprises seemingly opposing activities. On one hand, the tumor-promoting effect of chronic inflammation has long been recognized and mechanisms contributing to this activity, including proliferative and anti-apoptotic signaling, tissue remodeling, and mutagenesis, are well described. In contrast, tumor-specific immune responses mediated by a variety of cell types and soluble factors have been shown to inhibit the progression of cancer. A full understanding of the interplay between these opposing forces will be required before clinical manipulation of the tumor immune environment can achieve consistent improvement in the outcomes for patients with cancer. The focus of this chapter is the influence of genomic instability on the pro- and anti-tumor immune activities that impact on cancer development at multiple stages of progression.

Keywords Inflammation • Immunoediting • Immunosurveillance • Microsatellite instability • Escape variant • DDR • NKG2D • Innate • Adaptive • Antigens

Abbreviations

Activation-induced cytidine deaminase
Base excision repair
Chromosomal instability
Damage-associated molecular patterns
Dendritic cells
DNA damage response
Deoxyribonucleic acid
Immunoglobulin
Killer-cell immunoglobulin-like receptors
Type 1 macrophages
Myeloid derived suppressor cells
Major histocompatibility complex
Migration inhibitory factor

G. Reid (🖂)

Department of Pediatrics, University of British Columbia, Vancouver, BC V5Z 4H4, Canada e-mail: grogreid@mail.ubc.ca

[©] Springer International Publishing Switzerland 2015

C. Maxwell, C. Roskelley (eds.), *Genomic Instability and Cancer Metastasis*, Cancer Metastasis - Biology and Treatment 20, DOI 10.1007/978-3-319-12136-9_8

MMR	Mismatch repair
MSI	Microsatellite instability
MSI-H	High microsatellite instability
NK	Natural killer
TLR	Toll-like receptor
PAMP	Pathogen-associated molecular patterns
RONS	Reactive oxygen and nitrogen species
ROS	Reactive oxygen species
TAMS	Tumor-associated macrophages
Tfh	T follicular helper
T _{REGS}	Regulatory T cells

Introduction

Immune cells represent a significant component of the tumor microenvironment and as such can exert considerable influence over cancer progression. On one hand, it has been recognized since the observations of Rudolf Virchow in 1863 that cancer often arises at sites of prolonged inflammation, implicating immune mechanisms in the oncogenic process [1]. It is now estimated that approximately 25% of human cancers are associated with chronic inflammation and while mechanisms that may drive this process have been identified, the precise contribution of immune-mediated events to transformation and tumor progression remain incompletely understood [2].

In contrast to the tumorigenic effects of chronic inflammation proposed by Virchow, Paul Ehrlich postulated early in the twentieth century that the immune system could recognize and eliminate transformed cells and thus protect against the development of cancer [3]. Experimental support for immune recognition of cancer cells, obtained from the classical adoptive transfer studies of the 1950s, provided the foundation for Burnet and Thomas's seminal cancer immunosurveillance model, in which lymphocytes were responsible for reducing the incidence of cancer by eliminating continuously arising transformed cells [4]. Although this model has been extensively challenged for many years over the absence of empirical evidence, the basic premise has gained significant support from the results of experiments performed since the advent of molecularly defined immune-deficient mice [5, 6].

The clinical relevance of immune interactions with tumors, and insights into the mechanisms underlying the different progression scenarios, is provided by the strong correlation between immune cell infiltration and cancer prognosis. First reported for colorectal carcinoma [7–9], the influence of qualitative and quantitative differences in the composition of infiltrating immune cells on tumor progression is now recognized for several cancers [10–13]. In general, the presence of IFN- γ producing T cells at the invasive border and within the tumor stroma provides the greatest survival benefit, while a lack of infiltrating T cells correlates with poor prognosis. Indeed, in the case of colorectal cancer, the infiltration of T cells into the tumor is the most predictive for patient survival of all histologic and clinical criteria. With ever more sophisticated evaluations of immune cell infiltration of tumors being reported, it is apparent that there is considerable variation between tumors and that the interaction is far from static, evolving significantly throughout tumor progression [14].

Although we now know much more about the cellular and molecular mechanisms of an immune response than Virchow and Ehrlich, a unified model of the interaction between the host immune system and the developing tumor remains elusive. A primary goal of current tumor immunology research is to resolve the apparent contradiction between the oncogenic and immunosurveillance activities of the immune system. The success of this endeavor will lead to more effective strategies to manipulate the immune microenvironment to achieve better outcomes for cancer patients. This chapter focuses on the mechanisms that balance pro-tumor chronic inflammation, which may enhance genomic instability, and anti-tumor immune activity associated with cancer prevention and an improved prognosis.

Inflammation and Cancer

Inflammation is a normal physiological response to infection, irritation or tissue damage that is associated with extensive cytokine and chemokine signaling, vascular changes, immune cell infiltration, and tissue remodeling. Granulocytes are the predominant cells infiltrating sites of acute inflammation, which generally resolve in a few days. Chronic inflammation, dominated by macrophage infiltration, results from conditions of incomplete pathogen or irritant clearance, or unchecked inflammatory signaling, and predisposes the host to a range of pathologies, including cancer [2, 15].

A variety of underlying causes of chronic inflammation have been linked to cancer, including infectious agents (e.g. Hepatitis viruses for liver cancer and helicobacter pylori for gastric cancer) [16, 17], irritants (e.g. asbestos for lung cancer) [18], autoimmunity (e.g. inflammatory bowel disease for colorectal cancer) [19], as well as unidentified ones (e.g. prostatitis for prostate cancer) [20]. In these cases, the chronic inflammatory response initiates and promotes the transition from premalignant state to full transformation and thus increases the incidence of cancer by establishing conditions beneficial to the expansion of mutated cells. However, inflammation is also associated with cancers that do not arise at sites of chronic inflammation. In these cases, oncogenic signaling itself, in the absence of external stimuli, may initiate the chronic inflammatory response that generates conditions that promote tumor progression (e.g. RET oncogene for thyroid cancer) [21].

Acute inflammation is driven by cells of the innate immune system, in concert with endothelial and epithelial cells and fibroblasts, which detect infection or tissue damage through germline-encoded pattern recognition receptors. These receptors, including the toll-like receptor (TLR) family, bind to conserved motifs present on infectious agents, known as pathogen-associated molecular patterns (PAMPs), or local cell products, the damage-associated molecular patterns (DAMPs) [22].

Signaling via TLR is mediated through MyD88 and culminates in the activation of the NF-kb transcription factor, a central regulator of inflammation that induces the production of cytokines and chemokines, adhesion molecules, matrix metalloproteases, and other inflammatory mediators.

The production of cytokines and chemokines and changes in vascular permeability lead to the rapid accumulation of leukocytes, primarily neutrophils but also macrophages, natural killer (NK) and dendritic cells (DC), at the injured site [23]. Activation of these infiltrating cells by cytokines and PAMP/DAMP recognition perpetuates the inflammatory response through the continued production of mediators such as reactive oxygen species (ROS). ROS play a central role in the inflammatory response, functioning both as signaling molecules and as a host-defense mechanism [24]. In addition, activation of DCs leads to stimulation of adaptive immune responses through the presentation of antigens from the injured site to T cells in the draining lymph nodes [25, 26]. This later event culminates in the accumulation of activated lymphocytes and the exertion of adaptive immune effector mechanisms at the injured site, contributing to the eradication of the underlying insult and resolution of inflammation.

An inability to successfully resolve the initial inflammatory response can lead to the development of chronic inflammation. Lasting in some cases for years, chronic inflammation sites are dominated by infiltration of macrophages and lymphocytes. Chronic inflammation is sustained by the persistent activation of NF- κ B and STAT signaling pathways, leading to prolonged production of pro-inflammatory cytokines (e.g. TNF- α and IL-6) and reactive oxygen and nitrogen species (RONS) [27]. The resulting inflammatory milieu provides an environment supportive of cell survival and proliferation, angiogenesis and tissue remodeling. Such an environment is also conducive to the initiation, promotion, and progression of cancer [15, 28, 29].

The similarities between the immune environments generated by chronic inflammation in the presence or absence of cancer and tumors at sites previously devoid of inflammation suggest a mechanistic overlap in tumor promoting activities. The tumor-supportive immune environment is characterized the accumulation of M2polarized tumor-associated macrophages (TAMs) [30], regulatory T cells (T_{REGS}) [31], and myeloid derived suppressor cells (MDSC) [32], and is rich in cytokines that promote cell survival and proliferation (e.g. IL-6 and IL-23) or impair anti-tumor immune activity (e.g. TGF- β and IL-10). This setting, initiated and maintained in large part by NF- κ B- and STAT-driven transcription, results in the proliferation of cells with increasing levels of DNA damage and genome instability, leading to dysplasia that, in many cases, progresses to tumor development and metastasis.

Inflammation and Genomic Instability

While the chronic inflammatory environment provides proliferative and anti-apoptotic signaling and promotes the metastatic changes necessary for cancer progression, it is the ability to generate DNA changes and genomic instability that is the driver of tumorigenesis. DNA lesions occur at low levels in most cells, but are more frequent at sites of chronic inflammation where the mutation rate is significantly higher than in normal tissues [33]. Elevated mutation rates are observed even in the absence of dysplasia, implicating this mutagenesis as a potential mechanism of tumor initiation [34]. In the absence of efficient repair, the accumulation of DNA alterations results in destabilization of the genome and generation of the mutator phenotype that is a hallmark of most cancers. Several inflammation-induced mechanisms contribute to the generation of genome instability.

(i) DNA Alteration Chronic inflammation is associated with the sustained production of RONS by activated phagocytes and epithelial cells [35, 36, 24]. The mutagenic activity of inflammatory phagocytic cells was originally defined using the classical Ames test [37, 38]. Revealingly, phagocytes derived from patients with defects in NADPH-oxidases did not demonstrate mutagenic activity, implicating ROS as primary mediators. Consistent with the similarities between inflammatory and tumor immune environments, TAMs were also mutagenic in the bacterial and mammalian assays [39, 40]. Subsequently, co-culture of an immortalized mouse fibroblast cell line with human neutrophils revealed the ability of the phagocytic cells to drive transformation of mammalian cells at a rate similar to that achieved by exogenous free radicals [41].

Although the most common DNA alterations are base changes, RONS can also generate gross changes in chromosomes via rearrangements, deletions, insertions and amplifications resulting from the cumulative effects of oxidation of DNA bases, proteins or lipids [36, 35]. Furthermore, the most common oxidation-induced base change, conversion of guanine to 8-hydroxyguanine, could change gene expression by altering regional methylation patterns [42]. Genetic alterations detected in the transformed cells generated by co-culture with TAMs included DNA strand breaks, sister chromatid exchange and mutations. These results were consistent with the long established correlation between the capacity of tumor promoting agents to induce cancer and their ability to induce inflammatory cell infiltration and production of RONS.

A frequently observed consequence of the activation of NF-κB signaling in epithelial cells by chronic inflammation is the ectopic expression of activation-induced cytidine deaminase (AID) [43]. A member of the apolipoprotein B mRNA-editing enzyme, catalytic polypeptide-like family, AID is involved in the generation of adaptive immune receptor diversity through hyper-mutation of the variable regions of immunoglobulin (Ig) genes [44]. In addition, AID also induces the site-specific double strand DNA breaks that are necessary for the Ig class switching that occurs during maturation of an immune response. While AID activity is normally restricted to the Ig gene locus, off-target mutation of non-Ig genes, including oncogenes and tumor suppressor genes, has been widely reported [45]. Evidence for the ability of ectopic AID expression to induce genomic instability is provided by the generation of lymphomas and carcinomas in a transgenic mouse model of ubiquitous AID expression [46]. Elevated expression of AID has been detected at the sites of most inflammation-associated carcinomas, including liver, gastric, colorectal and esophageal [43]. (ii) Inhibition of DNA Repair The low mutation rate in normal tissue is in part a reflection of the efficiency of DNA repair pathways, including mismatch repair (MMR) and base excision repair (BER). Chronic inflammation is associated with aberrations in several DNA repair pathways, which compounds the mutagenic activity at these sites and leads to genomic instability [36]. The reduced repair activity can be caused by RONS-mediated damage to repair enzymes or altered expression of repair genes as a result of inflammation-associated changes in methylation patterns and gene promoter activity. Inflammatory cytokines and RONS also induce expression of HIF-1 α that in turn reduces expression of MMR proteins. Conversely, inflammation induces an increase in epithelial expression of two genes involved in BER, the primary pathway for repair of RONS-mediated DNA damage. Surprisingly, the increased activity of the AAG and APE1 enzymes correlated strongly with genomic instability [47]. Dysregulation of the MMR and BER pathways generates microsatellite instability (MSI) in regions of DNA rich in short repetitive sequences. Frameshift mutations in genes that contain such microsatellites, such as Bax and TGF- β RII, is thought to contribute to tumor development [48, 49].

(iii) Dysregulated Cell Cycle Checkpoints In addition to MSI resulting from inefficient DNA repair, chronic inflammation can also drive the generation of chromosomal instability (CIN) [15]. One of the primary mediators of CIN is the loss of the physiological mitotic checkpoints that normally inhibit the cell cycle during DNA repair or induce apoptosis if damage is excessive [50]. The p53 pathway is a frequent target of inflammation-induced dysregulation [51]. For example, the pro-inflammatory cytokine macrophage migration inhibitory factor (MIF) has been shown to inhibit the activity of the tumor suppressor proteins, p53 and retinoblastoma [52], a finding consistent with the clinical correlation between high MIF expression and poor prognosis [53–55]. In addition, RONS can inhibit p53 function and generate p53 mutations, and AID activity targets the p53 locus [56–59].

The ability of the chronic inflammation to activate both pathways that induce DNA abnormalities and those that nurture cells harbouring such lesions underlies tumor promotion by the immune microenvironment. The cumulative effect of these concurrent activities is the generation of a genome-unstable tumor cell population that provides the heterogeneity and plasticity that is the foundation for progression and metastasis of cancer [60]. Importantly, however, the generation of DNA lesions also serves as a signal to activate host immune surveillance mechanisms. The tumorigenicity of chronic inflammation is therefore countered by the induction of immune mechanisms that inhibit cancer progression (Fig. 8.1).

Immune Responses Against Cancer

Based on the correlation of tumor infiltration by various immune cells with clinical prognosis and the results of years of laboratory study, it is generally accepted that the most effective immune response for eliminating transformed cells and inhibiting



Fig. 8.1 Cancer in the balance. Chronic inflammation generates a microenvironment that includes both tumor promoting (*shown in green*) and tumor inhibiting (*shown in red*) immune activities. Tumor initiation, promotion and progression are influenced by the complex and dynamic interaction between these opposing factors. During immune-mediated equilibrium these activities are balanced, resulting in the survival, but not outgrowth of the established tumor. Effective targeting of these mechanisms, individually or in combination, may lead to improved patient outcomes by tipping the balance in favor of tumor elimination or sustained equilibrium

tumor growth is one dominated by the production of IFN- γ by CD4⁺ and CD8⁺ T cells, known as a Th1 response [61]. In addition to T cells, the tumor microenvironment during a Th1 response contains significant infiltration by natural killer (NK) cells, dendritic cells (DC), and type 1 macrophages (M1) and generates conditions that support the induction of tumor antigen-specific T cell cytotoxicity [62]. The role of specific lymphocyte subsets (e.g. T cells and NK cells), cytokines (e.g. IFN- γ and IL-12), and cytotoxic mechanisms (e.g. perforin, TRAIL) in protection against malignancy has been revealed by the increased tumor incidence in single gene-targeted immune-deficient mice. [63, 5] Confoundingly, components characteristic of a tumor-promoting immune environment, such as IL-6 and IL-23, are also often detected during Th1 anti-tumor responses, while many cells associated with productive anti-tumor immune responses have been reported to suppress such activity [64–66]. These apparent contradictions highlight the importance of the overall context of the immune response in determining the outcome, with variables such as tumor cell type, location, host immune status, and others, many of which are as yet unidentified, exerting significant influence.

Immune Recognition and Genomic Instability

The ability of the immune system to specifically recognize transformed cells is critical to mounting an effectively targeted anti-tumor response. In this regard, genomic instability contributes to such recognition through both the activation of the DNA damage response (DDR) and the generation of mutated proteins. Detection of cells, via recognition of stress-related molecules by cells of the innate immune system [67], or through the presentation of tumor–associated antigens to adaptive immune cells [68], provides the means to discriminate altered tissue from normal tissue, enabling specific eradication of pre-malignant or malignant cells.

The role of NK cells as important innate immune effector cells in anti-tumor responses has been identified in many experimental systems and their presence in tumor infiltrates correlates with good prognosis [69]. Indeed, of all the immune cell subsets detected at cancer sites, NK cells are the least reported as being associated with tumor-promoting activities. Activation of NK cells is achieved when the balance of signaling through activating receptors, including NKG2D, NKp30 and NKp46, outweigh those received via inhibitory receptors, such as killer-cell immunoglobulin-like receptors (KIR) and CD94/NKG2A [70]. Following activation, NK cells exert a variety of effector mechanisms, including perforin/granzyme-mediated cell cytotxicity and the production of cytokines that promote the generation of a Th1 response.

The occurrence of DNA lesions in cancer cells, resulting from chronic inflammation, oncogene activation, or other genotoxic stressors, can induce the DDR pathway [71]. Activation of the DDR cascade causes cell cycle arrest and initiates DNA repair pathways, enabling re-entry into the cell cycle, or senescence or apoptosis if the repair is unsuccessful. In addition, initiation of DDR can result in interferon production and the expression of ligands for the NK cell activating receptors, NKG2D and DNAM [72, 73]. Expressed at low levels on normal cells, these molecules are upregulated by cells exposed to DNA damaging agents and are frequently observed at higher levels on tumor cells. In addition to NK cells, NKG2D is often expressed on CD8 T cells, γ/δ T cells, and NKT cells; these are all cytotoxic cell types that have been associated with productive anti-tumor immune responses [74]. The increased expression of these ligands increases the sensitivity of cancer cells to killing by immune effector cells [75-77], and mice deficient in NKG2D or DNAM exhibit increased susceptibility to cancer [78, 79]. NKG2D binding to its ligand RAE-1 on the surface of tumor cells has also been implicated in NK cell-mediated clearance of senescent tumor cells [80]. Surprisingly, RAE-1 was expressed, independent of p53 status, on most tumor cells, but NK cells only eliminated those that had undergone p53-induced senescence. Critically, the senescent tumor cells secreted several chemokines, of which CCL2 was responsible for the NKG2D-independent migration of NK cells to the senescent tumors cells. In the absence of CCL2 production, there was limited NK infiltration into growing tumors. This finding highlights the importance of mechanisms of recruitment as well as targeting for immune-mediated elimination of cancer cells.

In contrast to the induction of innate immune responses, the altered expression of proteins on the surface of tumor cells is insufficient to generate a Th1 adaptive response. Recognition of cancer by T cells requires the presentation of peptide fragments of endogenous proteins in the context of major histocompatibility complex (MHC) class I or class II on the surface of tumor cells [81]. Furthermore, for the activation of naïve T cells, an additional costimulatory signal and appropriate cytokine milieu must be provided at the time of the initial binding of the T cell receptor to the MHC-peptide complex [82]. As most tumor cells do not express costimulatory molecules or MHC Class II, cross-presentation of tumor antigens by dendritic cells is necessary for induction of an optimal T cell response. At the tumor site, cell death, often as a result of innate immune responses, initiates the recruitment of immune cells, including immature DCs that take up tumor proteins. Following antigen processing and migration to the draining lymph node, DCs present MHC Class I and II associated tumor antigens, along with costimulatory signals, to CD8 and CD4 T cells respectively [83]. Activated tumor-specific T cells then migrate to the tumor site and, in the case of a productive Th1 response, infiltrate the stroma and eliminate transformed cells based on their expression of appropriate MHC-peptide complexes.

The peptide fragments presented by MHC complexes dictate the tumor specificity of a T cell response. T cell responses directed at many tumor antigens have been detected, including responses against antigens expressed by both normal and transformed cells, and those that are unique to cancer cells [68]. The mutations produced by the chronic inflammatory environment could provide a rich source of potentially unique peptides that would be specific to the transformed cells. Indeed, MSI generated by impaired MMR leads to the generation of multiple frameshift mutations that represent neo-antigens to the host immune system [84–87]. Intriguingly, tumors with high MSI (MSI-H) demonstrate greater immunogenicity than microsatellite stable tumors, with greater infiltration by activated CD8 T cells, and this has been suggested as the mechanism responsible for the reduced incidence of metastasis and better prognosis of the MSI-H tumors [88–92]. Consistent with this hypothesis, T cells specific for frameshift–generated epitopes have been detected in colorectal cancer patients with microsatellite unstable malignant cells [87, 93].

Genomic Instability and Immune Editing

As described in the preceding sections, the interaction between the immune system and transformed cells can lead to a range of possible outcomes, of which Virchow's cancer promotion and Ehrlich's eradication represent the two extremes. The threestage immune editing model of cancer progression was proposed in 2002 to reconcile the disparate influences of the immune system on cancer development [5]. The first stage, the elimination phase, resembles classical immunosurveillance, where newly arising transformed cells are specifically removed by the immune system. However, if this elimination is not complete, surviving transformed cells then enter the second stage, the equilibrium phase. During the equilibrium phase, there is a balance between tumor growth, often promoted by inflammatory mechanisms, and immune-mediated cytotoxicity that results in tumors that are present but not apparent. In the third stage of the model, the escape phase, loss of immune control over the nascent tumor leads to outgrowth and development of a clinically apparent cancer. In this model, loss of equilibrium is the result of selection of tumor cell clones that have lost their sensitivity to, or block the generation of, protective immune mechanisms; the evolving tumor is thus edited by the immune system to be less immunogenic as it progresses to overt disease.

The majority of the experimental support for the immune editing hypothesis has come from tumor progression experiments in mice with targeted immune deficiencies [77, 94–96]. However, a considerable body of data consistent with this model has emerged from human cancer studies. The development of antibody and T cell responses against tumor antigens in patients has been widely reported, although these alone do not prove that immune control was exerted at any stage in cancer progression. However, many mechanisms that subvert anti-tumor immunity have been detected in emergent tumor cells, a scenario consistent with the immunemediated selection of less immunogenic clones. These escape mechanisms include loss of MHC expression, upregulation of T cell inhibitory receptors, and release of soluble decoy ligands for NK cells [97–100]. Further evidence of an early equilibrium phase comes from patients with paraneoplastic syndromes in which immune responses directed at tumor antigens that are also expressed on neuronal cells trigger an autoimmune response [101, 102]. That the resultant neurologic dysfunction often presents significantly earlier than the underlying cancer suggests that the immune responses are being generated relatively early in tumorigenesis and may impede cancer progression. Finally, the previously described prognostic significance of tumor infiltrating immune cells implicates immune control as a significant modifier of cancer progression.

So where do the various immune modulatory effects of inflammation and genomic instability fit within the immune editing model of cancer progression? It is apparent that the onset of chronic inflammation precedes dysplasia in many instances, positioning this immune environment as an initiator of the tumor formation process. However, as mentioned earlier, some of the downstream effects of chronic inflammation may contribute to the induction of anti-tumor immune responses at this early stage. While it is obviously difficult to obtain clear evidence of successful cancer elimination, there are indications that anti-tumor immune activity is active during the very early stages of tumorigenesis. In the case of ulcerative colitis, an inflammatory condition that is highly associated with colon cancer, T cell responses specific for antigens expressed in dysplastic epithelium have been detected [103, 104]. This finding suggests that the immune activity against pre-malignant lesions may underlie the significantly lower incidence of colorectal colon cancer than ulcerative colitis [105, 104]. Importantly, this finding indicates that productive anti-tumor immune responses can be generated even after a chronic inflammatory environment has been established, an essential scenario if immune control of inflammation-driven cancer is to be clinically relevant. In keeping with early immune responses being induced by transformation events, T cells specific for frameshift mutations have been detected in cancer-free individuals with a genetic predisposition for colorectal cancer [106]. In the context of the immune editing model, the diagnosis of overt cancer is indicative that tumor escape has occurred. This is somewhat at odds with the prognostic significance of lymphocytic infiltration, which

suggests ongoing immune influence over cancer progression. While the precise nature of the association between immune infiltration and prognosis remains to be defined, it is an intriguing possibility that treatment of the cancer can re-establish a state of equilibrium, perhaps by restoring previously effective immune mechanisms or by the preferential elimination of escape variants [107].

Chronic inflammation provides several pathways for the escape of a tumor from immune control. Foremost amongst these is the generation of an environment that is inhibitory to productive anti-tumor immune responses. As described earlier, the presence of suppressive cell types, including MDSC and T_{REGs} , and cytokines, such as TGF- β and IL-10, may significantly undermine sustained T and NK cell-mediated cytotoxicity, allowing escape by simply removing the restraints. Such a scenario would not necessarily involve selection of resistant clones, and thus the tumor may still be amenable to immune control if the inhibitory environment could be altered [108]. Furthermore, the inflammatory environment provides additional features, such as vascular changes and extracellular protease activity, which could contribute to escape from local immune pressures through facilitation of metastasis.

The frequent loss or down-regulation of MHC class I expression on carcinoma cells suggests the outgrowth of immune-selected clones [97, 100]. MHC class I loss has been correlated with reduced T cell infiltration and poor prognosis [109]. While several mechanisms leading to MHC class I loss have been identified, frameshift mutations in β 2-microglobulin, an essential component of class I complexes, are frequently observed in MSI-H colorectal cancer cells early in tumor progression [110–112]. While these early occurring mutations generate a heterogeneous pool of class I positive and negative tumor cells, subsequent progression and metastasis is associated with selection and uniform outgrowth of the class I-deficient tumors (Fig. 8.2).

Genomic instability has also been implicated in the reduction of lymphocyte infiltration in colon cancer patients, where chromosomal instability led to the loss of CXCL13 expression, which resulted in lower densities of B cells and T follicular helper (Tfh) cells at the invasive margins [14]. High expression of B cell and Tfh cell markers correlated with extended disease-free survival. The apparent contradiction between the increased immunogenicity observed for MSI-H tumors and the ability of genomic instability to generate escape variants highlights how little we currently understand the dynamic nature of the interaction between cancer cells and the immune system and emphasizes the need for longitudinal studies of individual tumor evolution to provide clearer insights.

Immune evasion strategies employed in one tumor environment may not always be advantageous in another, a scenario that may be particularly relevant to metastasis. The frequent loss of MHC class I by carcinoma cells may represent such an event. While loss of MHC expression reduces the efficacy of anti-tumor T cellmediated cytotoxicity, a lack of class I increases the susceptibility of tumor cells to NK cell-mediated killing [113, 114]. As exposure to NK cells is likely to be increased during metastatic transit through blood or lymph, survival of the tumor cells during this migration may involve additional selective events. Interestingly, there is increasing evidence that platelets inhibit NK killing of tumor cells, both through the



Fig. 8.2 MSI-H and immune editing. The immune-driven selection of tumor cells is a multi-step process, one that MSI-H can influence in multiple ways. MSI-H induced mutations can serve as tumor-specific antigens for T cell responses (*step 1*). In response to T cell-driven selective pressure, tumor cells that lose MHC class I expression, for example by MSI-H-mediated mutations, have a growth advantage (*step 2*) and become the dominant clones (*step 3*) During metastasis of these escape variants, those that acquire additional survival advantages to escape the new immune pressures encountered during transit (*step 4*) will form a tumor at distant sites (*step 5*)

release of soluble factors and by physical protection of the tumor cells; increased adhesion to platelets by metastatic tumor cells may represent a selected trait that enhances tumor escape variant survival during transit and overall metastatic success (Fig. 8.2) [115–117].

Conclusions

Cells of the immune system are a significant component of the tumor microenvironment, capable of either promoting or inhibiting the progression of cancer. While immune mechanisms that contribute to each of these outcomes have been identified, the dynamic interplay between the components of the immune environment during the progression of disease remains poorly understood. The complexity of these interactions is highlighted by the differing contributions made by the same components in different tumor settings or at different times in the tumorigenic process; this complexity will make the consistent manipulation of the immune environment for clinical benefit a challenging proposition.

Given the divergent effects of immune responses in the tumor environment, two broad strategies are being investigated to enhance patient outcomes; inhibition of tumor promoting immune activity and boosting of anti-tumor responses. Reduced cancer incidence associated with the use of anti-inflammatory agents provides encouraging proof-of-principal evidence that the tumor-promoting inflammatory process can be clinically targeted [118–121]. While it is likely that achieving such outcomes against established cancer will be significantly harder, many approaches targeting the central pathways of chronic inflammation are currently under investigation for the ability to prevent cancer progression and metastasis [122–125].

The recent clinical successes with immune checkpoint blockade antibodies represent a significant milestone in the development of cancer immune therapies [126]. The improved patient outcomes achieved with monoclonal antibodies targeted to mediators of tumor-induced immune suppression demonstrates that evasion strategies can be overcome to re-activate protective anti-tumor immune responses. This success paves the way for the evaluation of strategies to address other escape pathways. Furthermore, the identification of commonly arising mutations in MSI-H cancers suggests that vaccine-based approaches to enhance T cell responses against these immunogenic tumors may be feasible [127]. Overall, our increasing understanding of the process of tumor progression has provided insights into how to target this process in patients. While we currently do not know enough to predict how well these approaches will work in each of the varied and complex tumor environments, there are encouraging signs that this work will result in improved outcomes for patients.

Acknowledgements The author is the recipient of a Canadian Cancer Society Career Development Award in Prevention.

References

- Virchow RLK (1863) Cellular pathology as based upon physiological and pathological histology.../by Rudolf Virchow. Translated from the 2d ed. of the original by Frank Chance. With notes and numerous emendations, principally from MS. notes of the author. 1–562
- 2. Mantovani A, Allavena P, Sica A, Balkwill F (2008) Cancer-related inflammation. Nature 454:436-444
- Ehrlich P (1909) Über den jetzigen stand der karzinomforschung. Ned Tijdschr Geneeskd 5:273–290
- Burnet M (1957) Cancer: a biological approach. III. Viruses associated with neoplastic conditions. IV. Practical applications. Br Med J 1:841–847
- Dunn GP, Bruce AT, Ikeda H, Old LJ, Schreiber RD (2002) Cancer immunoediting: from immunosurveillance to tumor escape. Nat Immunol 3:991–998
- Mittal D, Gubin MM, Schreiber RD, Smyth MJ (2014) New insights into cancer immunoediting and its three component phases–elimination, equilibrium and escape. Curr Opin Immunol 27:16–25
- Galon J, Costes A, Sanchez-Cabo F et al (2006) Type, density, and location of immune cells within human colorectal tumors predict clinical outcome. Science 313:1960–1964
- Galon J, Pagès F, Marincola FM et al (2012) Cancer classification using the Immunoscore: a worldwide task force. J Transl Med 10:205
- Pagès F, Kirilovsky A, Mlecnik B et al (2009) In situ cytotoxic and memory T cells predict outcome in patients with early-stage colorectal cancer. J Clin Oncol 27:5944–5951

- Adams S, Gray RJ, Demaria S et al (2014) Prognostic value of tumor-infiltrating lymphocytes in triple-negative breast cancers from two phase III randomized adjuvant breast cancer trials: ECOG 2197 and ECOG 1199. J Clin Oncol. doi: 10.1200/JCO.2013.55.0491
- 11. Khan H, Pillarisetty VG, Katz SC (2014) The prognostic value of liver tumor T cell infiltrates. J Surg Res. doi: 10.1016/j.jss.2014.06.001
- 12. Schatton T, Scolyer RA, Thompson JF, Mihm MC (2014) Tumor-infiltrating lymphocytes and their significance in melanoma prognosis. Methods Mol Biol 1102:287–324
- Webb JR, Milne K, Nelson BH (2014) Location, location, location: CD103 demarcates intraepithelial, prognostically favorable CD8(+) tumor-infiltrating lymphocytes in ovarian cancer. Oncoimmunology 3:e27668
- 14. Bindea G, Mlecnik B, Tosolini M et al (2013) Spatiotemporal dynamics of intratumoral immune cells reveal the immune landscape in human cancer. Immunity 39:782–795
- Colotta F, Allavena P, Sica A, Garlanda C, Mantovani A (2009) Cancer-related inflammation, the seventh hallmark of cancer: links to genetic instability. Carcinogenesis 30:1073–1081
- 16. Sun B, Karin M (2013) Inflammation and liver tumorigenesis. Front Med 7:242-254
- 17. Wang F, Meng W, Wang B, Qiao L (2014) Helicobacter pylori-induced gastric inflammation and gastric cancer. Cancer Lett 345:196–202
- Nagai H, Toyokuni S (2010) Biopersistent fiber-induced inflammation and carcinogenesis: lessons learned from asbestos toward safety of fibrous nanomaterials. Arch Biochem Biophys 502:1–7
- Hartnett L, Egan LJ (2012) Inflammation, DNA methylation and colitis-associated cancer. Carcinogenesis 33:723–731
- Nakai Y, Nonomura N (2013) Inflammation and prostate carcinogenesis. Int J Urol 20:150–160
- Borrello MG, Degl'Innocenti D, Pierotti MA (2008) Inflammation and cancer: the oncogenedriven connection. Cancer Lett 267:262–270
- 22. West AP, Koblansky AA, Ghosh S (2006) Recognition and signaling by toll-like receptors. Annu Rev Cell Dev Biol 22:409–437
- 23. Mantovani A, Cassatella MA, Costantini C, Jaillon S (2011) Neutrophils in the activation and regulation of innate and adaptive immunity. Nat Rev Immunol 11:519–531
- 24. Wu Y, Antony S, Meitzler JL, Doroshow JH (2014) Molecular mechanisms underlying chronic inflammation-associated cancers. Cancer Lett 345:164–173
- Moretta A, Marcenaro E, Sivori S, Della Chiesa M, Vitale M, Moretta L (2005) Early liaisons between cells of the innate immune system in inflamed peripheral tissues. Trends Immunol 26:668–675
- Steinman RM (2012) Decisions about dendritic cells: past, present, and future. Annu Rev Immunol 30:1–22
- Fan Y, Mao R, Yang J (2013) NF-κB and STAT3 signaling pathways collaboratively link inflammation to cancer. Protein Cell 4:176–185
- 28. Porta C, Larghi P, Rimoldi M, Totaro MG, Allavena P, Mantovani A, Sica A (2009) Cellular and molecular pathways linking inflammation and cancer. Immunobiology 214:761–777
- 29. Coussens LM, Werb Z (2002) Inflammation and cancer. Nature 420:860-867
- Martinez FO, Sica A, Mantovani A, Locati M (2008) Macrophage activation and polarization. Front Biosci 13:453–461
- 31. Whiteside TL (2014) Regulatory T cell subsets in human cancer: are they regulating for or against tumor progression? Cancer Immunol Immunother 63:67–72
- Ostrand-Rosenberg S, Sinha P (2009) Myeloid-derived suppressor cells: linking inflammation and cancer. J Immunol 182:4499–4506
- Loeb LA, Loeb KR, Anderson JP (2003) Multiple mutations and cancer. Proc Natl Acad Sci U S A 100:776–781
- 34. McLean MH, Murray GI, Stewart KN et al (2011) The inflammatory microenvironment in colorectal neoplasia. PLoS One 6:e15366
- 35. Wiseman H, Halliwell B (1996) Damage to DNA by reactive oxygen and nitrogen species: role in inflammatory disease and progression to cancer. Biochem J 313(Pt 1):17–29

- 8 Immunomodulation and Genomic Instability
- Kidane D, Chae WJ, Czochor J, Eckert KA, Glazer PM, Bothwell ALM, Sweasy JB (2014) Interplay between DNA repair and inflammation, and the link to cancer. Crit Rev Biochem Mol Biol 49:116–139
- Weitzman SA, Stossel TP (1981) Mutation caused by human phagocytes. Science 212:546– 547
- Weitzman SA, Stossel TP (1982) Effects of oxygen radical scavengers and antioxidants on phagocyte-induced mutagenesis. J Immunol 128:2770–2772
- Fulton AM, Loveless SE, Heppner GH (1984) Mutagenic activity of tumor-associated macrophages in Salmonella typhimurium strains TA98 and TA 100. Cancer Res 44:4308–4311
- Chong YC, Heppner GH, Paul LA, Fulton AM (1989) Macrophage-mediated induction of DNA strand breaks in target tumor cells. Cancer Res 49:6652–6657
- 41. Weitzman SA, Weitberg AB, Clark EP, Stossel TP (1985) Phagocytes as carcinogens: malignant transformation produced by human neutrophils. Science 227:1231–1233
- 42. Cerda S, Weitzman SA (1997) Influence of oxygen radical injury on DNA methylation. Mutat Res 386:141–152
- Shimizu T, Marusawa H, Endo Y, Chiba T (2012) Inflammation-mediated genomic instability: roles of activation-induced cytidine deaminase in carcinogenesis. Cancer Sci 103:1201– 1206
- 44. Honjo T, Kinoshita K, Muramatsu M (2002) Molecular mechanism of class switch recombination: linkage with somatic hypermutation. Annu Rev Immunol 20:165–196
- Fear DJ (2013) Mechanisms regulating the targeting and activity of activation induced cytidine deaminase. Curr Opin Immunol 25:619–628
- Okazaki I, Hiai H, Kakazu N, Yamada S, Muramatsu M, Kinoshita K, Honjo T (2003) Constitutive expression of AID leads to tumorigenesis. J Exp Med 197:1173–1181
- 47. Hofseth LJ, Khan MA, Ambrose M et al (2003) The adaptive imbalance in base excisionrepair enzymes generates microsatellite instability in chronic inflammation. J Clin Invest 112:1887–1894
- Imai K, Yamamoto H (2008) Carcinogenesis and microsatellite instability: the interrelationship between genetics and epigenetics. Carcinogenesis 29:673–680
- Yashiro M, Hirakawa K, Boland CR (2010) Mutations in TGFbeta-RII and BAX mediate tumor progression in the later stages of colorectal cancer with microsatellite instability. BMC Cancer 10:303
- Abbas T, Keaton MA, Dutta A (2013) Genomic instability in cancer. Cold Spring Harb Perspect Biol 5(3):a0129914. doi: 10.1101/cshperspect.a012914
- Schetter AJ, Heegaard NHH, Harris CC (2010) Inflammation and cancer: interweaving microRNA, free radical, cytokine and p53 pathways. Carcinogenesis 31:37–49
- 52. Petrenko O, Moll UM (2005) Macrophage migration inhibitory factor MIF interferes with the Rb-E2 F pathway. Mol Cell 17:225–236
- 53. Hira E, Ono T, Dhar DK, El-Assal ON, Hishikawa Y, Yamanoi A, Nagasue N (2005) Overexpression of macrophage migration inhibitory factor induces angiogenesis and deteriorates prognosis after radical resection for hepatocellular carcinoma. Cancer 103:588–598
- Shun C-T, Lin J-T, Huang S-P, Lin M-T, Wu M-S (2005) Expression of macrophage migration inhibitory factor is associated with enhanced angiogenesis and advanced stage in gastric carcinomas. World J Gastroenterol 11:3767–3771
- 55. White ES, Flaherty KR, Carskadon S, Brant A, Iannettoni MD, Yee J, Orringer MB, Arenberg DA (2003) Macrophage migration inhibitory factor and CXC chemokine expression in non-small cell lung cancer: role in angiogenesis and prognosis. Clin Cancer Res 9:853–860
- Hussain SP, Amstad P, Raja K et al (2000) Increased p53 mutation load in noncancerous colon tissue from ulcerative colitis: a cancer-prone chronic inflammatory disease. Cancer Res 60:3333–3337
- 57. Matsumoto Y, Marusawa H, Kinoshita K, Endo Y, Kou T, Morisawa T, Azuma T, Okazaki I-M, Honjo T, Chiba T (2007) Helicobacter pylori infection triggers aberrant expression of activation-induced cytidine deaminase in gastric epithelium. Nat Med 13:470–476

- Shinmura K, Igarashi H, Goto M et al (2011) Aberrant expression and mutation-inducing activity of AID in human lung cancer. Ann Surg Oncol 18:2084–2092
- 59. Wei J, Noto J, Zaika E, Romero-Gallo J, Correa P, El-Rifai W, Peek RM, Zaika A (2012) Pathogenic bacterium Helicobacter pylori alters the expression profile of p53 protein isoforms and p53 response to cellular stresses. Proc Natl Acad Sci U S A 109:E2543–E2550
- 60. Hanahan D, Weinberg RA (2011) Hallmarks of cancer: the next generation. Cell 144:646-674
- Bindea G, Mlecnik B, Fridman W-H, Galon J (2011) The prognostic impact of anti-cancer immune response: a novel classification of cancer patients. Semin Immunopathol 33:335–340
- Schreiber RD, Old LJ, Smyth MJ (2011) Cancer immunoediting: integrating immunity's roles in cancer suppression and promotion. Science 331:1565–1570
- Dunn GP, Old LJ, Schreiber RD (2004) The immunobiology of cancer immunosurveillance and immunoediting. Immunity 21:137–148
- Bui JD, Schreiber RD (2007) Cancer immunosurveillance, immunoediting and inflammation: independent or interdependent processes? Curr Opin Immunol 19:203–208
- DeNardo DG, Andreu P, Coussens LM (2010) Interactions between lymphocytes and myeloid cells regulate pro- versus anti-tumor immunity. Cancer Metastasis Rev 29:309–316
- Ruffell B, DeNardo DG, Affara NI, Coussens LM (2010) Lymphocytes in cancer development: polarization towards pro-tumor immunity. Cytokine Growth Factor Rev 21:3–10
- Marcus A, Gowen BG, Thompson TW, Iannello A, Ardolino M, Deng W, Wang L, Shifrin N, Raulet DH (2014) Recognition of tumors by the innate immune system and natural killer cells. Adv Immunol 122:91–128
- Coulie PG, Van den Eynde BJ, van der Bruggen P, Boon T (2014) Tumour antigens recognized by T lymphocytes: at the core of cancer immunotherapy. Nat Rev Cancer 14:135–146
- 69. Waldhauer I, Steinle A (2008) NK cells and cancer immunosurveillance. Oncogene 27:5932–5943
- Long EO, Kim HS, Liu D, Peterson ME, Rajagopalan S (2013) Controlling natural killer cell responses: integration of signals for activation and inhibition. Annu Rev Immunol 31:227–258
- Sancar A, Lindsey-Boltz LA, Unsal-Kaçmaz K, Linn S (2004) Molecular mechanisms of mammalian DNA repair and the DNA damage checkpoints. Annu Rev Biochem 73:39–85
- 72. Cerboni C, Fionda C, Soriani A, Zingoni A, Doria M, Cippitelli M, Santoni A (2014) The DNA Damage Response: A Common Pathway in the Regulation of NKG2D and DNAM-1 Ligand Expression in Normal, Infected, and Cancer Cells. Front Immunol 4:508
- Gasser S, Orsulic S, Brown EJ, Raulet DH (2005) The DNA damage pathway regulates innate immune system ligands of the NKG2D receptor. Nature 436:1186–1190
- Mincheva-Nilsson L, Baranov V (2014) Cancer exosomes and NKG2D receptor-ligand interactions: Impairing NKG2D-mediated cytotoxicity and anti-tumour immune surveillance. Semin Cancer Biol. doi: 10.1016/j.semcancer.2014.02.010
- Suzuki T, Terao S, Acharya B, Naoe M, Yamamoto S, Okamura H, Gotoh A (2010) The antitumour effect of {gamma} {delta} T-cells is enhanced by valproic acid-induced up-regulation of NKG2D ligands. Anticancer Res 30:4509–4513
- Nanbakhsh A, Pochon C, Mallavialle A, Amsellem S, Bourhis JH, Chouaib S (2014) c-Myc regulates expression of NKG2D ligands ULBP1/2/3 in AML and modulates their susceptibility to NK-mediated lysis. Blood 123:3585–3595
- 77. Smyth MJ, Swann J, Cretney E, Zerafa N, Yokoyama WM, Hayakawa Y (2005) NKG2D function protects the host from tumor initiation. J Exp Med 202:583–588
- Iguchi-Manaka A, Kai H, Yamashita Y, Shibata K, Tahara-Hanaoka S, Honda S, Yasui T, Kikutani H, Shibuya K, Shibuya A (2008) Accelerated tumor growth in mice deficient in DNAM-1 receptor. J Exp Med 205:2959–2964
- Guerra N, Tan YX, Joncker NT et al (2008) NKG2D-deficient mice are defective in tumor surveillance in models of spontaneous malignancy. Immunity 28:571–580
- Iannello A, Raulet DH (2014) Immunosurveillance of senescent cancer cells by natural killer cells. Oncoimmunology 3:e27616
- Blum JS, Wearsch PA, Cresswell P (2013) Pathways of antigen processing. Annu Rev Immunol 31:443–473
- 8 Immunomodulation and Genomic Instability
- Smith-Garvin JE, Koretzky GA, Jordan MS (2009) T cell activation. Annu Rev Immunol 27:591–619
- Guermonprez P, Valladeau J, Zitvogel L, Théry C, Amigorena S (2002) Antigen presentation and T cell stimulation by dendritic cells. Annu Rev Immunol 20:621–667
- Garbe Y, Maletzki C, Linnebacher M (2011) An MSI tumor specific frameshift mutation in a coding microsatellite of MSH3 encodes for HLA-A0201-restricted CD8 + cytotoxic T cell epitopes. PLoS One 6:e26517
- Maletzki C, Schmidt F, Dirks WG, Schmitt M, Linnebacher M (2013) Frameshift-derived neoantigens constitute immunotherapeutic targets for patients with microsatellite-instable haematological malignancies: frameshift peptides for treating MSI + blood cancers. Eur J Cancer 49:2587–2595
- Ishikawa T, Fujita T, Suzuki Y, Okabe S, Yuasa Y, Iwai T, Kawakami Y (2003) Tumorspecific immunological recognition of frameshift-mutated peptides in colon cancer with microsatellite instability. Cancer Res 63:5564–5572
- Saeterdal I, Bjørheim J, Lislerud K et al (2001) Frameshift-mutation-derived peptides as tumor-specific antigens in inherited and spontaneous colorectal cancer. Proc Natl Acad Sci U S A 98:13255–13260
- Buckowitz A, Knaebel H-P, Benner A, Bläker H, Gebert J, Kienle P, von Knebel Doeberitz M, Kloor M (2005) Microsatellite instability in colorectal cancer is associated with local lymphocyte infiltration and low frequency of distant metastases. Br J Cancer 92:1746–1753
- Nakata B, Wang YQ, Yashiro M, Nishioka N, Tanaka H, Ohira M, Ishikawa T, Nishino H, Hirakawa K (2002) Prognostic value of microsatellite instability in resectable pancreatic cancer. Clin Cancer Res 8:2536–2540
- Lü B-J, Lai M, Cheng L, Xu J-Y, Huang Q (2004) Gastric medullary carcinoma, a distinct entity associated with microsatellite instability-H, prominent intraepithelial lymphocytes and improved prognosis. Histopathology 45:485–492
- Dolcetti R, Viel A, Doglioni C, Russo A, Guidoboni M, Capozzi E, Vecchiato N, Macri E, Fornasarig M, Boiocchi M (1999) High prevalence of activated intraepithelial cytotoxic T lymphocytes and increased neoplastic cell apoptosis in colorectal carcinomas with microsatellite instability. Am J Pathol 154:1805–1813
- Guidoboni M, Gafà R, Viel A et al (2001) Microsatellite instability and high content of activated cytotoxic lymphocytes identify colon cancer patients with a favorable prognosis. Am J Pathol 159:297–304
- Bauer K, Nelius N, Reuschenbach M et al (2013) T cell responses against microsatellite instability-induced frameshift peptides and influence of regulatory T cells in colorectal cancer. Cancer Immunol Immunother 62:27–37
- O'Sullivan T, Saddawi-Konefka R, Vermi W et al (2012) Cancer immunoediting by the innate immune system in the absence of adaptive immunity. J Exp Med 209:1869–1882
- Koebel CM, Vermi W, Swann JB, Zerafa N, Rodig SJ, Old LJ, Smyth MJ, Schreiber RD (2007) Adaptive immunity maintains occult cancer in an equilibrium state. Nature 450:903–907
- Matsushita H, Vesely MD, Koboldt DC et al (2012) Cancer exome analysis reveals a T-celldependent mechanism of cancer immunoediting. Nature 482:400–404
- Chang C-C, Ferrone S (2007) Immune selective pressure and HLA class I antigen defects in malignant lesions. Cancer Immunol Immunother 56:227–236
- Chretien A-S, Le Roy A, Vey N, Prebet T, Blaise D, Fauriat C, Olive D (2014) Cancer-Induced Alterations of NK-Mediated Target Recognition: Current and Investigational Pharmacological Strategies Aiming at Restoring NK-Mediated Anti-Tumor Activity. Front Immunol 5:122
- 99. Gajewski TF (2007) Failure at the effector phase: immune barriers at the level of the melanoma tumor microenvironment. Clin Cancer Res 13:5256–5261
- 100. Seliger B (2005) Strategies of tumor immune evasion. BioDrugs 19:347-354
- Ferlito A, Elsheikh MN, Manni JJ, Rinaldo A (2007) Paraneoplastic syndromes in patients with primary head and neck cancer. Eur Arch Otorhinolaryngol 264:211–222

- Kanaji N, Watanabe N, Kita N, Bandoh S, Tadokoro A, Ishii T, Dobashi H, Matsunaga T (2014) Paraneoplastic syndromes associated with lung cancer. World J Clin Oncol 5:197– 223
- Karlsson M, Lindberg K, Karlén P, Ost A, Thörn M, Winqvist O, Eberhardson M (2010) Evidence for immunosurveillance in intestinal premalignant lesions. Scand J Immunol 71:362–368
- Kiran RP, Ali UA, Nisar PJ et al (2014) Risk and location of cancer in patients with preoperative colitis-associated dysplasia undergoing proctocolectomy. Ann Surg 259:302–309
- Zisman TL, Bronner MP, Rulyak S et al (2012) Prospective study of the progression of lowgrade dysplasia in ulcerative colitis using current cancer surveillance guidelines. Inflamm Bowel Dis 18:2240–2246
- Schwitalle Y, Kloor M, Eiermann S, Linnebacher M, Kienle P, Knaebel HP, Tariverdian M, Benner A, von Knebel Doeberitz M (2008) Immune response against frameshift-induced neopeptides in HNPCC patients and healthy HNPCC mutation carriers. Gastroenterology 134:988–997
- Zitvogel L, Galluzzi L, Smyth MJ, Kroemer G (2013) Mechanism of action of conventional and targeted anticancer therapies: reinstating immunosurveillance. Immunity 39:74–88
- Topalian SL, Weiner GJ, Pardoll DM (2011) Cancer immunotherapy comes of age. J Clin Oncol 29:4828–4836
- Garrido F, Ruiz-Cabello F (1991) MHC expression on human tumors-its relevance for local tumor growth and metastasis. Semin Cancer Biol 2:3–10
- Bernal M, Ruiz-Cabello F, Concha A, Paschen A, Garrido F (2012) Implication of the β2microglobulin gene in the generation of tumor escape phenotypes. Cancer Immunol Immunother 61:1359–1371
- Kloor M, Michel S, Buckowitz B et al (2007) Beta2-microglobulin mutations in microsatellite unstable colorectal tumors. Int J Cancer 121:454–458
- 112. Tikidzhieva A, Benner A, Michel S, Formentini A, Link K-H, Dippold W, von Knebel Doeberitz M, Kornmann M, Kloor M (2012) Microsatellite instability and Beta2-Microglobulin mutations as prognostic markers in colon cancer: results of the FOGT-4 trial. Br J Cancer 106:1239–1245
- 113. Purdy AK, Campbell KS (2009) Natural killer cells and cancer: regulation by the killer cell Ig-like receptors (KIR). Cancer Biol Ther 8:2211–2220
- 114. Tu MM, Mahmoud AB, Wight A, Mottashed A, Bélanger S, Rahim MMA, Abou-Samra E, Makrigiannis AP (2014) Ly49 family receptors are required for cancer immunosurveillance mediated by natural killer cells. Cancer Res 74:3684–3694
- 115. Placke T, Kopp H-G, Salih HR (2011) Modulation of natural killer cell anti-tumor reactivity by platelets. J Innate Immun 3:374–382
- 116. Palumbo JS, Barney KA, Blevins EA et al (2008) Factor XIII transglutaminase supports hematogenous tumor cell metastasis through a mechanism dependent on natural killer cell function. J Thromb Haemost 6:812–819
- Placke T, Salih HR, Kopp H-G (2012) GITR ligand provided by thrombopoietic cells inhibits NK cell antitumor activity. J Immunol 189:154–160
- Giardiello FM, Hamilton SR, Krush AJ, Piantadosi S, Hylind LM, Celano P, Booker S V, Robinson CR, Offerhaus GJ (1993) Treatment of colonic and rectal adenomas with sulindac in familial adenomatous polyposis. N Engl J Med 328:1313–1316
- Labayle D, Fischer D, Vielh P, Drouhin F, Pariente A, Bories C, Duhamel O, Trousset M, Attali P (1991) Sulindac causes regression of rectal polyps in familial adenomatous polyposis. Gastroenterology 101:635–639
- 120. Phillips RKS, Wallace MH, Lynch PM et al (2002) A randomised, double blind, placebo controlled study of celecoxib, a selective cyclooxygenase 2 inhibitor, on duodenal polyposis in familial adenomatous polyposis. Gut 50:857–860
- 121. Burn J, Bishop DT, Chapman PD et al (2011) A randomized placebo-controlled prevention trial of aspirin and/or resistant starch in young people with familial adenomatous polyposis. Cancer Prev Res (Phila) 4:655–665

- Quail DF, Joyce JA (2013) Microenvironmental regulation of tumor progression and metastasis. Nat Med 19:1423–1437
- Reisfeld RA (2013) The tumor microenvironment: a target for combination therapy of breast cancer. Crit Rev Oncog 18:115–133
- 124. Ataie-Kachoie P, Pourgholami MH, Morris DL (2013) Inhibition of the IL-6 signaling pathway: a strategy to combat chronic inflammatory diseases and cancer. Cytokine Growth Factor Rev 24:163–173
- Aggarwal BB, Vijayalekshmi R V, Sung B (2009) Targeting inflammatory pathways for prevention and therapy of cancer: short-term friend, long-term foe. Clin Cancer Res 15:425–430
- Pardoll DM (2012) The blockade of immune checkpoints in cancer immunotherapy. Nat Rev Cancer 12:252–264
- 127. Von Knebel Doeberitz M, Kloor M (2013) Towards a vaccine to prevent cancer in Lynch syndrome patients. Fam Cancer 12:307–312

Chapter 9 Synthetic Genetic Approaches in Colorectal Cancer: Exploiting and Targeting Genome Instability

Babu V. Sajesh, Amy L. Cisyk and Kirk J. McManus

Abstract Colorectal cancer is the third leading cause of cancer-related deaths throughout the world. Surgery is effective against early stage tumors but advanced stage tumors lack an effective targeting strategy. For nearly 50 years, 5-fluorouracil has been the standard of care for advanced disease, but the overall 5-year survival rate remains at only 6%. Accordingly, novel therapeutic strategies are urgently needed to decrease morbidity and mortality rates. Synthetic genetic approaches are well established in model organisms, and have recently garnered much attention in humans for their potential implications in cancer targeting. Synthetic lethality and synthetic dosage lethality are innovative strategies designed to specifically exploit and kill cancer cells based on the loss-of-function associated with tumor suppressors or the gain-of-function associated with oncogenes, respectively. By definition, these approaches are highly specific and restricted to tumor cells, and are expected to decrease side effects associated with current strategies. Both synthetic genetic approaches have been applied extensively in pre-clinical studies and numerous candidate drug targets have been identified, including some that have entered clinical trials. The focus of this chapter is to present the pathways that drive tumorigenesis in colorectal cancer, and describe how synthetic lethality and synthetic dosage lethality can exploit these origins for enhanced killing of tumor cells. Finally, we summarize the current status of the field and relate how these novel strategies can be custom-tailored to target advanced stage colorectal cancer as we enter the personalized medicine era.

Keywords Colorectal cancer · Metastatic disease · Chromosome instability · Genome instability · Therapeutic targeting · Treatment · Synthetic genetic approaches · Synthetic lethality · Synthetic dosage lethality · Personalized medicine

Abbreviations

5-FU 5-Flurouracil

APC Adenomatous polyposis coli

K. J. McManus $(\boxtimes) \cdot B$. V. Sajesh $\cdot A$. L. Cisyk

Manitoba Institute of Cell Biology, Department of Biochemistry and Medical Genetics, University of Manitoba, Winnipeg, MB R3E 0V9, Canada e-mail: Kirk.McManus@med.umanitoba.ca

© Springer International Publishing Switzerland 2015

C. Maxwell, C. Roskelley (eds.), *Genomic Instability and Cancer Metastasis*, Cancer Metastasis - Biology and Treatment 20, DOI 10.1007/978-3-319-12136-9_9

CIMP	CpG island methylator phenotype
CIN	Chromosome instability
CRC	Colorectal cancer
DNA	Deoxyribonucleic acid
KRAS	Kirsten rat sarcoma viral oncogene homolog
MMR	Mismatch repair
MSI	Microsatellite instability
nCIN	Numerical CIN
sCIN	Structural CIN

Introduction

The International Agency for Research on Cancer (World Health Organization) estimated that in 2012, there was 14.1 million newly diagnosed cancer cases in the world [1]. Amongst these cases, colorectal cancer (CRC) was the third most common cancer accounting for ~9.7% of all cases, or ~1.4 million new cases and ~700,000 deaths annually. CRC has significant burden for both males and females. In males, it ranks third behind lung and prostate cancer with ~746,000 new cases and 374,000 deaths annually while, in females, it ranks second behind breast cancer, with ~614,000 new cases and 320,000 deaths (Table 9.1). These statistics highlight the immediate need for novel therapeutic strategies and targets to better combat the disease.

CRC typically begins as a benign adenomatous polyp within the colon and/or rectum. Through the accrual of genetic insults, the polyps can develop into advanced stage adenomas with high-grade dysplasia that can progress into an invasive cancer, and may ultimately metastasize to lymph nodes and distant organs [2]. CRC staging is used as a metric to estimate the amount of penetration of the particular cancer. Staging is employed for diagnostic and prognostic purposes, and to determine the best treatment option. Five stages (stage 0, I, II, III and IV) are employed to describe

Location	Estimated	incidence		Estimated	mortality	
	Males	Females	Total	Males	Females	Total
World (2012) [1]	746,000	614,000	1,360,000	374,000	320,000	694,000
Australia ^a (2012/10) [147]	8760	7080	15,840	2205	1777	3982
Canada (2013) [148]	13,200	10,600	23,900	5000	4200	9200
United Kingdom ^b (2011) [149]	23,171	18,410	41,581	8520	7139	15,659
United States (2014) [150]	71,830	65,000	136,830	26,270	24,040	50,310

Table 9.1 Incidence and mortality rates of colorectal cancer

^a Bowel Cancer; Estimated incidence and actual deaths reported for 2012 and 2010, respectively ^b Bowel Cancer (Colon, Rectum and rectosigmoid junction); UK = England, Wales, Scotland and Northern Ireland the extent of local invasion, the degree of lymph node involvement and whether the tumor has metastasized [3]. In stage 0 disease (i.e., carcinoma *in situ*), the cancer is confined to the innermost lining of the colon or rectum, whereas stage I tumors have grown into the inner wall of the colon (or rectum), but have not yet invaded beyond the wall. Stage II tumors have extended more deeply into or through the inner wall, and may involve proximal tissues, but do not show lymph node involvement. Stage III cancers exhibit lymph node involvement, while stage IV tumors are those that have begun to metastasize to distal sites including the liver and lungs [4].

Although the overall 5-year survival rate for newly diagnosed individuals in the United States is ~65%, survival rates differ substantially depending on the stage at initial diagnosis (Table 9.2) [5]. As expected, individuals with tumors diagnosed at early stages (I and II) fair better than those diagnosed at late stages (IV). For example, stage I and II disease is often cured through surgical intervention alone, while up to 74% of stage III disease is curable by surgery in combination with adjuvant chemotherapy [6, 7]. Although, recent advances in chemotherapy have improved survival and/or quality of life, stage IV disease is typically incurable [7] and has a 5-year overall survival rate of only 6% [5].

The clinical behavior of a CRC is multi-factorial as there are a myriad of environmental and host factors that contribute to the pathogenesis of the disease. A major challenge in the field has been to identify and characterize the molecular factors and pathways that initiate, promote and drive tumor formation. Indeed over the past 50 years, substantial progress has been made and many environmental, germ-line and somatic factors have been identified including the aberrant genes and biological pathways implicated in the pathogenesis of CRC. Consequently, one of the remaining challenges lies in the evolution and/or development of next generation therapeutics to better combat the disease.

Genome Instability and Its Role in Colorectal Cancer Development

For over a century, genome instability has been suspected to underlie tumorigenesis. In 1890, David von Hansemann, first reported abnormal mitotic figures in several epithelial cancers [8]. In 1914, Theodor Boveri postulated that tumors arise as a consequence of abnormal chromosome segregation [9]. He also theorized that tumor growth was based on incorrect chromosome combinations, and likely accounted

Table 9.2Observed 5-yearsurvival rates for colon andrectal cancer (Based onUnited States data)

Stage	Observed 5-year surv	ival rate [5]
	Colon cancer (%)	Rectal cancer (%)
Ι	74	74
II	37–67	32–65
III	28–73	33–74
IV	6	6

for the abnormal growth characteristics passed on to daughter cells. During the subsequent ~ 100 years, a wealth of fundamental and clinical research has substantiated the correlation between aberrant chromosome numbers in countless tumor types, yet very little information is available that describes the underlying aberrant biological processes and genes accounting for genome instability.

Genome instability is a hallmark of virtually all tumor types, including CRC. There is a growing body of evidence that suggests genome instability is a driver of the tumorigenic process and underlies the acquisition of tumor-associated genetic alterations [4]. Genome instability generally arises due to random mutations within critical genes, chromosome rearrangements [10–12] and epigenetic changes [13] that collectively provide a growth advantage to cells permitting them to proliferate, survive and spread [14]. However, the genetic and epigenetic aberrations that cause genome instability and drive tumor development may be the very targets that can be exploited through novel therapeutic strategies collectively referred to as synthetic genetic approaches (described below). What is clear, however, is that CRCs exhibiting genome instability can be classified into at least one of three general categories: 1) microsatellite instability (MSI), 2) chromosome instability (CIN), or 3) CpG island methylator phenotype (CIMP). Not only do these pathways contribute to the pathogenesis of tumors, but they also serve to differentiate the tumor cells from normal surrounding cells and tissues. Thus, if therapeutic strategies can be identified and developed that leverage and exploit these inherent differences, the therapeutic effect would be restricted to tumors and thus minimize side effects within the normal surrounding tissues. Accordingly, an awareness of these genome instability pathways, and the molecular players implicated therein, is necessary to develop synthetic genetic approaches.

a. Microsatellite Instability and DNA Mismatch Repair MSI arises due to defects DNA mismatch repair (MMR), which result in the accumulation of genomic mutations [15–17]. MSI is typically detected by measuring changes within microsatellite DNA [18], which are highly repetitive DNA sequences scattered throughout the genome. The highly repeated core sequence within a given microsatellite typically ranges from 1–6 base pairs in length, and is prone to DNA replication errors. Under normal conditions, single base pair mismatches and/or DNA slippage occurring within microsatellites are normally repaired by the MMR system, which, when compromised, leads to insertions and/or deletions that contribute to the tumorigenic process. While the number of repeats contained within a microsatellite varies from person to person, a given microsatellite within an individual remains constant. Consequently, extensive length variations for a given microsatellite in an individual is suggestive of an underlying MMR defect [18].

There are four predominant genes that encode functions within the MMR pathway, namely *MSH2*, *MLH1*, *MSH6* and *PMS2* (reviewed in [19]). These genes encode proteins that collectively function to identify and coordinate the repair of DNA mismatches, including those within microsatellites. However, if these functions become compromised, genome instability ensues. Because most of the genomic alterations stemming from MMR defects are typically small in size, and usually involve one to several bases, most MSI tumors maintain a karyotypically

stable or diploid to near-diploid state [20, 21]. Thus, MSI is also frequently referred to as a mutator pathway or phenotype [22].

MSI was first identified as an aberrant phenotype associated with CRC in 1993 [20, 23, 24]. Since then, numerous studies have determined that both inherited and somatic defects (i.e., somatic mutations and epigenetic silencing) within the MMR genes are contributing factors in the etiology of CRC (reviewed in [25]). For example, in Lynch syndrome, which accounts for 1-2% of all CRCs, individuals harbor a predisposition to develop CRC (and other tumor types) due to the inheritance of a defective MMR allele [26, 27]. DNA sequencing of Lynch Syndrome patients have shown that the majority of cases are associated with inherited defects in *MSH2* (~60%) or *MLH1* (~30%) with a minor proportion of cases attributed to defects in *MSH6* (5–10%) or *PMS2* (<5%) [28-30]. Interestingly, although MSI is associated with ~15% of sporadic CRCs, up to 60% of those cases are due to epigenetic silencing of *MLH1* rather than acquired mutations within the gene [31–34], indicating divergent molecular origins for inherited versus sporadic CRCs.

b. Chromosome Instability Is Highly Associated with Colorectal Cancer CIN is defined by an increase in the rate at which whole chromosomes, or large parts thereof, are gained or lost. CIN is an aberrant phenotype associated with up to 80% of all tumor types [35, 36], and is arguably best described in CRC, where it is associated with up to 85% of all sporadic and the vast majority of heritable forms of the disease [37–40]. Conceptually, CIN impacts tumor suppressors and oncogenes by decreasing or increasing gene copy numbers respectively, and/or by inducing structural rearrangements of chromosomes.

In general, tumors exhibiting CIN can be classified into two categories, numerical (nCIN) or structural (sCIN) (reviewed in [41]). Tumors with nCIN typically have increases in chromosome numbers, and most late stage CRCs typically exhibit between 60 and 90 chromosomes per cell [39]. Although the underlying defects leading to nCIN are not well understood, they often involve mutations in genes that regulate chromosome biology including, chromosome congression and segregation, DNA replication and cytokinesis [42]. On the other hand, tumors exhibiting sCIN frequently harbor abnormal chromosome numbers, but also display structural defects resulting from segmental duplications, deletions, or translocations. Like nCIN, the molecular basis of sCIN is poorly understood, but can be associated with aberrant telomere biology, *MYC* amplification or DNA repair defects [43, 44].

The terms CIN and aneuploidy are often used interchangeably, but they are in fact different terms. While aneuploidy refers to a "state" of abnormal DNA content (i.e., chromosome numbers), CIN describes the "rate" leading to aneuploidy. Although studies dating back to Boveri have demonstrated a correlation between aneuploidy and cancer [9], no empirical data clearly demonstrated an increase in the 'rate' at which aneuploidy occurred. In other words, the association of aneuploidy with certain cancers merely suggested an underlying CIN, but its actual nature and magnitude were never examined. In 1997, Vogelstein and colleagues [39] utilized a panel of diploid and aneuploid CRC cells to conclusively demonstrate that the aneuploidy observed in a large subset of CRCs reflected an underlying and persistent error in chromosome segregation that was accounted for by elevated rates in

chromosomal changes. This decisive work identified gains or losses in excess of 10^{-2} chromosomes per generation within the aneuploid CRC lines that did not occur within the diploid lines. These data suggest that cancer cells could be classified into MSI or CIN categories based solely on the observation of an aneuploid karyotype. This seminal work also addressed the long-standing debate over whether CIN contributed to tumorigenesis or was merely the result of the tumorigenic process (i.e., initiator vs. bystander). The observation that the MSI cell lines remained diploid and did not develop a CIN phenotype argued strongly against a bystander effect. Although the above information highlights the robust association between CIN and CRC, very little information is available that addresses its causal nature.

The classical CIN pathway involves the multi-step acquisition of specific genetic alterations and was first characterized by Vogelstein et al. [45]. It was initially proposed that an early mutation occurred within APC (adenomatous polyposis coli), a tumor suppressor that was believed to be the principle determinant of CIN [46, 47]. This belief was supported by previous DNA sequencing efforts that identified frequent APC mutations within CIN tumors [48]. However, subsequent sequencing efforts have identified CIN tumors without APC mutations, and non-CIN tumors with APC mutations [49]. These observations suggest that APC is unlikely to be the sole determinant of CIN, and accumulating evidence now suggests that there is a spectrum of genes, rather than a single gene, which when aberrantly expressed contribute to the CIN phenotype [12, 37, 50, 51]. Genes encoding proteins that function within pathways such as the spindle assembly checkpoint, sister chromatid cohesion, centrosome biology, DNA damage response, telomere regulation, and cell cycle checkpoints are strongly implicated in the pathogenesis of CIN [52]. Indeed, a large number of the genes that regulate these pathways are somatically mutated in CRCs exhibiting CIN (see [53]), and further supports their underlying involvement in CIN and the tumorigenic process.

In addition to its causal role in the pathogenesis of CRC, the presence of CIN (and aneuploidy) has significant clinical and treatment implications for the disease. Many features associated with CIN (e.g., loss of heterozygosity, the presence of an aneuploid karyotype, or gross chromosomal re-arrangements) are associated with metastatic potential in sporadic tumors [54]. The presence of CIN is also employed as a prognostic indicator, as it correlates with poor patient prognosis [55–58]. Finally, CIN has been shown to confer intrinsic multi-drug resistance in CRC cell lines [59] and many tumor types [60, 61].

c. CpG Island Methylator Phenotype (CIMP) and Its Role in Colorectal Cancer CIMP is an epigenetic phenomenon associated with gene regulation. Unlike MSI and CIN, CIMP does not inherently change the information encoded within the genome, but rather it imparts a level of regulation onto the genome. CIMP is characterized by the hypermethylation of CpG islands that typically reside within the promoter regions of genes, and results in transcriptional silencing (reviewed in [62]). CpG islands are DNA sequences, ranging from 500–2000 base pairs in size, that are rich in CG di-nucleotide repeats that localize within the 5' promoter region of genes [63]. In general, CpG hypermethylation recruits additional factors, such as histone deacetylases and chromatin remodeling proteins, that generally function to

compact the DNA, and render it resistant to transcriptional factors thus preventing gene expression [13].

CpG island methylation can occur naturally or associate with the progression and development of cancer. Type-A methylation refers to the natural <u>age</u>-related increase in global methylation that occurs as colonic epithelial cells age [64, 65], while type-C methylation is the <u>cancer</u>-associated form that is frequently observed in tumor samples. Extensive DNA methylation patterns are attributed to gene inactivation in numerous tumor types including CRC where hypermethylation is frequently observed in hundreds of genes, including DNA repair (e.g., *MLH1* in sporadic CRC) and tumor suppressor genes [64]. Although tumor hypermethylation patterns appear specific, it remains to be determined why certain loci become preferentially hypermethylated [64].

Overall, CIMP is associated with $\sim 30\%$ of sporadic CRCs and a modest increase in DNA methyltransferase activity correlates with CRC progression. However, the underlying mechanism(s) remain poorly understood [66] as silencing in cancerrelated genes can occur independent of heterochromatic DNA compaction [67]. It should be noted that DNA methylation patterns are extremely stable. Therefore, they can be maintained through cellular division and transmitted to all daughter progeny [68]. Thus, it is likely that certain hypermethylation patterns, particularly those that confer a survival advantage, will be selected with time and will contribute to the development of CRC.

Current Therapeutic Strategies to Combat Colorectal Cancer

Overall, there are a number of current therapeutic strategies to combat CRC, but they can be generally classified into one of two categories: local or systemic. Local therapies include surgery and radiation therapy, and are designed to remove or destroy the cancer in or near the colon and rectum. Colonoscopies are the preferred method of surveillance for lesions, and permit the removal of polyps by surgical resection (e.g., snare polypectomy, cryosurgery, radiofrequency ablation therapy, etc.). Alternatively, laparoscopic or open surgeries can be performed to remove a portion of the bowel containing the tumor, which is frequently accompanied by the removal of proximal lymph nodes. These types of local therapies are most effective in early stages of the disease (0-II). Local therapies have limited applicability in advanced stages of the disease (Stages III or IV) especially those involving distal metastases (i.e., lymph node or tissues). Radiation therapy is often employed for palliative purposes, particularly in advanced rectal cancers or those with liver metastases [69–71].

More aggressive treatments are generally employed to treat advanced stage disease (stages III or IV), and are collectively referred to as systemic therapies. The overriding principle of systemic therapies is that a therapeutic agent is administered systemically so that it enters the circulatory system and can target cancer cells at the primary site(s) within the colon and rectum, and also at metastatic sites. Systemic therapies include chemical (e.g., small molecule inhibitor, antimetabolite, DNA damaging agent, etc.) and biological (e.g., monoclonal antibody) agents (Table 9.3). Unfortunately, the therapeutic activity associated with these agents is not restricted to the tumor cells. Rather, many of the agents employed (e.g., 5-fluorouricil, irinotecan, oxaliplatin, etc. [see below]) interfere with DNA replication and are thus potentially toxic to all dividing cells. However, cancer cells are predominantly affected due to their rapid proliferation rates and heavy reliance on DNA replication. Due to a lack of target specificity, a number of side effects are often associated with these agents. These side effects can include alopecia (hair loss), anemia (low red cell counts leading to shortness of breath and fatigue), effects on the cells lining the digestive tract (nausea, vomiting, diarrhea, etc.), neutropenia (low white blood cell counts leading to increased risk of infection), organ damage or even the development of second cancers [72–75].

a. Current Systemic Approaches Although current chemical or biological treatment options in CRC are typically reserved for stage III and IV disease, some individuals with stage II disease receive treatments for prophylactic purposes. For over 50 years, 5-fluorouracil (5-FU) has been the mainstay of CRC treatments (reviewed in [76]). 5-FU is an anti-metabolite that was first synthesized in 1957 by Charles

Chemotherapeutic agent	Therapeutic target ^a	Mechanism of action
Oxaliplatin	DNA	Prevents replication by forming inter-and intra-strand (DNA) crosslinks to prevent replication
Cetuximab	EGFR	Binds to EGFR and prevents signaling
Panitumumab	EGFR	Binds to EGFR and prevents signaling
5-Fluorouracil (5-FU)	TS	Pyrimidine analog and antimetabolite that binds and irreversibly inhibits TS to inhibit DNA replication
Capecitabine	TS	Pro-drug that is enzymatically converted to 5-FU; inhibits DNA replication
Tegafur	TS	Pro-drug that is enzymatically converted to 5-FU; inhibits DNA replication
Leucovorin ^b	TS (FU)	Stabilizes FU binding to TS
Bevacizumab	VEGF	Binds VEGF to inhibit angiogenesis
Ziv-Aflibercept	VEGF	Binds VEGF to inhibit angiogenesis
Irinotecan	TOP1	Inhibits DNA replication and transcription
Regorafenib	Multiple kinases ^c	Inhibits multiple kinases

Table 9.3 List of chemotherapeutic agents used to treat colorectal cancers

^a VEGF Vascular Endothelial Growth Factor, TS Thymidylate Synthetase, EGFR Epidermal Growth Factor Receptor, TOP1 Topoisomerase 1

^b Employed as an adjuvant therapy to FU

° Shown to inhibit the activity of RET, VEGFR1, VEGFR2, VEGFR3, KIT, PDGFR- α , PDGFR- β , FGFR1, FGFR2, TIE2, DDR2, TRKA, EPH2A, RAF-1, BRAF, BRAF, BRAF^{V600E}, SAPK2, PTK5 and ABL at clinically relevant doses

Heidelberger [77]. The therapeutic activity of 5-FU stems from its ability to inhibit thymidylate synthetase, an enzyme required for the synthesis of thymidine, an essential nucleoside precursor required for DNA synthesis during replication or repair. More specifically, 5-FU prevents the formation of dTMP (deoxythymidine monophosphate), which prevents cell proliferation. Alternatively, capecitabine and tegafur, pro-drugs that are enzymatically converted into 5-FU, can be substituted. Although 5-FU primarily prevents DNA synthesis, it is also metabolized into fluorouridine triphosphate (FUTP) and can be readily incorporated within RNA, which leads to the disruption of normal RNA processing and function [76].

Currently, 5-FU is used extensively in the treatment of advanced stage CRC, but its efficacy as a single agent appears limited ($\sim 10-15\%$) [78]. To enhance treatment efficacy, many new combinatorial approaches involving 5-FU and additional compounds have been identified (Table 9.3). Additional chemical agents also include leucovorin (or folinic acid), which stabilizes 5-FU binding to TS [79], or irinotecan, which is a topoisomerase 1 inhibitor that functions by preventing DNA from becoming unwound during replication. Finally, oxaliplatin has been used in advanced disease, and is a platinum-based antineoplastic that forms DNA cross-links and inhibits DNA replication and transcription.

In addition to the systemic, chemical therapies described above, biological therapies (e.g., monoclonal antibodies) are frequently employed (Table 9.3). The prevailing concept is that the antibody binds to its cognate epitope, often a growth factor or cell surface receptor, to prevent ligand binding, protein dimerization or signaling from occurring. In CRC, biological therapies have begun to garner attention, and can generally be classified into one of three categories: 1) those that impact mitogenic signaling through the epidermal growth factor receptor (EGFR) (e.g., cetuximab and panitumumab), 2) those that bind vascular endothelial growth factor (VEGF) to prevent angiogenesis (e.g., bevacizumab and ziv-aflibercept), and 3) those that inhibit a large number of kinases and presumably affect numerous signaling cascades, including mitogenic, anti-apoptotic and pro-survival pathways (e.g., regorafenib). Many of these biological agents are now used in combination with 5-FU (or related) approaches and are beginning to show some efficacy [80–83].

Evolving Synthetic Genetic Approaches for Targeting Advanced Stage Colorectal Cancers

Many current chemical and biological therapeutic regimens involve the systemwide administration of an agent, whose requisite activity affects all dividing cells. These agents generally function by inhibiting various biological processes required for cell cycle progression and/or preventing oncogenic signaling pathways from functioning. As a result, all dividing cells, whether cancerous or not, are targeted, which is often associated with a diverse array of side effects. Accordingly, novel therapeutic strategies are required that minimize side effects through the more selective targeting of genetic and/or epigenetic factors that underlie tumor initiation, progression and development. Over the past decade, there has been a fundamental shift away from the traditional pantropic approaches detailed above to more personalized approaches that target specific aberrant events driving tumor formation. Somatic mutations in genes that encode tumor suppressors, or oncogenes, whose disruption underlies genome instability (e.g., MSI, CIN and CIMP) are now widely recognized as significant genetic predispositions to tumorigenesis [10–12]. Typically, tumor suppressor genes (including DNA repair genes) are epigenetically silenced, deleted or accumulate non-synonymous mutations that adversely impact protein expression and/ or function, while oncogenes are amplified or accumulate mutations that enhance the activity of the encoded protein (i.e., constitutive activity).

Synthetic genetic approaches can be subdivided into two categories: 1) synthetic lethal approaches, which exploit the loss-of-function associated with tumor suppressor genes (and DNA repair genes); and, 2) synthetic dosage lethal approaches, which exploit the gain-of-function associated with oncogenes. While these strategies are in their infancy, and many examples are in a pre-clinical development phase, a select few are already in various phases of clinical trials. Below are descriptions of the principles, concepts and examples of synthetic lethality and synthetic dosage lethality, particularly as they pertain to advanced stage CRC.

a. Synthetic Lethality—Introduction and Concept The term synthetic lethality was first coined by Dobzhansky in 1946, and describes the lethal genetic interaction observed when two independently viable homologous chromosomes were allowed to recombine in *Drosophila pseudoobscura* [84]. Synthetic lethality now simply defines the lethal combination of two independently viable gene mutations or deletions (Fig. 9.1). In essence, it describes a genetic interaction in which the outcome of a particular mutation or deletion is influenced by the presence of a pre-existing genetic predisposition or perturbation. Conceptually, synthetic lethal interactions occur via three basic mechanisms (Fig. 9.2): (1) partial ablation of two proteins contained within the same essential biological pathway (i.e., epistasis group) such that the pathway is non-functional; 2) ablation of two proteins contained within parallel pathways, which together impinge on an essential biological process.



Fig. 9.1 The paradigm of synthetic lethality. Synthetic lethality defines a rare and lethal genetic interaction occurring between two genes. Mutually exclusive mutations occurring within either *Gene 1* or *Gene 2* are viable. A synthetic lethal interaction is defined if the simultaneous combination of *Gene 1* and *Gene 2* mutations results in death.



Fig. 9.2 Conceptual models of pathways underlying synthetic lethal interactions. Conceptual models depicting the mechanism(s) of synthetic lethal interactions: numbered/lettered rectangles represent gene products, somatically acquired (i.e., cancer) mutations are identified by red bounding boxes, while synthetic lethal interactors (i.e., drug targets) are identified with blue bounding boxes. **a** Partial ablation (*grey crosses*) of two independent functions encoded within a single essential pathway (i.e., epistasis group), such that the pathway is no longer functional. **b** Ablation (*black crosses*) of two functions encoded within two distinct parallel pathways. For example, a DNA replication defect would lead to DNA errors requiring repair, and inadequate repair will lead to death. **c** Ablation (*black crosses*) of two functions encoded within two separate pathways that together impinge on a single essential process. For example, DNA double strand breaks can only be repaired through two pathways, namely homologous recombination repair and non-homologous end joining—defects in both lead to cellular cytotoxicity.

Synthetic lethal studies are performed in a variety of genetically tractable model systems, including worms, flies and yeast (Fig. 9.3). Most recent approaches have employed deletion mutant arrays of budding yeast strains (collections of ~4700 non-essential gene deletion strains) to systematically interrogate all pair-wise gene combinations and produce comprehensive synthetic lethal interaction network data [85-89]. Over the past 15 years, these efforts have provided critical insights into biological function and pathway architecture, and have helped define molecular complex and epistasis group members [90-94]. However, the potential for these data to predict conserved interactions in human cancer has not yet been fully realized.

In 1997, Hartwell and colleagues [95] posited that cancer cells represent genetically sensitized cells that may be susceptible to drug therapies selectively targeting a second unlinked gene product (Fig. 9.4a). They suggested that synthetic lethal partners identified in model organisms could be used to identify conserved candidate interactions in a human cancer context (Fig. 9.4b). In support of this hypothesis, Dixon et al. [96] recently demonstrated a significant overlap between

B. V. Sajesh et al.



Fig. 9.3 Synthetic lethality in budding yeast. A schematic diagram depicting two independently viable yeast stains that are mutant for *gene1* and *gene2*. However, if these two genes are synthetic lethal interactors, the presence of both mutations within a single organism will produce a lethal phenotype. If slowed growth occurs, a synthetic sickness or synthetic growth defect is observed.

Fig. 9.4 Exploiting synthetic lethality for therapeutic benefit in a cancer context. a A conceptual model depicting the synthetic genetic therapeutic strategy in cancer. A specific cancerassociated mutation in a tumor suppressor gene by inhibiting the activity of a synthetic lethal interactor through the use of a small molecule inhibitor (SMI). **b** A normal cell is converted into a cancerous cell through the acquisition of mutations, some of which occur within defined tumor suppressor or DNA repair genes (GENE1). The hypomorphic function encoded by (epi-)genetic defects in GENE1 can be therapeutically exploited by identifying and targeting a synthetic lethal interactor (GENE2).



synthetic lethal interaction data sets for two distantly related eukaryotes, while McManus et al. [97] provided the first empirical evidence of a conserved synthetic lethal interaction occurring in humans that was first identified in yeast (described below). Because many essential biological processes required to maintain genome integrity are evolutionarily conserved, synthetic lethal networks identified within model organisms may represent major untapped repositories.

a.i. Synthetic Lethality in Practice Perhaps the best studied and prototypic example of a human synthetic lethal interaction is that observed between *BRCA1/BRCA2* and *PARP1 (poly ADP-ribose polymerase 1)*. *BRCA1* and *BRCA2* are breast and ovarian cancer susceptibility genes that are also mutated in a large number of additional tumor types [98–102]. BRCA1 and BRCA2 function in the homologous recombination repair pathway, which is also referred to as the "error-free" DNA double-strand break repair and, although it was traditionally believed to function in base excision repair [104, 105], it may participate in single-strand repair independent of base excision repair [106]. Nevertheless, PARP1 generally functions to detect DNA single-strand breaks and elicit a response through the ADP-ribosylation at the site of damage (reviewed in [107]).

In 2005, two independent research teams reasoned that PARP1 inhibition would lead to DNA single-strand breaks that would be converted into double-strand breaks during replication [108, 109]. They further posited that cellular cytotoxicity would ensue if the homologous recombination repair pathway was compromised by defects in *BRCA1* or *BRCA2*. As predicted, relative to controls, PARP1 silencing and inhibition enhanced killing within the *BRCA1*-and *BRCA2*-deficient cells, which was further substantiated in embryonic stem cells [109] and animal models [108], thus confirming PARP1 as a novel candidate drug target.

Since the initial characterization of a synthetic lethal interaction between *BRCA1/BRCA2* and *PARP1*, a number of small molecule inhibitor screens have been performed (see [3]). The efficacy of many PARP inhibitors (e.g., veliparib, CEP-9722, rucaparib, E7016, BMN-673, olaparib, etc.) are now being evaluated as either a mono- or combination therapy [110]. Olaparib (KuDOS Pharmaceuticals; KU-0059436 or AstraZeneca; AZD-2281) is perhaps the best studied amongst these inhibitors, and AstraZeneca recently initiated a Phase III clinical trial for ovarian cancer patients with *BRCA* mutations. Although many of the studies focus on hereditary breast and ovarian cancers, PARP inhibitors have clinical relevance in many tumor types. In CRC, for example, *BRCA1* and *BRCA2* are somatically altered in ~2.8 and ~6.6% of sporadic cases [99], respectively, which potentially represents ~3250 and ~7675 newly diagnosed Americans annually. Thus, the clinical potential of olaparib (and other PARP inhibitors) in managing cancers with *BRCA1* and *BRCA2* defects has broad-spectrum appeal.

As indicated above, McManus et al. [97] were the first to utilize synthetic lethal datasets from budding yeast to identify novel drug targets in human CRC. In 2009, they employed a cross-species approach to identify a conserved synthetic lethal interaction in a CRC context. Using reverse genetics, biochemistry and microscopy, they showed that diminished FEN1 (flap endonuclease 1) expression induced

cellular cytotoxicity specifically within *RAD54B*-deficient CRC cells, but not in isogenic controls, which identified FEN1 as a novel candidate drug target. *RAD54B* encodes an evolutionarily conserved helicase that functions in the homologous recombination repair pathway, and it is somatically altered in ~8.2% of CRCs (~9500 newly diagnosed Americans annually) [99]. FEN1 is an evolutionarily conserved flap endonuclease that functions in the removal of 5' overhangs during DNA repair, and in the processing of Okazaki fragments on the lagging strand during DNA replication. Subsequent work by van Pel et al. [111] expanded the synthetic lethal network of FEN1 to include additional genes that are somatically mutated in CRC, including *CDC4*, *MRE11A*, *SMC1A*, *SMC3* and *RNF20* [50]. The authors also performed a screen of 30,000 compounds and identified 13 novel FEN1 inhibitors with *in vitro* activity [111]. Of these, three were validated using cell-based assays and are now lead candidate compounds requiring additional pre-clinical testing and optimization prior to entering clinical trials.

More recently, the initial cross species candidate gene approaches have been expanded upon and additional synthetic lethal interactors (i.e., drug targets) have begun to be identified. For example, Sajesh et al. [112] employed hierarchical clustering on 692 yeast genes and identified the top 500 genetic interactions. Based on yeast datasets, several data-rich regions were identified including one that included all three evolutionarily conserved members of a superoxide radical (i.e., reactive oxygen species) detoxification pathway. SOD1, CCS and PRDX2 encode proteins required to remove excess superoxide radicals through a two-step process. First, SOD1 dismutates superoxide radicals into hydrogen peroxide, an enzymatic reaction that requires the copper chaperone of SOD1 (CCS) to provide Cu^{2+} [113]. Second, PRDX2 reduces hydrogen peroxide into water and oxygen. The authors reasoned that SOD1 silencing or inhibition would cause the accumulation of excessive superoxide radicals that would produce DNA double-strand breaks. In cells with defective homologous recombination repair (e.g., RAD54B-deficient CRC cells), this damage would not be adequately repaired and death would result. In agreement with this hypothesis, the authors showed that RAD54B-deficient cells were hypersensitive to SOD1 silencing or inhibition, and thus confirmed SOD1 as a novel candidate therapeutic target.

Beyond the cross-species approaches detailed above, large-scale unbiased approaches have also been conducted to undercover novel candidate drug targets (i.e., synthetic lethal interactors) for many additional tumor suppressor genes. *TP53* is somatically altered in $\sim 50\%$ of CRC [114], and encodes a protein that normally functions to preserve genome integrity by inducing cell cycle arrests or apoptosis in the presence of DNA damage [115]. From a clinical perspective, the loss of P53 function is associated with poor prognosis and increased resistance to chemotherapeutic treatments [116–118].

Xie et al. [119] recently performed a genome-wide screen for synthetic lethal interactors of *TP53* in CRC cells. They identified 103 putative candidates of which two, *ATR* (Ataxia Telangiectasia and Rad3 related) and *ETV1* (ETS Translocation Variant 1), were confirmed as synthetic lethal interactors of *TP53*. While ATR is a serine/threonine kinase that functions in the DNA damage response [120], ETV1

is a transcription factor that regulates numerous biological processes including cell growth, proliferation and angiogenesis [121]. Using reverse genetics, cell-based and xenograft models, the authors showed that cellular proliferation was dramatically impeded within the *TP53*-deficient cells relative to controls [119]. Thus, both ATR and ETV1 were identified as novel candidate drug targets in *TP53*-deficient CRC cells.

Many of the synthetic lethal interactors detailed above warrant additional preclinical study and lead chemicals must be further optimized for *in vivo* efficacy and delivery using relevant animal models. Nevertheless, the identification of synthetic lethal interactors in CRC represents a critical first step towards migrating maximal numbers of candidate drugs into pre-clinical studies and clinical trials for advanced stage CRC.

b. Synthetic Dosage Lethality—Concept and Approach Oncogenic transformation provides a key growth advantage that drives tumor development. However, it may also represent an Achilles' heel that can be selectively targeted through synthetic dosage lethality. In 1996, Kroll and colleagues [122] developed a variation of the traditional yeast synthetic lethal screen in which they demonstrated that increased protein expression and/or activity caused lethality in a genetically sensitized, mutant yeast strain. The concept, termed synthetic dosage lethality, was based on previous observations in which lethality was observed following enhanced gene expression/function in specific mutant yeast strains. More specifically, mcm3 (minichromosome maintenance 3) overexpression enhanced defects observed in mcm2 mutants [123], while orc6 (origin of replication complex 6) overexpression lowered the non-permissive temperature associated with cdc46–1 mutants [124]. Thus, synthetic dosage lethality, similar to synthetic lethality, may hold therapeutic potential in human cancers (Fig. 9.5).

Many human oncogenes (e.g., *KRAS*, *B-RAF*, *EGFR*, etc.) are either amplified or mutated in such a manner as to produce a constitutively active oncoprotein. This not only results in hypermorphic expression/function, but also serves to differentiate tumor cells from cells in normal surrounding tissues. Therefore, the genetic and epigenetic insults underlying the gain-of-functions associated with oncogenes may also be therapeutically exploited through the identification of a synthetic dosage lethal interactor. Although strategies employed to identify synthetic dosage lethal interactors are detailed elsewhere [125], what follows are brief examples of synthetic dosage lethal interactors identified for *KRAS*, an oncogene altered in CRC.

b.i. Synthetic Dosage Lethality in Practice KRAS (Kirsten Rat Sarcoma Viral Oncogene Homolog) is a proto-oncogene that encodes a membrane-associated guanosine triphosphate/diphosphate binding protein that is widely expressed in human cells. KRAS normally functions in intracellular signaling cascades, particularly through EGFR-signaling activation to regulate cell division, differentiation and apoptosis. Mutant *KRAS* is correlated with oncogenic transformation in CRC, and is observed in 35–45% of all cases (reviewed in [126, 127]). Approximately 95% of the *KRAS* mutations involve single bases within codon 12 or 13, which underlies single amino acid substitutions (e.g., G12D, G12V or G13D) [128]. These



substitutions render KRAS in a constitutively active GTP-bound state that produces a constant proliferation signal [129]. Because of its strong association with CRC, synthetic dosage lethal targets that can exploit oncogenic *KRAS* mutations are of extreme clinical interest.

In 2009, several large-scale screens identified novel candidate drug targets (i.e., synthetic dosage lethal interactors) that specifically exploited hypermorphic expression and/or function associated with oncogenic *KRAS* mutations [130–132]. Scholl et al. [132] performed high-throughput RNAi-based screens in a panel of human cancer cell lines harboring either wild type or mutant *KRAS*, and identified STK33 as an interactor. STK33 is a serine/threonine kinase that functions in cytoskeletal regulation through the phosphorylation of vimentin [133]. The authors showed that STK33 expression is essential for the survival and proliferation of various cell lines harboring *KRAS* mutations, but is dispensable in cell lines with wild-type *KRAS* [132]. Accordingly, they suggested that STK33 inhibition might be a strategy for therapeutic intervention in a broad spectrum of cancers. However, more recent studies have shown that STK33 expression is dispensable in certain cellular contexts with oncogenic *KRAS* mutations [134].

Luo et al. [131] performed a genome-wide RNAi-based screen, and uncovered multiple synthetic dosage lethal interactors of *KRAS*. Key amongst these genes were those that encoded mitotic-specific functions, including *CCNA2* (cyclin A2),

CDCA8 (borealin), *CASC5* (KNL-1), *KIF2C* (MCAK), components of the anaphase promoting complex (*ANAPC1*, *ANAPC4*, *CDC16* and *CDC27*), *SMC4* and *PLK1*. The large number of genes with established roles in mitosis led the authors to speculate that cells harboring *KRAS* mutations may experience heightened mitotic stress, and may exhibit hypersensitivity towards chemicals affecting mitosis. Indeed PLK1 silencing and inhibition (BI-2536) were both associated with increased sensitivity in *KRAS* mutant cell lines relative to isogenic controls, and was further validated in xenograft tumor models. These data suggest that PLK1, and perhaps many other mitotic proteins, may be candidate therapeutic targets. However, the clinical value of targeting PLK1 is currently unclear as PLK1 is an essential mitotic kinase under normal conditions [135, 136]. Thus, the use of PLK1 inhibitors in a treatment setting may be associated with adverse side effects. Nevertheless, a large number of additional candidates were identified that may hold therapeutic potential in *KRAS* mutant CRC contexts.

Sun et al. [137] employed a kinome-centered screen to identify candidates that could synergize with MEK inhibitors to selectively target cells harboring *KRAS* mutations. *KRAS* mutations reduce the sensitivity of cells to MEK inhibitors [138], and thus identifying conditions that re-sensitize tumors to MEK inhibition are highly desired. In this study, the authors showed that the mechanism accounting for the lack of MEK sensitivity was dependent upon the upregulation of ERBB3, a member of the EGFR family of tyrosine kinase receptors. Armed with the knowledge that ERBB3 heterodimerizes and forms active kinase complexes with other members of the ERBB family (e.g., ERBB1 [EGRF] or ERBB2), they showed that dual ERBB1/ERBB2 inhibitors (afatinib and dacomitinib) synergized with MEK inhibitors (selumetinib or trametinib) and resulted in decreased growth in cell-based CRC models. Subsequent *in vivo* xenograft models validated the combination therapy at least over the ~4-week study period. Thus, the authors conclude that *KRAS* mutated tumors may be resensitized to MEK inhibitors through the co-administration of ERBB1/ERBB2 inhibitors, although this remains to be tested clinically.

In addition to the above examples, a large number of studies have identified conditions (i.e., targets and chemicals) that can exploit oncogenic *KRAS* mutations [130, 134, 137, 139–142]. It is interesting to note that very few targets identified in any individual study are shared between studies. This may be due to the inherent differences in the experimental systems employed (i.e., RNAi-based libraries, experimental conditions, specific assays, etc.), the levels of sensitivity required to identify a positive candidate, or the genetic heterogeneity of the cellular contexts employed. Nevertheless, these studies have identified a myriad of synthetic dosage lethal interactors of *KRAS*, which will ensure that maximal targets are evaluated in the hopes of identifying as many new clinically-relevant targets and compounds as possible.

Conclusions—Evolution of Therapeutic Strategies at the Dawn of the Personalized Medicine Era

Novel therapeutic strategies and drug targets are needed to not only decrease morbidity and mortality rates associated with CRC, but to better target the tumor cells so as to minimize side effects within healthy cells and tissues. Synthetic genetic strategies potentially represent significant advancements over traditional pantropic approaches for targeting advanced stage CRCs. Synthetic lethality and synthetic dosage lethality are designed to specifically kill cancer cells based on the loss-of tumor suppressor function or the gain-of oncogene function. Both approaches have been applied extensively in cell-based screens and numerous candidate drug targets have been identified. In either case, the penultimate goal of the synthetic genetic screens is to identify candidates for which small molecule inhibitors will ultimately be developed (see [125]). Once a target is identified, an appropriate chemical screen can be devised to identify lead chemical compounds for subsequent validation, optimization and pre-clinical studies, prior to initiating clinical trials.

In theory, synthetic genetic approaches offer the ability to custom tailor the therapy to the individual and the tumor itself. With the decreasing costs and availability of next generation sequencing, it will become possible to generate sequence-specific data from tumor biopsies or circulating tumor cells that identify an aberrant molecular signature (i.e., aberrant tumor suppressor genes and/or aberrant on-cogenes) that may be exploited through synthetic lethal and/or synthetic dosage lethal approaches. As we approach the dawn of the personalized medicine era, it will become possible to obtain critical information about the individual and their tumor(s) to better select a therapeutic strategy. It is conceivable that by identifying the specific defects in tumor suppressor genes and oncogenes, drugs can be identified and employed that exploit those very defects. This therapeutic concept will no longer treat advanced stage disease as a single entity with pantropic agents, but rather will treat each patient as an individual case.

Synthetic genetic approaches also represent significant advancements over traditional chemical and biological strategies as they potentially offer enhanced specificity. In this regard, the drug (i.e., synthetic genetic interactor) specifically exploits defects inherent to the tumor, and often seeks to target the molecular origins associated with genome instability (i.e., MSI, CIN or CIMP). Accordingly, these approaches not only serve to target and restrict the cytotoxic effects within tumor cells, but also decrease or eliminate side effects within healthy cells and tissues. Since metastatic disease often shares 'ancestral' aberrant genetic and epigenetic events with the primary tumor (particularly the loss of tumor suppressor functions), a major benefit of this approach is the potential to impact metastatic disease.

It may also become possible to identify appropriate combinatorial therapies that simultaneously exploit a number of the underlying genetic insults that may improve the current 6% 5-year overall survival rates for stage IV cancers [5]. For example, combining FEN1 and SOD1 inhibitors may produce additive or synergistic effects in tumors with *RAD54B*-defects. Alternatively, combinations targeting both syn-

thetic lethal (e.g., PARP1, FEN1, SOD1, etc.) and synthetic dosage lethal (e.g., ERBB2, MEK1, etc.) interactors could be developed that may even include traditional approaches, such as 5-FU (Table 9.3). Although highly speculative, these concepts must be formally evaluated in appropriate pre-clinical models before clinical trials can be initiated.

In summary, tumor heterogeneity and the development of drug resistance is a major concern in combating any tumor. With the advancement of DNA sequencing platforms and decreases in the associated costs, genetic information may direct patient care and treatment. For example, extensive DNA and chromatin immunoprecipitation sequencing (reviewed in [143]) efforts have already begun on CRC cell lines and for a select few patient samples [99], and is providing clinically-relevant insight into the genetic and epigenetic events associated with CRC [144–146]. However, the notion of a personalized medicine approach is in its infancy, as much of the pre-clinical studies are still ongoing and will have to make their way into clinical trials. Nevertheless, a select few clinical studies evaluating the efficiency of synthetic genetic approaches (e.g., PARP inhibitors) are currently underway. Thus, the clinical efficacy of these drugs and others that employ a synthetic genetic paradigm will remain to be determined.

Acknowledgements We thank members of the McManus laboratory for helpful discussions and suggestions. We are grateful for operational support from CIHR (MOP 115179), Colon Cancer Canada, Manitoba Health Research Council and the Cancer Care Manitoba Foundation. ALC is a recipient of a Flying Officer George Finkle Scholarship and a GETS award (University of Manitoba). We acknowledge the strong support of the Manitoba Institute of Cell Biology and Cancer Care Manitoba Foundation.

References

- Ferlay J, Soerjomataram I, Ervik M, Dikshit R, Eser S, Mathers C et al (2013) GLOBOCAN 2012 v1.0, Cancer Incidence and Mortality Worldwide: IARC CancerBase No. 11 [Internet]. Lyon, France: International Agency for Research on Cancer
- 2. Kinzler KW (2002) Colorectal tumors. In: Vogelstein BK (ed) The genetic basis of human cancer. McGraw-Hill, New York
- Rouleau M, Patel A, Hendzel MJ, Kaufmann SH, Poirier GG (2010) PARP inhibition: PARP1 and beyond. Nat Rev Cancer 10:293–301
- Markowitz SD, Bertagnolli MM (2009) Molecular origins of cancer: molecular basis of colorectal cancer. N Engl J Med 361:2449–2460
- Howlader NNA, Krapcho M, Garshell J, Neyman N et al (2014) SEER Cancer Statistics Review, 1975–2011. National Cancer Institute, Bethesda
- Andre T, Boni C, Mounedji-Boudiaf L, Navarro M, Tabernero J, Hickish T et al (2004) Oxaliplatin, fluorouracil, and leucovorin as adjuvant treatment for colon cancer. N Engl J Med 350:2343–2351
- Compton CHE, Grochow L, Lee F, Ritter M et al (2008) Colon cancer. In: Abeloff MD, Niederhuber JE; Kastan MB; McKenna GW (eds) Abeloff's clinical oncology. Churchill Livingstone, Philadelphia
- von Hansemann D (1890) Ueber asymmetrische Zelltheilung in epithel Krebsen und deren biologische Bedeutung. Virchows Arch A Pathol Anat Histopathol 119:299–326
- 9. Boveri T (1914) Zur Frage der Entstehung maligner Tumoren. G. Fischer, Jena

- Cahill DP, Kinzler KW, Vogelstein B, Lengauer C (1999) Genetic instability and darwinian selection in tumours. Trends Cell Biol 9:M57–M60
- 11. Lengauer C, Kinzler KW, Vogelstein B (1998) Genetic instabilities in human cancers. Nature 396:643–649
- 12. Rajagopalan H, Jallepalli PV, Rago C, Velculescu VE, Kinzler KW, Vogelstein B et al (2004) Inactivation of hCDC4 can cause chromosomal instability. Nature 428:77–81
- 13. Issa JP (2004) CpG island methylator phenotype in cancer. Nat Rev Cancer 4:988-993
- 14. Hanahan D, Weinberg RA (2011) Hallmarks of cancer: the next generation. Cell 144:646–674
- Boyer JC, Umar A, Risinger JI, Lipford JR, Kane M, Yin S et al (1995) Microsatellite instability, mismatch repair deficiency, and genetic defects in human cancer cell lines. Cancer Res 55:6063–6070
- Loeb LA (1991) Mutator phenotype may be required for multistage carcinogenesis. Cancer Res 51:3075–3079
- 17. Loeb LA, Bielas JH, Beckman RA (2008) Cancers exhibit a mutator phenotype: clinical implications. Cancer Res 68:3551–3557; discussion 7
- Chung DC, Rustgi AK (1995) DNA mismatch repair and cancer. Gastroenterology 109:1685– 1699
- 19. Jiricny J (2006) The multifaceted mismatch-repair system. Nat Rev Mol Cell Biol 7:335-346
- 20. Aaltonen LA, Peltomaki P, Leach FS, Sistonen P, Pylkkanen L, Mecklin JP et al (1993) Clues to the pathogenesis of familial colorectal cancer. Science 260:812–816
- Bocker T, Schlegel J, Kullmann F, Stumm G, Zirngibl H, Epplen JT et al (1996) Genomic instability in colorectal carcinomas: comparison of different evaluation methods and their biological significance. J Pathol 179:15–19
- Boland CR, Goel A (2010) Microsatellite instability in colorectal cancer. Gastroenterology 138:2073–87e3
- Ionov Y, Peinado MA, Malkhosyan S, Shibata D, Perucho M (1993) Ubiquitous somatic mutations in simple repeated sequences reveal a new mechanism for colonic carcinogenesis. Nature 363:558–561
- 24. Thibodeau SN, Bren G, Schaid D (1993) Microsatellite instability in cancer of the proximal colon. Science 260:816–819
- 25. Vilar E, Gruber SB (2010) Microsatellite instability in colorectal cancer-the stable evidence. Nat Rev Clin Oncol 7:153–162
- Salovaara R, Loukola A, Kristo P, Kaariainen H, Ahtola H, Eskelinen M et al (2000) Population-based molecular detection of hereditary nonpolyposis colorectal cancer. J Clin Oncol 18:2193–2200
- Peel DJ, Ziogas A, Fox EA, Gildea M, Laham B, Clements E et al (2000) Characterization of hereditary nonpolyposis colorectal cancer families from a population-based series of cases. J Natl Cancer Inst 92:1517–1522
- Fishel R, Lescoe MK, Rao MR, Copeland NG, Jenkins NA, Garber J et al (1993) The human mutator gene homolog MSH2 and its association with hereditary nonpolyposis colon cancer. Cell 75:1027–1038
- Miyaki M, Konishi M, Tanaka K, Kikuchi-Yanoshita R, Muraoka M, Yasuno M et al (1997) Germline mutation of MSH6 as the cause of hereditary nonpolyposis colorectal cancer. Nat Genet 17:271–272
- Nicolaides NC, Papadopoulos N, Liu B, Wei YF, Carter KC, Ruben SM et al (1994) Mutations of two PMS homologues in hereditary nonpolyposis colon cancer. Nature 371:75–80
- Cunningham JM, Christensen ER, Tester DJ, Kim CY, Roche PC, Burgart LJ et al (1998) Hypermethylation of the hMLH1 promoter in colon cancer with microsatellite instability. Cancer Res 58:3455–3460
- 32. Kuismanen SA, Holmberg MT, Salovaara R, de la CA, Peltomaki P (2000) Genetic and epigenetic modification of MLH1 accounts for a major share of microsatellite-unstable colorectal cancers. Am J Pathol 156:1773–1779

- 9 Synthetic Genetic Approaches in Colorectal Cancer
- Herman JG, Umar A, Polyak K, Graff JR, Ahuja N, Issa JP et al (1998) Incidence and functional consequences of hMLH1 promoter hypermethylation in colorectal carcinoma. Proc Natl Acad Sci U S A 95:6870–6875
- 34. Kane MF, Loda M, Gaida GM, Lipman J, Mishra R, Goldman H et al (1997) Methylation of the hMLH1 promoter correlates with lack of expression of hMLH1 in sporadic colon tumors and mismatch repair-defective human tumor cell lines. Cancer Res 57:808–811
- 35. Weaver BA, Cleveland DW (2006) Does an euploidy cause cancer? Curr Opin Cell Biol 18:658–67
- Foijer F, Draviam VM, Sorger PK (2008) Studying chromosome instability in the mouse. Biochim Biophys Acta 1786:73–82
- Cahill DP, Lengauer C, Yu J, Riggins GJ, Willson JK, Markowitz SD et al (1998) Mutations of mitotic checkpoint genes in human cancers. Nature 392:300–303
- Danes BS, Alm T (1979) In vitro studies on adenomatosis of the colon and rectum. J Med Genet 16:417–422
- Lengauer C, Kinzler KW, Vogelstein B (1997) Genetic instability in colorectal cancers. Nature 386:623–627
- Gardner EJ, Rogers SW, Woodward S (1982) Numerical and structural chromosome aberrations in cultured lymphocytes and cutaneous fibroblasts of patients with multiple adenomas of the colorectum. Cancer 49:1413–1419
- Gordon DJ, Resio B, Pellman D (2012) Causes and consequences of aneuploidy in cancer. Nat Rev Genet 13:189–203
- 42. Rajagopalan H, Nowak MA, Vogelstein B, Lengauer C (2003) The significance of unstable chromosomes in colorectal cancer. Nature Rev Cancer 3:695–701
- Goncalves Dos Santos Silva A, Sarkar R, Harizanova J, Guffei A, Mowat M, Garini Y et al (2008) Centromeres in cell division, evolution, nuclear organization and disease. J Cell Biochem 104:2040–2058
- Mai S (2010) Initiation of telomere-mediated chromosomal rearrangements in cancer. J Cell Biochem 109:1095–1102
- Vogelstein B, Fearon ER, Hamilton SR, Kern SE, Preisinger AC, Leppert M et al (1988) Genetic alterations during colorectal-tumor development. N Engl J Med 319:525–532
- Kaplan KB, Burds AA, Swedlow JR, Bekir SS, Sorger PK, Nathke IS (2001) A role for the Adenomatous Polyposis Coli protein in chromosome segregation. Nat Cell Biol 3:429–432
- Fodde R, Kuipers J, Rosenberg C, Smits R, Kielman M, Gaspar C et al (2001) Mutations in the APC tumour suppressor gene cause chromosomal instability. Nat Cell Biol 3:433–438
- 48. Powell SM, Zilz N, Beazer-Barclay Y, Bryan TM, Hamilton SR, Thibodeau SN et al (1992) APC mutations occur early during colorectal tumorigenesis. Nature 359:235–237
- Jallepalli PV, Lengauer C (2001) Chromosome segregation and cancer: cutting through the mystery. Nat Rev Cancer 1:109–117
- Barber TD, McManus K, Yuen KW, Reis M, Parmigiani G, Shen D et al (2008) Chromatid cohesion defects may underlie chromosome instability in human colorectal cancers. Proc Natl Acad Sci U S A 105:3443–3448
- 51. Wang Z, Cummins JM, Shen D, Cahill DP, Jallepalli PV, Wang TL et al (2004) Three classes of genes mutated in colorectal cancers with chromosomal instability. Cancer Res 64:2998–3001
- 52. Gollin SM (2005) Mechanisms leading to chromosomal instability. Sem Cancer Biol 15:33-42
- 53. Rao CV, Yamada HY (2013) Genomic instability and colon carcinogenesis: from the perspective of genes. Front Oncol 3:130
- Chang SC, Lin JK, Lin TC, Liang WY (2005) Loss of heterozygosity: an independent prognostic factor of colorectal cancer. World J Gastroenterol 11:778–784
- 55. Lurje G, Zhang W, Lenz HJ (2007) Molecular prognostic markers in locally advanced colon cancer. Clin Colorectal Cancer 6:683–690

- 56. Sheffer M, Bacolod MD, Zuk O, Giardina SF, Pincas H, Barany F et al (2009) Association of survival and disease progression with chromosomal instability: a genomic exploration of colorectal cancer. Proc Natl Acad Sci USA 106:7131–7136
- 57. Walther A, Houlston R, Tomlinson I (2008) Association between chromosomal instability and prognosis in colorectal cancer: a meta-analysis. Gut 57:941–950
- Watanabe T, Kobunai T, Yamamoto Y, Matsuda K, Ishihara S, Nozawa K et al (2012) Chromosomal instability (CIN) phenotype, CIN high or CIN low, predicts survival for colorectal cancer. J Clin Oncol 30:2256–2264
- Lee AJ, Endesfelder D, Rowan AJ, Walther A, Birkbak NJ, Futreal PA et al (2011) Chromosomal instability confers intrinsic multidrug resistance. Cancer Res 71:1858–1870
- 60. McClelland SE, Burrell RA, Swanton C (2009) Chromosomal instability: a composite phenotype that influences sensitivity to chemotherapy. Cell Cycle 8:3262–3266
- 61. Swanton C, Tomlinson I, Downward J (2006) Chromosomal instability, colorectal cancer and taxane resistance. Cell Cycle 5:818–823
- 62. Baylin SB, Jones PA (2011) A decade of exploring the cancer epigenome-biological and translational implications. Nat Rev Cancer 11:726–734
- 63. Bird AP (1986) CpG-rich islands and the function of DNA methylation. Nature 321:209-213
- 64. Toyota M, Ahuja N, Ohe-Toyota M, Herman JG, Baylin SB, Issa JP (1999) CpG island methylator phenotype in colorectal cancer. Proc Natl Acad Sci U S A 96:8681–8686
- 65. Curtin K, Slattery ML, Samowitz WS (2011) CpG island methylation in colorectal cancer: past, present and future. Patholog Res Int 2011:902674
- Issa JP, Vertino PM, Wu J, Sazawal S, Celano P, Nelkin BD et al (1993) Increased cytosine DNA-methyltransferase activity during colon cancer progression. J Natl Cancer Inst 85:1235–1240
- 67. Easwaran HP, Van Neste L, Cope L, Sen S, Mohammad HP, Pageau GJ et al (2010) Aberrant silencing of cancer-related genes by CpG hypermethylation occurs independently of their spatial organization in the nucleus. Cancer Res 70:8015–8024
- Esteller M (2007) Epigenetic gene silencing in cancer: the DNA hypermethylome. Hum Mol Genet 16(Spec No 1):R50–R59
- 69. Topkan E, Onal HC, Yavuz MN (2008) Managing liver metastases with conformal radiation therapy. J Support Oncol 6(1):9–13, 15
- Rodriguez-Bigas MA, Chang GJ, Skibber JM (2010) Multidisciplinary approach to recurrent/ unresectable rectal cancer: how to prepare for the extent of resection. Surg Oncol Clin N Am 19:847–859
- Herman J, Messersmith W, Suh WW, Blackstock W, Cosman BC, Mohiuddin M et al (2010) ACR Appropriateness Criteria: rectal cancer-metastatic disease at presentation. Curr Probl Cancer 34:201–210
- 72. Rodriguez AM, Kuo YF, Goodwin JS (2014) Risk of colorectal cancer among long-term cervical cancer survivors. Med Oncol 31:943
- Musunuru H, Mason M, Murray L, Al-Qaisieh B, Bownes P, Smith J et al (2014) Second primary cancers occurring after I-125 brachytherapy as monotherapy for early prostate cancer. Clin Oncol (R Coll Radiol) 26:210–215
- Murray L, Henry A, Hoskin P, Siebert FA, Venselaar J, PgotG ESTRO (2014) Second primary cancers after radiation for prostate cancer: a systematic review of the clinical data and impact of treatment technique. Radiother Oncol 110:213–228
- Sautter-Bihl ML, Sedlmayer F (2013) [Second primary cancers after radiotherapy in breast cancer patients]. Strahlentherapie und Onkologie: Organ der Deutschen Rontgengesellschaft [et al.] Strahlenther Onkol 189:902–903
- Longley DB, Harkin DP, Johnston PG (2003) 5-fluorouracil: mechanisms of action and clinical strategies. Nat Rev Cancer 3:330–338
- 77. Heidelberger C, Chaudhuri NK, Danneberg P, Mooren D, Griesbach L, Duschinsky R et al (1957) Fluorinated pyrimidines, a new class of tumour-inhibitory compounds. Nature 179:663–666
- Johnston PG, Kaye S (2001) Capecitabine: a novel agent for the treatment of solid tumors. Anticancer Drugs 12:639–646

- 9 Synthetic Genetic Approaches in Colorectal Cancer
- 79. Wyeth-Canada. Lederle LEUCOVORIN® calcium folinate tablets Product Monograph. Montreal, Quebec2004
- Iwamoto S, Hazama S, Kato T, Miyake Y, Fukunaga M, Matsuda C et al (2014) Multicenter phase II study of second-line cetuximab plus folinic acid/5-fluorouracil/irinotecan (FOL-FIRI) in KRAS wild-type metastatic colorectal cancer: the FLIER study. Anticancer Res 34:1967–1973
- Morganti AG, Mignogna S, Caravatta L, Deodato F, Macchia G, Plantamura NM et al (2014) FOLFIRI-bevacizumab and concurrent low-dose radiotherapy in metastatic colorectal cancer: preliminary results of a phase I-II study. J Chemother :1973947813Y0000000163 [Epub ahead of print]
- Douillard JY, Siena S, Cassidy J, Tabernero J, Burkes R, Barugel M et al (2014) Final results from PRIME: randomized phase 3 study of panitumumab with FOLFOX4 for first-line treatment of metastatic colorectal cancer. Ann Oncol 25:1346–1355
- Loree JM, Mulder KE, Ghosh S, Spratlin JL (2014) CAPOX associated with toxicities of higher grade but improved disease-free survival when compared with FOLFOX in the adjuvant treatment of stage III colon cancer. Clin Colorectal Cancer 13:172–177
- 84. Dobzhansky T (1946) Genetics of natural populations. Xiii. Recombination and variability in populations of drosophila pseudoobscura. Genetics 31:269–290
- Davierwala AP, Haynes J, Li Z, Brost RL, Robinson MD, Yu L et al (2005) The synthetic genetic interaction spectrum of essential genes. Nat Genet 37:1147–1152
- Measday V, Baetz K, Guzzo J, Yuen K, Kwok T, Sheikh B et al (2005) Systematic yeast synthetic lethal and synthetic dosage lethal screens identify genes required for chromosome segregation. Proc Natl Acad Sci U S A 102:13956–13961
- 87. Pan X, Yuan DS, Xiang D, Wang X, Sookhai-Mahadeo S, Bader JS et al (2004) A robust toolkit for functional profiling of the yeast genome. Mol Cell 16:487–496
- Tong AH, Evangelista M, Parsons AB, Xu H, Bader GD, Page N et al (2001) Systematic genetic analysis with ordered arrays of yeast deletion mutants. Science 294:2364–2368
- Tong AH, Lesage G, Bader GD, Ding H, Xu H, Xin X et al (2004) Global mapping of the yeast genetic interaction network. Science 303:808–813
- Collins SR, Miller KM, Maas NL, Roguev A, Fillingham J, Chu CS et al (2007) Functional dissection of protein complexes involved in yeast chromosome biology using a genetic interaction map. Nature 446:806–810
- 91. Fiedler D, Braberg H, Mehta M, Chechik G, Cagney G, Mukherjee P et al (2009) Functional organization of the S. cerevisiae phosphorylation network. Cell 136:952–963
- Lin YY, Qi Y, Lu JY, Pan X, Yuan DS, Zhao Y et al (2008) A comprehensive synthetic genetic interaction network governing yeast histone acetylation and deacetylation. Genes Dev 22:2062–2074
- Montpetit B, Thorne K, Barrett I, Andrews K, Jadusingh R, Hieter P et al (2005) Genomewide synthetic lethal screens identify an interaction between the nuclear envelope protein, Apq12p, and the kinetochore in Saccharomyces cerevisiae. Genetics 171:489–501
- 94. Zhao R, Davey M, Hsu YC, Kaplanek P, Tong A, Parsons AB et al (2005) Navigating the chaperone network: an integrative map of physical and genetic interactions mediated by the hsp90 chaperone. Cell 120:715–727
- Hartwell LH, Szankasi P, Roberts CJ, Murray AW, Friend SH (1997) Integrating genetic approaches into the discovery of anticancer drugs. Science 278:1064–1068
- 96. Dixon SJ, Fedyshyn Y, Koh JL, Prasad TS, Chahwan C, Chua G et al (2008) Significant conservation of synthetic lethal genetic interaction networks between distantly related eukaryotes. Proc Natl Acad Sci U S A 105:16653–16658
- McManus KJ, Barrett IJ, Nouhi Y, Hieter P (2009) Specific synthetic lethal killing of RAD54B-deficient human colorectal cancer cells by FEN1 silencing. Proc Natl Acad Sci U S A 106:3276–3281
- Cancer Genome Atlas Network (2012) Comprehensive molecular portraits of human breast tumours. Nature 490:61–70
- 99. Cancer Genome Atlas Network (2012) Comprehensive molecular characterization of human colon and rectal cancer. Nature 487:330–337

- 100. Cancer Genome Atlas Research Network (2008) Comprehensive genomic characterization defines human glioblastoma genes and core pathways. Nature 455:1061–1068
- Cancer Genome Atlas Research Network (2011) Integrated genomic analyses of ovarian carcinoma. Nature 474:609–615
- Cancer Genome Atlas Research Network (2012) Comprehensive genomic characterization of squamous cell lung cancers. Nature 489:519–525
- Roy R, Chun J, Powell SN (2012) BRCA1 and BRCA2: different roles in a common pathway of genome protection. Nat Rev Cancer 12:68–78
- El-Khamisy SF, Masutani M, Suzuki H, Caldecott KW (2003) A requirement for PARP-1 for the assembly or stability of XRCC1 nuclear foci at sites of oxidative DNA damage. Nucleic Acids Res 31:5526–5533
- Okano S, Lan L, Caldecott KW, Mori T, Yasui A (2003) Spatial and temporal cellular responses to single-strand breaks in human cells. Mol Cell Biol 23:3974–3981
- Strom CE, Johansson F, Uhlen M, Szigyarto CA, Erixon K, Helleday T (2011) Poly (ADPribose) polymerase (PARP) is not involved in base excision repair but PARP inhibition traps a single-strand intermediate. Nucleic Acids Res 39:3166–3175
- Kraus WL, Hottiger MO (2013) PARP-1 and gene regulation: progress and puzzles. Mol Aspects Med 34:1109–1123
- Bryant HE, Schultz N, Thomas HD, Parker KM, Flower D, Lopez E et al (2005) Specific killing of BRCA2-deficient tumours with inhibitors of poly(ADP-ribose) polymerase. Nature 434:913–917
- Farmer H, McCabe N, Lord CJ, Tutt AN, Johnson DA, Richardson TB et al (2005) Targeting the DNA repair defect in BRCA mutant cells as a therapeutic strategy. Nature 434:917–921
- Papeo G, Casale E, Montagnoli A, Cirla A (2013) PARP inhibitors in cancer therapy: an update. Expert Opin Ther Pat 23:503–514
- 111. van Pel DM, Barrett IJ, Shimizu Y, Sajesh BV, Guppy BJ, Pfeifer T et al (2013) An evolutionarily conserved synthetic lethal interaction network identifies FEN1 as a broad-spectrum target for anticancer therapeutic development. PLoS Genet 9:e1003254
- Sajesh BV, Bailey M, Lichtensztejn Z, Hieter P, McManus KJ (2013) Synthetic lethal targeting of superoxide dismutase 1 selectively kills RAD54B-deficient colorectal cancer cells. Genetics 195:757–767
- Culotta VC, Klomp LW, Strain J, Casareno RL, Krems B, Gitlin JD (1997) The copper chaperone for superoxide dismutase. J Biol Chem 272:23469–23472
- 114. Cerami E, Gao J, Dogrusoz U, Gross BE, Sumer SO, Aksoy BA et al (2012) The cBio cancer genomics portal: an open platform for exploring multidimensional cancer genomics data. Cancer Discov 2:401–404
- 115. Reinhardt HC, Schumacher B (2012) The p53 network: cellular and systemic DNA damage responses in aging and cancer. Trends Genet 28:128–136
- 116. Hamelin R, Laurent-Puig P, Olschwang S, Jego N, Asselain B, Remvikos Y et al (1994) Association of p53 mutations with short survival in colorectal cancer. Gastroenterology 106:42–48
- 117. Leroy K, Haioun C, Lepage E, Le Metayer N, Berger F, Labouyrie E et al (2002) p53 gene mutations are associated with poor survival in low and low-intermediate risk diffuse large B-cell lymphomas. Ann Oncol 13:1108–1115
- Wattel E, Preudhomme C, Hecquet B, Vanrumbeke M, Quesnel B, Dervite I et al (1994) p53 mutations are associated with resistance to chemotherapy and short survival in hematologic malignancies. Blood 84:3148–3157
- 119. Xie L, Gazin C, Park SM, Zhu LJ, Debily MA, Kittler EL et al (2012) A synthetic interaction screen identifies factors selectively required for proliferation and TERT transcription in p53-deficient human cancer cells. PLoS Genet 8:e1003151
- Marechal A, Zou L (2013) DNA damage sensing by the ATM and ATR kinases. Cold Spring Harb Perspect Biol 5(9):a012716
- 121. Oh S, Shin S, Janknecht R (2012) ETV1, 4 and 5: an oncogenic subfamily of ETS transcription factors. Biochim Biophys Acta 1826:1–12

- Kroll ES, Hyland KM, Hieter P, Li JJ (1996) Establishing genetic interactions by a synthetic dosage lethality phenotype. Genetics 143:95–102
- 123. Yan H, Gibson S, Tye BK (1991) Mcm2 and Mcm3, two proteins important for ARS activity, are related in structure and function. Genes Dev 5:944–957
- Li JJ, Herskowitz I (1993) Isolation of ORC6, a component of the yeast origin recognition complex by a one-hybrid system. Science 262:1870–1874
- Sajesh BV, Guppy BJ, McManus KJ (2013) Synthetic genetic targeting of genome instability in cancer. Cancers (Basel) 5:739–761
- 126. Normanno N, Tejpar S, Morgillo F, De Luca A, Van Cutsem E, Ciardiello F (2009) Implications for KRAS status and EGFR-targeted therapies in metastatic CRC. Nat Rev Clin Oncol 6:519–527
- Pylayeva-Gupta Y, Grabocka E, Bar-Sagi D (2011) RAS oncogenes: weaving a tumorigenic web. Nat Rev Cancer 11:761–774
- Tan C, Du X (2012) KRAS mutation testing in metastatic colorectal cancer. World J Gastroenterol 18:5171–5180
- Neumann J, Zeindl-Eberhart E, Kirchner T, Jung A (2009) Frequency and type of KRAS mutations in routine diagnostic analysis of metastatic colorectal cancer. Pathol Res Pract 205:858–862
- Barbie DA, Tamayo P, Boehm JS, Kim SY, Moody SE, Dunn IF et al (2009) Systematic RNA interference reveals that oncogenic KRAS-driven cancers require TBK1. Nature 462:108–112
- Luo J, Emanuele MJ, Li D, Creighton CJ, Schlabach MR, Westbrook TF et al (2009) A genome-wide RNAi screen identifies multiple synthetic lethal interactions with the Ras oncogene. Cell 137:835–848
- Scholl C, Frohling S, Dunn IF, Schinzel AC, Barbie DA, Kim SY et al (2009) Synthetic lethal interaction between oncogenic KRAS dependency and STK33 suppression in human cancer cells. Cell 137:821–834
- 133. Brauksiepe B, Mujica AO, Herrmann H, Schmidt ER (2008) The Serine/threonine kinase Stk33 exhibits autophosphorylation and phosphorylates the intermediate filament protein Vimentin. BMC Biochem 9:25
- Babij C, Zhang Y, Kurzeja RJ, Munzli A, Shehabeldin A, Fernando M et al (2011) STK33 kinase activity is nonessential in KRAS-dependent cancer cells. Cancer Res 71:5818–5826
- Spankuch-Schmitt B, Bereiter-Hahn J, Kaufmann M, Strebhardt K (2002) Effect of RNA silencing of polo-like kinase-1 (PLK1) on apoptosis and spindle formation in human cancer cells. J Natl Cancer Inst 94:1863–1877
- Liu X, Erikson RL (2003) Polo-like kinase (Plk)1 depletion induces apoptosis in cancer cells. Proc Natl Acad Sci U S A 100:5789–5794
- 137. Sun C, Hobor S, Bertotti A, Zecchin D, Huang S, Galimi F et al (2014) Intrinsic resistance to MEK inhibition in KRAS mutant lung and colon cancer through transcriptional induction of ERBB3. Cell Rep 7:86–93
- 138. Bernards R (2012) A missing link in genotype-directed cancer therapy. Cell 151:465-468
- Ebi H, Faber AC, Engelman JA, Yano S (2014) Not just gRASping at flaws: finding vulnerabilities to develop novel therapies for treating KRAS mutant cancers. Cancer Sci 105:499– 505
- 140. Migliardi G, Sassi F, Torti D, Galimi F, Zanella ER, Buscarino M et al (2012) Inhibition of MEK and PI3K/mTOR suppresses tumor growth but does not cause tumor regression in patient-derived xenografts of RAS-mutant colorectal carcinomas. Clin Cancer Res 18:2515–2525
- 141. Steckel M, Molina-Arcas M, Weigelt B, Marani M, Warne PH, Kuznetsov H et al (2012) Determination of synthetic lethal interactions in KRAS oncogene-dependent cancer cells reveals novel therapeutic targeting strategies. Cell Res 22:1227–1245
- 142. Corcoran RB, Cheng KA, Hata AN, Faber AC, Ebi H, Coffee EM et al (2013) Synthetic lethal interaction of combined BCL-XL and MEK inhibition promotes tumor regressions in KRAS mutant cancer models. Cancer Cell 23:121–128

- 143. Furey TS (2012) ChIP-seq and beyond: new and improved methodologies to detect and characterize protein-DNA interactions. Nat Rev Genet 13:840–852
- Neff T, Armstrong SA (2009) Chromatin maps, histone modifications and leukemia. Leukemia 23:1243–1251
- 145. Robertson AG, Bilenky M, Tam A, Zhao Y, Zeng T, Thiessen N et al (2008) Genome-wide relationship between histone H3 lysine 4 mono-and tri-methylation and transcription factor binding. Genome Res 18:1906–1917
- Varley KE, Gertz J, Bowling KM, Parker SL, Reddy TE, Pauli-Behn F et al (2013) Dynamic DNA methylation across diverse human cell lines and tissues. Genome Res 23:555–567
- 147. Australian Institute of Health and Welfare & Australasian Association of Cancer Registries (2012) Cancer in Australia: an overview, 2012. Cancer series no. 74. Cat. no. CAN 70. Canberra, Australia: AIHW
- Canadian Cancer Society's Advisory Committee on Cancer Statistics (2013) Canadian cancer statistics 2013. Toronto: Canadian Cancer Society
- 149. Cancer Research UK (2014) Cancer statistics report: cancer incidence and mortality in the UK. London: Cancer Research UK
- 150. American Cancer Society.(2014) Colorectal cancer facts & figs. 2014–2016. Atlanta: American Cancer Society

Chapter 10 Nanomedicine—Nanoparticles in Cancer Imaging and Therapy

Alexandra M. N. Hauser-Kawaguchi and Leonard G. Luyt

Abstract Nanomedicine refers to the application of nanotechnology in medicine, and endeavors to diagnose, treat, and/or monitor disease on a nanoscale. Cancer nanotechnology is a quickly evolving field of interdisciplinary research that involves the biomedical application of nanoparticles, which are nanoscale devices that are able to overcome biological barriers, specifically recognize a single type of cancer cell, and accumulate preferentially in tumors. Medical applications with nanoparticles are growing, as they have the potential to offer novel methods of non-invasive cancer detection, diagnosis, and treatment. Tumor targeting ligands, such as antibodies, peptides, or small molecules, can be attached to nanoparticles for targeting of tumor antigens and vasculatures with high affinity and specificity. In addition, diagnostic agents (i.e. optical, radiolabels, or magnetic) and chemotherapeutic drugs can be integrated into their design for more efficient imaging and treatment of the tumor with fewer side effects. Recent advances in nanomedicine raise exciting possibilities for future nanoparticle applications in personalized cancer therapy.

Keywords Nanomedicine \cdot Nanoparticles \cdot Nanoplatforms \cdot Cancer \cdot Cancer therapy \cdot Targeted delivery \cdot Drug delivery \cdot Molecular imaging \cdot Extravasation \cdot Tumor targeting

Abbreviations

CEST	Chemical exchange saturation transfer
DOTA	1,4,7,10-tetraazacyclodocecane-N, N'N", N"'-tetraacetic acid
DOX	Doxorubicin
EPR	Enhanced permeability and retention
GRP	Gastrin releasing peptide

L. G. Luyt (🖂)

London Regional Cancer Program, Cancer Research Laboratory Program, 790 Commissioners Road E, London, ON N6A 4L6, Canada

Departments of Oncology, Chemistry, and Medical Imaging, The University of Western Ontario, London, ON N6A 5C1, Canada

e-mail: lluyt@uwo.ca

A. M. N. Hauser-Kawaguchi Department of Chemistry, The University of Western Ontario, London, ON N6A 5C1, Canada

© Springer International Publishing Switzerland 2015

C. Maxwell, C. Roskelley (eds.), *Genomic Instability and Cancer Metastasis*, Cancer Metastasis - Biology and Treatment 20, DOI 10.1007/978-3-319-12136-9_10

HSA	Human serum albumin
ID/g	Injected dose per gram
MRI	Magnetic resonance imaging
MTX	Methotrexate
NIRF	Near infrared fluorescence
NPs	Nanoparticles
P-gp	P-glycoprotein
PEG	Polyethylene glycol
PET	Positron emission tomography
PLGA	D, L-lactide co-glycolide
PTX	Paclitaxel
RES	Reticuloendothelial system
RGD	Arginine-glycine-aspartic acid
SPECT	Single photon emission computed tomography
VAP	Vapreotide
VEGF/R	Vascular endothelial growth factor/receptor

Types of Nanoparticle Platforms

Gold NP
Cross-linked Iron Oxide NPs
Carbon nanotube
Cowpea mosaic virus
Iron oxide NPs
Manganese-doped magnetism-engineered iron oxide
Poly(amidoamine) dendrimer
Quantum Dot
Superparamagnetic iron oxide NPs
Single-walled carbon nanotube
Viral Nanoparticle

Introduction

Nanomedicine refers to the application of nanotechnology in medicine, and endeavors to diagnose, treat, and/or monitor disease on a nanoscale. Specifically, cancer nanotechnology is a quickly evolving field of interdisciplinary research that involves elements of biology, chemistry, engineering, and medicine, with the aim of producing novel methods of noninvasive cancer detection, diagnosis, and treatment [1–3]. This is done with nanoparticles (NPs), which are nanoscale devices that are able to overcome biological barriers, specifically recognize a single type of cancer cell, and accumulate preferentially in tumors.

NPs are submicron-sized synthetic particles that range from one to hundreds of nanometers in diameter. They are considered to have great potential for medicinal applications because they are stable, have large payloads, and have the capacity for multiple, simultaneous applications due to their size and high surface area:volume ratio [4]. In addition, their distinct size—bigger than many biological molecules, such as proteins, receptors, and antibodies, but 100 to 1000 times smaller than human cells—allows them to act uniquely when they are introduced *in vivo* for imaging and therapeutic purposes. The applications of nanoparticles in medicine are numerous: they can be administered by all routes; they can be developed and formulated rapidly; they increase the aqueous solubility of the attached drug; they protect the drug from degradation; they can release the drug at a controlled rate; they improve the bioavailability of the drug; they enable targeted delivery of the drug; and they decrease the adverse side effects of the drug. Indeed, these nanoscale particles are capable of unique interactions with biological molecules both at the surface and inside cells, which offers unlimited possibilities for future applications in cancer therapy.

Contemporary diagnostic classifications do not encompass the diverse heterogeneity of tumors and are incapable of identifying an appropriate method of treatment or projecting a patient's outcome. Current drug-related cancer treatments originated from traditional methods of drug-based disease treatments. However, most current anticancer drugs are nonselective, which results in their accumulation in both cancerous and normal cells. In addition, a tumor is often diagnosed at a later stage of its development, after it has metastasized to other parts of the body. Increasing concern over the toxicity of non-targeted chemotherapeutic drugs, which can cause severe tissue damage when localized in non-diseased tissue, has resulted in an interest in designing NPs capable of recognizing and targeting tumor tissue specifically. In doing so, NPs are now capable of delivering imaging and therapeutic agents to the target of interest—the tumor—and offer the potential for individualized cancer treatment.

This review will provide an overview of the most well studied nanoplatforms for imaging of tumors and the delivery of therapeutic drugs that are currently being employed. The most well studied nanoplatforms include quantum dots (QDs), liposomes, dendrimers, polymeric NPs, iron oxide nanoparticles (IONPs) and their derivatives, gold nanoparticles, and carbon nanotubes. This review will also explore the current challenges and exciting new approaches being utilized in the design of NPs for cancer imaging and therapy.

Nanoparticles with Medical Applications

Nanoparticles are constructed of a wide range of materials, from organic molecules such as liposomes, dendrimers or carbon nanotubes, through to inorganic structures including cadmium, iron oxide and gold. Table 10.1 describes the predominant nanoplatforms currently in use and provides examples of their applicability to nanomedicine.

Table 10.1 Nar	noparticle characteristics and	medicinal applications			
NP platform	Composition and structure	Characteristics	Application	Examples	Ref
Quantum Dot	CdSe, CdTe	Intense intrinsic fluorescence and photostability	Imaging: optical fluorescence	1. PEG-conjugated QDs	[26]
		Toxicity concerns with material of construction		2. Antibody-conjugated QDs	
Liposome	Phospholipid	Amphiphilic, biocompatible	Imaging	1. Doxil (PEGylated liposo- mal DOX)	[26–30]
	Closed colloidal structure composed of lipid bilayer	Ease of modification	Drug delivery	2. Myocet (Non-PEGylated liposomal DOX)	
		Suppresses tumor growth and inhibits metasta- sis of lung cancer		3. DauoXome (Liposomal daunorubicin)	
Dendrimer	Various organic monomers	Highly branched for high ligand density	Imaging	1. PAMAM-MTX	[27, 31, 32]
	Hyperbranched synthetic polymer with regular pat- tern and repeated units	Ease of functionalization	Drug delivery	2. PAMAM-platinate	
		Biodistribution can be tuned			
		Structural and chemical homogeneity			
		Controlled degradation			
Polymeric NP	Linear polymer	Water-soluble	Drug delivery	1. Abraxane (Albumin Taxol)	[33–35]
		Biodegradable		2. Xyotax (PGA-Taxol)	
		Surface modification		3. PK1(HPMA-DOX)	
		Selective targeting of and accumulation in tumor			

208

Table 10.1 (col	ntinued)				
NP platform	Composition and structure	Characteristics	Application	Examples	Ref
Iron oxide NP	Iron oxide	Magnetic surface	Imaging: con- trast agent for MRI	 TNT AntiEpCAM (poly- mer-coated IONP) 	[36]
		Inherent MRI contrast agent		2. SPIO/DOX-NP	
		Biocompatible		3. SPIO/PTX-NP	
Gold NP	Gold	Easily conjugated surface	Imaging: con- trast agent for MR	1. Aurimmune (CYT-6091)	[26]
		Spectroscopic advantages for sensitive tumor detection	Drug delivery	2. Verigene platform	
		Nontoxic	Radiation sensi- tizer for therapy		
		Biocompatible			
Carbon Nanotube	Carbon	Water-soluble (if modified)	Imaging	1. CNT-MTX	[27, 37, 38]
	Carbon cylinders made of graphene	Cytotoxic (without chemical modification)	Drug delivery	2. CNT-amphotericin B	
		Multifunctional			

Quantum Dots (QDs)

QDs are inorganic semiconductor nanocrystals that are the most widely studied nanoplatform for optical imaging applications due to their intrinsic fluorescence. They benefit from small size (as small as 2 nm), versatile surface chemistry, high quantum yields, and outstanding optical properties, including high resolution, high sensitivity, and high inherent photostability and brightness [5–7]. In addition, they have continuous absorption spectra that range from UV to near-infrared wavelengths, as well as long fluorescence lifetimes (>10 h), and large effective Stoke shifts [5–7]. Despite the potential for the toxic release of Cd from their core, *in vivo* studies have shown that they do not exhibit acute toxicity when their surfaces are properly coated. Thus, a number of *in vitro* and cell-based applications exist for QDs, including high-resolution cellular imaging, long-term *in vivo* cell tracking observation, diagnostic work, and tumor targeting [5–9].

Liposomes

Liposomes are self-assembling closed vesicles that have a spherical shape, in which natural phospholipid bilayers surround a central aqueous core that houses a therapeutic drug molecule. Liposomes are biocompatible, biodegradable, biologically inert, and weakly immunogenic. Liposomes are disintegrated by electrostatic, hydrophobic, and van der Waals forces, and can undergo opsonization, resulting in their rapid clearance from the body after their surfaces are coated with protein. As a result, they require steric stabilization, which is achieved by surface coatings with inert polymers, such as polyethylene glycol (PEG). The addition of PEG to the surface of liposomes has resulted in "STEALTH" technology, whereby the intact NP has been known to circulate throughout the body for days without clearance. This phenomenon is particularly useful for the delivery of therapeutic drug molecules, especially when combined with the capability of functionalizing liposome surfaces with targeting ligands or antibodies that assist in directing the nanoplatform to the target of interest (i.e. the tumor). Consequently, targeted liposomal vesicles are able to release the drug molecule in high concentrations at the cell membrane.

Dendrimers

Dendrimers are a unique class of polymeric macromolecules. They are treelike structures that are made up of multiple repeating perfectly branched polymeric monomers emerging from a central core. The branching units may be synthesized from the central core and emerge radially outwards (divergent method), or from the periphery inwards towards the core (convergent method). Depending on how the particular dendrimer is synthesized, it is possible to achieve precise control over its molecular shape, dimensions, density, polarity, flexibility, and solubility by choosing different building/branching units and surface functional groups [10, 11]. In addition, the nanoparticle's chemistry permits that several modifications can take place to incorporate certain imaging agents, targeting ligands, and other components to increase tumor specificity.

Polymeric NPs

Polymeric NPs offer a surface with high potential for modification and functionalization with different targeting ligands. In addition, they are biocompatible, biodegradable, and demonstrate good pharmacokinetic control in the body [12]. Depending on the method of preparation, polymeric NPs can take different shapes, including nanoparticles, nanospheres, or nanocapsules. Nanospheres are spherical structures made up of a matrix-like system, in which the drug can be entrapped, adsorbed at their surface, or encapsulated in the matrix. Nanocapsules, on the other hand, are a vesicular system that is made up of a polymeric shell with a central cavity or core, which houses the drug [13–15]. The core may be made of solids, liquids, or gas, with the environment often being aqueous or oily [13].

Iron Oxide NPs (IONPs)

IONPs are the predominant class of inorganic nanoparticle being used for imaging tumors [16]. They offer an advantage over other nanoplatforms because they can be visualized by magnetic resonance imaging (MRI) due to their magnetic property and are an intrinsic contrast agent. In addition, IONPs can be guided towards the target of interest (i.e. the tumor) by an external magnetic field, or can be heated to provide hyperthermia for cancer therapy [17]. Several types of nanosize iron oxides have been studied, including those constructed of magnetite, Fe₃O₄, and maghemite, γ -Fe₂O₃. Magnetite is a common construction for the nanoplatform for medical applications as its biocompatibility has been well studied and proven [18]. With proper surface coatings, IONPs can be dissolved in solvents, which results in a homogenous suspension, called ferro-fluids [19]. In this state, the magnetic particles can be used for a number of different *in vitro* and *in vivo* applications, as they interact with an external magnetic field and if directed to a specific location, are able to facilitate medical diagnosis by MRI.

Gold NPs (AuNPs)

Gold NPs are another class of commonly employed inorganic metal NPs. The advantages of gold NPs are multifold: the gold core is relatively inert and non-toxic, making them biocompatible [20, 21]; they can be easily synthesized by simple, reliable, and low-cost methods, with core sizes ranging from 1 to 150 nm by changing
simple parameters; their surfaces can be easily functionalized by a number of biomolecules due to the presence of a negative charge; and their photophysical properties allow therapeutic drugs to be released in remote places [22]. Gold NPs are useful for medical applications due to their unique physicochemical properties, which include extremely small size, large surface area-to-mass ratio, intriguing optical properties, and superior surface reactivity. Although gold NPs include various physical dimensions and shapes, including nanospheres, nanorods, nanoshells, and nanocages, the most common type is the gold nanosphere, which has a bright red colour in aqueous solution [23]. In this review, the term "gold nanoparticle" will be used as a general term to describe the collection of gold nanoplatforms, unless a specific subtype is given.

Carbon Nanotubes

Carbon nanotubes are carbon cylinders that are made of graphene. They can be single or multi-walled, and have a hollow cage-like architecture [24]. Carbon nanotubes are completely insoluble in all solvents, resulting in toxicity concerns; however, when they are chemically modified, they can become water-soluble for introduction into the body [25]. Atoms may be trapped inside the cylinder, while their surfaces may be functionalized with a number of different active molecules, including peptides, antibodies, and therapeutic drugs [25]. In addition, carbon nanotubes are able to carry multiple covalent functionalizations on their sidewall and on the tips of the cylinders, giving them the advantage of being able to carry several molecules at one time.

Nanoparticles as a Platform for Nanocarrier Design

Nanoparticles benefit from a combination of different physical and chemical properties that facilitate their use with biological systems for real-time molecular imaging, intracellular uptake, and drug release. Specifically, their compact size and versatile surface modification strategies enable them to be a powerful platform for nanocarrier design.

Nanoscale Dimensions

NPs are synthetic structures on the nanometer scale. However, the different NPs used in nanomedicine vary in size, shape, and functionality. One of the most studied and prototypic NP is the QD, which consist of hundreds to a few thousand atoms, and possess a very small core size of only 2–10 nm in diameter, and are therefore,

one of the smallest platforms for NP-based drug delivery vehicle engineering [39]. Particles of this size offer advantages because they can be non-intrusively incorporated within larger drug delivery vehicles as tracers for imaging and monitoring intracellular trafficking and biodistribution, and they can be released from larger carriers, providing insight into the redistribution and eventual clearance of a drug or other NP component [39].

In practice, however, most nanocarriers are of a larger size, as they can better accommodate a wider range of materials, can provide more space for drug loading, and can integrate additional functionalities. For example, liposomes and dendrimers are popular drug delivery vehicles, and are "soft" and flexible NPs, that are able to penetrate biological membranes due to their flexibility [40]. Liposomes are made up of a lipid bilayer surrounding a water core hosting a drug, and they range in size from a minimal diameter of 30 nm to several microns [40]. Their advantage lies in their versatility: they have the capacity to carry diverse cargo; have proven to be stable in blood circulation; and they have on-demand drug release in response to intracellular or external stimuli [41-43]. Dendrimers, on the other hand, are the main polymeric architectures that are used in nanomedicine [40]. They are a unique class of repeatedly branched polymeric molecules with a nearly perfect 3-dimesional geometric pattern. Dendrimers range in size, and can be as small as 1.9 nm for a first generation dendrimer and 4.4 nm for a fourth generation dendrimer [44], where the generation number refers to the number of repeating branching units that are added during its synthesis. In addition, their chemistry permits several modifications to incorporate certain imaging agents, targeting ligands, and other components to increase tumor specificity.

Versatile Surface Chemistry

A number of different modifications to the surface of NPs facilitate specific targeting, such as to a tumor. These modifications complement the core of the NP and create a highly amenable platform for nanocarrier design.

Polyethylene Glycol

PEG, a coiled polymer that is made up of repeating ethylene ether units, is typically added to the surface of any NP that is injected intravenously for tumor-targeting, as PEG assists in extending the circulation time of NPs *in vivo*. To increase steric stability *in vitro* (in buffers, for storage), and *in vivo* (longevity in biological circulation), the FDA-approved protective layer can be formed on the outer surface of NPs made from the hydrophilic PEG polymer layer, as solubility in buffer or serum increases due to the ethylene glycol subunits [45]. In addition, PEG prevents NP uptake by macrophages and the reticuloendothelial system (RES), and inhibits

their interaction with plasma proteins by reducing charge-based interactions that are typical of proteins or other small molecules [41, 45–47]. PEG is often the material of choice because it demonstrates increased hydrophilicity and flexibility [48]. The size and density of the PEG layer greatly influences NP circulation time and accumulation in tumors [40,45].

NP type is the most important consideration for stability and circulation time in vivo, and it is most affected by factors such as size, composition, and charge of the nanoplatform [45]. For example, NPs with positive surface charges and diameters >100 nm are rapidly cleared from the body despite careful strategies in PEG modification. In addition, PEG modification strategies vary between different NP types; PEG types suitable for liposomes cannot be successfully used for solid, metal-based NPs, as liposomes mimic naturally-occurring entities that circulate throughout the body better than other NP types, and therefore, require less strict PEGylation procedures [45]. As a result, liposomes are the most widely studied PEG-modified NP platform to date [45]. PEG-lipid conjugates can be incorporated into the lipid film of liposomes by several methods: during hydration of the liposome, PEG polymers can be incorporated directly into its lipid film; *post-conjugation*, which involves functionalized PEG being covalently attached to the pre-formed NP; and post-insertion, which requires that the pre-formed liposome be incubated with PEG-lipid conjugates in aqueous solution, resulting in micellar formation of the PEG-lipids due to the amphiphilic nature of PEG, and their subsequent insertion into the liposomes [49, 50].

Inorganic NPs, on the other hand, are subject to different methods of coating the NP surface with the layers necessary for colloidal stability. The NPs are frequently co-precipitated with a multitude of polymers or cross-linked polymers, which enhances the NPs monodispersity [51]. IONPs have been successfully coated with dextran, albumin, and PEG (MW 5000) for improved biocompatibility, which resulted in an improved $t_{1/2}$ by up to 200 min [51–53]. Gold NPs, on the other hand, are better suited for PEG grafting; thiol-PEG, bifunctional PEG, and sulfydrylated PEG have all been used in coating the NP surface for improving colloidal stability and biocompatibility [54]. In addition, PEG bidentate ligands (PEG-thioctic acid and PEG-dihyrolipoic acid) were recently shown to significantly improve the stability in biological media of both QDs and gold NPs [55]. Unfortunately, the effects of the size of PEG and the density of a layer are still unknown for inorganic NPs, particularly with respect to NP kinetics and accumulation in tumors, as inorganic NPs are significantly smaller than lipid-based NPs.

The efficiency of a PEG layer in improving steric stability of the NP depends on the arrangement of the individual PEG polymers on the NP surface [56–59]. PEG molecules are made up of one end that attaches to the NP surface (designated R1), while the other distal terminal end interacts with the solvent environment (designated R2). A number of ethylene glycol residues complete the space between R1 and R2 to create PEG of varying lengths. Poor PEG organization can impede NP interaction and uptake by tumor cells. While a denser PEG layer is achieved by incorporating a greater number of PEG-lipids, NPs bear a maximum molar percentage of PEG-lipids that can be incorporated into their lipid layer before their effects become unfavorable. For example, liposomal membranes are held together by non-covalent bonds, which can cause the PEG-lipid conjugates to become dissociated from the lipid layer when PEG densities are not ideal [40]. Similarly, PEG conformation is important for it to carry out its proper function of increasing the NPs retention time by preventing its interaction with plasma proteins. A *brush*like conformation, which occurs when the distance between the individual PEGlipids, D, is less than the volume that each flexible PEG polymer cloud occupies, more commonly referred to as the Flory dimension, Rf, (D<Rf) is superior to the *mushroom* configuration, which occurs when the opposite is true (D>Rf) [60–63]. Therefore, increasing the concentration of PEG-lipids increases the PEG density, which subsequently decreases the distance, D, between each PEG molecule, and ultimately results in the PEG chains adopting the undesired mushroom shape. The brush configuration, which is a more linear conformation than the mushroom configuration, and therefore, allows for denser polymer coatings, is preferred because it produces greater protein repulsion and results in a longer circulation time [64].

Targeted Ligands

A targeting moiety can be added onto the surface of a NP to enable the selective recognition of the target of interest, such as a tumor cell. Without a targeting ligand on its surface, NPs are only able to recognize and interact with cell membranes non-specifically; such interactions can be insignificant when the surface of the NP is covered with PEG.

Cancer cells exhibit many of the same characteristics as healthy cells, which poses as a major difficulty in targeting these unhealthy cells with specificity. Thus, ligands are designed to target receptors that may be highly over-expressed on tumor cells, but that are often less prominently expressed on normal cells. The targeting layer, which is the most peripheral component of the NP, interacts primarily with cell membranes, and therefore, must exhibit the right conformation and a high affinity for its specific target in order to interact and possibly be internalized by the cell. Consequently, it is important that ligands be conjugated onto NPs in the most favorable way that allows them to preserve their high affinity for their targeting receptors.

Targeting moieties such as antibodies (full and fragmented), peptides, small molecules, and aptamers, have all been proven to facilitate NP targeting of cancer cells. For example, antibodies against receptors known to be over-expressed in tumors, such as the HER-2 receptor, transferrin receptor or epidermal growth factor receptor (EGFR), can be attached to NPs to provide cancer targeting. Similarly, small peptides with receptor specificity can be added onto NPs for targeting. For example, the arginine-glycine-aspartic acid (RGD) peptide, which has a high affinity for $\alpha_v\beta_3$ integrin receptors over-expressed on angiogenic vasculatures, is a common example of a short peptide sequence used to functionalize NP surfaces. Specific examples of these and other targeting approaches will be discussed in the subsequent sections on imaging and therapy.

Successful receptor targeting generally requires that ligands be affixed to the end of the PEG spacer. The purpose of the PEG spacer is to provide distance between the solid NP platform and the targeting ligand, thereby providing the ligand with space and flexibility to interact with its receptor. In addition, the PEG chain length is an important consideration for effective targeting. A PEG spacer that has a longer chain than the PEG layer already present for improved colloidal stability can cause the ligand attached to its end to become buried within it, as it folds over to form a mushroom-like shape, rather than maintain its brush conformation [65]. As mentioned previously, the brush conformation is the desired configuration for PEG attached to a NP's surface, as it permits the ligand to extend past the NP core, and to better interact with proteins on the cellular membrane. Thus, ligands attached to long-chained PEG spacers are at risk of being less exposed to their target receptors than without the PEG spacer. For this reason, the ligand-conjugated PEG spacer should be of the same length as the non-conjugated free PEG chains.

While the presence of a PEG spacer improves targeting efficiency, the concentration of surface ligands is also important in the design of a targeting probe. A higher concentration of ligands attached to the surface of a NP is intuitively thought to increase the probability of interacting with and targeting receptors on cancer cells due to the positive effects of multivalency. However, the opposite may also be true, as higher concentrations of ligands may crowd out and negate the stabilizing effects of PEG, resulting in poorer NP delivery to the site of interest. NPs functionalized with high concentrations of ligands have been shown to accumulate in the liver and the spleen at faster rates and in higher concentrations than those NPs whose surfaces are conjugated with lower concentrations of ligands [66]. In addition, NPs with higher ligand densities localize in lower concentrations in tumor cells [66]. This indicates that high concentrations of non-PEG structures, such as targeting ligands, on NPs oppose the stabilizing effects of the PEG layer, and permit the unwanted recognition of those NPs by spleen-associated macrophages and plasma proteins.

Nanoparticles for Cancer Imaging

Many imaging modalities may be used to detect metastatic cancer and the use of NPs may improve the sensitivity and specificity for cancer imaging. Various nanoplatforms are available that allow for interaction with cancer biological targets either on the surface of or inside cells. Some NPs exhibit properties that provide them with intrinsic imaging capabilities, while others require surface functionalization in order for them to be used as imaging agents.

Molecular Imaging

Molecular imaging is a non-invasive method of characterizing biological processes and measuring their changes at cellular and molecular levels. In nanomedicine, NPs decorated with targeting ligands, imaging labels, or therapeutic drugs can be monitored in real-time as they move through a biological organism, and therefore,



Fig. 10.1 A schematic illustration of the different nanoplatforms used for multimodality imaging. Tools used for imaging are attached to the nanoparticle core through a linking molecule, such as a spacer, peptide linker, or antibody

provide a comprehensive understanding of the delivery, administration, and effects of a drug nanocarrier. A number of different imaging modalities exist for use with NPs, and each of which depends upon different surface modifications (Fig. 10.1).

Optical Imaging

Optical imaging is a technique used to characterize a biological system using nonionizing radiation and the spectral properties of photons. Most commonly, fluorescence is utilized, whereby excitation light is directed to the target of interest and the emission light is collected at a shifted wavelength, which produces an image. The benefit of using optical imaging is that it reduces patient exposure to harmful radiation, and is therefore, a safer method of imaging tumors. In addition, it produces images faster, and can therefore be applied to lengthy and repeated procedures to monitor the progression of a particular disease. On the other hand, optical imaging is limited by poor tissue penetration, and therefore, the imaging information is surface-weighted [67]. As a result, the imaging modality is generally not quantitative, and is used almost entirely to examine shallow lesions, and subcutaneous or surgically exposed tumors [23, 67]. QDs are the most well studied platform for optical imaging applications. Their natural fluorescence in two spectral windows at near-infrared wavelengths, 700–900 nm 1200–1600 nm [68], enables them to be used as a direct measure of brightness, and therefore, provides them with the potential to be a superior near-infrared fluorescence (NIRF) imaging probe. The use of QDs *in vivo* requires that they reach their specific target of interest without alteration; therefore, their surfaces are often outfitted with a variety of targeting molecules that assist in their delivery. However, specific targeting and imaging with QDs is difficult due to their relatively large size (>20 nm in hydrodynamic diameter) and their short circulation half-life. In one report, the size of QD705-RGD (approximately 20 nm in diameter) prevented efficient extravasation, and thus, mainly targeted tumor vasculature instead of tumor cells [69].

Specific targeting with QDs *in vivo* is possible by functionalizing their surfaces with peptides as targeting ligands, and the peptide-conjugated QDs can subsequently be tracked and imaged [70]. Cai et al. has studied QDs for biomedical applications extensively, and demonstrated the *in vivo* targeted imaging of tumor vasculature using QDs [69]. One of the most common examples of tumor-targeting with QDs is the use of the tripeptide sequence RGD to target integrin $\alpha_v \beta_3$, a cell adhesion molecule that is over-expressed on activated endothelial cells of tumor neovasculature and some tumor cells but that has limited expression on mature endothelial cells [71]. Studies performed by Cai et al. showed that peptides containing RGD that were conjugated to QD705, which has a maximum emission at 705 nm, were able to bind integrin $\alpha_v \beta_3$ specifically with high affinity *in vitro* and *ex vivo*, and that imaging with NIRF was successful [69].

In addition to peptide-conjugated QDs, antibodies can be conjugated to the NPs to create a nanoplatform that targets tumor cells specifically. A study by Yu et al. revealed that an anti-alpha-fetoprotein antibody, a marker for hepatocellular carcinoma cell lines, could be added to QD surfaces and be used to image tumors *in vivo* [72]. Tada et al. used high-resolution intravital confocal fluorescence microscopy to monitor tumor targeting by antibody-conjugated QDs in real-time [73]. The study demonstrated that vascular transport, extravasation, binding to cancer cells, and cellular internalization in a living mouse could be observed at a single NP level [73]. In addition, quantitative determination of nanocarrier transport kinetics was possible, which enabled the rate-determining step of nanoparticle drug delivery to be identified [73].

Radionuclide-Based Imaging

Radionuclide-based imaging, also commonly referred to as nuclear imaging, includes single photon emission computed tomography (SPECT) and positron emission tomography (PET). It uses internal radiation that is introduced through a targeted molecule that is labeled with a radioisotope at doses far below those that may cause pharmacologic effects. Nuclear imaging offers greater sensitivity than other imaging modalities with no tissue penetration limits [74], and it is quantitative. Greater sensitivity from nuclear imaging is the result of advancements in hardware development and improvements in image-processing algorithms [75]. However, despite its clear advantages, the disadvantage is that both SPECT and PET result in lower resolution images than other modalities. For this reason, radiolabeled NPs are used almost exclusively for studying pharmacokinetic properties of new devices or nanoparticle vehicles [76]. Most radiolabeled NPs are made up of a core, a targeting biomolecule, and the radioisotope. The most commonly employed radioisotopes used with NPs for PET imaging are ¹⁸F ($t_{1/2}$ =109.8 min), ⁶⁸Ga ($t_{1/2}$ =68.1 min), and ⁶⁴Cu ($t_{1/2}$ =12.7 h), and those used in SPECT imaging are ^{99m}Tc ($t_{1/2}$ =6 h), and ¹¹¹In ($t_{1/2}$ =2.8 days).

PET Imaging

⁶⁴Cu-labeled NPs

To this point, QDs, IONPs, single-walled carbon nanotube (SWNT) and gold NPs, have all been radiolabeled with ⁶⁴Cu ligands for PET/MRI or PET/NIRF imaging [77]. Combining two different methods of imaging permits the same molecular target to be imaged in order to obtain additional information, such as anatomical and molecular information, which would not normally be achieved by a single modality. Thus, the combination of modalities used to study the same target enhances diagnostic accuracy due to improved sensitivity and greater resolution [77].

QDs have recently been reported for use as a PET/NIRF probe. QDs have been conjugated with the RGD peptide sequence and 1,4,7,10–tetraazacyclodocecane-N, N'N", N"'-tetraacetic acid (DOTA) chelator for integrin $\alpha_{\nu}\beta_3$ imaging, as well as with VEGF and DOTA for VEGF imaging [78, 79]. In both cases, a strong correlation between the *in vivo* PET images obtained and the *ex vivo* NIRF images of uptake of both the ⁶⁴Cu-DOTA-QD-RGD and ⁶⁴Cu-DOTA-QD-VEGF probes was observed. In addition, it was demonstrated that the ⁶⁴Cu-DOTA-QD-RGD probe targets the tumor vasculature through RGD-integrin interaction, and with little extravasation [78]. It was concluded that this dual-modality probe is superior to solely using optical imaging, as there was little observed toxicity and as it was able to overcome the tissue penetration limits found in optical imaging.

IONPs have magnetic properties that have allowed them to be extensively employed as direct contrast agents in MRI. A dual-modality PET/MRI probe, ⁶⁴Cu-DO-TA-IONP-RGD (diameter 45±10 nm), was created to image integrin $\alpha_v\beta_3$ expression [80]. Each IONP probe was made up of 35 RGD peptides and 30 DOTA chelators [80]. In addition, superparamagnetic iron oxides (SPIO) were functionalized with cRGDfC and 1, 4, 7-triazacyclononane-N, N', N"-triacetic-thiol and labeled with ⁶⁴Cu, resulting in strong tumor-targeting capability and tumor contrast in U87MG cells when imaged by PET/MRI [81].

The biodistribution of SWNTs functionalized with PEG-DOTA, RGD, and subsequently labeled with ⁶⁴Cu was studied by Liu et al. (2007) by PET, *ex vivo* biodistribution, and Raman spectroscopy. The results indicated that these SWNT probes were highly stable *in vivo*, and that when efficiently PEGylated, they have a relatively longer circulation half life (~2 h) and are subject to lower uptake by the RES [82]. In addition, it was found that the ⁶⁴Cu-DOTA-PEG-SWNT-RGD probe was able to target integrin $\alpha_v\beta_3$ -positive tumors in mice with high efficiency, resulting in high tumor accumulation (~7% at 1 h) and that there was limited retention in the kidneys and efficient renal clearance [82].

¹⁸F-labeled NPs

Fluoride-18 is the most widely used PET isotope and numerous examples of ¹⁸F-labelled NPs have been reported. Devaraj et al. synthesized and studied a modified ¹⁸F-labeled trimodal IONP (¹⁸F-CLIO). The particles were made up of a SPIO core shell that was cross-linked with dextran molecules, and functionalized with ¹⁸F in high concentration by means of azide-alkyne cycloaddition click chemistry. *In vivo* characterization of the probe determined that it was suitable for PET, NIRF, and MRI. When applied for *in vivo* PET/MRI imaging of lymph nodes for cancer metastasis detection, the probe was able to identify small lymph nodes as well as the precise anatomical information [83]. Analogously, Jarrett et al. designed a probe using IONPs coated with dextran sulfate that could be labeled with ¹⁸F for PET/MRI to target vascular inflammation [84].

Recently, Zhu et al. reported on ¹⁸F-labeled gold NPs that were also modified to contain PEG chains and thus have improved water solubility suitable for *in vivo* use [85]. Through the addition of the cyclic peptide octreotate, this NP system has potential for imaging of neuroendocrine metastasis by targeting the somatostatin receptor.

SPECT Imaging

Integrin $\alpha_{v}\beta_{3}$, which is a marker of angiogenic vascular tissue, has been targeted and imaged with ^{99m}Tc-labeled RGD peptides conjugated to gold NPs [86]. Specifically, binding assays performed in C6 glioma cancer cells, which overexpress $\alpha_{v}\beta_{3}$ receptors, proved that the ^{99m}Tc-AuNP-RGD specifically recognized the receptors [87]. The gastrin-releasing peptide (GRP) receptor, which is overexpressed in prostate cancer and breast cancer, has been targeted with ^{99m}Tc-Lys³-bombesin [88, 89]. For example, ^{99m}Tc-AuNP-Lys³-bombesin was shown to have specific recognition for GRP receptors in PC3 cancer cells [90]. Studies with ^{99m}Tc- and ¹¹¹In-labeled carbon nanotubes [91–94], ¹²⁵I-labeled silver NPs [95] and ^{99m}Tc-labeled IONPs [96] for SPECT imaging have also been reported.

Molecular MRI

MRI is a non-invasive method of imaging biological organisms by placing the subject in a magnetic field, and is based on the interaction of particular nuclei, such as protons, with one another and with molecules found in the tissue of interest [97]. The image that results from an MRI scan is quite detailed, and enables the different tissues to be visualized and studied with a high degree of accuracy. Each of the different tissues is subject to different relaxation times after the magnetic field has been removed, which can produce endogenous contrast. In addition, exogenous contrast agents, such as gadolinium chelates, can be applied for some procedures in order to further increase the contrast by selectively correcting for the optimal T1 (longitudinal) or T2 (transverse) relaxation time [98, 99]. Recently, compounds using particular NP designs have been applied as novel contrast agents, as they have larger relaxivities, such as paramagnetic gadolinium-containing liposomes/micelles and SPIO nanoparticles [98, 100]. Unlike radionuclide-based imaging, MRI has the advantage of not using radiation to arrive at an image as well as a higher spatial resolution. However, it is less sensitive than other modalities, which can only be partially corrected for by working at higher magnetic field strengths, applying exogenous contrast agents, and/or imaging for a longer period of time. The need for a high concentration of a targeted MRI contrast agent in order to provide meaningful signal, gives the potential for a pharmacological response to the agent and additional care must be taken in the toxicological evaluation of the imaging agent during development.

Non-Targeted MR Contrast Agents

Iron-oxide nanoparticles (IONPs) are one of the most common nanoplatforms used as MR contrast agents. They have been shown to be detectable at such low concentrations that single-particle detection has been proven effective [101, 102]. In addition, non-targeted IONPs are usually used to image the liver [103, 104], the spleen [105, 106], and the lymph node [107, 108] because their small size causes them to be nonspecifically taken up by the RES [109], and to accumulate in lymphatic tissue, which surrounds those organs. Furthermore, IONPs have been found to accumulate in tumors due to the leaky vasculature that arises at tumor sites, and therefore, they have been used to image brain tumors successfully [110]. Another important approach to creating NPs for MRI is to incorporate gadolinium onto the surface of the NP through the addition of a metal chelator. For example, Milne et al. reported a water soluble AuNP functionalized with Gd-DOTA metal complexes and found this nanoplatform to be suitable for preclinical in vivo MRI [111]. In addition, chemical exchange saturation transfer (CEST) is believed to be a promising approach for MRI contrast. Endogenous or exogenous compounds containing exchangeable protons or molecules, such as amide, amine, or hydroxyl groups, are selectively saturated and subsequently detected with enhanced sensitivity [112,

113]. For example, Castelli et al. created liposome-based CEST probes, which when stimulated by endogenous (variation in pH) and externally applied (nonfocused ultrasound) stimuli, successfully imaged the selective release of material from the nanoparticle [114].

Molecular MRI of Integrin Expression

The most well studied target of molecular MRI is integrin $\alpha_{\nu}\beta_3$ [115–117]. The first study in which integrin $\alpha_{\nu}\beta_3$ expression was imaged by MRI used antibody-coated paramagnetic liposomes that contained Gd³⁺ ions [118]. Paramagnetic liposomes coated in LM609, a mouse anti-human integrin $\alpha_{\nu}\beta_3$ monoclonal antibody, were targeted to the angiogenic vasculature of squamous cell carcinomas in rabbit [118], resulting in successful imaging by molecular MR.

Integrin $\alpha_{\nu}\beta_3$ -targeted magnetic NPs have been shown to improve imaging sensitivity and quality by increasing MR signal significantly. For example, when used to image a Vx-2 squamous cell carcinoma model at 1.5T, MR signal improved in the tumor periphery at 2 h post-injection [119]. Interestingly, these NPs were able to enter into the leaky tumor neovasculature, but did not reach the interstitium. In another model, athymic nude mice with human melanoma xenografts were successfully imaged using $\alpha_{\nu}\beta_3$ integrin-targeted paramagnetic NPs [120].

Molecular MRI of Other Targets

NP size and magnetic properties are important characteristics that influence MR signal. A comparison was made between two types of NPs, manganese-doped magnetism-engineered iron oxide (MnMEIO) and cross-linked iron oxide (CLIO), both conjugated with trastuzumab, an anti-HER2 monoclonal antibody [121, 122]. Fluorescent-activated cell sorting analysis was performed and demonstrated that both NPs had similar targeting abilities *in vitro*. However, MnMEIO was found to be far superior in imaging small HER-2 positive tumors implanted in mice than CLIO. It is believed that the reasons for CLIO-trastuzumab conjugates resulting in lower MR contrast enhancement when compared to conjugates with MnMEIO are two-fold: CLIO is a larger particle than MnMEIO, and therefore, leads to poorer extravasation from the leaky tumor vasculature, and also it has poorer magnetic properties, which hinders MR contrast significantly, especially at low concentrations [121, 122].

In many studies to date, *ex vivo* analyses have not been performed to determine if the targeted NPs are actually targeting the desired tumor vasculature and/ or cells, or if they are accumulating nonspecifically in the interstitial space. Due to the large overall size of the targeted IONPs (>20 nm in diameter), which includes surface polymer coating and targeting ligands, they tend to travel as far as the vasculature. However, with the development of smaller NPs bearing longer circulation times, NPs used as MR contrast agents are showing improved extravasation from the leaky tumor vasculature. In addition, surface conjugation with peptides or small molecules instead of antibodies may also improve the ability of these NPs to reach their target, due to the possibility of modifying the NP surface with more targeting ligands, while maintaining a smaller overall size. However, the primary issue that arises when imaging with MR is inherent low sensitivity. Therefore, the ideal NP design would be a MR contrast agent that provides improved imaging sensitivity and that has a targeting mechanism for tumor specificity.

Multimodality Imaging

Each individual imaging modality has its strengths, and therefore, is used for specific applications; however, no single modality is perfect or capable of sufficiently gaining all of the necessary information. Therefore, multimodality imaging exists and has been used extensively to study the movement of engineered NPs in vivo. A single NP that has been functionalized at its surface can be imaged using different modalities. For example, Xie et al. developed a triple functional IONP probe for use with PET imaging, NIRF imaging, and MRI [123]. In this study, IONPs were modified with dopamine to provide them with moderate polarity, and they were subsequently inserted into human serum albumin (HSA) matrices. The HSA-IONPs were then labeled with Cy5.5 dye and ⁶⁴Cu DOTA chelates to be used as imaging agents using NIRF and PET, respectively. The inherent magnetic property of IONPs enables them to serve as contrast agents in MRI. Thus, the trimodality probe was developed and its pharmacokinetics were investigated in xenograft U87MG tumors (Fig. 10.2). The combination of multiple imaging modalities within a single nanoplatform offers synergistic advantages over a single modality alone, as it is possible to gain more detailed and accurate information from the resulting image.



Fig. 10.2 *In vivo* optical-PET-MR trimodality imaging of IONP probe after 1, 4, 18 h post-injection in a mouse model. **a** *in vivo NIR image* of mice. **b** *in vivo PET images* of mice. **c** *in vivo MR images* of mice acquired pre-injection and 18 h post-injection. *White arrows* indicate xenograft U87MG tumors. Reproduced with permission from Xie J, Chen K, Huang J, Lee S, Wang J, Gao J, Li X, Chen X (2010) PET/NIRF/MRI triple functional iron oxide nanoparticles. Biomaterials 31(11):3016–3022. Copyright © 2010, with permission from Elsevier

Targeted Cancer Therapy

Upon administration to a patient, classical anticancer agents are dispersed throughout the whole body and are unable to distinguish between normal cells and tumor cells. Therefore, the unwanted delivery of an anticancer drug to healthy cells can result in significant adverse effects and toxicity. In addition, rapid clearance from the body, enzyme degradation, nonspecific delivery of the drug, and suboptimal accumulation of the drug in the targeted tissue require that the dosage of the drug be increased significantly, which is neither economical nor medically safe, as it leads to further adverse effects and systemic toxicity. Nanomedicine presents an improved approach to administering anticancer therapeutics, as it is able to provide a targeted mechanism for cancer therapeutics and a controlled release of a therapeutic. Those nanoparticles that are capable of drug loading, and that have optical, magnetic, or photothermal properties, provide a strong framework for therapeutic treatment.

Drug Delivery

Currently, a number of different nanoplatforms exist for use as drug delivery vehicles. They each have a different architecture, and are of different sizes, shape, and material. As a result, they have different drug loading capabilities (by encapsulation, surface attachment, or entrapment), drug release rates, targeting abilities, and vary in their *in vivo* stability. The use of nanoparticles as drug delivery systems is highly advantageous. Firstly, their submicron size allows them to penetrate across barriers into small cells, thereby enabling them to arrive and accumulate at the target site. As a result, toxic side effects in nonspecific areas of the body are reduced [124]. In addition, NPs offer a scaffold for targeted delivery of a drug, as its surface can be modified with stabilizing and targeting molecules. Furthermore, NPs are often used because they offer superior drug stabilization in the body, including improved bio-availability, increased aqueous solubility, decreased degradation, and are capable of producing a prolonged release of the drug. This chapter focuses on only a few of the nanoplatforms currently being employed.

Polymer-Based Drug Nanocarriers

Polymer-based drug nanocarriers include polymeric NPs, micelles, and dendrimers. Depending on how the nanoplatform is prepared, the drug can either be physically entrapped inside, or covalently bound to the NP surface, producing structures that are capsules (polymeric NPs), amphiphilic cores (polymeric micelles) or hyper-branched structures (dendrimers) [125].

Polymeric NPs

The use of biodegradable materials in polymeric NP preparations is a strategic means of allowing sustained drug release at the target site for a longer duration. For example, biodegradable NPs prepared from D, L-lactide co-glycolide (PLGA) and polyactide have been studied for drug delivery [15, 126]. Recent studies have been done using NPs formulated with paclitaxel. For example, paclitaxel-loaded PLGA NPs demonstrated greater and sustained anti-proliferative activity in HeLa cells [127]. In this study, PLGA NP-encapsulated paclitaxel was shown to be a promising controlled drug-delivery system after they were found to increase apoptosis of HeLa cells. In addition, the use of naturally occurring polymers, such as albumin, chitosan, and heparin, has been explored as drug conjugates for the delivery of oligonucleotides, DNA, protein, and drugs. For example, NP serum albumin-bound paclitaxel (Abraxane) has been used in the clinic in order to treat metastatic breast cancer, non-small cell lung cancer and metastatic adenocarcinoma of the pancreas [33].

Polymeric Micelles

Polymeric micelles are made up of a hydrophobic core, which houses hydrophobic drugs, and a hydrophilic shell, which stabilizes the structure and enables their interaction with the liquid environment surrounding the micelle structure. As a result, they are water-soluble, and are ideal for the delivery of water insoluble drugs by intravenous injection [128, 129]. Drugs can be loaded by either physical encapsulation [130] or covalent attachment [131]. Polymeric micelles formulations of paclitaxel have recently been studied. PEG-poly(D, L-lactide)-paclitaxel (Genexol-PM) is a cremophor-free micelle-formulated paclitaxel, which has been applied in phase I and pharmacokinetic clinical trials in patients with advanced refractory malignancies [132]. It is believed that multifunctional polymeric micelles that contain targeting ligands as well as imaging and therapeutic agents will become an important model of micellar formulations [133].

Dendrimers

Dendrimers are treelike structures that are made up of multiple perfectly branched polymeric monomers emerging from a central core. Dendrimers bear advantageous properties for drug delivery, including monodisperse size, ease of surface functionality, multivalency, water solubility, and drug-loadable central core [44]. Dendrimers are a multifunctional platform, as their surfaces can be conjugated with several molecules, including imaging agents, targeting ligands, and therapeutic drugs [44]. An early example of dendrimers in drug delivery involved complexing cisplatin (Fig. 10.3a), an antitumor drug, to the surface of a polyamidoamines (PAMAM) dendrimer, which resulted in slower release and higher accumulation in



Fig. 10.3 Chemical structures for three chemotherapeutic drugs commonly administered with nanoparticles

solid tumors, and lower toxicity when compared with the free drug [32]. Doxorubicin has been covalently attached to a poly(ethylene oxide)-dendrimer through a hydrazine linkage, allowing for release of the drug in the more acidic environment of lysosomal compartments [134].

Lipid-Based Drug Nanocarriers

Liposomes

Liposomes may be classified by the number of bilayers they have—unilamellar systems have an aqueous core for encapsulation of water-soluble drugs, and multilamellar systems encapsulate lipid-soluble drugs. Liposomal anticancer drugs were the first to gain approval for use in cancer therapy, and are typically applied in the multilamellar system for the transport of lipid-soluble drugs. For example, liposomal formulations of the anthracyclines doxorubicin, which can be either PEGylated (Doxil in the United States and Caelyx outside of the United States) or non-PE-Gylated (Myocet), and PEGylated liposomal daunorubicin (DaunoXome) are approved for the treatment of metastatic breast cancer [30, 28, 135]. After extravasation, Doxil liposomes disintegrate and doxorubicin is released. It has been reported that the drug accumulation is 10 fold greater in tumor tissue when delivered by a targeted NP than when it is administered on its own by conventional methods [136]. DaunoXome demonstrated delayed opsonization and avoided rapid RES clearance, resulting in increased plasma circulation [137].

Metal-Based Drug Carriers

Gold Nanoparticles

Gold NPs can be synthesized as very small particles (2–50 nm in diameter), which enhances their ability to carry higher drug dosages due to their relatively large surface

area-to-mass ratio. Gold NPs also offer a number of advantages for cancer therapy, including simple and reliable methods of synthesizing the NP, easy functionalization by various biomolecules due to their negative surface charge, biocompatibility, and low toxicity (if greater than 3 nm in diameter) [21]. In drug delivery, gold NPs conjugated with oxaliplatin were synthesized and studied by Brown et al. and shown to enhance cytotoxicity in all cell lines that were tested, and to penetrate the nucleus in lung cancer cells [138]. In addition, functionalized gold NPs have demonstrated efficient drug delivery to drug-resistant tumor cells when 3-mercaptopropionic acid capped gold NPs efficiently delivered drugs to drug-resistant leukemia K562/ADM cells [139]. Thus, functionalized gold NPs are a potential platform to inhibit multidrug resistance in targeted tumor cells.

Other Methods of Drug Delivery

Image-Guided Drug Delivery

Drug delivery vehicles must be able to successfully accumulate in the tumor at concentrations above a therapeutic threshold, and be distributed evenly throughout the tumor. However, a challenge in drug delivery arises when several of its fundamental determinants, including vessel density, permeability, and expression of the tumor markers, differ among tumor regions, disease stages, and in patients. For this reason, imaging tools can be used to guide drugs to the targeted site, enabling individualized treatment.

Image-guided drug delivery is used to monitor the biodistribution, blood circulation, and tumor accumulation of administered drugs, resulting in greater knowledge of a tumor's response to treatment. To accomplish this, drugs are loaded onto or into NPs that can be used for medical imaging purposes. SPIOs are a popular choice, as they offer excellent MRI contrast and biocompatibility, and can be easily loaded with chemotherapeutic drugs, such as doxorubicin and paclitaxel [133, 140–142] (Fig. 10.3b). Dextran-coated SPIOs conjugated with siRNA, and with a NIR dye, Cy5.5, attached have also been employed to image *in vivo* siRNA delivery using MRI/NIRF dual modality imaging [143].

Magnetic Drug Targeting

Magnetic NPs act like nanosized magnets, and therefore, experience a force when placed in a magnetic field. The force depends on the magnetic moment of the NP and the gradient of the magnetic field [144]. Therefore, by carefully planning the conditions of the magnetic field, magnetic NPs that carry anticancer drugs can theoretically be guided towards the tumor [145, 146].

The first use of magnetic targeting *in vivo* was carried out with magnetic albumin microspheres in the 1970s, in which magnetite particles of 10–20 nm diameter were clustered with albumin [147]. In the early 1980s, the same magnetic particles were used to deliver doxorubicin (Fig. 10.3c) to Yoshida sarcoma tumors in a rat model,

and resulted in complete remission of the disease [148]. In addition, MTC-DOX, another magnetic microparticle, proved successful in animal models; however, it failed to reach clinical trials in 2004 [149]. Studies were later refocused to involve magnetic NPs of <10 nm diameter, as they offer superior biocompatibility, biodistribution, and deeper tumor penetration than microparticles. A number of magnetic NPs, including micelles, liposomes, and SPIOs, have been reported to have successful magnetic targeting of chemotherapeutic drugs to tumor sites in animal models [140, 150].

Thermal Therapy

Cancer cells are more vulnerable to treatment at elevated temperatures, as local heating is able to increase tumor vasculature extravasation of drug carriers. For this reason, combining chemotherapy treatments with thermal therapy is believed to provide a synergistic effect for cancer intervention [151]. Metal NPs, such as gold NPs, and those NPs with magnetic properties, are highly effective for thermal therapies, and are also suitable for loading drugs into their interior, thereby providing the benefit of both drug-delivery and thermal therapy combinations [151].

As heat can potentially degrade the anticancer drug, or the NP itself, controlled drug release is important for thermal therapy. The drug molecule is stored in a heat-responsive structure, which triggers the slow release of the drug at the target site when local heating is exerted. Hu et al. demonstrated that local heating of NPs made up of magnetite and silica cores in an oscillating magnetic field was able to break the magnetic shell, releasing the drug molecule at the tumor [152]. Thomas et al. further reported the rapid release of the anticancer drug, doxorubicin, from zinc-doped iron oxide nanocrystals encapsulated in mesoporous silica under focused hyperthermic conditions [153].

Avoiding the Reticuloendothelial System (RES)

Unmodified NPs typically demonstrate short circulation times in the body due to their small size and surface properties. They are generally removed from circulation by the RES, with the greatest accumulation in the liver, spleen, and lungs [13]. In order to extend circulation times, NPs should be as small as possible; studies have demonstrated that NPs of 100 nm in diameter or less have the longest circulation times. In addition, NPs with hydrophilic surface coatings have been shown to remain in the circulatory system longer than those that are more hydrophobic. Hydrophilic polymers placed at the NP surface, such as PEG (described in Sect. 3.2.1), are able to repel plasma proteins and other blood macrophages [154]. Unlike hydrophobic NPs, which are the prime targets of the RES, hydrophilic NPs were revealed to have less than 1% uptake by the spleen and liver, with 8–10% still circulating in blood 8 h post-injection [155].

Passive Targeting

While normal tissue vasculature is lined with endothelial cells that are tightly aligned, which prevents extravasation of NPs, tumor vasculature grows at an uncontrollable rate and the endothelial cells that line it adopt an abnormal shape, resulting in a leaky and imperfect architecture that is hyperpermeable to small particles [156–159]. In addition, tumor vessels exhibit poor lymphatic drainage, which when coupled with rapid vascularization, results in the preferential accumulation of NPs in the tumor interstitial space. This concept is referred to as the enhanced permeability and retention (EPR) effect. The EPR effect is the primary means of carrying out passive NP tumor targeting, as it facilitates the movement of macromolecules, such as drug-loaded NPs, into tumor tissues.

Active Targeting

Active targeting differs from passive targeting in that it does not rely on the natural accumulation of particles at the desired tumor site. Instead, targeting is achieved by functionalizing the particle surface with a targeting moiety that facilitates its delivery to the site of interest. Thus, active targeting is based on the specific interactions of a ligand to a receptor, such as antibody-antigen or peptide-receptor interactions.

Tumor-Specific Targeting

Overexpression of receptors on tumor cell surfaces makes targeting of these tumor environments possible. Therefore, one plausible mechanism for tumor cell targeting is through the conjugation of specific ligands, such as antibodies, peptides, small molecules, etc., onto the surfaces of NPs to recognize and selectively bind the tumor cell with the overexpressed receptor on its surface. However, targeting effectiveness depends on the NPs ability to reach the cell surface after extravasation across the tumor vasculature endothelium. Recently, it was shown that NPs with similar characteristics, such as size and surface charge, can have different extravasation behavior in vivo [160], suggesting that there is little true understanding of conditions under which the extravasation of NPs occurs. For example, Smith et al. determined that QDs extravasate in LS174T tumors significantly better than SWNTs, while SWNTs extravasate better in U87MG tumors than ODs, but that neither ODs nor SWNTs demonstrate extravasation in xenograft SKOV-3 tumors [160]. In addition, NPs must overcome a number of biological barriers following extravasation, including high interstitial fluid pressure and a dense collagen matrix [161], and must travel tens to hundreds of micrometers before reaching and binding the tumor cell surface [162]. Thus, targeting the tumor cell surface with functionalized NPs faces many challenges. For example, the use of antibody targeting of lipid-based NPs demonstrated greater targeting capability in vitro, but showed little improvement over

non-targeted NPs in reaching solid tumors *in vivo* [163]. However, other examples demonstrate successful *in vivo* targeting and appear to be dependent upon the NP platform being used and the specific targeting mechanism employed. Vapreotide (VAP)-modified core-shell liposome NPs targeting the somatostatin receptor and containing VEGF siRNA as well as paclitaxel, were reported to significantly inhibit tumor growth in a murine model as compared to a non-targeted control NP [164].

Targeting the Tumor Vasculature

Tumor angiogenesis is the rapid and uncontrolled formation of new vessels from the pre-existing vasculature. In normal tissue, the formation of new blood vessels is controlled by the release of anti-angiogenic molecules, such that angiogenic stimulatory molecules are in equilibrium with the inhibitory molecules. However, during tumor neovascularization, this balance is interrupted, causing the increased secretion of the stimulatory molecules, and ultimately, results in the spread of new blood vessels that cause the tumor mass to grow uncontrollably. Consequently, tumors characteristically have poorly vascularized areas with large amounts of necrosis, tumor vessels that are highly branched and abnormal in shape, and a leaky vasculature due to abnormalities and holes in the basement membrane. As a result, molecules are able to move more freely through the blood vessel wall into the interstitial space surrounding the tumor cell.

Molecules that are selectively overexpressed on the tumor vasculature that are involved in angiogenesis are thought to be potential targets for treatment. For example, the VEGF/VEGFR signaling pathway is important for vasculature development in both normal and diseased tissues [165]. Similarly, integrins, which are a family of cell adhesion molecules, are involved in interactions between cells and the extracellular matrix, and can therefore, control cell migration and survival during angiogenesis [166]. Therefore, recent efforts have been directed towards inhibiting tumor growth during angiogenesis. For example, Feng et al. developed a core-shell type NP co-encapsulating VEGF targeted siRNA (siVEGF) and paclitaxel, with VAP targeting peptides on their surface for interaction with somatostatin receptors, referred to as VAP-PLPC/siRNA NPs [164]. The nanocarrier was designed to produce a synergistic inhibition of tumor growth due to RNA interference, which assists in down regulating VEGF, and the simultaneous delivery of the drug, paclitaxel, into cells. Their study demonstrated that the targeted NP had stronger drug distribution in tumor tissue and was more efficient at inhibiting tumor growth and neovascularization in vivo [164]. In another study, Jiang et al. showed that integrinmediated poly(trimethylene carbonate)-based stealth NPs, c(RGDyK)-NP, loaded with paclitaxel had greater tumor penetration, accumulation, and growth inhibitory effect than other conventional NPs [167].

Challenges and Future Outlook

NPs have a number of potential cancer applications, particularly in personalized oncology, which allows the detection, diagnosis, and treatment to be tailored to each individual case, as well as in predictive oncology, which allows molecular markers to be used as guides for disease development, progression, and clinical outcomes [3]. Recent advances in nanotechnology have demonstrated the promise of NPs in cancer research, such as in the delivery of siRNA, overcoming drug resistance, and applications in multivalent targeting; however, there is still a great need for concentrated effort to improve the field and applications of nanotechnology, and to help guide it towards a future of increasing clinical relevance.

Drug resistance is one of the major obstacles that restricts efficient cellular delivery of chemotherapeutic agents. Several mechanisms exist at the cellular level that are responsible for drug resistance. The most well known example is membrane bound P-glycoprotein (P-gp), a drug efflux protein on the cell membrane, which decreases the intracellular concentration of chemotherapeutic drugs [168–170]. In addition, P-gp may be present on the nuclear membrane, which can limit the transport of drugs into the nucleus [171]. Moreover, the drugs that do gain intracellular access are often trapped in cytoplasmic vesicles and then degraded, or are externalized directly from the cell by exocytosed vesicles [172].

Several strategies exist to overcome P-gp mediated resistance. For example, agents that inhibit P-gp, such as verapamil and cyclosporine, are administered simultaneously with a cytotoxic drug. It is believed that NPs can evade recognition by the P-gp efflux pump by being engulfed in an endosome upon entering the cell, resulting in greater drug concentration inside the cell [173]. Therefore, NPs can be loaded with both a cytotoxic drug and a P-gp inhibiting agent for simultaneous delivery into the cell [174]. Furthermore, ligand-targeted strategies are useful overcoming P-gp mediated resistance, as the ligands are internalized due to receptormediated endocytosis. For this reason, NPs functionalized with ligands and PEG polymers have the potential to deliver elevated drug concentrations to the plasma membrane, thereby saturating the plasma membrane and reversing the effects of P-gp [169]. A transferrin-conjugated NP loaded with paclitaxel [175] and a folate-receptor targeted, pH-sensitive polymeric micelle that contained doxorubicin [176] were shown to have greater ability to inhibit drug resistance in MCF-7 cells and xenografts than their non-targeted counterparts.

Despite novel designs for cancer-targeted NPs, very few NP-based chemotherapeutics are currently used in the clinic, as current NP technologies fail to universally improve the efficacy of a drug. The reasons for this are multifold:

 Tumor vasculature targeting results in low absolute tumor accumulation of the NP (typically <10% ID/g) when compared with accumulation in the liver (>20% ID/g) [161]. It has been widely accepted and proven that the nanoplatform's size, chemical composition, surface charge, and density of PEG and targeting ligands affect its biodistribution and accumulation at the site of interest; however, general rules governing the design of NPs for vasculature targeting have yet to described [161]. It has been suggested that a side-by-side comparison of *in vivo* tumor vasculature targeting be conducted with different nanoplatforms of different size, composition, ligand density, etc., as no summary of this kind exists in literature [161].

- ii. The therapeutic drug is often poorly loaded onto the NP for delivery to the tumor, with the drug making up <5% of the NPs weight. This runs the risk of having the drug concentration be too low to be pharmacologically active upon delivery, or the need for NP concentrations to exceed pharmacologically safe limits, resulting in unwanted side effects and high toxicity levels.
- iii. Drugs encapsulated in NPs, or adsorbed onto their surface, are susceptible to being released prematurely or too quickly (termed "burst release") upon administration. Consequently, a large portion of the drug is often released before reaching its target in the body, which results in lower activity at the tumor, and ultimately, can produce greater side effects.
- iv. The risk for toxicity and side effects exists with the use of NPs for tumor targeting with imaging agents and therapeutic drugs. Toxicity can result from NP introduction into the body by adsorption through the skin, inhalation, and ingestion [76]. This can result in the accumulation of NPs in unintended organs, potentially causing adverse effects. In addition, NPs may not be cleared from the body by hepatic and/or renal pathways at a rate that escapes toxic build-up. Similarly, the body must be able to respond quickly enough to dispose of the NP in order to evade toxicity and adverse side effects [76]. Moreover, there are concerns about toxicities resulting from the NP material itself, such as CdSe and CdTe in QDs [9].

Therefore, greater efforts need to be directed towards better and safer technologies for optimizing loading capacities and release of challenging therapeutic drugs from the different nanoplatforms. In addition, there is a need for improving *in vivo* targeting efficiency in order to determine the optimal conditions for tumor targeting, and to decrease the unwanted accumulation of drugs in the RES, thereby improving the tumor to liver ratio. Similarly, improved design strategies in order to develop NPs with enhanced clearance properties are also required.

Multifunctional NPs have the potential to be an exciting future direction in nanomedicine. Important factors include imaging (i.e. single- vs. dual-modality), therapy (i.e. single drug or combination of drugs) and targeting (i.e. one or more targeting ligands). With each additional NP functionalization, NPs can be used for novel applications due to their added capabilities.

NPs that carry multiple contrast moieties already exist and are currently being employed as multimodality imaging agents using different modalities for efficient signal enhancement or for comparing *in vivo* to *ex vivo* specimens. In addition, NPs can be designed to use a single platform to combine therapeutics and imaging for personalized patient management ("Nanotheranostics") [177], such that a single NP may be capable of identifying and targeting cancerous cells by way of a specific targeting ligand on its exterior, visualizing their location in the body in real-time, delivering chemotherapeutic drugs in a controlled manner with minimal adverse side effect, and finally, monitoring the treatment in real-time. Some nanoplatforms that have intrinsic imaging capabilities, such as IONPs, gold NPs and QDs, can be advantageous in developing new strategies for multifunctional NPs, as the composition of the particle itself produces imaging contrast. Similarly, NPs with inherent therapeutic properties may decrease the need for as complex modifications to the particle. For example, magnetic NPs (i.e. IONPs, gold NPs, SWNT NPs) have thermoablative properties, which can lead to carcinoma destruction when localized heat, from sources such as infrared lamps, ultrasound or lasers, is applied [77]. Unfortunately, it is difficult to contain the heat in a localized area; however, the problem can be avoided with NPs that are designed to absorb NIR light, such as that emitted from laser irradiation. When NIR light is absorbed by these NPs, the target site increases in temperature from about 40 to 50 °C, potentially resulting in irreversible selective cellular destruction [178].

Another aspect of multifunctional NPs focuses primarily on the delivery of multiple therapeutic drugs simultaneously. Most of the NP platforms studied to date, including liposomes, micelles, and dendrimers, are incapable of loading and releasing multiple drugs due to the complexities of ratiometric delivery and synchronized release of multiple drugs from a single NP scaffold [179-182]. Liao et al. successfully synthesized the first example of a NP platform that is capable of controlled loading and synchronized release of precise molar ratios of three chemotherapeutic drugs, doxorubicin, camptotheticin, and cisplatin [183]. In delivering multiple drugs simultaneously, it is important to consider the toxicity profiles of the individual drugs, in order that the drugs do not adversely interact with one another. These three drugs were chosen because their toxicity profiles do not overlap with one another [184, 185]. In this study, they present a novel strategy wherein convergent methods were applied for the synthesis of the NP platform, such that the drug molecules are used as building blocks for the particle itself. In doing so, complexities that arise from drug conjugation and encapsulation reactions are eliminated. In addition, it was believed that the concentration of each drug at (or near) the maximum tolerated dose would lead to maximum therapeutic index. The study demonstrated that the three-drug-loaded nanoplatform with concentrations of each drug in multiples of each drug's maximum tolerated dose successfully outperformed their one- and twodrug-loaded NP counterparts in vitro with ovarian cancer cells [183]. Although in vivo studies have yet to be performed, the approach carried out in this study is not believed to be limited by the number and ratio of molecular species that could be built into particles. As a result, this first example of triplex drug delivery provides the framework for future combination drug studies. Combining this approach to drug delivery along with a cancer specific targeting mechanism would fully utilize the capabilities of the NP platform and has great potential for targeted anti-cancer therapeutics.

Nanomedicine would also benefit from the use of different nanoplatforms, particularly those that are already developed in nature. Viral nanoparticles (VNPs) are naturally occurring bio-nanomaterials that have recently gained much attention because they offer numerous advantages over synthetic platforms: they are made of biocompatible and biodegradable materials; they are derive from bacteriophages and plant or animal viruses, which can be infectious or non-infectious; they are deemed safe for human application; they can be produced quickly and in large quantities; they can be modified and functionalized with ligands with high specificity and accuracy because their structures are known to atomic resolution; and finally, functionalization can be carried out by way of genetic engineering and chemical bioconjugation, resulting in relatively flexible construction [186, 187]. VNPs derived from plant viruses and bacteriophages are of particular interest because they are the least pathogenic in humans, and therefore, are believed to produce less adverse side effects [186].

Efficient therapy requires that the drug is able to avoid interaction with normal cells, and that it accumulates in the tumor. Targeted NPs have been shown to have affinity for particular receptors in diseased tissue, and therefore, reduced accumulation in nontargeted tissue. Cowpea mosaic virus (CPMV) is a plant-derived VNP that has a natural affinity for and is internalized by mammalian endothelial cells *in vivo* [188]. Animal toxicity studies have revealed that CPMV show no clinical symptoms, and therefore, can be applied in therapeutic drug delivery [189]. Studies have shown that CPMV possess a lysine residue with enhanced reactivity in each asymmetric unit [190] (Fig. 10.4).

Steinmetz et al. synthesized and characterized a novel multifunctional VNP using CPMV for the detection and visualization of human prostate cancer. In this study, CPMV was modified to display the targeting peptide bombesin, PEG polymers, and the NIR dye Alexa Fluor 647 (AF647) for specific targeting and imaging of GRP receptors, which are over-expressed in prostate cancer [187]. Unlike previous studies of this kind, the bombesin peptide was attached at the outer ends of the PEG chains, distal to the VNP core. It was found that the inclusion of the hydrophilic PEG linker in the hydrophobic bombesin peptide increased the solubility of the particle, and improved its stability, without resulting in aggregation. The CPMV-PEG-bombesin



Fig. 10. 4 a Molecular model of CPMV nanoparticle generated by the VIPER website from protein data bank file 1NY7. Lys sidechains are highlighted in *red*. Image was created with Jmol. **b** A closer view of a *Lys-rich* region. Lys residues are shown as 'Ball and stick' and colored by Atoms. Image was created with Ds 3.5 client

targeted particle was shown to accumulate in a human PC-3 tumor model, and its uptake in the tumor was visualized and quantified in real-time [187]. In addition, the targeted particle demonstrated more uniform uptake throughout the tumor and was retained at significantly higher levels in the tumor as compared with its nontargeted analogue (CPMV-PEG) [187].

The versatility of NPs and their applications in nanomedicine make for promising tools in the medical world. The combination of their applications in nanoscale drug delivery systems and the development of nanoscale imaging indicates that the potential exists for future multifunctional, multivalent nanoplatforms that can be used in personalized oncology. Furthermore, there is the possibility of new nanoplatforms that can be applied for simultaneous *in vivo* imaging and treatment of the disease. While nanomedicine still requires a great deal of maturation before it can be routinely applied in a clinical setting, considerable progress has already been made with respect to cancer research, and therefore, the use of different NPs in individualized cancer treatment may soon become an exciting reality.

References

- Ferrari M (2005) Cancer nanotechnology: opportunities and challenges. Nat Rev Cancer 5(3):161–171
- Srinivas PR, Barker P, Srivastava S (2002) Nanotechnology in early detection of cancer. Lab Invest 82(5):657–662
- Nie S, Xing Y, Kim GJ, Simons JW (2007) Nanotechnology applications in cancer. Annu Rev Biomed Eng 9:257–288
- Jain PK, Huang X, El-Sayed IH, El-Sayed MA (2008) Noble metals on the nanoscale: optical and photothermal properties and some applications in imaging, sensing, biology, and medicine. Acc Chem Res 41(12):1578–1586
- Alivisatos P (2004) The use of nanocrystals in biological detection. Nat Biotechnol 22(1): 47–52
- Michalet X, Pinaud FF, Bentolila LA, Tsay JM, Doose S, Li JJ, Sundaresan G, Wu AM, Gambhir SS, Weiss S (2005) Quantum dots for live cells, in vivo imaging, and diagnostics. Science 307(5709):538–544
- Medintz IL, Uyeda HT, Goldman ER, Mattoussi H (2005) Quantum dot bioconjugates for imaging, labelling and sensing. Nat Mater 4(6):435–446
- Li Z-B, Cai W, Chen X (2007) Semiconductor quantum dots for in vivo imaging. J Nanosci Nanotechnol 7(8):2567–2581
- 9. Cai W, Hsu AR, Li Z-B, Chen X (2007) Are quantum dots ready for in vivo imaging in human subjects? Nanoscale Res Lett 2(6):265–281
- Caminade A-M, Laurent R, Majoral J-P (2005) Characterization of dendrimers. Adv Drug Deliv Rev 57(15):2130–2146
- Tomalia DA, Naylor AM, Goddard WA (1990) Starburst dendrimers: molecular-level control of size, shape, surface chemistry, topology, and flexibility from atoms to macroscopic matter. Angew Chem Int Ed Engl 29(2):138–175
- 12. Sahoo SK, Ma W, Labhasetwar V (2004) Efficacy of transferrin-conjugated paclitaxel-loaded nanoparticles in a murine model of prostate cancer. Int J Cancer 112(2):335–340
- 13. Brigger I, Dubernet C, Couvreur P (2002) Nanoparticles in cancer therapy and diagnosis. Adv Drug Deliv Rev 54(5):631–651
- Sahoo SK, Labhasetwar V (2003) Nanotech approaches to drug delivery and imaging. Drug Discov Tod 8(24):1112–1120

- 15. Panyam J, Labhasetwar V (2003) Biodegradable nanoparticles for drug and gene delivery to cells and tissue. Adv Drug Deliv Rev 55(3):329–347
- 16. Peng X-H, Qian X, Mao H, Wang AY (2008) Targeted magnetic iron oxide nanoparticles for tumor imaging and therapy. Int J Nanomed 3(3):311
- Yu MK, Jeong YY, Park J, Park S, Kim JW, Min JJ, Kim K, Jon S (2008) Drug-loaded superparamagnetic iron oxide nanoparticles for combined cancer imaging and therapy in vivo. Angew Chem Int Ed Engl 47(29):5362–5365
- 18. Schwertmann U, Cornell RM (2008) Iron oxides in the laboratory: preparation and characterization. Wiley, Hoboken
- Babincova M, Babinec P, Bergemann C (2000) High-gradient magnetic capture of ferrofluids: implications for drug targeting and tumor embolization. Z Naturforschung C, J Biosci 56(9–10):909–911
- 20. Connor EE, Mwamuka J, Gole A, Murphy CJ, Wyatt MD (2005) Gold nanoparticles are taken up by human cells but do not cause acute cytotoxicity. Small 1(3):325–327
- Hainfeld JF, Slatkin DN, Focella TM, Smilowitz HM (2006) Gold nanoparticles: a new X-ray contrast agent. Br J Radiol 79(939):248–253. doi:10.1259/bjr/13169882
- Skirtach AG, Muñoz Javier A, Kreft O, Köhler K, Piera Alberola A, Möhwald H, Parak WJ, Sukhorukov GB (2006) Laser-induced release of encapsulated materials inside living cells. Angew Chem Int Ed Engl 45(28):4612–4617
- Bao G, Mitragotri S, Tong S (2013) Multifunctional nanoparticles for drug delivery and molecular imaging. Annu Rev Biomed Eng 15:253–282
- Bianco A, Kostarelos K, Partidos CD, Prato M (2005) Biomedical applications of functionalised carbon nanotubes. Chem Commun 7(5):571–577
- Bianco A, Kostarelos K, Prato M (2005) Applications of carbon nanotubes in drug delivery. Curr Opin Chem Biol 9(6):674–679
- Parveen S, Misra R, Sahoo SK (2012) Nanoparticles: a boon to drug delivery, therapeutics, diagnostics and imaging. Nanomedicine 8(2):147–166
- Cho K, Wang X, Nie S, Shin DM (2008) Therapeutic nanoparticles for drug delivery in cancer. Clin Cancer Res 14(5):1310–1316
- Rosenthal E, Poizot-Martin I, Saint-Marc T, Spano J-P, Cacoub P, Group DS (2002) Phase IV study of liposomal daunorubicin (DaunoXome) in AIDS-related Kaposi sarcoma. Am J Clin Oncol 25(1):57–59
- 29. Rivera E (2003) Current status of liposomal anthracycline therapy in metastatic breast cancer. Clin Breast Cancer 4:S76–83
- Markman M (2006) Pegylated liposomal doxorubicin in the treatment of cancers of the breast and ovary. Expert Opin Pharmacother 7(11):1469–1474. doi:10.1517/14656566.7.11.1469
- Kukowska-Latallo JF, Candido KA, Cao Z, Nigavekar SS, Majoros IJ, Thomas TP, Balogh LP, Khan MK, Baker JR (2005) Nanoparticle targeting of anticancer drug improves therapeutic response in animal model of human epithelial cancer. Cancer Res 65(12):5317–5324
- 32. Malik N, Evagorou EG, Duncan R (1999) Dendrimer-platinate: a novel approach to cancer chemotherapy. Anticancer Drugs 10(8):767–776
- 33. Gradishar WJ, Tjulandin S, Davidson N, Shaw H, Desai N, Bhar P, Hawkins M, O'Shaughnessy J (2005) Phase III trial of nanoparticle albumin-bound paclitaxel compared with polyethyl-ated castor oil-based paclitaxel in women with breast cancer. J Clin Oncol 23(31):7794–7803
- Sabbatini P, Aghajanian C, Dizon D, Anderson S, Dupont J, Brown JV, Peters WA, Jacobs A, Mehdi A, Rivkin S (2004) Phase II study of CT-2103 in patients with recurrent epithelial ovarian, fallopian tube, or primary peritoneal carcinoma. J Clin Oncol 22(22):4523–4531
- 35. Vasey PA, Kaye SB, Morrison R, Twelves C, Wilson P, Duncan R, Thomson AH, Murray LS, Hilditch TE, Murray T (1999) Phase I clinical and pharmacokinetic study of PK1 [N-(2-hydroxypropyl) methacrylamide copolymer doxorubicin]: first member of a new class of chemotherapeutic agents-drug-polymer conjugates. Clin Cancer Res 5(1):83–94
- 36. Schleich N, Sibret P, Danhier P, Ucakar B, Laurent S, Muller R, Jérôme C, Gallez B, Préat V, Danhier F (2013) Dual anticancer drug/superparamagnetic iron oxide-loaded PLGA-based nanoparticles for cancer therapy and magnetic resonance imaging. Int J Pharm 447(1):94–101

- 10 Nanomedicine—Nanoparticles in Cancer Imaging and Therapy
- Wu W, Wieckowski S, Pastorin G, Benincasa M, Klumpp C, Briand JP, Gennaro R, Prato M, Bianco A (2005) Targeted delivery of amphotericin B to cells by using functionalized carbon nanotubes. Angew Chem Int Ed Engl 44(39):6358–6362
- Pastorin G, Wu W, Wieckowski S, Briand J-P, Kostarelos K, Prato M, Bianco A (2006) Double functionalisation of carbon nanotubes for multimodal drug delivery. Chem Commun 21(11):1182–1184
- Probst CE, Zrazhevskiy P, Bagalkot V, Gao X (2013) Quantum dots as a platform for nanoparticle drug delivery vehicle design. Adv Drug Deliv Rev 65(5):703–718
- 40. Wang M, Thanou M (2010) Targeting nanoparticles to cancer. Pharmacol Res 62(2):90-99
- Immordino ML, Dosio F, Cattel L (2006) Stealth liposomes: review of the basic science, rationale, and clinical applications, existing and potential. Int J Nanomedicine 1(3):297–315
- Obata Y, Tajima S, Takeoka S (2010) Evaluation of pH-responsive liposomes containing amino acid-based zwitterionic lipids for improving intracellular drug delivery in vitro and in vivo. J Control Release 142(2):267–276
- Katagiri K, Imai Y, Koumoto K, Kaiden T, Kono K, Aoshima S (2011) Magnetoresponsive on-demand release of hybrid liposomes formed from Fe3O4 nanoparticles and thermosensitive block copolymers. Small 7(12):1683–1689
- Svenson S, Tomalia DA (2012) Dendrimers in biomedical applications-reflections on the field. Adv Drug Deliv Rev 64:102–115
- Jokerst JV, Lobovkina T, Zare RN, Gambhir SS (2011) Nanoparticle PEGylation for imaging and therapy. Nanomedicine 6(4):715–728
- 46. Allen C, Dos Santos N, Gallagher R, Chiu G, Shu Y, Li W, Johnstone S, Janoff A, Mayer L, Webb M (2002) Controlling the physical behavior and biological performance of liposome formulations through use of surface grafted poly (ethylene glycol). Biosci Rep 22:225–250
- Woodle MC (1995) Sterically stabilized liposome therapeutics. Adv Drug Deliv Rev 16(2):249–265
- Beroström K, Österberg E, Holmberg K, Hoffman AS, Schuman TP, Kozlowski A, Harris JM (1995) Effects of branching and molecular weight of surface-bound poly (ethylene oxide) on protein rejection. J Biomater Sci Polym Ed 6(2):123–132
- Hoarau D, Delmas P, Roux E, Leroux J-C (2004) Novel long-circulating lipid nanocapsules. Pharm Res 21(10):1783–1789
- Kostarelos K, Miller AD (2005) Synthetic, self-assembly ABCD nanoparticles; a structural paradigm for viable synthetic non-viral vectors. Chem Soc Rev 34(11):970–994
- Gupta AK, Gupta M (2005) Synthesis and surface engineering of iron oxide nanoparticles for biomedical applications. Biomaterials 26(18):3995–4021
- Berry CC, Wells S, Charles S, Curtis AS (2003) Dextran and albumin derivatised iron oxide nanoparticles: influence on fibroblasts in vitro. Biomaterials 24(25):4551–4557
- Xie J, Xu C, Kohler N, Hou Y, Sun S (2007) Controlled PEGylation of monodisperse Fe3O4 nanoparticles for reduced non-specific uptake by macrophage cells. Adv Mater 19(20):3163–3166
- 54. Shenoy D, Fu W, Li J, Crasto C, Jones G, DiMarzio C, Sridhar S, Amiji M (2006) Surface functionalization of gold nanoparticles using hetero-bifunctional poly(ethylene glycol) spacer for intracellular tracking and delivery. Int J Nanomedicine 1(1):51–57
- Mei BC, Susumu K, Medintz IL, Mattoussi H (2009) Polyethylene glycol-based bidentate ligands to enhance quantum dot and gold nanoparticle stability in biological media. Nat Protoc 4(3):412–423
- Torchilin VP, Omelyanenko VG, Papisov MI, Bogdanov AA Jr, Trubetskoy VS, Herron JN, Gentry CA (1994) Poly (ethylene glycol) on the liposome surface: on the mechanism of polymer-coated liposome longevity. Biochimi Biophys Acta 1195(1):11–20
- 57. Simões S, Slepushkin V, Gaspar R, de Lima M, Duzgunes N (1999) Successful transfection of lymphocytes by ternary lipoplexes. Biosci Rep 19:601–609
- Woodle MC (1998) Controlling liposome blood clearance by surface-grafted polymers. Adv Drug Deliv Rev 32(1):139–152

- Vonarbourg A, Passirani C, Saulnier P, Benoit J-P (2006) Parameters influencing the stealthiness of colloidal drug delivery systems. Biomaterials 27(24):4356–4373
- Jeon S, Lee J, Andrade J, De Gennes P (1991) Protein-surface interactions in the presence of polyethylene oxide: I. Simplified theory. J Colloid Interface Sci 142(1):149–158
- 61. Szleifer I (1997) Polymers and proteins: interactions at interfaces. Curr Opin Solid State Mater Sci 2(3):337–344
- 62. Gbadamosi J, Hunter A, Moghimi S (2002) PEGylation of microspheres generates a heterogeneous population of particles with differential surface characteristics and biological performance. FEBS Lett 532(3):338–344
- 63. Yan X, Scherphof GL, Kamps JA (2005) Liposome opsonization. J Liposome Res 15(1-2): 109–139
- De Gennes P (1987) Polymers at an interface; a simplified view. Adv Colloid Interface Sci 27(3):189–209
- Sawant RR, Sawant RM, Kale AA, Torchilin VP (2008) The architecture of ligand attachment to nanocarriers controls their specific interaction with target cells. J Drug Target 16(7– 8):596–600
- 66. Gu F, Zhang L, Teply BA, Mann N, Wang A, Radovic-Moreno AF, Langer R, Farokhzad OC (2008) Precise engineering of targeted nanoparticles by using self-assembled biointegrated block copolymers. Proc Natl Acad Sci U S A 105(7):2586–2591
- 67. Weissleder R, Mahmood U (2001) Molecular imaging. Radiology 219(2):316-333
- Lim YT, Kim S, Nakayama A, Stott NE, Bawendi MG, Frangioni JV (2003) Selection of quantum dot wavelengths for biomedical assays and imaging. Mol Imaging 2(1):50–64
- Cai W, Shin D-W, Chen K, Gheysens O, Cao Q, Wang SX, Gambhir SS, Chen X (2006) Peptide-labeled near-infrared quantum dots for imaging tumor vasculature in living subjects. Nano Lett 6(4):669–676
- Åkerman ME, Chan WC, Laakkonen P, Bhatia SN, Ruoslahti E (2002) Nanocrystal targeting in vivo. Proc Natl Acad Sci U S A 99(20):12617–12621
- Cai W, Chen X (2006) Anti-Angiogenic Cancer Therapy Based on Integrin v3 Antagonism. Anticancer Agents Med Chem (Formerly Current Medicinal Chemistry-Anti-Cancer Agents) 6(5):407–428
- Yu X, Chen L, Li K, Li Y, Xiao S, Luo X, Liu J, Zhou L, Deng Y, Pang D, Wang Q (2007) Immunofluorescence detection with quantum dot bioconjugates for hepatoma in vivo. J Biomed Opt 12(1):014008. doi:10.1117/1.2437744
- Tada H, Higuchi H, Wanatabe TM, Ohuchi N (2007) In vivo real-time tracking of single quantum dots conjugated with monoclonal anti-HER2 antibody in tumors of mice. Cancer Res 67(3):1138–1144
- 74. Ferro-Flores G, de M RF, Melendez-Alafort L, Santos-Cuevas C (2010) Peptides for in vivo target-specific cancer imaging. Mini Rev Med Chem 10(1):87–97
- 75. Madsen MT (2007) Recent advances in SPECT imaging. J Nucl Med 48(4):661-673
- Cai W, Chen X (2007) Nanoplatforms for targeted molecular imaging in living subjects. Small 3(11):1840–1854
- Morales-Avila E, Ferro-Flores G, Ocampo-García BE, de María Ramírez F (2012) Radiolabeled nanoparticles for molecular imaging. In: Schaller B (ed) Molecular imaging, ISBN: 978-953-51-0359-2, InTech, doi: 10.5772/31109. Available from: http://www.intechopen. com/books/molecular-imaging/radiolabeled-nanoparticles-for-molecular-imaging
- Cai W, Chen K, Li Z-B, Gambhir SS, Chen X (2007) Dual-function probe for PET and nearinfrared fluorescence imaging of tumor vasculature. J Nucl Med 48(11):1862–1870
- Chen K, Li Z-B, Wang H, Cai W, Chen X (2008) Dual-modality optical and positron emission tomography imaging of vascular endothelial growth factor receptor on tumor vasculature using quantum dots. Eur J Nucl Med Mol Imaging 35(12):2235–2244
- Lee H-Y, Li Z, Chen K, Hsu AR, Xu C, Xie J, Sun S, Chen X (2008) PET/MRI dual-modality tumor imaging using arginine-glycine-aspartic (RGD)-conjugated radiolabeled iron oxide nanoparticles. J Nucl Med 49(8):1371–1379

- Yang X, Hong H, Grailer JJ, Rowland IJ, Javadi A, Hurley SA, Xiao Y, Yang Y, Zhang Y, Nickles RJ (2011) cRGD-functionalized, DOX-conjugated, and ⁶⁴Cu-labeled superparamagnetic iron oxide nanoparticles for targeted anticancer drug delivery and PET/MR imaging. Biomaterials 32(17):4151–4160
- Liu Z, Cai W, He L, Nakayama N, Chen K, Sun X, Chen X, Dai H (2007) In vivo biodistribution and highly efficient tumour targeting of carbon nanotubes in mice. Nat Nanotechnol 2(1):47–52
- Devaraj NK, Keliher EJ, Thurber GM, Nahrendorf M, Weissleder R (2009) 18F labeled nanoparticles for in vivo PET-CT imaging. Bioconjugate Chem 20(2):397–401
- Jarrett BR, Frendo M, Vogan J, Louie AY (2007) Size-controlled synthesis of dextran sulfate coated iron oxide nanoparticles for magnetic resonance imaging. Nanotechnology 18(3):035603
- Zhu J, Chin J, Wängler C, Wängler B, Lennox RB, Schirrmacher R (2014) Rapid 18Flabeling and loading of pegylated gold nanoparticles for in vivo applications. Bioconjugate Chem 25(6):1143–1150. doi:10.1021/bc5001593
- Liu S (2009) Radiolabeled cyclic RGD peptides as integrin αvβ3-targeted radiotracers: maximizing binding affinity via bivalency. Bioconjugate Chem 20(12):2199–2213
- Morales-Avila E, Ferro-Flores G, Ocampo-García BE, De León-Rodríguez LM, Santos-Cuevas CL, García-Becerra R, Medina LA, Gómez-Oliván L (2011) Multimeric system of 99mTc-labeled gold nanoparticles conjugated to c [RGDfK (C)] for molecular imaging of tumor α (v) β (3) expression. Bioconjugate Chem 22(5):913–922
- Ferro-Flores G, de Murphy CA, Rodrguez-Cortes J, Pedraza-Lopez M, Ramrez-Iglesias MT (2006) Preparation and evaluation of 99mTc-EDDA/HYNIC-[Lys3]-bombesin for imaging gastrin-releasing peptide receptor-positive tumours. Nucl Med Commun 27(4):371–376
- Santos-Cuevas CL, Ferro-Flores G, de Murphy CA, Pichardo-Romero PA (2008) Targeted imaging of gastrin-releasing peptide receptors with 99mTc-EDDA/HYNIC-[Lys3]-bombesin: biokinetics and dosimetry in women. Nucl Med Commun 29(8):741–747
- Mendoza-Sánchez AN, Ferro-Flores G, Ocampo-García BE, Morales-Avila E, Ramírez FdM, De León-Rodríguez LM, Santos-Cuevas CL, Medina LA, Rojas-Calderón EL, Camacho-López MA (2010) Lys 3-Bombesin Conjugated to 99 m Tc-labelled gold nanoparticles for in vivo gastrin releasing peptide-receptor imaging. J Biomed Nanotechnol 6(4):375–384
- Guo J, Zhang X, Li Q, Li W (2007) Biodistribution of functionalized multiwall carbon nanotubes in mice. Nucl Med Biol 34(5):579–583
- 92. Chan HB, Ellis BL, Sharma HL, Frost W, Caps V, Shields RA, Tsang SC (2004) Carbonencapsulated radioactive 99mtc nanoparticles. Adv Mater 16(2):144–149
- Singh R, Pantarotto D, Lacerda L, Pastorin G, Klumpp C, Prato M, Bianco A, Kostarelos K (2006) Tissue biodistribution and blood clearance rates of intravenously administered carbon nanotube radiotracers. Proc Natl Acad Sci U S A 103(9):3357–3362
- McDevitt MR, Chattopadhyay D, Kappel BJ, Jaggi JS, Schiffman SR, Antczak C, Njardarson JT, Brentjens R, Scheinberg DA (2007) Tumor targeting with antibody-functionalized, radiolabeled carbon nanotubes. J Nucl Med 48(7):1180–1189
- Chrastina A, Schnitzer JE (2010) Iodine-125 radiolabeling of silver nanoparticles for in vivo SPECT imaging. Int J Nanomedicine 5:653–659. doi:10.2147/ijn.s11677
- Torres M de RR, Tavaré R, Glaria A, Varma G, Protti A, Blower PJ (2011) 99mTc-bisphosphonate-iron oxide nanoparticle conjugates for dual-modality biomedical imaging. Bioconjugate Chem 22(3):455–465
- Pathak AP, Gimi B, Glunde K, Ackerstaff E, Artemov D, Bhujwalla ZM (2009) Molecular and functional imaging of cancer: advances in MRI and MRS. Methods Enzyol 2004:3–60
- Zhaoda Z, Nair SA, McMurry TJ (2005) Gadolinium meets medicinal chemistry: MRI contrast agent development. Curr Med Chem 12(7):751–778
- Pautler RG, Fraser SE (2003) The year (s) of the contrast agent-micro-MRI in the new millennium. Curr Opin Immunol 15(4):385–392

- Thorek DL, Weisshaar CL, Czupryna JC, Winkelstein BA, Tsourkas A (2011) Superparamagnetic iron oxide-enhanced magnetic resonance imaging of neuroinflammation in a rat model of radicular pain. Mol Imaging 10(3):206–214
- Kawasaki ES, Player A (2005) Nanotechnology, nanomedicine, and the development of new, effective therapies for cancer. Nanomedicine 1(2):101–109
- 102. Sahoo SK, Parveen S, Panda JJ (2007) The present and future of nanotechnology in human health care. Nanomedicine 3(1):20–31
- Reimer P, Jähnke N, Fiebich M, Schima W, Deckers F, Marx C, Holzknecht N, Saini S (2000) Hepatic lesion detection and characterization: value of nonenhanced mr imaging, superparamagnetic iron oxide-enhanced MR imaging, and spiral ct-roc analysis 1. Radiology 217(1):152–158
- Shamsi K, Balzer T, Saini S, Ros P, Nelson R, Carter E, Tollerfield S, Niendorf H (1998) Superparamagnetic iron oxide particles (SH U 555 A): evaluation of efficacy in three doses for hepatic MR imaging. Radiology 206(2):365–371
- Weissleder R, Hahn PF, Stark DD, Elizondo G, Saini S, Todd L, Wittenberg J, Ferrucci J (1988) Superparamagnetic iron oxide: enhanced detection of focal splenic tumors with MR imaging. Radiology 169(2):399–403
- Weissleder R, Stark D, Rummeny E, Compton C, Ferrucci J (1988) Splenic lymphoma: ferrite-enhanced MR imaging in rats. Radiology 166(2):423–430
- 107. Anzai Y, Piccoli CW, Outwater EK, Stanford W, Bluemke DA, Nurenberg P, Saini S, Maravilla KR, Feldman DE, Schmiedl UP (2003) Evaluation of neck and body metastases to nodes with ferumoxtran 10-enhanced MR imaging: phase iii safety and efficacy study 1. Radiology 228(3):777–788
- Mack MG, Balzer JO, Straub R, Eichler K, Vogl TJ (2002) Superparamagnetic iron oxide-enhanced MR imaging of head and neck lymph nodes. Radiology 222(1):239–244. doi:10.1148/radiol.2221010225
- Chavanpatil MD, Khdair A, Panyam J (2006) Nanoparticles for cellular drug delivery: mechanisms and factors influencing delivery. J Nanosci Nanotechnol 6(9–10):9–10
- Enochs WS, Harsh G, Hochberg F, Weissleder R (1999) Improved delineation of human brain tumors on MR images using a long-circulating, superparamagnetic iron oxide agent. J Magn Reson Imaging 9(2):228–232
- 111. Milne M, Gobbo P, McVicar N, Bartha R, Workentin MS, Hudson RH (2013) Water-soluble gold nanoparticles (AuNP) functionalized with a gadolinium (III) chelate via Michael addition for use as a MRI contrast agent. J Mater Chem B 1(41):5628–5635
- 112. Langereis S, Keupp J, van Velthoven JL, de Roos IH, Burdinski D, Pikkemaat JA, Grüll H (2009) A temperature-sensitive liposomal 1H CEST and 19F contrast agent for MR imageguided drug delivery. J Am Chem Soc 131(4):1380–1381
- 113. Aime S, Castelli DD, Lawson D, Terreno E (2007) Gd-loaded liposomes as T 1, susceptibility, and CEST agents, all in one. J Am Chem Soc 129(9):2430–2431
- 114. Castelli DD, Boffa C, Giustetto P, Terreno E, Aime S (2014) Design and testing of paramagnetic liposome-based CEST agents for MRI visualization of payload release on pH-induced and ultrasound stimulation. J Biol Inorg Chem 19(2):207–214. doi:10.1007/s00775-013-1042-0
- Cai W, Rao J, Gambhir SS, Chen X (2006) How molecular imaging is speeding up antiangiogenic drug development. Mol Cancer Ther 5(11):2624–2633
- Sosnovik DE, Weissleder R (2007) Emerging concepts in molecular MRI. Curr Opin Biotechnol 18(1):4–10
- Cai W, Sam Gambhir S, Chen X (2005) Multimodality tumor imaging targeting integrin alphavbeta3. Biotechniques 39(6 Suppl):S14–25. doi:10.2144/000112091
- 118. Sipkins DA, Cheresh DA, Kazemi MR, Nevin LM, Bednarski MD, Li KC (1998) Detection of tumor angiogenesis in vivo by $\alpha\nu\beta$ 3-targeted magnetic resonance imaging. Nat Med 4(5):623–626
- Winter PM, Caruthers SD, Kassner A, Harris TD, Chinen LK, Allen JS, Lacy EK, Zhang H, Robertson JD, Wickline SA (2003) Molecular imaging of angiogenesis in nascent Vx-2

rabbit tumors using a novel $\alpha\nu\beta$ 3-targeted nanoparticle and 1.5 tesla magnetic resonance imaging. Cancer Res 63(18):5838–5843

- 120. Schmieder AH, Winter PM, Caruthers SD, Harris TD, Williams TA, Allen JS, Lacy EK, Zhang H, Scott MJ, Hu G (2005) Molecular MR imaging of melanoma angiogenesis with αvβ3-targeted paramagnetic nanoparticles. Magn Reson Med 53(3):621–627
- 121. Jun Y-w, Huh Y-M, Choi J-s, Lee J-H, Song H-T, Kim S, Kim S, Yoon S, Kim K-S, Shin J-S (2005) Nanoscale size effect of magnetic nanocrystals and their utilization for cancer diagnosis via magnetic resonance imaging. J Am Chem Soc 127(16):5732–5733
- Lee J-H, Huh Y-M, Jun Y-w, Seo J-w, Jang J-t, Song H-T, Kim S, Cho E-J, Yoon H-G, Suh J-S (2007) Artificially engineered magnetic nanoparticles for ultra-sensitive molecular imaging. Nat Med 13(1):95–99
- 123. Xie J, Chen K, Huang J, Lee S, Wang J, Gao J, Li X, Chen X (2010) PET/NIRF/MRI triple functional iron oxide nanoparticles. Biomaterials 31(11):3016–3022. doi:10.1016/j.
- Yih T, Al-Fandi M (2006) Engineered nanoparticles as precise drug delivery systems. J Cell Biochem 97(6):1184–1190
- Rawat M, Singh D, Saraf S, Saraf S (2006) Nanocarriers: promising vehicle for bioactive drugs. Biol Pharm Bull 29(9):1790–1798
- Panyam J, Zhou WZ, Prabha S, Sahoo SK, Labhasetwar V (2002) Rapid endo-lysosomal escape of poly(DL-lactide-co-glycolide) nanoparticles: implications for drug and gene delivery. FASEB J 16(10):1217–1226. doi:10.1096/fj.02-0088com
- 127. Yang H, Li K, Liu Y, Liu Z, Miyoshi H (2009) Poly (D, L-lactide-co-glycolide) nanoparticles encapsulated fluorescent isothiocyanate and paclitaxol: preparation, release kinetics and anticancer effect. J Nanosci Nanotechnol 9(1):282–287
- Adams ML, Lavasanifar A, Kwon GS (2003) Amphiphilic block copolymers for drug delivery. J Pharm Sci 92(7):1343–1355
- Moghimi SM, Hunter AC, Murray JC (2005) Nanomedicine: current status and future prospects. FASEB J 19(3):311–330
- Batrakova E, Dorodnych TY, Klinskii EY, Kliushnenkova E, Shemchukova O, Goncharova O, Arjakov S, Alakhov VY, Kabanov A (1996) Anthracycline antibiotics non-covalently incorporated into the block copolymer micelles: in vivo evaluation of anti-cancer activity. Br J Cancer 74(10):1545–1552
- Nakanishi T, Fukushima S, Okamoto K, Suzuki M, Matsumura Y, Yokoyama M, Okano T, Sakurai Y, Kataoka K (2001) Development of the polymer micelle carrier system for doxorubicin. J Control Release 74(1):295–302
- 132. Kim T-Y, Kim D-W, Chung J-Y, Shin SG, Kim S-C, Heo DS, Kim NK, Bang Y-J (2004) Phase I and pharmacokinetic study of Genexol-PM, a cremophor-free, polymeric micelle-formulated paclitaxel, in patients with advanced malignancies. Clin Cancer Res 10(11):3708–3716
- 133. Nasongkla N, Bey E, Ren J, Ai H, Khemtong C, Guthi JS, Chin S-F, Sherry AD, Boothman DA, Gao J (2006) Multifunctional polymeric micelles as cancer-targeted, MRI-ultrasensitive drug delivery systems. Nano Lett 6(11):2427–2430
- 134. Padilla De Jesús OL, Ihre HR, Gagne L, Fréchet JM, Szoka FC (2002) Polyester dendritic systems for drug delivery applications: in vitro and in vivo evaluation. Bioconjugate Chem 13(3):453–461
- 135. O'brien M, Wigler N, Inbar M, Rosso R, Grischke E, Santoro A, Catane R, Kieback D, Tomczak P, Ackland S (2004) Reduced cardiotoxicity and comparable efficacy in a phase III trial of pegylated liposomal doxorubicin HCl (CAELYXTM/Doxil®) versus conventional doxorubicin for first-line treatment of metastatic breast cancer. Ann Oncol 15(3):440–449
- 136. Lasic DD, Martin FJ (1995) Stealth liposomes, vol 20. CRC, Boca Raton
- Hofheinz R-D, Gnad-Vogt SU, Beyer U, Hochhaus A (2005) Liposomal encapsulated anticancer drugs. Anticancer Drugs 16(7):691–707
- 138. Brown SD, Nativo P, Smith J-A, Stirling D, Edwards PR, Venugopal B, Flint DJ, Plumb JA, Graham D, Wheate NJ (2010) Gold nanoparticles for the improved anticancer drug delivery of the active component of oxaliplatin. J Am Chem Soc 132(13):4678–4684

- Li J, Wang X, Wang C, Chen B, Dai Y, Zhang R, Song M, Lv G, Fu D (2007) The enhancement effect of gold nanoparticles in drug delivery and as biomarkers of drug-resistant cancer cells. ChemMedChem 2(3):374–378
- Nobuto H, Sugita T, Kubo T, Shimose S, Yasunaga Y, Murakami T, Ochi M (2004) Evaluation of systemic chemotherapy with magnetic liposomal doxorubicin and a dipole external electromagnet. Int J Cancer 109(4):627–635
- 141. Park J-H, von Maltzahn G, Ruoslahti E, Bhatia SN, Sailor MJ (2008) Micellar hybrid nanoparticles for simultaneous magnetofluorescent imaging and drug delivery. Angew Chem Int Ed Engl 47(38):7284–7288. doi:10.1002/anie.200801810
- 142. Kim J, Kim HS, Lee N, Kim T, Kim H, Yu T, Song IC, Moon WK, Hyeon T (2008) Multifunctional uniform nanoparticles composed of a magnetite nanocrystal core and a mesoporous silica shell for magnetic resonance and fluorescence imaging and for drug delivery. Angew Chem Int Ed Engl 47(44):8438–8441
- Medarova Z, Pham W, Farrar C, Petkova V, Moore A (2007) In vivo imaging of siRNA delivery and silencing in tumors. Nat Med 13(3):372–377
- 144. Furlani EP, Ng KC (2008) Nanoscale magnetic biotransport with application to magnetofection. Phys Rev E Stat Nonlin Soft Matter Phys 77(6):061914
- Yellen BB, Forbes ZG, Halverson DS, Fridman G, Barbee KA, Chorny M, Levy R, Friedman G (2005) Targeted drug delivery to magnetic implants for therapeutic applications. J Magn Magn Mater 293(1):647–654
- 146. Shapiro B (2009) Towards dynamic control of magnetic fields to focus magnetic carriers to targets deep inside the body. J Magn Magn Mater 321(10):1594–1599
- Widder KJ, Senyei AE, Scarpelli DG (1978) Magnetic microspheres: a model system for site specific drug delivery in vivo. Exp Biol Med 158(2):141–146
- Widder KJ, Morris RM, Poore GA, Howard DP, Senyei AE (1983) Selective targeting of magnetic albumin microspheres containing low-dose doxorubicin: total remission in Yoshida sarcoma-bearing rats. Eur J Cancer Clin Oncol 19(1):135–139
- Rudge S, Peterson C, Vessely C, Koda J, Stevens S, Catterall L (2001) Adsorption and desorption of chemotherapeutic drugs from a magnetically targeted carrier (MTC). J Control Release 74(1):335–340
- Jain TK, Richey J, Strand M, Leslie-Pelecky DL, Flask CA, Labhasetwar V (2008) Magnetic nanoparticles with dual functional properties: drug delivery and magnetic resonance imaging. Biomaterials 29(29):4012–4021
- 151. Kong G, Braun RD, Dewhirst MW (2001) Characterization of the effect of hyperthermia on nanoparticle extravasation from tumor vasculature. Cancer Res 61(7):3027–3032
- 152. Hu SH, Chen SY, Liu DM, Hsiao CS (2008) Core/Single-Crystal-Shell nanospheres for controlled drug release via a magnetically triggered rupturing mechanism. Adv Mater 20(14):2690–2695
- 153. Thomas CR, Ferris DP, Lee J-H, Choi E, Cho MH, Kim ES, Stoddart JF, Shin J-S, Cheon J, Zink JI (2010) Noninvasive remote-controlled release of drug molecules in vitro using magnetic actuation of mechanized nanoparticles. J Am Chem Soc 132(31):10623–10625
- Storm G, Belliot SO, Daemen T, Lasic DD (1995) Surface modification of nanoparticles to oppose uptake by the mononuclear phagocyte system. Adv Drug Deliv Rev 17(1):31–48
- Gaur U, Sahoo SK, De TK, Ghosh PC, Maitra A, Ghosh P (2000) Biodistribution of fluoresceinated dextran using novel nanoparticles evading reticuloendothelial system. Int J Pharm 202(1):1–10
- 156. Matsumura Y, Maeda H (1986) A new concept for macromolecular therapeutics in cancer chemotherapy: mechanism of tumoritropic accumulation of proteins and the antitumor agent smancs. Cancer Res 46(12 Part 1):6387–6392
- 157. Jain R (2001) Delivery of molecular medicine to solid tumors: lessons from in vivo imaging of gene expression and function. J Control Release 74(1):7–25
- Jain RK (1999) Understanding barriers to drug delivery: high resolution in vivo imaging is key. Clin Cancer Res 5(7):1605–1606

- Duncan R (2003) The dawning era of polymer therapeutics. Nat Rev Drug Discov 2(5):347– 360
- Smith BR, Kempen P, Bouley D, Xu A, Liu Z, Melosh N, Dai H, Sinclair R, Gambhir SS (2012) Shape matters: intravital microscopy reveals surprising geometrical dependence for nanoparticles in tumor models of extravasation. Nano Lett 12(7):3369–3377
- Chen F, Cai W (2014) Tumor vasculature targeting: a generally applicable approach for functionalized nanomaterials. Small 10(10):1887–1893. doi:10.1002/smll.201303627
- Nichols JW, Bae YH (2012) Odyssey of a cancer nanoparticle: from injection site to site of action. Nano Today 7(6):606–618
- 163. Kirpotin DB, Drummond DC, Shao Y, Shalaby MR, Hong K, Nielsen UB, Marks JD, Benz CC, Park JW (2006) Antibody targeting of long-circulating lipidic nanoparticles does not increase tumor localization but does increase internalization in animal models. Cancer Res 66(13):6732–6740
- 164. Feng Q, Yu M-Z, Wang J-C, Hou W-J, Gao L-Y, Ma X-F, Pei X-W, Niu Y-J, Liu X-Y, Qiu C (2014) Synergistic inhibition of breast cancer by co-delivery of VEGF siRNA and paclitaxel via vapreotide-modified core-shell nanoparticles. Biomaterials 35(18):5028–5038
- 165. Ferrara N (2002) VEGF and the quest for tumour angiogenesis factors. Nat Rev Cancer 2(10):795–803
- Desgrosellier JS, Cheresh DA (2010) Integrins in cancer: biological implications and therapeutic opportunities. Nat Rev Cancer 10(1):9–22
- 167. Jiang X, Xin H, Gu J, Xu X, Xia W, Chen S, Xie Y, Chen L, Chen Y, Sha X (2013) Solid tumor penetration by integrin-mediated pegylated poly (trimethylene carbonate) nanoparticles loaded with paclitaxel. Biomaterials 34(6):1739–1746
- Li C (2002) Poly (L-glutamic acid)-anticancer drug conjugates. Adv Drug Deliv Rev 54(5):695–713
- 169. Krishna R, Mayer LD (2000) Multidrug resistance (MDR) in cancer: mechanisms, reversal using modulators of MDR and the role of MDR modulators in influencing the pharmacokinetics of anticancer drugs. Eur J Pharm Sci 11(4):265–283
- 170. Brown R, Links M (1999) Clinical relevance of the molecular mechanisms of resistance to anti-cancer drugs. Expert Rev Mol Med 1(15):1–21
- Calcabrini A, Meschini S, Stringaro A, Cianfriglia M, Arancia G, Molinari A (2000) Detection of P-glycoprotein in the nuclear envelope of multidrug resistant cells. Histochem J 32(10):599–606
- Vasir JK, Labhasetwar V (2005) Targeted drug delivery in cancer therapy. Technol Cancer Res Treat 4(4):363–374
- 173. Wong HL, Bendayan R, Rauth AM, Xue HY, Babakhanian K, Wu XY (2006) A mechanistic study of enhanced doxorubicin uptake and retention in multidrug resistant breast cancer cells using a polymer-lipid hybrid nanoparticle system. J Pharmacol Exp Ther 317(3):1372–1381
- 174. Emilienne Soma C, Dubernet C, Bentolila D, Benita S, Couvreur P (2000) Reversion of multidrug resistance by co-encapsulation of doxorubicin and cyclosporin A in polyalkylcyanoacrylate nanoparticles. Biomaterials 21(1):1–7
- Sahoo SK, Labhasetwar V (2005) Enhanced antiproliferative activity of transferrin-conjugated paclitaxel-loaded nanoparticles is mediated via sustained intracellular drug retention. Mol Pharmaceutics 2(5):373–383
- Lee ES, Na K, Bae YH (2005) Doxorubicin loaded pH-sensitive polymeric micelles for reversal of resistant MCF-7 tumor. J Control Release 103(2):405–418
- 177. Arias JL, Reddy LH, Othman M, Gillet B, Desmaele D, Zouhiri F, Dosio F, Gref R, Couvreur P (2011) Squalene based nanocomposites: a new platform for the design of multifunctional pharmaceutical theragnostics. ACS Nano 5(2):1513–1521
- Pissuwan D, Valenzuela SM, Cortie MB (2006) Therapeutic possibilities of plasmonically heated gold nanoparticles. Trends Biotechnol 24(2):62–67

- 179. Sengupta S, Eavarone D, Capila I, Zhao G, Watson N, Kiziltepe T, Sasisekharan R (2005) Temporal targeting of tumour cells and neovasculature with a nanoscale delivery system. Nature 436(7050):568–572
- Lammers T, Subr V, Ulbrich K, Peschke P, Huber PE, Hennink WE, Storm G (2009) Simultaneous delivery of doxorubicin and gemcitabine to tumors in vivo using prototypic polymeric drug carriers. Biomaterials 30(20):3466–3475
- 181. Kolishetti N, Dhar S, Valencia PM, Lin LQ, Karnik R, Lippard SJ, Langer R, Farokhzad OC (2010) Engineering of self-assembled nanoparticle platform for precisely controlled combination drug therapy. Proc Natl Acad Sci U S A 107(42):17939–17944
- Aryal S, Hu C-MJ, Zhang L (2011) Polymeric nanoparticles with precise ratiometric control over drug loading for combination therapy. Mol Pharmaceutics 8(4):1401–1407
- 183. Liao L, Liu J, Dreaden EC, Morton SW, Shopsowitz KE, Hammond PT, Johnson JA (2014) A convergent synthetic platform for single-nanoparticle combination cancer therapy: ratiometric loading and controlled release of cisplatin, doxorubicin, and camptothecin. J Am Chem Soc 136 (16):5896–5899. doi:10.1021/ja502011 g
- DeVita VT, Serpick AA, Carbone PP (1970) Combination chemotherapy in the treatment of advanced Hodgkin's disease. Ann Intern Med 73(6):881–895
- Al-Lazikani B, Banerji U, Workman P (2012) Combinatorial drug therapy for cancer in the post-genomic era. Nat Biotechnol 30(7):679–692
- Steinmetz NF (2010) Viral nanoparticles as platforms for next-generation therapeutics and imaging devices. Nanomedicine 6(5):634–641
- 187. Steinmetz NF, Ablack AL, Hickey JL, Ablack J, Manocha B, Mymryk JS, Luyt LG, Lewis JD (2011) Intravital imaging of human prostate cancer using viral nanoparticles targeted to gastrin-releasing peptide receptors. Small 7(12):1664–1672
- Lewis JD, Destito G, Zijlstra A, Gonzalez MJ, Quigley JP, Manchester M, Stuhlmann H (2006) Viral nanoparticles as tools for intravital vascular imaging. Nat Med 12(3):354–360
- Singh P, Prasuhn D, Yeh RM, Destito G, Rae CS, Osborn K, Finn M, Manchester M (2007) Bio-distribution, toxicity and pathology of cowpea mosaic virus nanoparticles in vivo. J Control Release 120(1):41–50
- Wang Q, Kaltgrad E, Lin T, Johnson JE, Finn M (2002) Natural supramolecular building blocks: wild-type cowpea mosaic virus. Chem Biol 9(7):805–811

Index

A

Aneuploidy, 37 Animal models genetically engineered mouse model, 97, 102, 115 breast cancer, 99, 103, 104 prostate cancer, 104, 105 mouse xenograft implantation sites, 106, 107, 110, 111 metastasis, 107-109 patient-derived xenograft advantages, 5, 16, 17, 27, 31, 86, 101, 112, 211, 213 applications, 112-115 implantation sites, 106, 107, 110-112 metastasis, 101, 111 Zebrafish xenotransplantation advantages, 98, 100, 101, 112 disadvantages, 101 history, 98

B

Biomarker cancer risk, 85 Bioassay, 86 Boveri, 34

С

Cancer predisposition, 54, 82 Centrosome, 23, 25, 27, 56, 128, 184 Checkpoints cell cycle, 17, 21, 24, 164 spindle assembly, 184 Clinical translation general, 18 Colorectal cancer development genomic instability, 27, 43, 44, 128, 144, 147, 182–185 staging, 180 therapeutic strategies current, 3, 4, 51–54, 185, 187, 191–193, 196, 197 Convergent phenotypic evolution, 18 CpG Islands, 38 Crosstalk, 21, 144

D

Defective repair pathologies, 82 Diagnosis large scale aberrations, 27 small scale aberrations, 29, 30 Double strand break assavs comet, 78, 86 gamma-H2AX, 81 pulse field electrophoresis, 78 radiation induced foci, 78 kinetics, 79, 80 probability, 81, 83, 84 rate of production age dependence, 84 spontaneous, 78, 79 repair homologous recombination, 49, 50 sources errors, 77 ionizing radiation, 77 reactive oxygen species, 76 VDJ recombination, 77 DNA amplifications, 29, 163 damage

© Springer International Publishing Switzerland 2015 C. Maxwell, C. Roskelley (eds.), *Genomic Instability and Cancer Metastasis*, Cancer Metastasis - Biology and Treatment 20, DOI 10.1007/978-3-319-12136-9 245

response, 17, 33, 41, 78, 166, 184, 192 deletions, 29, 52, 182 methylation, 38 repair base excision, 36 nucleotide excision, 36 replication, 22, 23, 42, 48, 56, 77, 147, 182, 183, 186, 187, 192 translocations, 29, 53, 183 Dynamic heterogeneity, 15

Е

Epidemiology a-bomb survivors, 83

F

Factors extracellular, 144 compression, 144 growth factors, 101, 110, 127, 129, 141 interstitial fluid pressure, 144, 229 matrix, 19, 100, 132, 141, 142, 144, 146 mechanical load, 141 physical force, 141 intracellular cytoskeleton, 132, 142-146, 149, 151 gene expression, 100 integrins, 140, 142, 146 mechanosensors, 23, 141-145, 147, 151 Fanconi anemia bone marrow failure (BMF), 54 cancer predisposition, 54 developmental, 54 pathway, 54 replication-dependent interstrand crosslink repair coordination of repair, 55 translesion synthesis, 58

G

Genomic instability, 10, 84, 169

H

Histone modification, 39

I

Immune cells prognosis, 160, 164, 167, 169 tumor cell interactions, 84, 100 Inflammation acute

damage associated molecular patterns, 161 pathogen associated molecular patterns, 161 toll-like receptors, 129, 161, 162, 165, 167 anti-tumor Th1 response, 165-167 genomic instability cell cycle checkpoints, 24-26, 164 DNA repair, 17, 18, 21, 45, 77, 128, 164, 182 mutation rate, 162, 163 genomic stability immune editing, 127, 129, 150, 162, 165, 167-169 stress-related molecules, 166 tumor-associated antigens, 43, 147, 164, 167-169, 182 pro-tumor chronic, 161 Instability chromosomal. 37 epigenome, 38 genetic, 14, 15 genomic, 34, 36 microsatellite, 36, 37 nucleotide, 36, 37 crosstalk with genomic, 40 Interstrand crosslinks, 41

L

Localized disease, 11 Loss of heterozygosity, 37

M

Maintenance or integrity epigenome, 33 genome, 33 mechanisms of, 41 telomere, 33, 37 mechanisms of, 41 Mathematical modeling agent based models morphogenesis, 84 stem cell kinetics, 85 radiation induced carcinogenesis, 86 initiation, 84, 85 progression, 84 promotion, 84, 85 transformation, 84 two-stage clonal expansion genomic instability, 84

Index

Mechanotransduction, 132, 140, 141, 143, 145-149, 151 Metastasis, 97, 99, 100, 103, 106, 112, 143 complexity of, 9 plasticity of, 16 process of, 11 selective pressure on, 16 treatment of, 17, 20, 21 Mismatch repair alkylating agents therapy-related myeloid neoplasms, 45, 59 deficiency microsatellite instability, 44 Lynch Syndrome, 44 eukaryotic, 42, 43 prokarvotic, 42 strand discrimination, 43 Mitosis, 20-25, 148, 149, 195 Molecular classification, 19

Ν

Nanoparticles imaging molecular, 216 molecular MRI, 221-223 multimodality, 223 optical, 210, 217, 218 radionuclide-based, 219, 220 medical applications, 207, 211, 212, 218 platform nanocarrier design, 212, 213, 215 therapy drug delivery, 224, 226, 227 reticuloendothelial system, 213, 228 targeting, 229 Nonhomologous end joining cancer, 49 Nucleosome remodeling, 39

0

Outlook mechanism, 150

P

Personalized medicine, 21 Prevention, 26, 57, 86 Prognosis, 18, 20, 26, 30, 48, 53, 151, 160, 161, 164, 167, 169, 192 Progression evolution, 14 linear, 13 natural selection, 14 Proliferation biochemical cues P38, 150 PTEN, 149 Reactive oxygen species, 150 YAP and TAZ, 149 micropatterning, 148 mitosis, 22, 23, 24, 128, 147 stroma restructuring, 150

R

Reactive oxygen species, 26, 41, 76, 82, 128, 150, 162

S

Sequencing, 19, 20, 35 Single strand break repair base excision & end processing, 47 cancer, 48 gap filling & ligation, 47 recognition, 46

Т

Targeting, 187, 196, 215, 216, 218, 222, 224 Telomerase, 38 Therapy adjuvant, 10, 12 local, 10, 12 risk of relapse to, 10 sequential, 13 Tumor dormancy, 11, 12 Tumor microenvironment promotion angiogenic response, 127 cancer associated fibroblasts, 126, 131 epigenetic dysregulation, 129 epithelial to mesenchymal transition, 20, 100, 106, 127, 130-132, 145, 149, 150, 193 extracellular matrix, 30, 84, 97, 126, 141 hypoxia, 10, 18, 42, 43, 49, 50, 59, 147, 164, 169, 182, 191, 192 immune filtration, 127, 160, 161, 163, 169 immunomodulation, 129, 162 mechanotransduction, 127, 162 mesenchymal to epithelial transition, 16, 131, 142 soluble factors, 127, 132, 145, 150, 170 suppression, 59, 128 Tumor subtypes, 19